National Criminal Justice Reference Service



This microfiche was produced from documents received for inclusion in the NCJRS data base. Since NCJRS cannot exercise control over the physical condition of the documents submitted, the individual frame quality will vary. The resolution chart on this frame may be used to evaluate the document quality.



MICROCOPY RESC TION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Microfilming procedures used to create this fiche comply with the standards set forth in 41CFR 101-11.504.

Points of view or opinions stated in this document are those of the author(s) and do not represent the official position or policies of the U. S. Department of Justice.

National Institute of Justice United States Department of Justice Washington, D. C. 20531

3/13/85



### U.S. Department of Justice National Institute of Justice

B . . D

This document has been reproduced exactly as received from the person or organization originating it. Points of view or opinions stated in this document are those of the authors and do not necessarily represent the official position or policies of the National Institute of Justice.

Permission to reproduce this copyrighted material has been

Public Domain/Nat'l Inst. on Drug Abuse/U.S. Dept. of Justice

to the National Criminal Justice Reference Service (NCJRS)

Further reproduction outside of the NCJRS system requires permission of the convright owner.



OF

This report was primarily funded by the National Institute on Drug Abuse (RO1 DA 01926-05). Additional funding was also provided by an Interagency Agreement between the National Institute on Drug Abuse (R01-DA-02355) and the Law Enforcement Assistance Administration (LEAA-J-IAA-005-8), U.S. Department of Justice under the Omnibus Crime Control and Safe Streets Act of 1968, as amended.

Additional support was also provided by the New York State Division of Substance Abuse Services, Narcotic and Drug Research, Inc., and by a research agreement program (80-IJ-CX-0049 S1) to Narcotic and Drug Research, Inc. by the National Institute of Justice, U.S. Department of Justice, under the Omnibus Crime Control and Safe Street Act of 1968, as amended.

Points of view or opinions in this document do not necessarily represent the official position or policies of the U.S. Government, New York State Division of Substance Abuse Services, or Narcotic and Drug Research, Inc.

# THE ECONOMIC BEHAVIOR

# STREET OPIATE USERS

# FINAL REPORT

To the NATIONAL INSTITUTE ON DRUG ABUSE and NATIONAL INSTITUTE OF JUSTICE

by

Bruce D. Johnson, Ph.D., Paul Goldstein, Ph.D., Edward Preble, M.A.

James Schmeidler, Ph.D., Douglas S. Lipton, Ph.D.,

Barry Spunt, M.A., Nina Duchaine, M.A., Reuben Norman, M.A.

Thomas Miller, Nancy Meggett, Andrea Kale, and Deborah Hand

Narcotic and Drug Research, Inc. New York State Division of Substance Abuse Services

> Two World Trade Center, 67th Floor New York, N.Y. 10047

> > September, 1983

# DEDICATION

-i-

to

# EDWARD PREBLE

1922 - 1982

Co-principal investigator and valued colleague. His contri-butions to this research were critical. This project would not have been possible without his wide contacts among street opiate users and the local community. His life work has enriched our understanding of the lives of heroin and other drugs users.

Co

See Appendix C for a summary of his contributions to this field.



TITLE PAGE **DEDICATION** Table of Con EXECUTIVE SU Acknowledger List of Vigr

# Chapter

1	INTRODUCTIO
II	RESEARCH DE
III	CHARACTERIS
ΙV	DEVELOPING
Y	DRUG USE AN
ΥI	NONDRUG CRI
VII	DRUG BUSINE
VIII	HOW MUCH CR STREET OP
IX	NONCRIME IN
X	NONDRUG EXP
ΧI	TOTAL INCOM
XII	ECONOMIC VA
XIII	CRIMINAL OF
XIV	WHICH STREE
XV	THE SOCIAL
XVI	SUMMARY AND
Appendix A	- A METHODO OF STREET
Appendix B	- TAKING CA
Appendix C	- A TRIBUTE

REFERENCES

## THE ECONOMIC BEHAVIOR OF STREET OPIATE USERS -ii-FINAL REPORT

# TABLE OF CONTENTS

TITLE PAGE	Page
DEDICATION TO EDWARD PREBLE	i
Table of Contents	ii
EXECUTIVE SUMMARY	iii
Acknowledgements	iv
List of Vignettes	xxiv
List of Tables and Graphs	××۷

N AND OBJECTIVES	1
SIGN AND METHODOLOGY	8
TICS OF RESPONDENTS	29
A HEROIN USER TYPOLOGY	37
D PURCHASE AMONG HEROIN USER TYPES	48
ME AMONG HEROIN USER TYPES	67
SS CRIMES AMONG HEROIN USER TYPES	98
IME AND CRIMINAL INCOME AMONG IATE USERS?	116

NCOME AMONG HEROIN USER TYPES	131
PENDITURES AMONG HEROIN USE TYPES	142
ME AND EXPENDITURES BY HEROIN USER TYPES	153
ALUES AMONG STREET OPIATE USERS	162
FFENDERS, HEROIN USERS, AND CRIME RATES	178
ET OPIATE USERS WERE NOT IN METHADONE TREATMENT	202
IMPACT OF CRIMINAL AND HEROIN USER LIFESTYLES	213
D POLICY IMPLICATIONS	232
OLOGICAL HISTORY OF THE ECONOMIC BEHAVIOR	243

Edditine mistory of the Economic Benarior	240
OPIATE USERS PROJECT	
RE OF RESEARCH BUSINESS	287
TO EDWARD PREBLE, M.A.	312

318

The Economic Behavior of Street Opiate Users: Executive Summary -iii-ECONOMIC BEHAVIOR OF STREET OPIATE USERS

FINAL REPORT

# EXECUTIVE SUMMARY

In 1976, the Panel on Drug Use and Criminal Behavior (1976) reviewed the literature and commissioned several scientific papers. This Panel report generated considerable controversy (see Clayton, 1981) when it concluded that "convincing empirical data on drug use and crime are generally unavailable--the principal reason being the lack of a long term, well coordinated, policy relevant research program in the area." This Panel report listed several questions about the relationship of heroin use to crime and suggested many hypotheses and methodologies which needed exploration.

To conduct the basic recommended research and to address the questions given in this Panel report, an application was submitted to the National Institute on Drug Abuse in 1977 to study the "Economic Behavior of Street Addicts," Bruce D. Johnson and Edward Preble, principal investigators. Two years of funding for pilot research were provided by the National Institute on Drug Abuse and by the National Institute of Justice. During 1978-79, the staff developed innovative methodologies for locating, recruiting, and systematically interviewing active street heroin and cocaine abusers (see below and in Appendices A and B.) Many complex problems emerged, and innovative techniques were developed to obtain detailed information about the daily criminal behavior, drug use, income, and expenditures of these persons. In 1979, these investigators submitted a new application to NIDA for three additional years of funding to conduct a major study of the



7 6

The Economic Behavior of Street Opiate Users: Executive Summary -iv-"Economic Behavior of Street Opiate Users." The main report describes that research effort, and provides answers to many of the questions asked in the 1976 Panel report. This executive summary highlights the major findings and refers the reader to appropriate sections of the main report.

East Harlem was selected as the initial research site because it had many methadone patients and street opiate users, a long history of being a high addiction community, and because Edward Preble knew many heroin users in the community. Two years (1978-79) of pilot research and instrument development demonstrated that reliable and valid data could be collected from opiate users recruited from the streets.

During these years, staff encountered many subjects whose physical dependence upon heroin or opiates could not be ascertained, but who injected ("mainlined") heroin or cocaine on at least several days per month. Thus, we adopted the term "street opiate users" rather than addicts." In 1980, efforts were undertaken to locate a scientific probability sample of opiate-using respondents from randomly selected blocks in East Harlem, but this effort had to be abandoned for many practical reasons itemized in Appendix A.

On the other hand, such persons were a highly visible presence on many streets in East and Central Harlem. They were accustomed to being approached by strangers (or contacted by others to buy drugs). Former heroin users with good street reputations were hired. These field workers went to several different blocks in East Harlem (and in 1981-82 to locations in Central Harlem). They recruited "street heroin users" who: a) had a history of heroin use and were currently injecting heroin or using methadone (both legal and illicit); b) spent most of their time on the street or at illegal institutions ('shooting galleries,' after hour clubs, etc.); and c) engaged routinely in criminal activities, especially robbery, burglary, theft, and drug sales, and other drug distribution crimes.

Design, Methodologies, and Respondents (Chapters II & III)

The Economic Behavior of Street Opiate Users: Executive Summary -V-After locating subjects, the field workers brought them to a research storefront for interviews. After obtaining the subject's informed consent, research staff conducted the first of a series of interviews in which subjects were asked to report about the previous day: a) their crimes and the dollar income, b) the drug(s) they used, purchased, sold, or distributed, c) their cash income from all sources, d) their cash expenditures for all purposes, e) their involvement in methadone or other treatment, and f) whether they received drugs or services at no cost (i.e., "in-kind") during the previous day. Respondents were paid \$10 and invited to return for an additional 8 interviews. During the next four days, they completed "daily" interviews. We then asked them to return once during each of the next four weeks. At these "weekly" meetings, they were interviewed about their behavior during the preceding seven days. This generated 33 person days (5 "dailies" plus 28 days on the four "weeklies") of data for each subject. During the first cycle of interviewing, we also interviewed each person about his/her life history.

In East Harlem, 132 subjects\* completed this first cycle. As part of the same study, street opiate users were recruited from Central (predominately Black) Harlem in 1981-82; 69 subjects were interviewed about 33 consecutive days. In addition, 110 subjects were invited to return for additional cycles of four more "weekly" interviews (28 days) which were scheduled three to six months apart. The actual interval between cycles varied greatly from subject to subject across the two years of data collection. These 201 subjects contributed collectively 11,417 person-days of data (an average of 57 days per subject) during almost 2,500 separate interviews. This data collection period ended in May, 1982.

The Economic Behavior of Street Opiate Users: Executive Summary -vi-Among the 201 subjects, 75% were male; 55% were black, 44% Hispanic, and 1% white. While 56% were over age 30 and only 13% were under 25, this was similar to the age distribution of clients in several methadone programs in New York City. No important differences existed between the East Harlem and Central Harlem respondents on almost all background variables except that almost all Central Harlem subjects were Black and somewhat older than East Harlem subjects. Over 60% were high school dropouts and 60% claimed to have been daily heroin users in the year prior to interview. They had 10 or more years of heroin use since age 15. Almost 85% percent reported having been arrested. They had an average of 8 arrests and 3 incarcerations. About half claimed to support their drug use mainly by theft, a quarter by involvement in drug business and 16% claimed to work as a major source of income.

These 201 street opiate users were among the most criminal persons ever studied (Chapter VIII). The average street opiate users in this study had the following major characteristics: 1) they had no or very little legal income (work or welfare); 2) three quarters had no methadone treatment during their reporting period; those admitted tend to drop out or not abide by methadone clinic rules; 3) almost all subjects committed crimes on one to five days per week; 4) most of their cash income was derived from crime; and 5) all subjects use drugs on a near daily basis, although there was considerable variation by type of drug. They spent most of their time on the streets, in abandoned buildings, and in the company of other street opiate users like themselves. In short, they were a highly deviant group.

The recruitment procedures probably seriously underrepresented methadone clients who were moderately conventional. Because our subjects were not recruited as part of a scientific sample, the findings cannot be generalized to a larger population of street opiate users -- even to other New York City opiate users. Nevertheless, their characteristics seem similar to many persons in methadone programs and in other research studies.

<sup>\*</sup> An additional 37 subjects provided fewer than 33 days of data and have been excluded from this analysis. Almost 85% of those giving their informed consent returned to the storefront on 9 different occasions to provide the 33 days of data; a very high response and reinterview rate among such street opiate users.

The Economic Behavior of Street Opiate Users: Executive Summary -vii-MAJOR QUESTIONS AND MAIN FINDINGS

The project was designed to address several questions raised by the Panel on Drug Use and Criminal Behavior (1976) about the behavior of street opiate users and heroin addicts. Many of these original questions can now be answered with findings from this project, summarized below. These issues will be discussed at length in the indicated chapters of the main report.

### a) How, if possible, can accurate data on income, expenditures, drug consumption and criminal activities of street opiate users be measured on a short interval basis? (Chapter II, Appendices A and B)

Street opiate users can be located and recruited in high addiction neighborhoods. They willingly answer virtually any question(s) when assured that their answers will remain confidential and when they receive a modest payment for their time. These subjects report details about specific crimes, their drug consumption, and the money they received or expended, as well as various forms of bartering for drugs, food, or pervices. Requests for information about a brief time period (such as a day) elicit detailed, high quality data that would be otherwise unavailable.

### b) How accurate, reliable, and valid are the data obtained? (Chapter II, Appendices A and B)

Such data, obtained on a day-by-day basis during interviews on differing days, were at least as reliable, valid, and accurate as the data in any prior study. Such data were generally reliable in that internal checks about a given day's income and expenditures generally balance. Moreover, the patterns of behavior reported in the first few interviews were typically reported in later interviews, although considerable variability existed about the details of events and the dollar amounts reported on different days.



new information.

c) ł	low n	nuch r	ieroi	n,
ā l	and a	other	drug	IS
Ī	burch	lased	comp	are
I	V &	V)		
Subje	ects	were	cla	ssi
approxima	ite r	umber	of	day
				-
week as:	d	aily	her'o	in
days/week	(), a	nd ir	regu	lar

- used.
- used and purchased.
- consumption.

Chapter V and subsequent chapters include a series of vignettes or brief descriptions of the drug use patterns, criminal lifestyles and typical crime events, as well as the typical living patterns of selected subjects. These vignettes convey the complexity of respondent lifestyles and provide insights which cannot be presented in this executive summary.

The Economic Behavior of Street Opiate Users: Executive Summary -viji-The data were also valid in the sense that field workers and interviewers observed subjects selling stolen goods, or having large amounts of cash, being under the influence of drugs, and participating in some of the activities they

described in interviews. Two different respondents occasionally reported the same crime or drug use event. The accuracy of their estimates of the value of drugs received for working in drug distribution, and other in-kind income, were harder to assess. But since such "cashless" income had not been systematically measured before, the data in the main report now provide major

### cocaine, illicit methadone, legal methadone, do street opiate users consume? How much is ed to the amounts received in-kind? (Chapters

fied into a Heroin User Typology according to the ys during the reporting periods they used heroin per users (6-7 days/week), regular heroin users (3-5 heroin users (0-2 days/week). Major findings were:

Except for their heroin use, daily heroin users were not significantly more involved with cocaine, alcohol, marijuana, or pills, than the less regular heroin users, although there were some differences in the proportion of person-days consuming specific drugs and the dollar amounts

Daily heroin users consumed over \$17,000 worth of drugs per year (mostly heroin) compared to about \$5,000 by irregular heroin users.

Heroin and cocaine accounted for almost 90% of the dollar value of drugs

Cash purchases of drugs accounted for approximately 55-90% of the drugs consumed, depending upon the specific drug and measure of purchase or The Economic Behavior of Street Opiate Users: Executive Summary -ix-

d) How much crime and cash income did addicts earn from various types of predatory crimes (robbery, burglary, theft), victimless crimes (prostituting, pimping, conning), and drug distribution crimes? (Chapters VI & VII)

Several important findings emerged from an analysis of nondrug crimes:

- Daily heroin users were criminally active on more days than regular and irregular heroin users, but such differences were important mainly for robbery, burglary, shoplifting, and other larcenies.
- Daily heroin users had more nondrug offenses and criminal cash income than regular or irregular heroin users; this was entirely due to the fact that daily heroin users were criminally active on more days, and not because they committed more crimes per day or gained larger cash returns per crime.
- The heroin user types did not differ in forgery, con games, prostitution/ pimping, shoplifting-own use.
- Cash income from nondrug crime for all subjects was relatively modest at \$16/day (including days without crimes).

Drug distribution crimes were much more complex than previously described in the professional literature. Several street opiate users "deal" (sell drug directly to the buyer), but more engaged in steering, touting, and copping (henceforth STC) or other drug-related crimes (renting works, running a shooting gallery, tasting drugs--see Chapter VII and Goldstein, 1981). Respondents gained "drug income" by working in the drug business, as well as by stealing from, burglarizing, or robbing other drug distributors. In addition, respondents obtained drugs by "avoided expenditures" -- as when they obtained gifts, shared someone else's drugs, or "copped short." Other important findings also emerge:

- While virtually all subjects engaged in drug distribution crimes, daily heroin users were active on more days and obtained greater income in the form of cash or drugs than irregular heroin users.
- Subjects engaged in steering, touting, and copping (STC) on a more regular basis and received more "drug income" for their labor in helping to distribute drugs than from direct sales of drugs to buyers.
- Daily and regular heroin users committed an average of over 800 drug distribution crimes per year; irregular heroin users committed only about 250/year.



The Economic Behavior of Street Opiate Users: Executive Summary -xe) How many different criminal offenses do street opiate users commit while on the street? With what returns? (Chapter VIII)

These street opiate users exhibited an extremely high volume of criminal activity, generally higher than prior studies -- although not in every offense By annualizing the number of crimes per day the offending rates category. (lambda) of unapprehended criminals was estimated. Their cash income from such crimes, however, was much more modest. These data showed:

- times as many as irregular heroin users.
- irregular heroin users.
- (daily heroin users) more offenses.
- than \$5,000 among irregular heroin users.
- about \$6,000.
- Chaiken, 1982).
- from their crimes.

Daily heroin users committed about twice as many robberies (12/year) and burglaries (34/year) per year as regular heroin users, and about five

Daily heroin users committed 209 nondrug crimes per year compared with 162 among regular heroin users and 116 among irregular heroin users.

Regular and daily heroin users committed almost 1,000 major offenses per year (including drug distribution crimes) compared with 360/year by

Minor crimes (fare evasion, drug thefts, illegal drug transfers between friends) added an additional 147/year (irregular heroin users) to 311/yr

Daily heroin users gained over \$11,000 in criminal cash income versus less

When drug income (the economic value of drugs received without cash payment) was included, the annual economic value from crime received by daily heroin users exceded \$18,000 while irregular heroin users gained

The annualized offense rates among these street opiate users was generally higher than among Miami heroin users (Inciardi, 1981), California prisoners (Peterson, et al., 1980), a national treatment cohort (TOPS -Collins, et al., 1982abc), and as high or higher than among incoming prison and jail inmates in Texas, Michigan, and California (Chaiken and

These subjects, especially the daily heroin users, exhibited levels of criminal behavior which was among the highest ever recorded. This was especially true when the many drug distribution crimes and other minor offenses were systematically counted and included.

Despite the very high rates of criminal offending, however, these respondents (with some exceptions) obtained relatively modest cash income

The Economic Behavior of Street Opiate Users: Executive Summary -xi-

f) How much noncriminal income do street opiate users generate and from what sources? (Chapter IX)

Street opiate users obtained very limited legitimate income from employment, from welfare and other transfer payments, or from other noncriminal sources. Noncriminal income showed no significant variation among daily, regular, and irregular heroin users. Specifically:

- Less than half of these street opiate users obtained any income from legitimate employment or from welfare/unemployment. These sources each provided about \$2,000 of cash income to all subjects with no variation by heroin user type.
- These street opiate users obtained about \$1,400/year in "avoided expenditures" (meals, shelter, and other goods provide by family and friends).
- Cash income from employment, public transfer payments, friends, family, respondent payments, and other income sources, provided slightly over \$4,000/year.
- Clearly, these street opiate users lived at or below the federal poverty line (\$4,729/yr for a single person).
- Criminal cash income was greater than the noncriminal income among the daily and regular heroin users.
  - g) How much cash do street opiate users expend upon nondrug items? (Chapter X)

Nondrug expenditures exhibited no differences by heroin user types. A11

subjects had very low levels of cash expenditures for nondrug purchases (about

\$5,000/year).

- . Subjects spent about \$1,000/year for food, about \$700 for shelter, about \$700 upon their family, about \$700 for alcohol and cigarettes, and very little for other nondrug purposes.
- Very few subjects had an apartment of their own and for which they paid rent. Relatives, girlfriends, or others frequently paid the basic shelter costs and permitted the respondent to live there in exchange for an occasional cash gift, or purchase of a few groceries.
- Expenditures for food were generally for snacks and food items consumed during the day with available funds. "Meals" were generally eaten with relatives, girlfiends, and others.
- Thus, subjects carefully minimized their expenditures for basic necessities such as food and shelter; their cash expenditures were low even by poverty standards and in comparison with their low income counterparts.



(Chapter XI)

The income of these respondents was more complex than previously reported. Four analytically different kinds of "income" were received by street opiate users: 1) cash income from crime (both drug distribution and nondrug crime); 2)cash income from noncriminal sources; 3) "drug income" was obtained when drugs having a real economic value were received but without a cash purchase (generally, labor in or knowledge of the drug business was exchanged for the drugs used); and 4) "avoided expenditures" were a form of income to respondents in that someone else paid cash for food, shelter, or drugs so the subject avoided necessary cash expenditures. The data showed:

- among the irregular heroin users.

- heroin plus cocaine.

- consumption of drugs.

# The Economic Behavior of Street Opiate Users: Executive Summary -xiih) How much income do street opiate users have and from what sources? How much cash income do street opiate users expend on various items?

Daily heroin users had a total cash income of almost \$15,000 per year of street time (assuming no incarceration) compared to over \$8,000 per year

Daily heroin users obtained a total income (from all four types of income) of almost \$24,000 per year, compared to about \$16,000 among the regular heroin users, and about \$11,000 for the irregular heroin users.

These differences in total income were due mainly to the fact that daily heroin users obtained over twice as much income as the irregular heroin users -- from cash income from crime (\$11,292 vs. \$4,451), from avoided expenditures (\$4,857 vs. \$2,484, and about nine times as much in drug payments (\$4,016 vs. about \$605).

Regular heroin users had incomes which were more similar to the irregular heroin users than to the daily heroin users.

The average respondent had somewhat over \$11,500 in cash expenditures per year, of which over a third (\$4,200) involved cash purchases of heroin.

Over half (\$5,990) of these cash expenditures were cash purchases of

The average respondent spent over one and a half times as much upon drugs (\$7,252/\$4,257) as upon food, shelter, and other nondrug expenditures.

Daily heroin users had greater cash expenditures than regular or irregular heroin users, only because they purchased so much heroin. They had virtually the same expenditures for cocaine, other drugs, and nondrug purposes as the less regular heroin users.

Noncriminal income was just about equal to nondrug expenditures. Likewise, criminal income (including drug payments and avoided expenditures for drugs and alcohol) was just about equal to annual The Economic Behavior of Street Opiate Users: Executive Summary -xiii-

### i) How much were the stolen goods and illegal services worth? That is, what were the direct and indirect economic values associated with heroin users? (Chapter XI)

Despite their limited cash income from crime, illegal activities had important economic values. Estimates were made of the "fence" factor (i.e. the fair retail value of merchandise stolen during burglary, robbery, and theft). Additional estimates were made of 33 different components of economic values, including the "value added" to the illegal drug distribution system by the labor performed in the drug business by these subjects. These estimates showed:

- The average street opiate user had direct economic values of almost \$20,000 per year, of which almost \$14,000 was from the goods or cash obtained during nondrug crimes, mainly shoplifting or burglary, \$2,600 was from noncriminal sources, and \$2,900 was from avoided expenditures.
- Daily heroin users had three times (almost \$30,000) greater direct economic values than the irregular heroin users (almost \$10,000).
- Indirect economic values (drug sales, STC returns, income tax evasion, and the "value added" to the illegal drug distribution system) were also substantial (about \$14,400/year), with the daily heroin users (\$25,400) having five times more indirect economic values than irregular heroin users (\$5,000).
- When direct and indirect economic values were combined, daily heroin users had total economic values of about \$55,000, regular heroin users about \$32,000, and irregular heroin users about \$15,000.
- These estimates of economic values did <u>not</u> include many other factors typically included in prior research about social costs such as: foregone productivity of legitimate work, police, court, correction, probation/parole, treatment costs, private anti-theft costs, fear of crime, and the psychic pain to vitims.



j)

Chaiken and Chaiken (1982) recently developed a new way to classify criminal offenders which was reproduced as closely as possible among these New York street opiate users. The central findings of their study were replicated in almost all essential respects, and extended to include information about criminal income. Street heroin users classified into the most serious categories of this hierarchy, when compared with subjects in the less serious categories:

- number of different kinds of crime).
- offense.
- many or more person days.
- offenses.
- criminal income from all crimes.
- in STC as those who sell drugs.
- typology. :
  - users were robbers.
- from all major crimes.

•

The Economic Behavior of Street Opiate Users: Executive Summary -xiv-What types of criminal offenders were involved in heroin abuse? How does heroin use among such criminal offenders affect their crime rates and criminal income? (Chapter XIII)

Exhibited greater breadth of criminal involvement (i.e. commit a larger

Were as likely or more likely to report involvement in any given specific

Were apt to commit the most serious (robbery and burglary) crimes on a greater proportion of person days, and to commit less serious crimes on as

Committed as many or more crimes per year.

Obtained a high annual cash income from definitional offenses and from all

The most serious offenders, the robber-dealers, committed robbery and drug sales at high rates and also committed burglaries, property crimes, and steering/touting/copping at high rates also. The robber-dealers also had high cash income from most offenses, and had the highest overall

Almost all subjects engaged in steering, touting, and copping; this behavior appeared to be relatively independent of "drug dealing;" criminal offender types not selling drugs appeared about as likely engage

This criminal offender typology was strongly related to the heroin user

. Half of the daily heroin users were classified as robbers, while a quarter of the regular heroin users and 13% of the irregular heroin

. Irregular heroin users were especially concentrated among those classified in the theft and lower level drug dealer categories.

Among a specific category of criminal offender, the more regular the heroin use, the greater the annual crime rate and criminal cash income

### The Economic Behavior of Street Opiate Users: Executive Summary -xv-

Robber-dealers (11% of all subjects) committed 60% of all robberies, 26% of all drug sales, 22% of all burglaries, 10% of the theft offenses, and 14% of the STC offenses. In short, the most serious offender types committed a disproportionate share of the total volume of all crimes, but especially the most serious crimes.

### k) How do subjects with some methadone treatment compare to those without it? (Chapter XIV)

A quarter of these street opiate users reported having "some" methadone treatment during their reporting period; they exhibited much variation in the number of days enrolled, whether they took their medication as prescribed, or distributed it to others. Moreover, our recruitment techniques located the least compliant methadone patients. Several important findings emerged:

- Although it might be anticipated that street opiate users without current methadone treatment might be more deviant than those with such treatment. few differences in their demographic characteristics or prior criminal or drug use history were found.
- Subjects with some methadone treatment were a third as likely to be daily heroin users and twice as likely to be irregular heroin users as subjects without methadone treatment.
- Some evidence suggested that subjects without methadone treatment were avoiding it and other kinds of treatment. Only 10% of those classified as robbers reported some methadone treatment, and only one such subject consumed legal methadone during two-thirds of his reporting period.
- Very few subjects mentioned a desire to go to treatment to our staff; the rare request for help to gain treatment entry was seldom followed up by the subject.
- With a few exceptions, the evidence suggested that few (if any) of the subjects not in methadone treatment would voluntarily enter treatment in the near future. This avoidance of treatment appeared most pronounced among the most seriously criminal who also were daily heroin users.



Four dimensions of social impact were developed: the annual number of offenses, criminal severity scores (after Wolfgang and Figlio, 1982), criminal income, economic values, and the seriousness of illicit drug use. In addition, an Intensive Criminality Typology was developed and had five categories: a) robbers and daily heroin users were "intensives;" b) nonrobbers but daily heroin users were "highs;" c) robbers and nondaily heroin users were "actives;" d) nonrobbers and regular heroin users were "inactives;" e) nonrobbers and irregular heroin users were "lows." This typology was essentially a four-fold classification along the robber-nonrobber and daily-nondaily heroin user dimension, except that the large cell of nonrobbers/ nondaily heroin users (54% of all subjects) was subclassified into nonrobbers who were regular and irregular heroin users. Major findings

emerged:

- robber.

The Economic Behavior of Street Opiate Users: Executive Summary -xvi-1) What combinations of heroin use and criminality have the greatest social impact? (Chapter XV).

The intensive criminals were not significantly different than their less criminal counterparts on most major background characteristics nor did they report greater drug use or criminal activity and income (excepting robbery and burglary) during the year previous to interview.

Nevertheless, intensive criminals (and sometimes the highs) had significantly greater values than their less criminal counterparts on a variety of measures of social impact, especially those involving criminal income and economic values. Specificially, intensive criminals had three to six times greater social impacts than the lows and about twice as much criminal income and economic values as their inactive and active counterparts. While the intensive criminals always had the highest mean values on these social impact measures, their values usually did not differ significantly from the highs, with the important exception of the economic values associated with their nondrug crime.

Daily heroin users had social impacts which were significantly greater than nondaily heroin users regardless of whether the person was also a

Despite their greater criminal income and economic harm, however, intensive criminals did not report significantly more arrests or years of incarceration than their less criminal counterparts.

The Economic Behavior of Street Opiate Users: Executive Summary -xvii-SUMMARY AND POLICY IMPLICATIONS (Chapter XVI)

"Policy implications" were understood to be a set of goals or objectives flowing naturally from the research findings. Treatment or criminal justice practitioners may be able to design programs, practices, or techniques to accomplish such objectives. These "implications," however, were not "policy recommendations" which suggest specific steps about how to implement changes in existing arrangements.

Evidence from this study and from other recent research documented a central major policy implication for American Society:

## The most criminally active street opiate users are "slipping between the cracks" of the criminal justice and treatment systems.

The most criminally active persons in this study, the "intensive criminals" were defined as robbers and daily heroin users. Similar findings were reported by Ball, et al. 1981; Chaiken and Chaiken (1982); Chaiken (1983); Moore, et al., (1981), Johnson, (1981); Johnson, et al, (1983a). This study and these sources demonstrated that additional information about daily and/or high cost heroin use may assist in identifying high risk street opiate users. Such intensive criminals were rarely in methadone treatment and other drug treatment was uncommon. In short, intensive criminals "slip away" from or avoid treatment.

The criminal justice system largely ignores the drug use patterns of arrestees in making prosecutorial and sentencing decisions. Moreover, the prior arrest and incarceration histories did not distinguish intensive criminals from their less criminal counterparts. Thus, they appear to "slip through" the criminal justice system with jail and prison sentences which were typically short and generally not more severe than their less criminal counterparts. Five major policy implications summarized below are documented at more length in the main report:



### Policy Implications

This study cannot definitely answer the difficult question, "what then can be done with these highly criminal street opiate users?" Suggested directions implied by the data, however, involve closer coordination and cooperation between the criminal justice system and treatment systems. New social policies and institutional arrangements need to be carefully developed so that more systematic, probably daily, pressure is placed upon criminally active street opiate users to detoxify and enter long term drug abuse treatment programs. Once arrested and convicted of crimes, including minor offenses, they should undergo routine monitoring for drugs by urinalysis, and be required to attend treatment programs daily and report for long counseling Other innovative approaches also need careful exploration and sessions. implementation. Whatever solutions are developed, the high criminality levels among street opiate users demand to be addressed more directly than during the five year study period of the Economic Behavior Project.

The Economic Behavior of Street Opiate Users: Executive Summary -xviii-

A-- Daily heroin users who committed robbery were the most criminally active. These "intensive criminals" had high crime rates, criminal severity scores, criminal income, and economic values and should be a major focus of criminal justice agencies and social policies to address criminality among street opiate users.

B - Intensive criminals report prior arrests and incarcerations that did not differ greatly from their less criminal counterparts; thus they will be difficult to systematically identify from current information maintained by the criminal justice system.

C - Despite their very high crime levels, however, a social policy of incarcerating street opiate users does not appear to be a socially appropriate or economically reasonable solution for their criminality.

D - Despite widespread patterns of multiple drug use among street opiate users, the drugs imposing major economic problems were heroin (primarily) and cocaine. Social policies designed to reduce by half the regularity or dollar amount of heroin (or cocaine) consumed, especially by daily users, would have substantial benefits for both society and these street opiate users.

E - While additional capacity is needed for all persons voluntarily seeking methadone treatment, new social policies need to be designed having the objective of effectively identifying criminally active street opiate users and pressuring and monitoring them to reduce their heroin and other drug abuse patterns. Such policies must be especially directed towards street opiate users currently avoiding methadone or other treatment.

# The Economic Behavior of Street Opiate Users: Acknowledgements -xix-ACKNOWLEDGEMENTS

This report contains the results of almost seven consecutive years of effort. The research was a pioneering effort to develop ways of obtaining information about crime, drug use, income, and expenditure among street opiate users while at liberty on the street.

During these seven years, many people have influenced the course of this project and many have worked on various phases of the research. I cannot fully express my appreciation for their contributions, but these contributions deserve recognition.

Staff associated with the National Institute on Drug Abuse made this research possible. In late 1976, I attended a workshop held by NIDA on the Social Costs of Drug Abuse and was thinking of a similar study. After the conference, I stopped at Dan Letteri's office to explore some possible grant ideas; he gently suggested some good reasons why my ideas were unlikely to get funded. As we talked about the work Edward Preble was conducting on the Ethnography of Drug Use Among White Ethnic Groups, however, Letteri mentioned a report which had just become available and of which I was unaware. I went to the National Technical Information Service and purchased a copy of the Report of the Panel on Drug Use and Criminal Behavior (1976). Two days later I had read the whole report, cover to cover. It pointed me directly to the kinds of questions which needed research -- and it was clear that only Ed Preble would be able to recruit the research subjects needed. Several conversations with Fred Goldman of Columbia University reiterated how little was really known about the economic aspects of the lives of heroin users.

I enthusiastically told Ed Preble about this report, and he quickly agreed to become a co-Principal Investigator. We started work on the



The Economic Behavior of Street Opiate Users: Acknowledgements  $_{-XX-}$  proposal and submitted it in February, 1977. The initial review group at NIDA realized that what we planned to do was extremely important. But they had serious reservations about whether such detailed information could be collected from active heroin users. The site visit occurred on a sunny plaza in San Francisco at the American Sociological Association where I met with Michael Agar. We discussed various options and he suggested that the review committee would provide 18 months of funding to find out answers to basic methodological questions. The project started in December, 1977 with the first staff (Preble, Miller, and Goldstein) hired about that time.

Richard Clayton was assigned as the NIDA project officer for the grant. As a visiting scientist, he had the mission of building a research agenda about drug use and crime for NIDA -- and this grant was a key part of that agenda. Both of us learned a great deal about federal grantsmanship. He helped me over two hurdles. Obtaining a federal certificate of confidentiality was essential for this research to procede. While a new law had just been passed, no regulations and no procedures for obtaining this certificate were available. But eventually Clayton found the right person(s) and got the signatures.

In addition, he phoned one day in 1978 to tell me that the National Institute of Justice (NIJ) might interested in funding part of the project. This led to an interagency grant between NIDA and NIJ which provided funding in 1979 which permitted this grant to extend an 18 month pilot study into a 24 month pilot study. Richard Barnes and Bernard Gropper were responsible for the NIJ contributions.

After Clayton left NIDA to return to the University of Kentucky, the project officer became Louise Richards. She continued to keep me informed about emerging changes at NIDA, helped me obtain funding for the main study in 1979, and worked to maintain interest in drugs and crime at NIDA as the administration changed in Washington. She has made this report possible by obtaining a no cost extension after the grant funding terminated. The Economic Behavior of Street Opiate Users: Acknowledgements -xxi-At the Division of Substance Abuse Services (DSAS), Douglas Lipton, brought myself and Edward Preble into the agency and helped both of us make this project work. By freeing me from my State responsibilities, I was able to write and supervise this grant. He helped me iron out numerous problems and carefully read the final manuscript.

Narcotic and Drug Research, Inc. is a private, nonprofit research organization affiliated with DSAS and administers grants such as these.Hugh Schrader, Marion Ludlum, and Valarie Murphy of NDRI handled the business aspects of the grant efficiently so that full effort could be concentrated on the research.

This research could not have been accomplished without the consistent efforts of the staff affiliated with the Economic Behavior Project. I had major responsibility for grant writing, administration, and much data analysis and report writing. The responsibility for doing the field work and running the storefront was handled by Edward Preble. Paul Goldstein had major responsibility for developing interview schedules, interviewing, coding, data analysis, and report writing. Thomas Miller was the senior interviewer, and is now one of the best in the business. His many brief descriptions of crimes have been lifted from the interview forms directly into the "vignettes;" they bring ethnographic richness to the statistical data. These four key staff remained on the project for five years; only death has separated Edward Preble from us. (See Appendix C).

During the data collection phase of this research, several persons have participated as field workers and part time interviewers. In Appendix A and B, we indicate how crucial field workers were to the success of this project--thanks are due to Enrique Ruiz, Melvin Davis, Rebecca Mitchell, "Top," Angela Jacquez, and Alton Bates. This project ultimately rests upon their ability to locate street opiate users and encourage them to come to the storefront for the first and subsequent interviews.



Similar contributions Nancy Meggett was a stell processor. Others have Washington, Patricia Miller Nina Duchaine was the much of the coding. Andre schedules and entered the major responsibility for assisted by Ken Robertson.

James Schmiedler suggested many simple but important ways of analyzing the data as well as appropriate statistical tests. In addition, his extensive programming skills have made this report possible. As Ed Preble once told me: "he crunches the numbers, you just write about them!" (as if that were so easy!)

Barry Spunt and Deborah Hand have helped compile all these numbers, checked them carefully against computer printouts, and the text against tables. They have located and written the vignettes, coded materials, and helped in many other ways.

In addition to the above persons associated with the Economic Behavior project, several other persons at the research bureau of DSAS have contributed greatly to this project. Blanch Frank's comments on a draft of this report were very helpful. Don Des Jarlais, William Hopkins, Dana Hunt, Douglas Goldsmith, Mike Miranda, Sherry Deren, Ken Robertson, Jonathan Krieger, Phil Appel, Herman Joseph, David Strug, and others have all assisted this project in some important way.

In particular, Eric Wish, who was intially hired for the Economic Behavior Project but transferred to another similar project, has been a strongly supportive and a constructive critic. His careful comments have been incorporated into this report. He also directs a project which builds

The Economic Behavior of Street Opiate Users: Acknowledgements -xxii-Similar contributions were also made by staff at the World Trade Center. Nancy Meggett was a stellar secretary and became an expert on the WANG word processor. Others have functioned as transcribers: Alice Traub, Deborah Washington, Patricia Miller, and Elaine Fillippini.

Nina Duchaine was the primary research assistant who supervised and did much of the coding. Andrea Kale also did much of the editing of the weekly schedules and entered the data for computerized analysis. Reuben Norman had major responsibility for setting up computerized data files, and was ably assisted by Ken Robertson.

The Economic Behavior of Street Opiate Users: Acknowledgements -xxiiidirectly from the research undertaken by the Economic Behavior Project. Wish and I are conducting studies of the role of drugs and alcohol in crime events among the same types of street opiate users studied here.

Other office staff at DSAS have also contributed to this project at various times. Connie Ross has helped keep files organized, helped clerical staff understand the pressures of grant writing, and trained staff in word processing. Marva Bruzal did the graphs. Sharon Hodges, Kelita Jones, and others have done some clerical work on this project.

I particularly want to thank my wife, Theresa, and daughter, Amanda, for their support while completing this report. It has meant many late nights and long weekends away from home. They have found other constructive things to do and supported me during those times.

My greatest expressions of appreciation, however, are reserved for the persons who shared their lifes with us. While our research subjects must remain anonymous, the many days, months, and years of seeing them. interviewing them, laughing with them, hanging around with them, and being a part of their lives has been one of the richest experiences we have had. In addition, they have also provided the kinds of rich data and information that is otherwise unavailable. Not only do we thank them, but the sociology and criminology is vastly richer for their participation. To paraphrase Ed Preble (1980), what originally appeared to be a mean motive, their desire for money, has been transformed into a rewarding research relationship which will contribute to society for many years to come, and possibly improve the social response to both drug abuse and crime in the future. Our relationships with these street opiate users has been rewarding for us and for society.

To all of the above, I express my sincere appreciation for the many contributions which have brought this research to a conclusion and this report to the public. I look forward to further analyses of these data in the forthcoming years.

Bruce D. Johnson, New York City September, 1983

# VIGNETTE A В С D Ε G Η Ι J Μ N 0 0 R S Т U V

DESCRIPTION

Heroin and Opiate Daily, Regular, Cocaine Use by St Alcoholic Heroin Daily Patterns of A Street Robber D Burglars Doing Bur "Boosters" and "Ca Multiple Offenders Forgery Events Con Games and Thre Prostitutes and Pi "Prostitutes" as Li Other Illegal Acts Dealers, Jugglers, Steerers, Touts, ar Shooting Galleries Role Diversity in t The Big 'Score' --Getting Drugs for " Avoiding Expenditure "Getting by" Shelter An Intensive Crimina

# THE ECONOMIC BEHAVIOR OF STREET OPIATE USERS -xxiv-FINAL REPORT

# INDEX OF ALL VIGNETTES

	Page
e Use and Purchase Patterns Among , and Irregular Heroin Users	54
reet Opiate Users	56
Users	57
Substance Use Among A Regular Heroin User	60
oing Robberies	79
rglaries	70
attle Rustlers"	79
Doing Other Larcenies	80
, ,	81
e Card Monte	82
mps	82
	83
ares for "Chump" Robberies and Larcenies	83
	84
and House Connections	99
nd Cop Men	100
	101
he Drug Business	102
Robberies and Burglaries of Dealers	103
Nothing"	105
es	134
r Cests	146
al Diversity of Criminality	222

# The Economic Behavior of Street Opiate Users -xxv-THE ECONOMIC BEHAVIOR OF STREET OPIATE USERS FINAL REPORT

# INDEX OF ALL TABLES AND GRAPHS

Support and the second second

Table/Graph	<u>Contents</u>
Graph II. 1	Methadone Clients in Northern Manhattan in 1979 by ZIP Code of Residence
Table III. ]	Number of Subjects and Days of Reporting and Reasons for Noncompletion
Table III. 2	Characteristics of Respondents in the Economic Behavior
Graph IV. 1	Project Amount of Heroin Used Per Day (Including Nonuse Days) by Persons Classified According to Their Frequency of Heroin Use
Graph IV. 2	Dollar Amount Consumed Per Day of Heroin Use by Persons Classified According to Their Frequency of Heroin Use
Table IV. 1	Proportion of Subjects and Number of Heroin Use Days
Graph V. 1	Annual Dollar Amounts of Drugs Used and Purchased Among Heroin User Types
Table V. 1	Percentage of <u>Respondents</u> Using and Purchasing Brugs by Heroin User Typology
Table V. 2	Percentage of <u>Person-Days</u> Using or Purchasing Drugs by Heroin User Typology
Table V. 3	Dollar Value of Drug(s) <u>Used</u> or <u>Purchased</u> Per Day by Heroin User Typology
Table V. 4	Dollar Values Consumed Per <u>Day of Use</u> or <u>Value</u> <u>Purchased</u> Per <u>Day of Purchase</u> by Heroin User Typology
Table V, 5	Dollar Value of Drug(s) <u>Used</u> or <u>Expended</u> Per <u>Year</u> by Heroin User Typology
Table VI. 1	Percentage of <u>Respondents</u> Committing Nondrug Crimes by Heroin User Typology
Table VI. 2	Percentage of <u>Person-Days</u> on Which Nondrug crimes Were Committed by Heroin User Typology
Table VI. 3	Nondrug Crime: Offending Rates Per Day and Per Criminally Active Day by Heroin User Typology

		Ta
		Та
<u>Page</u> 13		Ta
31		Ta
33	and the second se	Ta
39		Та
41		Ta
44	and the second sec	Ta
52	and the second s	Ta
62	and an and a second	Ta
63	and a second	Ta
64		Gr
65	and and a second se	Tal
66	Same	Tal
91		Gra
92	C Catilla C	IGI
93		Tat
		Tat

States i

Second March

The	Ecor
Table/Graph	
Table VI. 4	Cas Per
Table VI. 5	Cas Typ
Table VI. 6	Cas Use
Table VI. 7 –	Ann Inc
Table VI. 8	Num spe (th Cal
Table VII. 1	Per Dru
Table VII. 2	Per Dru
Table VII. 3	Ann Dru Her
Table VII. 4	Ann Valu
Table VIII. 1	Anni Burg Crim
Graph VIII. 1	Numl
Table VIII.2	Annı Hero
Table VIII. 3	Annı Herc
Graph IX。1	Annı
Table IX. 1	Perc and by H
Table IX. 2	Perc and Hero
Table IX. 3	Mean Annu Sour

nomic Behavior of Street Opiate Users -xxvi-	
Contents	Page
sh Returns from Nondrug Crime: <u>Mean Cash Income</u> r <u>Day</u> by Heroin User Typology	94
<u>sh Returns Per Nondrug Offense</u> by Heroin Use pology	95
sh Returns Per <u>Criminally Active Day</u> by Heroin er Typology	96
nualized Crime Days, Offense Rates, and Cash come from Crime(s) by Heroin User Typology	97
nber of offenses per street year involved in ecific crimes: Comparison of heroin users his study) with prison and jail inmates in lifornia, Michigan, and Texas.	75
rcent of <u>Respondents</u> Reporting Involvement in Ig Distribution Activities by Heroin User Typology	112
rcent of <u>Person Days</u> Reporting Involvement in Ig Distribution Activities by Heroin User Typology	113
nualized Dollar Amount of <u>Cash Income</u> and <u>Ig Income</u> from Drug Distribution Activities by oin User Typology	114
nualized <u>Number of Drug Transactions</u> and Dollar ue of <u>Drugs Distributed</u> by Heroin User Typology	115
nualized Offense Rates (Lambda) for Robbery, glary, Thefts, and Drug Sales Among Studies of minal Offending.	121
ber of Criminal Offenses Per Year	122
ualized Criminal Offending Rate (Lambda) by oin User Typology	129
ualized Criminal Income for Each Source by the oin User Typology	130
ual Noncriminal Income by Heroin User Type	136
centage of <u>Respondents</u> Reporting Cash Income Avoided Expenditures from Various Sources Heroin	139
centage of <u>Person Days</u> Reporting Cash Income Avoided Expenditures from Various Sources by oin User Typology	140
n Dollar Amounts of <u>Cash Income Per Year</u> and ual Value of Avoided Expenditures from Various rces by Heroin User Typology	141

	·	
· Tire	Economic Behavior of Street Opiate Users -xxvii-	Dago
Table/Graph	Contents	raye
Graph X. 1	Nondrug Expenditures by Heroin User Typology	144
Table X. 1	Percentage of <u>Respondents</u> Reporting Cash Expenditures for Various Purposes by Heroin User Typology	150
Table X. 2	Percentage of <u>Person Days</u> Reporting Cash Expenditures from Various Sources by Heroin User Typology	151
Table X. 3	Mean Dollar <u>Expenditures Per Year</u> for Various Purposes by Heroin User Typology.	152
Graph XI. 1	Annual Income From All Sources by Herioin User Types	155
Graph XI. 2	Annual Cash Expenditures by Herioin User Types	157
Table XI. 1	Total Income per Year of Street Opiate Users	160
Table VI. 2	Total Cash Expenditures Per Year for Drugs and Nondrug Purposes by Heroin User Typology	161
Graph XII. 1	The Economic Values Associated with Heroin User Types.	167
Table XII. 1	Direct and Indirect Economic Values per Year among Street Opiate Users by the Heroin User Typology.	171
Table XIII. 1	Definitions of the CRIMINAL OFFENDER TYPOLOGY and Percentage of Sample So Classified	183
Table XIII. 2	Percentage of <u>Respondents</u> Active in Each Crime Type Among Each Criminal Variety Type	192
Table XIII. 3	Proportion of <u>Person Days</u> Active in Each Crime Type Among Each Criminal Variety Type	193
Table XIII. 4	Number of <u>Offenses (Lambda) Per Year</u> of Street Time in Each Crime Type Among Each Criminal Variety Type	194
Table XIII. 5	Annualized Criminal Income for Each Offense Among Each Criminal Offender Type	195
Table XIII. 6	Percentage of Respondents in the Criminal Offender Typology and Heroin User Typology	196
Table XIII. 7	Number of <u>Offenses (Lambda) Per Year</u> of Street Time for Robbery, Burglary, and Property Crimes Among Each Criminal Variety Type and Heroin Use Type	197
Table XIII. 8	Number of <u>Offenses Per Year (Lambda)</u> of Street Time for Drug Sales, STC, and All Major Crimes Among Each Criminal Variety Type and Heroin Use Type	198
Table XIII. 9	Annualized Criminal Income for Robbery, Burglary, and Property Crimes Among Each Criminal Variety Type and Heroin Use Type	199

\$

and the second second second

				The	Eco
-	Table/	'Grap	<u>oh</u>		
7	Table	XIII	10		Aı aı Ty
٢	ſable	XIII	. 11		Pe Co Va
ŗ	[ab]e	XIV.	1	-	Cł Me
Ţ	[ab]e	XIV.	2		Pe Tj Re
٦	[ab]e	XIV.	3 -	-	Pe ar Ti
٢	[able	XV.	1		Th He
٢	ſable	XV.	2		An Ty Ch
7	[ab]e	XV.	3		Ir Ty
1	[able	XV.	4		Pe Co Ci
. 1	[able	XV.	5		An Ty Wi

conomic Behavior of Street Opiate Users -xxviii-	
Contents	Page
Annualized Criminal Income for Drug Sales, STC, and All Major Crimes Among Each Criminal Variety Type and Heroin	200
Percentage of All Offenses of Each Crime Type Committed During Year by Subjects in Each Criminal Pariety Type	201
haracteristics of Subjects Without and With Some lethadone Treatment	204
Percentage of Criminal Offenders and Heroin Users Types with No Methadone Treatment During Their Reporting Period.	211
Percentage of All Repondents Among Criminal Offender and Heroin User Types by Whether Having Some Methadone Treatment or Not.	212
he Intensive Criminality Typology As Derived from the leroin User Typology by Robbery	227
mong Subjects Classified in the Intensive Criminal ypology, Percentages Exhibiting Certain Background haracteristics and Prior Drug/Criminal Histories.	228
ndices of Social Impact by the Intensive Criminality ypology	229
Percentage of Annualized Volume of Social Impact Committed by Those Classified in the Intensive riminality Typology.	230
mong Subjects Classified in the Intensive Criminal ypology, Percentages Having Self-Reported Contacts with The Criminal Justice System.	231

# Introduction and Objectives -1-CHAPTER I

E. M. M. M. M. S. S. S. S. S. S.

# INTRODUCTION AND OBJECTIVES

This monograph presents findings from a major research study about the economic behavior of street opiate users. In 1976, two factors were associated with the submission of a major proposal to the National Institute on Drug Abuse (NIDA). The Federal government had requested a group of scholars to review and provide recommendations about the linkages between drugs and crime, especially heroin and crime. The report by the Panel on Drug Use and Criminal Behavior (1976) generated considerable controversy (Clayton, 1981) because it reached conclusions that were generally divergent with the prevailing beliefs about the importance of heroin addiction as a cause of crime. Specifically the Panel Report (1976) concluded that "convincing empirical data on drug use and crime ... are generally unavailable - the principal reason being the lack of a long term, well coordinated, policy relevant research program in the area." This panel also generated many questions about the relationship of heroin use to crime and suggested many hypotheses and methodologies which needed exploration. These questions provided a guide for planning the research effort leading to this report.

A second factor in the decision to submit a proposal was the research being conducted by Edward Preble on a NIDA research grant, "The Ethnography of Drug Use Among Two White Ethnic Groups" (ETHNOS) (Preble and Johnson, Preble was collecting fascinating information from street heroin 1978). users in a ghetto community with high levels of opiate addiction. He appeared to be about the only researcher doing so on a systematic basis in 1977.



Thus, the Panel's review of the professional literature pointed to critical questions which needed systematic study and Preble was daily demonstrating that detailed data could be collected from hard-to-reach heroin Accordingly, Principal Investigators Johnson and Preble developed a abusers. research proposal which was funded for two years to demonstrate that systematic data could be obtained. During 1978 and 1979, the research staff explored a variety of methodologies for collecting systematic and quantifiable data from street opiate users about their: criminal activity and income, drug consumption and purchases, income from all sources, expenditures for all purposes, arrests during the reporting period, and involvement in drug or alcohol treatment.

During the two years of pilot research, the staff found that street opiate users could be easily recruited to participate in this research effort when assured that all data would be kept strictly confidential. During the course of a half to one hour interview, they would willingly answer almost any question(s) asked by the interviewer for a payment of \$5 to \$10. They would reveal in great detail the commission of both serious and minor crimes, and about how they managed to subsist with minimal expenditures for food or shelter. One difficulty was developing data collection instruments and codes which could systematically collect data about the complex lifestyles followed by these respondents. Assigning economic values to the noncash (barter) exchanges involving drugs, goods and services which involved no cash expenditures was problematic.

Nevertheless, by the end of the two pilot years, 51 subjects had been interviewed on a daily basis for 30 or more consecutive days. Based upon the success of the pilot years, Johnson and Preble submitted a second proposal in 1979 for a three year research effort to study the Economic Behavior of Street Opiate Users which was funded for three years (1980-1982). This report provides an analysis of the data collected during that period.

Introduction and Objectives -2-

### Introduction and Objectives -3-

ALCONT OF THE SALES

The second s

The rationale and background section of the 1979 proposal identified several major questions and issues which this report can now answer in a systematic fashion. The literature available in 1979 (and most of it in subsequent years - see Gandossy, et al. 1980; Inciardi, 1981; Johnson, 1981) revealed that little or no quantitative data existed which described the social characteristics, criminal activities and income, and the economic behavior of street opiate users when they were actually active on the street. Almost all estimates in the published literature on habit size, crimes committed, or dollar values stolen, were derived from retrospective interviews with addicts in treatment programs, jail, or followup studies of samples from these institutions.

Moreover, no information or data existed about addict economic behavior (total income and expenditures, regardless of the source of income or patterns of expenditures). Although economic supply and demand theories have been used to formulate drug policies, basic economic data about addicts were almost totally absent. From the beginning of this research effort, plans were made to describe the behavior of street opiate users and not to test hypotheses about the linkages of drugs to crime. Thus, this report will be primarily descriptive; it documents empirical regularities rather than test hypotheses.

The following questions were formulated to guide this research -- relevant data are presented in subsequent chapters:

- a) How, if possible, can accurate data on income, expenditures, drug consumption, and criminal activities of street opiate users be measured on a short interval basis?
- b) How accurate and valid are data obtained by various techniques?
- c) How much heroin, cocaine, methadone, alcohol, and other drugs do street opiate users consume? How does the amount of drugs purchased compare to the amount of drugs received as gifts or in-kind payments?



- drugs?
- their time on the street?
- drug use and drug using lifestyle?
- victims of their crime and upon society?
- programs?
- increase in criminal activity?

This last question, when broadened to include all other economic behaviors (drug consumption, income, expenditures, treatment) will provide a guiding focus for this entire report. In chapter IV, all respondents have been classified into typology of heroin users according to the number of days during their reporting period that they used heroin. They are classified as "daily heroin users" if they consumed heroin on 6 or 7 days per week (or over 78% of the time) as "regular heroin users" if they use heroin on 3 - 5 days per week (or 36% to 77%) of the time; and as "irregular heroin users" if they used heroin on 2 or less days per week or under 35% of the time). This classification of respondents according to their frequency of heroin use will be employed in all chapters as a major independent variable.

Introduction and Objectives -4-

d) How much cash income do street opiate users earn from various types of predatory (robbery, theft, burglary) and victimless (prostituting, gambling, drug selling/touting, etc.) crimes?

e) How much income do street opiate users generate and from what sources?

f) How much cash do street opiate users expend upon various items? (food, shelter, clothing, opiates, other drugs, etc.)?

g) What proportion of street opiate user income is spent on heroin and other

h) How many different criminal offenses do street opiate users commit during

i) If street opiate users are classified according to their criminal lifestyle (i.e., major sources of income), how does this relate to their

j) How much are the stolen good(s) and/or illegal services worth to society? That is, how much economic cost do street opiate users have upon on the

k) Previous research has noted that a few criminals commit a disproportionately large share of all crime. How many and what proportion of the street opiate users are such intensive criminals and what social impact do these intensive criminals have on society?

1) What kinds of street opiate users are or are not in drug treatment

m) As heroin consumption becomes more regular and daily, is there an

# Introduction and Objectives -5-

Thus, the focus of this report is upon persons and not upon the process or day-to-day patterns of involvement. That is, respondents will be classified according to their heroin consumption (Chapter IV), their criminal lifestyle (Chapter XIII), and the regularity of their involvement in methadone treatment (Chapter XIV) which will be related to a variety of dependent variables of economic activity such as: the percent of respondents who are active, the number of person-days active, the annual number of offenses or acts, and the annualized cash value.

The central question raised throughout this report is: How different are daily heroin users from the regular and irregular heroin users in their use of nonheroin drugs, criminal activity and income, noncriminal income, nondrug expenditures, drug distribution activities, and in the economic costs and social impact upon victims and the larger society?

The data collected during this research, however, were also designed to permit analysis of the process of addiction and patterns of criminality on a day-by-day basis. Some data from the daily lives of selected subjects have been included as vignettes or brief scenarios to provide ethnographic examples of the behaviors which have been included in the quantitative data analyses. The vignettes are included in several chapter, especially those on drug use and crime (V - VII). Analyses of these processes will be provided in later papers from the Economic Behavior project. In forthcoming papers, the criminal patterns of street opiate users on days with and without heroin use can be examined, and the impact of methadone upon heroin consumption will be analyzed. Likewise, diverse patterns of criminal behavior will be documented.

### Organization of This Report

This report addresses most of the questions above. Chapter II presents the research design and methodology employed in collecting the data. Two lengthy appendices (A & B) provide detailed rationales and descriptions about how methodologies were developed and this research was carried into the field.



The chapter ends with remarks about how all data in the report have been statistically standardized. Chapter III presents information about the backgrounds and other criminal and drug using characteristics of the 201 respondents. The age, sex, and ethnicity of our subjects are also compared with clients in methadone treatment programs.

Chapter IV presents information about these respondent's heroin use levels and develops a Heroin User Typology which is employed as the major independent variable in all subsequent chapters. Chapter V presents findings about the patterns of drug consumption and purchase among these street opiate users. Chapter VI presents detailed information about the various types of nondrug crimes (robbery, burglary, theft, etc.) committed, the percentage of respondents and person days active in each offense type, the number of offenses, and the cash income generated by such crimes. Chapter VII presents information about respondent involvement in crimes involving drug distribution -- particularly drug sales, steering, touting, copping, and drug thefts. Chapter VIII provides annual estimates of the number of criminal offenses and criminal income obtained by these respondents from both nondrug and drug distribution crimes and compares these findings with those of similar studies.

Chapter IX presents information about the respondent's cash income from sources other than crime. Chapter X provides information about cash expenditures for purposes other than drugs. Chapter XI provides evidence about the total income and cash expenditures by these respondents.

By making several conservative assumptions about the actual value of stolen goods, the annualized economic costs imposed by different types of heroin users are presented in Chapter XII.

Introduction and Objectives -6-

### Introduction and Objectives -7-

In Chapter XIII, a typology of criminal behavior is developed and shown to relate strongly to the Heroin User Typology. In Chapter XIV, a typology of legal methadone consumption is developed and related to measures of criminality and heroin use. In Chapter XV, an Intensive Criminality Typology of respondents is developed and related to several measures of social impact including offense rates, criminal income, economic costs, criminal severity scores, and seriousness of involvement with drugs.

Chapter XVI summarizes the findings and outlines some of the policy implications of this research.



In order to answer questions listed in Chapter I, the Economic Behavior Project also had to develop pioneering methodologies and techniques for obtaining systematic and quantifiable information from seriously criminal drug users about their: criminal behavior and income; drug use, purchase, and sale; income from all sources; and expenditures for all purposes. When the project first began, there was initial skepticism that street opiate users and criminals would report details of their illegal activity to outside researchers. Even in 1983, many persons feel strongly that such information could not possibly be obtained.

Nevertheless, this report presents systematic information demonstrating that such data can be collected and analyzed in a coherent fashion. In some respects, the development of research methods for collecting data from street opiate users may be as fascinating as the substantive findings to many readers. A full exposition of our research techniques at this point would greatly detract from the central findings. Thus, this chapter provides only a very brief description of the research design and methodologies used to conduct the study so that the findings in subsequent chapters can be understood and interpreted clearly.

In addition to this chapter, over 90 pages of appendices (A & B) provide further details about how methodologies were developed for collecting detailed data on a day-by-day basis about the economic behavior of street opiate users. In fact, the current chapter is a condensed version (almost an abstract) of the major points more extensively documented in Appendix A. "A Methodological History." In addition, Appendix B, "Taking Care of [Research] Business" provides an ethnographic account of how this research project was carried into the field and describes staff experiences in dealing with street opiate user subjects and exaddict fieldworkers in a low income neighborhood.

BDJ:8256A;0740A;bj

「「「「「「「「」」」」」」

# Research Design and Methodology -8-CHAPTER II RESEARCH DESIGN AND METHODOLOGY

# Research Design and Methodology -9-A Brief Overview of the Economic Behavior Project

The main methodological objective was to obtain detailed descriptive information about street opiate users and their routine criminal behavior, drug use, purchase, and sale, income from all sources, and expenditures for all purposes. In order to obtain such data, staff selected two low income communities (East and Central Harlem) in New York City where heroin use has been endemic for decades. Research storefronts were located in these communities and exaddict, exoffender field workers employed to recruit criminally active street opiate users from different neighborhoods in the study area; they brought potential subjects back to the storefront for interviewing.

Respondents were interviewed about their economic behavior for five consecutive days, and then interviewed for four consecutive weeks about the seven preceding days, providing 33 days of data. About a third of the respondents returned for subsequent cycles (four weekly interviews covering 28 days) between three and six months after the first. Subjects were also interviewed about their history of criminality and drug abuse prior to this research.

The data were coded, entered, and computerized -- providing the data for this report. Although 238 subjects were recruited, 201 subjects (84%) completed the first interview cycle and provided 33 days or more person-days of data. The characteristics of these subjects are described in Chapter III.

### A. Street Opiate Users and Street Criminals

This research was originally (1977 application) designed to study street But during the pilot years (1978-79), staff encountered opiate addicts. many subjects whose physical dependence upon heroin or opiates could not be Surprisingly, many subjects reported several days without ascertained. heroin or opiate use, and consumed widely varied amounts of opiates on successive days (Johnson, 1979, 1983).



Nevertheless, if abuse is roughly equated with injecting drugs into a vein ("mainlining"), most subjects continued to abuse heroin, other opiates, or cocaine several times a month. They also reported long histories of heroin abuse and living in the streets. They were also observed living a lifestyle consistent with that of the public image of "addicts." Recognizing that "addict" is a chameleon concept (Johnson, 1978), the research staff began using the term "street opiate users" to describe our respondents.

Rather than try to impose a conceptual (or academic) definition upon respondents (and exclude those who did not apply), staff relied mainly upon field workers to recruit respondents who exhibited a variety of behaviors, some of which are specified below. Our main objective was to ensure that subjects exhibiting a variety of opiate using and criminal lifestyles were recruited. Nevertheless, the idea of "street opiate users" contains three major ideas.

Opiate use refers to the consumption of heroin and/or methadone. All subjects reported the use of one or both of these drugs during their reporting periods. In addition, although we did not ask direct questions about route of administration, almost all subjects appeared to have mainlined heroin and/or cocaine for several years. Recruiters were told to direct their attention towards finding persons who were currently injecting heroin, and to underrecruit methadone clients who were mainly alcoholics (this was based upon experience during the pilot years--see Appendix A).

These subjects were mainly on the streets. During most of the daytime and frequently late into the night, these subjects reported being physically on the streets or in other illegal locations (shooting galleries, after-hours clubs, apartments from which drugs were sold). They spend relatively little time at home, at work, or in other conventional pursuits. Although such persons form a small proportion of the total population, they are a major visible presence in certain "copping communities" (Hughes, 1977) in the study area.

Research Design and Methodology -10-

## Research Design and Methodology -11-

A third major criteria for recruitment was the field worker's opinion that the subject was engaged in some form of criminality. While emphasis was placed upon locating those committing robbery, burglary, and theft, several street drug dealers were also recruited. Field workers were asked to locate persons who exhibited a range of criminal behaviors. This they were able to do, but with some limitations.

The criteria of selecting "street opiate users" who were mainly involved in street crime had the effective result of excluding many persons who may have one of these criteria but not others. For example, persons who snorted heroin or cocaine, but did not generally mainline drugs, would generally have not been recruited. While many subjects were recruited from loiterers around methadone programs, field staff accepted only a few methadone alcoholics. Moreover, methadone clients who were employed, or were otherwise conventional, or who avoided hanging out on the streets were unlikely to be recruited. Hercin abusers who were "house connections" (sold drugs from apartments) but did not hang out on the streets were also unlikely to be recruited. High level dealers and importers were not located.

Nonheroin users who committed crimes were also unlikely to be recruited. That is, professional fences, loan sharks, marijuana dealers, numbers runners, bookies, safecrackers, truck hijackers, "hit men," members of organized crime, etc. were not recruited if they did not inject heroin or cocaine (although some of our subjects were occasionally involved in such activities).

In short, the recruitment procedures assured that street opiate users who engaged in street crime were effectively recruited; while persons lacking such characteristics, although engaging in opiate use or some forms of crime were effectively excluded. As a partial result of our recruitment techniques, our subjects are among the most criminal ever recruited and also exhibit varied patterns of heroin and other drug use, as the data in following chapters show.



There were three major reasons for selecting East and Central Harlem in Manhattan as the location for this research. First, these two communities have among the highests levels of opiate use in the country; many heroin users reside there or come to these neighborhoods to obtain drugs. Second, many of these opiate users spend most of their time on the streets and are available as potiential subjects. Third, co-principal investigator Edward Preble had many years of experience with street opiate users in East Harlem and had been able to recruit them for previous research activities. Thus, gaining access to and gaining the trust of heroin users in these communities was easily accomplished in a short time.

East Harlem, referred to as "Spanish Harlem" or "El Barrio," is the area from Fifth Avenue to the East River north of 96th Street. Demographically, the population is about 44% black (1977 data); the remainder is mainly of Hispanic origin, although a few whites also live in the area. East Harlem is generally high on several indicators of social disorganization.

Central Harlem includes the area from 5th Avenue to Morningside Avenue and St. Nicholas (on the West) from 110th to 135th Street. Over 95% of the population is black. Central Harlem's reputation throughout the country is that of the heroin capital of the U.S.A., if not the world. Almost all problems of ghetto life affect Harlem, but public and widespread heroin distribution and use has been a continuing problem in this community since the end of World War II. Several decades of enforcement efforts have been unable to prevent widespread street sales. East and Central Harlem have perhaps the largest number of street opiate users in the country.\*

The Lower East Side of Manhattan rivals these communities in the probable number of street opiate users and has drug "supermarkets" (where touts openly and aggressively solicit customers wishing to buy heroin, cocaine, marijuana and other drugs) rivaling those of Central and East Harlem. Moreover, drug dealers from white communities and the entire New York metropolitan area frequently "cop" (buy) high quality drugs at locations in the Lower East Side -- this also occurs in Harlem and East Harlem.

Research Design and Methodology -12-B. The Research Location -- East and Central Harlem

Research Design and Methodology -13-Graph II. 1 Methadone Clients in Northern Manhattan by Zip Code of Residence in 1979



6



Research Design and Methodology -14-One of the better indicators of the large numbers of opiate users in this community comes from the New York methadone central registry. Data presented in Figure II. 1 shows that the southern East Harlem ZIP code (10029) has more currently active MMTP clients than any of the nearby ZIP code zones and that Central and East Harlem Zip Codes (10026,10027,10030,10035) have many more MMTP clients than other ZIP codes. Almost 1,600 clients (or almost 6% of all MMTP clients in New York City) reside within the two East Harlem ZIP codes (10029, 10035), another 2,500 (or almost 9%) reside in Central Harlem ZIP codes (10026, 10027, 10030, 10037). Assuming that one former methadone client and one heroin injector who has never been in treatment exist for each current methadone client, the total street opiate user population may be 4,000 - 5,000 or more in East Harlem, with an equal or larger number in Central Harlem. The figures may actually be considerably higher.

C. The Storefront as a Place for Routine Data Collection

While different potential techniques for obtaining routine economic behavior data from respondents were tried in 1978-79, formal interviews at a field office or "storefront" proved to be a very efficient way of obtaining high quality economic behavior data from respondents. Other techniques were tried but found difficult to implement (See Appendix A). By locating the research staff in a field office that blends naturally into the study neighborhood, the subjects felt comfortable.

The major difficulties with using a storefront as an interview site were: 1) the poor condition of most rental space available, 2) congestion of respondents wanting to be interviewed during the morning hours, and 3) problems associated with staffing and supervision of the nonprofessional staff. These difficulties are specified in more detail in Appendix B. Nevertheless, the advantages of a neighborhood storefront for conducting a confidential and structured interview to obtain detailed information from street opiate users vastly outweighed the disadvantages.

# Research Design and Methodology -15-

D. Recruitment of Subjects

AND THE ALL AND A SHE AND A SHE

が主

A STORE

10 10

1

Locating research subjects was among the easiest tasks which this research faced, especially since respondent payments were available. As Preble (1980a) indicates, the need for money was continuous among street opiate users. What initially appeared to be a mean motive, the need for money, could be transformed into a fruitful research relationship by following careful procedures to protect confidentiality and by skillful rapport building and interviewing. The respondent quickly comes to provide high quality information about his activity in return for a modest payment (\$5 or \$10) for 30 minutes to 2 hours of his time.

During the main data collection period, 1980-1982, the research staff relied heavily upon exaddict/exoffender staff who were sent into the streets to locate subjects. These field workers approached unknown persons on the street, but more frequently found a previous acquaintance who informally introduced them to other potential respondents. In order to ensure a diversity of lifestyles among respondents, these staff were directed to go to different blocks in the study community with directions to avoid bringing in too many "low lifes," and to concentrate on finding heroin users who did robbery, burglary, frequent thefts, and drug sales, as well as to bring in about one female for every two or three males (our estimate of the probable sex ratio).

Although efforts were made to develop a quasi-scientific sampling frame permitting respondent selection with a known probability of selection, many problems were encountered and this effort was dropped (see Appendix A). Thus, this study does not and cannot generalize to a broader population of street opiate users.\* On the other hand, little evidence exists that our subjects are greatly different than other heroin users; their sex and ethnic distributions appear similar to clients in methadone programs (see Chapter III).

Hunt, et al. (1983) conducted a contemporary study of opiate users in and around methadone clinics in the Bronx, New Jersey, and Connecticut. Their subjects were frequently white and had stable employment.



In order to maintain initial and long-term contact with respondents, the initial week of interviewing was critical (see Appendix A for rationale). Short interviews with respondents on five successive days were important for building rapport. The interviewer and respondent developed a sense of trust, respect, and liking for one another. Respondents learned what information was being sought and paid somewhat more attention to the details of their income, expenditures, crime, and drug use. Thus, most of the persons completing five daily interviews completed 33 consecutive days of interviews, and many were found at a later time for additional cycles of interviews. Clearly, the initial effort at building rapport established a firm research relationship which paid major dividends (valid and reliable data, consistent future reporting, and ease of maintaining a long term relationship) with the street opiate users being studied in this project.

E. Development of Instrumentation

A major challenge facing this research was the development of instruments (interview schedules and coding schemes) by which the complexity of respondent behavior could be captured and measured in a systematic quantitative fashion. At the beginning of this research, the investigators decided to obtain detailed data only about common crimes where an economic value (i.e. money, drugs, or goods) was possible. All other crimes which could not provide an economic benefit to the respondent were excluded. Thus. respondents were not asked questions about assault (aggravated and simple), rape, homicide, arson, weapons possession, vandalism, disorderly conduct, property damage, etc. (see Appendix A for rationale).

Likewise, obtaining details about specific criminal or drug use episodes as units of analysis was not possible. Rather, the respondent would be asked to sum across all crime and drug-related episodes of a given kind for a single

Research Design and Methodology -16-

# Research Design and Methodology -17-

THE REAL PROPERTY AND A DESCRIPTION OF A

day, and report the total dollar income. This means that the smallest unit of analysis is the person-day (see below). Thus, very rich detail about crime events (and we had some very interesting crimes reported) would not be systematically obtained for each event. Rather, the interviewer was asked to write a brief scenario about the crime on the back side (the 'flip side') of the interview form.

Interviews with respondents were focused upon their economic behavior during a particular 24 hour period, called the person-day. Staff developed both a daily interview schedule (used during 1978-79), and a weekly interview schedule (used for the main study, 1980-82) to obtain detailed quantitative data about the crimes, drug use/purchase/sale, income, and expenditures during each reporting person-day. A variety of instruments were pilot tested and revised during 1978-79 (and are described in Appendix A).

Most of the information analyzed in subsequent chapters, however, comes from a weekly interview schedule that is presented at the end of this chapter. This weekly interview collected data about seven person-days of data in one interview. It took a half hour to one hour to complete, depending upon how active the respondent was and his ability to recall his economic behavior during the past seven days. This interview schedule was highly cost effective (\$10 per interview and about one hour of interviewer time).

During the main study (1980-82), each new subject gave his informed consent, and then was interviewed for five consecutive days (data were recorded on the weekly form), then the subject was asked to report on a weekly basis for the next four weeks. This generated 33 consecutive person-days of data (the first cycle) which was critical for calculating and annualizing the rates of criminal offending for each subject. The first cycle with 132 East Harlem subjects was conducted mainly in 1980-81, while the 69 Central Harlem subjects were interviewed mainly in 1981-82.



Among the East Harlem subjects, efforts were also made to conduct additional cycles of data collection in order to approximate a longitudinal design. Among the subjects recruited in 1980-81, respondents were asked to return every three to six months for additional cycles of four weekly interviews covering 28 consecutive days. In total, 201 subjects provided 11,417 person-days of data (an average of almost 57 person-days per subject). In order to obtain systematic information about the respondent's background, project staff also developed a relatively lengthy open-ended life history interview (LHI) schedule which obtained information about demographic characteristics (sex, ethnicity, age, marital status, education, etc.), self reported involvement in a variety of crimes and drug use (including age of initiation), and prior treatment and arrest/incarceration histories. In addition, a special set of questions asked subjects to estimate their prior year's income and expenditures. For a variety of reasons described in Appendix A, however, several respondents did not complete this LHI schedule

and a shortened version of it was administered to about 40 respondents.

A frequently asked question is: how can you trust what they (respondents) tell you? This question addresses the long standing problem of the validity and reliability of self-report data. Other researchers have carefully assessed the validity and reliability of self reported criminality and drug use; they almost always conclude that such self-reports are considerably better than any other source of information. When assured of confidentiality, street opiate users provided answers about their criminality that was generally better than police records, or most other source of external validation. This is particularly true when the researcher wishes to find out about "successful" crimes (those committed but no arrest or police contact occurs) (also see Ball, 1976; Chaiken and Chaiken, 1982; Wish. et al.. 1983; Hindelang, et al., 1982).

Research Design and Methodology -18-

### F. Obtaining Valid and Reliable Economic Behavior Data

# Research Design and Methodology -19-

Key Company and a set of the set of

Three major kinds of evidence suggest that the data obtained are generally valid and reliable. First, internal consistency checks were systematically built into the interview instrument, and respondents were asked to correct discrepancies which emerged. Second, by interviewing individuals on several different occasions during an extended time period, subjects re-reported similar types of crimes and drug consumption, although their accounts exhibit important variation about the details of specific events and the dollar returns. These points are delineated in more detail in Appendix A. Third, the data have high face validity because field workers and professional staff frequently observed respondents engaging in very kinds of behaviors that they reported in their interviews about their criminal activity.

One respondent who consistently reported stealing parts from cars was observed walking down the street with a car bumper on his shoulder.

Another subject claimed to have committed a burglary the previous day in which he obtained \$600 and bought \$200 worth of drugs; he shows the field worker \$300.

Another subject who claims to have robbed a gun-runner was observed carrying four 32 magnum pistols in a shopping bag.

A respondent who reported serving as a tout on the streets was observed day after day talking to anybody who looks like an addict while trying to drum up business for a local dealer; he was observed making sales between a customer and dealer who never meet (served as a "cop man").

Four persons who sold drugs from an apartment were observed in their apartment for several days; they made 15-30 transactions per day in front of research staff.

Both subjects and other neighborhood residents wandered into the storefront and offered to sell stolen merchandise to other subjects, field workers, and research staff (see Appendix B).

These observations could be extended many times from the personal experiences of field workers and professional research staff. Thus, such observations suggest that the data analyzed in this report are sufficiently reliable and valid to provide new insights about the economic behavior of street opiate users. While such evidence does not constitute statistically documented evidence of reliability and validity, such observations may be more convincing indicators of validity than comparison with arrest information.



# Standardization of Data

In the following chapters, for each major kind of activity (crime, drug dependent variables are the percentage of respondents involved, the percentage of person-days active, the rates of involvement, and the dollar amounts involved. Since our respondents had different numbers of person-days during which they were interviewed, all data have been statistically standardized so that each respondent contributes the same number of person-days to data presented in the tables. This was accomplished by computing (for example) the mean number of burglaries per day (or dollars/per day from burglary) for each respondent and then multiplying by 100 (for 100 person-days) or 365 (for an annualized rate). Detailed information about how the data was statistically standardized and statistical tests of variance among group means are

use/purchase, noncriminal income, nondrug expenditures, etc.), the major presented at the end of Chapter VI.

Since this report has been written to address several audiences (professional researchers, policy makers, lay readers, etc.), the main questions of substantive interest are directly stated and the central findings and data are clearly stated in the body of the chapter. Detailed tables and important methodological and definitional issues are attached at the end of each chapter.

### Summary

The Economic Behavior Project developed several important methodologies for obtaining daily data from street opiate users regarding their crimes and criminal returns, their use, purchase, and sale of various drugs, their income from all sources, their expenditures for all purposes, their involvement in treatment or arrest. Each respondent provided 33 or more consecutive person-days of data (i.e. they completed the first cycle). About a third of the subjects provided additional cycles (four weekly interviews or 28 person-days). Their self-reports about their drug use and criminal behavior appear to be quite reliable; the face validity also appears high as

Research Design and Methodology -20-

Research Design and Methodology -22-ECONOMIC BEHAVIOR STUDY WEEKLY/DAILY DATA COLLECTION FORM L. Subject Name 2. Subject Number 3. Date of Interview 5. Interviewer (initials) . I. Old you earn any money at all over the past seven days from legitimate employment? YES NO If YES: Day of Heek Date 10. Any Work (YES or NO) II. Job Type , 12. Earnings II. Did you earn any money at all over the past seven days from any kind of criminal activity? YES NO. If YES: (Enter cash earnings above line; number of victimizations below.) 13. Shoplifting(resale) YES NO 14. Shoplifting(own use)YES NO 15. Burglary YES NO YES NO 15. Robbery 17. Forgery YES NO 18. Prostitution YES NO 19. Pimping \_ YES NO 20. Con Games YES NO 21. Other Theft YES NO YES NO 22. Other

Research Design and Methodology -21-

and the second second

Street opiate users were recruited from different street locations in East and Central Harlem areas of New York City. These respondents were not recruited as part of a scientific sample, hence, the findings cannot be generalized to a larger population of street opiate users.

Unlike almost all previous research about drug use and crime, the Economic Behavior Project obtained detailed information from subjects within a day to a week after the behaviors occurred. Almost all the data about their criminality involved "successful" crimes in which the respondent obtained money, drugs, or goods without an arrest or any contact with police. They typically obtained, purchased, or sold drugs, especially heroin and cocaine, with minimual difficulty.

For the first time, high quality data is available from street opiate users about their criminal activities and drug consumption on a day-by day basis. From such reports of daily behavior, the questions listed in Chapter I can be systematically addressed in subsequent chapters. The demographic and other characteristics of our 201 subjects is provided in Chapter 3; our subjects are shown not to differ greatly from methadone clients in these communities. These subjects represent a wide range of criminal behaviors and frequencies (Chapters VI - VIII, XIII) and different levels of drug consumption (Chapters IV & V), but similar patterns of noncriminal income and nondrug expenditures (Chapter IX - XI) and economic costs (Chapter XII).

Three-quarters of these subjects were not in methadone treatment and appeared to be avoiding it during the reporting period (Chapter XIV). Daily heroin users and those who engaged in robbery or burglary had much more substantial social costs and criminal impact than their less criminal or nondaily heroin user counterparts (Chapter XII & XV). All of these findings have important policy implications which are delineated in Chapter XVI.

In the following chapter, the background characteristics of our respondents is presented; they are compared with persons currently in methadone treatment.

6.	Editor
 7.	Coder
8.	Type of Interview







(2)

AND THE REAL PROPERTY AND A DESCRIPTION OF A DESCRIPTIONO

III. Did you do any drug dealing over the past seven days?

YES NO



5

# Day of Week

Date

- 23. Cash Earnings (amount and type)
- 25. Number of Transactions (amount and type)
- 27. Principal Drug Earned (amount and type)
- 28. Secondary Drug Earned (amount and type)
- 29. Principal Drug Dealt (amount and type)
- 30. Secondary Drug Dealt (amount and type)
- IV. Did you do any steering, touting, or copping for other people over the past seven days? YES NO
  - If YES: (Note: Fill Out same way as above)

### STEERING, TOUTING, COPPING

- 31. Cash Earnings (amount and type)
- 33. Number of Transactions (amount and type)
- 35. Principal Drug Earned (amount and type)
- 36. Secondary Drug Earned (amount and type)
- 37. Principal Drug Dealt (amount and type)
- 38. Secondary Drug Dealt (amount and type)





۷.	Did you purch	ase an	d/o <del>r</del> u	ise a
	YES NO			
	If YES, enter	anoun	t used	abo
	Day of Week			-
	Date			•
			1	1
40.	Heroin	YES	NO	
41.	Other Opfates	YES	NO	
42.	Methadone (illicit)	YES	NO	2
43.	Cocaine	YES	NO	
44.	Amphetamines	YES	NO	
45.	Barbiturates	YES	NO	
45.	Marijuana	YES	NO	
47.	Alcahal	YES	NO	F
48.	Tranquilizers	YES	NO	
49.	Psychedelics	YES	NO	
50.	Other	YES	NO	
VI.	Oid you steal YES NO	any da	rugs a	ver
	A. Date drugs	stole	1	
	8. Amount and	type	ຫີ ດານ	ds s
	C. Were drugs	used,	resol	d, o



Research Design and Methodology -24-

(3)

any drugs in the past seven days?

we line and amount purchased below line.



the past seven days?

2 3 4 5 6 7

Research De	sign and	Methodology	-25-
-------------	----------	-------------	------

O THE STATE OF THE

S

(4)

VII. Were you a patient of a MMTP program over the past seven days?

YES NO							
If YES:			•				
Day of Week		<del>ساييونېس</del> .				<del>متابا البيسيني</del> .	
Date	-						
		2	3	4	5	6	7
54. Regular Methadone Dosage Received (Mgs.)							
55. Take Home Oosage Received (Mgs.)							
56. Amount Used (Mgs.)							
57. Amount Resold (Mgs.)							}
58. Amount Given Away (Mgs.)							

VIII. Oid you receive any other type of treatment for drug use since your last interview?

YES NO

1 G

 $\sim$ 

공

If YES, write a detailed account and indicate what type of treatment, e.g., T.C., Private Doctor, Narcotic Antagonist, Detox, and dates of treatment.





IX.	Were you arro YES NO If YES, write in jail, disp	ested e a de posit	since etailed ion etc
60.	For Coding U: Date of Arre	se On	1 <u>y</u>
<b>61</b> .	f of Days in		
62.	Charges	•	******
			***
Χ.	Did you rece YES NO If YES, enter below line. Day of Week Date	ive a	ny in-k h value
64.	Raom	YES	NO
, 65.	Meals	YES	NO
66.	Cigarettes	YES	Ю
67.	Orugs	YES	Ю
68.	Alcohol	YES	Ю
69.	Other	YES	Ю

Research Design and Methodology -26-

(5)

your last interview?

account of the arrest and include dates, changes, time spent

kind income over the past seven days?

above line, source of in-kind, e.g., friend, mother, fence, etc.



Research Design and Methodology -27-(6)

XI. Oid you earn any money from the following sources over the past seven days?

**CREAS** 

-----

	YES NO									
	If YES:						٠			
	Day of Week									
	Date	•		-		<u>مرد نکور نم</u>				
			,	1	2	3	4	5	<u>ه</u>	7
71.	Working	YES	NO	· .		, , ,				
72.	Crime	YES	NO							
73.	Orug Business	YES	ŅО							
74.	Family	YES	NO							
75.	Spouse/ Paramour	YES	NO							
76.	Welfare	YES	NO							
77.	Unemp loyment	YES	NO							
78.	Other Public Support	YES	NO							1
79	Friends	YES	NO							
80.	Panhandling	YES	NO							
81.	Gambling	YES	NO							
82.	Respondent Payment	YES	Ю							
83.	Other	YES	NO							

an and a second s -YES NO If YES: Day of Week Date 87. Living (rent, G.E. etc.) 88. Food 89. Family 90. Legal Fees 91. Clothes 92. Recreation 93. Drugs 94. Alcohol 95. Gambling 96. Cigarettes 97. Savings 98. Transportation YES NO 99. Other 9. Other Criminal Activity 84. Other Drugs

YES NO

# Research Design and Methodology -28-

(7)

XII. Did you spend any money over the past seven days?

	`	 -				
<u> </u>			-	5	0	· · · · · ·
		•				
				l 		
		<u> </u>				
	ł					

<b>(</b>		 1	1	
l l				
	 		<u> </u>	

Characteristics of Respondents -29-CHAPTER III CHARACTERISTICS OF RESPONDENTS

This chapter provides descriptive information about the 201 respondents participating in this study who provided at least 33 person days of data. Some information is also given about an additional 37 respondents who did not provide 33 days of data and who have been excluded from the analysis in this report.

As discussed in the methodology chapter (II), this project undertook efforts to develop a strategy for sampling of street opiate users in East Harlem. Unfortunately, no techniques were found to be successful in sampling respondents at a known probability (see Appendix A). Thus, direct recruitment of subjects from different street locations and snowball sampling was employed. While efforts were made to locate subjects who had diverse lifestyles (and the data below reflect this diversity), this project cannot make direct statements about how representative such respondents are of all street opiate users in East Harlem or Central Harlem.

Nevertheless, this chapter presents their demographic characteristics and other background factors and compares them with clients in several methadone programs in New York City. This comparison suggests that the sex and age of these 201 subjects were quite similar to their counterparts in local treatment programs. This similarity suggests that the behavioral patterns which are documented in more detail in subsequent chapters are likely to be similar to street opiate users not studied but living in the study communities.

The research design involved collecting data from two different samples of respondents -- in East Harlem\* and in Central Harlem.

<sup>\*</sup>Among the East Harlem subjects, efforts were made to maintain longitudinal contact; many of these respondents were reinterviewed on a quarterly or semiannual basis (This design effect is not considered in this report -- it will become more important in subsequent papers).



The original criteria for calling a person a "respondent" was that the respondent who had been recruited from the street must have received the project's informed consent procedures, agreed to participate in the study and have completed at least one interview (usually the first daily interview). A total of 238 persons met this criteria. The authors of this report planned to calculate annual offense rates and criminal income as well as other drug and economic variables; such calculations needed a stable number of person-days which was available when the subject completed one full cycle of interviews. Thus, staff decided to include only the 201 subjects who provided 33 person-days or more of economic behavior data. [Table III. ] about here. ]

This meant that 37 respondents completed at least one interview but did not provide 33 days of usable data. The reasons and number of person-days completed are given in Table III. 1. Among the the 37 excluded subjects, 23 either completed five daily interviews or less; only six persons provided over 22 days of data.

While about a quarter of these were removed by staff, death, or jail, the main reason for noncompletion of 33 person-days was that these respondents did not return for the next interview. Their termination usually occurred after completion of the five daily interviews, although several persons stopped after completing one or more weekly interviews. While efforts were made to locate them on the streets, the unstable living arrangements and high mobility of such respondents made followup efforts frustrating and unfruitful.

Characteristics of Respondents -30-

Characteristics of Respondents -31-Table III. 1 -- Number of Subjects and Days of Reporting and Reasons for

## Noncompletion

	Number of Subjects
Gave informed consent and completed one interview:	238
Provided 33 or more person-days of data:	201
Provided less data:	37
1 – 2 days of data:	6
<pre>3-5 days of data (completed daily interviews):</pre>	17
6 - 21 days of data (completed daily and about two weekly interview):	8
22-32 days of data:	6
Reasons for noncomple	tion:
Removed by Staff (not eligible, crime against staff)	5
Died	1
Jail/prison, hospitalized:	3
Known to have moved out of area:	0
Could not be relocated:	7
Contacted, refused to come back	1
Did not return, reasons unknown	20



Given that many street opiate users evade conflict and other obligations by disappearing, these 37 respondents were generally following well known patterns of behavior. Nonetheless, of the 77 Central Harlem subjects participating, 69 (or 90%) completed 33 or more days of reporting. Likewise, of 161 East Harlem subjects participating, 132 (82%) completed 33 or more days.

Thus, 201 (or 84.5%) of the 238 who gave their informed consent reported to the storefront on at least 9 different occasions (5 daily and 4 weekly) and completed interviews about their criminal and economic behavior during 33 or more days. This is a high completion rate given the kinds of respondents recruited and the number of different occasions (9 or more) they had to be interviewed in order to qualify for inclusion in this study.

# Characteristics of 201 Respondents

The demographic and other characteristics of these 201 subjects are given in Table III. 2. The data is presented for subjects from both Central and East Harlem samples and for all subjects combined.

Three-quarters of the respondents were males in both samples. Only a third of the East Harlem subjects were black, while almost all Central Harlem subjects were black. Overall, these ethnic distributions appeared to be a reasonable reflection of the ethnic composition of these two communities. Thus, slightly over half (55%) of the 201 subjects were black, 44% were Hispanic, and only 1% were white.

The East Harlem subjects appeared somewhat younger than the Central Harlem respondents, although the differences were not statistically significant. That is, about 48% of of the East Harlem subjects versus 25% of the Central Harlem respondents were age 30 and younger. Thus, 40% of the 201 subjects were age 30 and under.

Characteristics of Respondents -32-

[Table III. 2 about here.]

Characteristics of Respondents -33-

Table III. 2 -- Characteristics of Respondents in the Economic Behavior

A DESCRIPTION OF A

TANK MANYA MANYA TA

Project

		East	Central	A11	Metha	done Pro	grams:a	
	Number of	Harlem	Harlem	Harlem	ARTCP	Mt. Sinai	Beth Israel	
	Subjects/Clients	(132)	(69)	(201)	(1,268)	(612)	(6,829)	
Sex:	Male Female	75 25	75 25	75 25	68 32	67 33	72 28	•
Ethnici	ty: Black Hispanic White/Other	34 64 1	92 6 2	55 44 1	66 27 4	35 46 18	39 23 36	
Age at	end of 1981:							÷.
	Under 25 25 - 30 31 - 35	17 31 (22	6 19 (26	13 27 (23	8 33	7 36	4 25	
	36 - 40 51 41 and over Unknown	{ 14 15 1	69 { 19 ( 24 6	56 { 15 ( 18 3	59	56	71	
High Sc	hool Dropout-%	64	61	63				
Ever In	carcerated - %	58	72	62			r Marine	
Ever Ar	rested - %	80	93	84				u
Median	number of arrests:	3	4	3				
Median sin	number of years nce first heroin use	: 9	11	10				
Percent Her	: Claiming Daily oin Use in Past Yea	r 64	61	63				
Princip Sup	al Means of porting Drug Use Theft Drug Business Work	41 23 18	49 22 10	44 23 16				
	Family, Fublic Support, Other	17	20	18			and the second	

a - Division of Substance Abuse Services Management Information System on Client Characteristics of Methadone Maintenance Treatment Programs as of 3/31/82.

b - Addiction Research Treatment Corporation.

Almost two-thirds of all respondents were high school dropouts. While Central Harlem subjects were somewhat more likely to have been incarcerated and arrested, the differences are not substantial. Likewise, the median number of arrests and years since first heroin use were virtually identical. Over three-fifths of both groups reported that they used heroin daily during the year prior to participation in the study (this claim of daily heroin use in the previous year was about twice as high as the percentage who used heroin on 6-7 days per week during the reporting period -- see Chapter IV). Theft and drug business were the major ways of supporting drug consumption in both groups. Work appeared to be a more important source for supporting drug use in the subject's recollections of the prior year than was actual employment patterns during the data collection phase (see Chapter IX).

The clear conclusion of these data is that the Central Harlem and East Harlem subjects did not different from each other in most respects (other than ethnic composition). In addition, data not presented here show that employment history, marital status, education do not vary by community of recruitment. Likewise, during the year prior to recruitment, Central and East Harlem subjects report almost identical levels of use or addiction to heroin, cocaine, alcohol, and other drugs; the frequency of and income from burglary, robbery, shoplifting; total criminal income; and expenditures for drugs.

Thus, in future chapters, the information from the East Harlem and Central Harlem subjects have been combined and are analyzed together, with no distinction between the sample groups.

Characteristics of Respondents -34-

# Characteristics of Respondents -35-Comparison with Opiate Users in Methadone Programs

Table III. 2 also presents information taken from the New York State Division of Substance Abuse Service's (DSAS) management information system which continuously updates characteristics of clients in treatment. Three major programs with several methadone clinics in the East and Central Harlem areas are presented. Mt. Sinai methadone program is located in a hospital in East Harlem and serves clients from both the immediate community and from other areas of Manhattan. The Addiction Research Treatment Corporation (ARTC) has three clinics in the Harlem community as well as more clinics in black communities in Brooklyn. Beth Israel is the largest centrally administered methadone program in New York City. It has six clinics in East and Central Harlem, and an additional sixteen clinics elsewhere in Manhattan and the boroughs.

A comparison of the demographic characteristics of clients in these methadone programs clinics with the 201 subjects in this research project show the following. A somewhat higher proportion of females are in ARTC and Mt. Sinai (about a third) than among the Economic Behavior subjects (a quarter) and Beth Israel clients. Likewise, the Economic Behavior subjects have relatively lower proportion of whites (1%) than these methadone programs although ARTC has relatively few whites (4%). Mt. Sinai and Beth Israel have much higher proportions of whites because they enroll clients from all over the city.

Mt. Sinai has about the same proportion of blacks (35%) as among the Economic Behavior subjects recruited from East Harlem (34%), but relatively Actually the ethnic composition of clinics within the Beth fewer Hispanics. Israel and ARTC programs vary greatly according to the neighborhood in which they are located.

The age distribution of clients in the three programs, however, are very similar to each other and to the Economic Behavior clients. Approximately



Characteristics of Respondents -36two-fifths of clients at ARTC and Mt. Sinai and Economic Behavior subjects are age 30 and under. Beth Israel clients appear to be somewhat older; 71% are age 31 and older. If anything, the subjects recruited from East Harlem appear to be younger (48% are 30 and younger) than clients in MMTPs, while the Central Harlem subjects appear to be about as old as the Beth Israel clients.

### Summary

Subjects recruited for the Economic Behavior project (and whose behavior are analyzed in the following chapters) were very similar to opiate users enrolled in methadone treatment programs in terms of their sex and age distributions. The ethnic distributions are not as well balanced, but do appear to reflect the ethnic composition of the two study communities.

Given the lifestyles of the street opiate users recruited and the fact that they needed to be interviewed on nine or more different occasions, the proportion (84%) of subjects included in the analysis below is very high. The 201 subjects exhibit high levels of maladjustment and deviance in that over three-fifths were high school dropouts, and most had prior arrests and incarcerations. They claimed almost ten years of heroin use, and most claimed to be daily heroin users.

Data to be presented in subsequent chapters will show high levels of criminality and drug use and very low levels of legitimate income or public support. These respondents appear to be among some of the most criminally active ever studied (see Chapter VIII) in a research effort.

The analysis of their economic behavior can now begin by examining their heroin using behavior; a major product is the development of a major typology of respondents according to the regularity of their heroin use.

BDJ;8256A;0740A;bj

A Heroin User Typology -37-CHAPTER IV DEVELOPING A HEROIN USER TYPOLOGY

This report is designed to provide basic insights about the economic behavior of street level opiate users -- specifically their nondrug criminal activity and drug distribution crimes, drug use and purchase, income from all sources and expenditures for all purposes. Detailed information about these topics are explored in more detail in the following chapters.

The current chapter provides data about one critical dimension of opiate use -- the frequency with which subjects used heroin. In addition, a Heroin User Typology is developed and employed as a major independent variable measuring the frequency of heroin use in the analyses which follow.

At the initial stages of analysis, a variable is needed that is relatively simple to understand, clear in meaning, and measures a central dimension of the lifestyles of street opiate users. As will become clear in this and later chapters, the subjects in this study generally used either heroin or methadone (both legal and illicit). Only on rare occasions did they consume other opiates (demerol, morphine, raw opium, etc.). Since heroin was much more common than other opiates (see Chapter V), the central drug of analytic interest is heroin.

Moreover, previous research has demonstrated that classifying respondents according to their frequency of heroin use provides meaningful distributions and relatively strong associations with other variables, especially criminal behavior. Studies based upon retrospective reports of heroin use by respondents (Ball, et al., 1979,1981,1982,1983; Nurco, 1981abc, McGlothlin, et al., 1977, 1978; Chaiken and Chaiken, 1982ab; Johnson, 1978) show that daily or near daily users of heroin, when compared with less regular heroin users, generally report more serious types of crimes, have more crime activity, and obtain larger dollar amounts of criminal income; the daily heroin user may also be more active in a variety of other behaviors as well.



A Heroin User Typology -38-Unlike previous studies which have relied entirely upon respondent reports about heroin use averaged over relatively long period of time (i.e., weeks, months, or years), this study has obtained specific data from each respondent about his/her heroin (and other drug) use on a daily basis during 33 or more consecutive days and about the dollar amounts of heroin used on each of those Thus, respondents can now be classified according to the proportion days. of days in which they reported some heroin use. Then, the dollar amounts of heroin consumed can be computed and employed to make decisions about appropriate cutting points in order to classify respondents into low, medium, and high levels of heroin use.

# Calculating the proportion of person-days with heroin use.

As indicated in the Methodology Chapter (II), respondents had differing number of "person-days" during which they were interviewed. In this chapter and future chapters, all data have been standardized so that each respondent contributed the same number of days of information. For each subject, the number of person-days during which he reported heroin use was divided by the total number of person-days and multiplied by 100; this provided the percentage of person-days during which each subject used heroin.

Next, respondents were classified into eight groups corresponding to the number of days per week during which they used heroin when averaged across all days of reporting. That is, subjects who reported no (zero) days of heroin use (N = 6)\* plus subjects who reported up to 7.1% of days of heroin use (N=12) sere classified into a "near zero" use category. Likewise, similar calculations were performed for one, two, ... seven days per week of heroin use. The proportion of person-days, the cutting points, and approximate number of days per week with heroin use are presented in Graph IV. 1.

All of these six subjects were receiving legal methadone; they reported lengthy histories of heroin use.

A Heroin User Typology -39-

Graph IV.1

Dollar Amount of Heroin Used Per Day (Including Nonuse Days) By Persons Classified According To Their Frequency of Heroin Use



and the second second second





,

Graph IV. 1 shows a strong linear trend. Those with near zero days of heroin use, consumed less than \$1 per day, while those who used heroin on approximately seven days consumed over \$38 per day. While other categories fall between, natural cutting points appeared between days five ( $\frac{24}{day}$ ) and six ( $\frac{534}{day}$ ). Cutting points were less clear at less frequent levels of heroin use.

These data, of course, were heavily dependent upon the number of days without heroin use. This raised the question, did the dollar amount of heroin consumed differ on days <u>with</u> heroin use?

Data in Graph IV. 2 show some differences, but not as striking as might be anticipated. Subjects who used heroin on few days (near zero, one, two, and three days) used between \$21 to \$28 per day of heroin use. Those using heroin on four and five days consumed \$34 per day of heroin use, while those using on six and seven days consumed \$40 and \$39 per day of use. Again a natural cutting point appears between days five and six, and possibly between days three and four.

This eight category measure of drug use was also related to the crime, drug use, income, and expenditure variables reported in the next few chapters (detailed data are not presented). Given the small number of subjects in each of these eight categories, the variation between the eight groups was considerable. Thus, subjects who used heroin on seven days were not always higher than those who used on six days, although both these groups were generally higher than less frequent users. Likewise, subjects who used heroin on three days were more similar to subjects who use heroin on one or two days for some variables, but were more similar to subjects who use heroin on four or five days for other variables.

A Heroin User Typology -40-(Graph IV. 1 about here)

(Graph IV. 2 about here).


# Defining a Heroin User Typology

In developing a useful typology, a key group of subjects, the daily heroin users were first identified. The data above show that persons who use on over 78.6% of their person days (the six and seven days per week users) closely approximate the stereotype of daily heroin users in that they have the highest dollar consumption of heroin of all groups. Such persons constituted 31% of all subjects, and used heroin on an average of 91% of their person-days.

At the other extreme, subjects were sought who were relatively inactive in using heroin. Graphs IV. 1-2 show that persons who used heroin on 35.7% or less of the time (zero, one, and two days per week) clearly belong in this group and were classified as irregular heroin users. Due to sample size considerations and a desire to maximize differences between the irregular and daily heroin users, persons who used heroin on three, four, or five days (between 35.7% and 78.5% of their person days) were included in the intermediate category and were called regular heroin users. Thus, the eight categories were reduced to three to form a Heroin User Typology. Tables IV. 1 and 2 described the levels of heroin use by those in each catagory of this typology: (Table IV. 1 and 2 about here)

### Findings about the Heroin User Typology

When subjects are classified according to their regularity of heroin use, 1. Daily heroin users constituted 31% of all subjects; as a group, they had 52% of all days of heroin use in this sample; and consumed 63% of the annual dollar volume of heroin by all subjects.

- 2. Each daily heroin user on the average (\$10) bags of street heroin);

obtained in this study.

A Heroin User Typology -42-

consumed about \$40 per day of heroin use\* (or about 4 "dime" consumed an annual average of over \$13,000 worth of heroin.

\*- Information about the number of heroin use episodes per day was not

### A Heroin User Typology -43-

3. Regular heroin users constituted 39% of the subjects; as a group, they

E 12

-i - i - s

- contributed a similar proportion of person days (39%) of use, but
- consumed only 31% of the annual dollar volume of heroin.
- 4. Each regular heroin user on the average

AND STREET AND A S

- consumed about \$32 (an average of about three street bags) per day of heroin use; and
- consumed an annual average of over \$6,000 worth of heroin.
- 5. Irregular heroin users constitute 30% of all subjects, but as a group, thev
  - . had a very small proportion (9%) of all person days with heroin use:
  - consumed a smaller proportion (6%) of the annual dollar volume of heroin.
- 6. Each irregular heroin user on the average
  - consumed about \$23 (or about two street bags) per day of heroin use: and
  - consumed an annual average of less than \$1,500 of heroin.

In comparison with mass media sterotypes, the dollar amounts of heroin consumed by these subjects may seem low. That is, the grand mean for all subjects for all days was \$19; and was only \$35 per day with heroin use. Even among daily heroin users on days of heroin use (\$40), the amount seems low in comparison with the daily heroin habit sizes claimed by many addicts of \$50 to \$100 or more per day. Evidence in other papers (Goldstein, 1980, Goldstein, et al., 1982) showed that self-reports of "habit" sizes by street opiate users overstated actual consumption by a considerable degree.

Other possible factors may explain the discrepancies between claimed habit sizes and mean daily consumption (as calculated here). Heroin users may report as "habit sizes" the amount of heroin they would like to use. They may fail to recall days with no heroin use or days with considerably less heroin consumption than they would like. In addition, a few subjects do average over \$50 of heroin per day of use, but more daily heroin users also have averages of \$30 or less per day of use. Thus, the few subjects with large averages are offset by their more numerous counterparts who had substantially less consumption.

Definition       N       x       N	Heroin User	Typolo	gy Definition	SL	bjects		For 100	Person	Days/Su
Irregular (0,1,2 Days/Week) 61 30% 974 Regular (3,4,5 Days/Week) 78 39% 4,255 Daily (6 & 7 Days/Week) 62 31% 5,623 Totals 201 100% 10,853 1 Table IV. 2 Daily Amounts and Annualized Dollars of Heroin Consumed by Heroin User Typology Percentage Mean Dollars of Heroin Used % of \$ Amoun Heroin Days With Some Per Heroin Per Per Days With Some Per Heroin Per Per Irregular 16% 23,82 3.80 1,389 6% Regular 55% 32.30 17.62 6,431 31% Daily 91% 39.84 36.14 13,189 63% Total 54% 35.45 19.14 6,986 100% (21,009)** * Dollars Per Day and Per Year include days without heroin use. ** Sum of three groups in column "Per Year;" base for percentages above.		<u>_</u>	Jermition	<u></u> !N	<i>k</i>	-	Use Day	s Among	<u>n</u> -
Regular(3,4,5 Days/Week)7839%4,255Daily(6 & 7 Days/Week)6231%5,623Totals201100%10,8531Table IV. 2Daily Amounts and Annualized Dollars of Heroin Consumed by Heroin User TypologyPercentageMean Dollars of Heroin Used Bays With Some Heroin Consumption% of \$ Amoun Heroin Qays With Some Per Heroin Per Days With Some Per Heroin Per Per Za.82% of \$ Amoun Heroin Qsed Each Categor During YearIrregular16%23.823.801,3896%Regular55%32.3017.626,43131% (21,009)**Daily91%39.8436.1413,18963% (21,009)*** Dollars Per Day and Per Year include days without heroin use. ** Sum of three groups in column "Per Year;" base for percentages above.	Irregula	ır (0,	1,2 Days/Wee	k) 61	30%	5	9	974	
Daily(6 & 7 Days/Week)6231%5,623Totals201100%10,8531Table IV. 2Daily Amounts and Annualized Dollars of Heroin Consumed by Heroin User TypologyPercentage UserMean Dollars of Heroin Used Per Heroin Per Days With Some Heroin Consumption% Consumption% Soft \$ Amoun Heroin Used Day*% Consumption% Consumption Year*% Consumption% Consumption% Consumption During Year 	Regular	(3,4	4,5 Days/Wee	k) 78	3 39%	, 2	4,:	255	
Totals201100%10,8531Table IV. 2 Daily Amounts and Annualized Dollars of Heroin Consumed by Heroin User TypologyPercentage (Jeroin Consumption 	Daily	(6 8	& 7 Days/Wee	k) 62	2 31%	2	5,0	523	
Table IV. 2Daily Amounts and Annualized Dollars of Heroin Consumed by Heroin User TypologyPercentage User TypologyMean Dollars of Heroin Used Per Heroin Per Days With Some Heroin Consumption Use Day IrregularMean Dollars of Heroin Used Per Heroin Per Day* Year* During Year During YearIrregular16% Regular23.82 32.303.80 1,3891,389 6% 6% RegularDaily91% 39.8436.14 36.1413,189 6.986 (21,009)***Dollars Per Day and Per Year include days without heroin use. ** Sum of three groups in column "Per Year;" base for percentages above.		Totals		201	100%	,	10,	353	1
Inform consumption       ose day       Day       lear       During lear         Irregular       16%       23.82       3.80       1,389       6%         Regular       55%       32.30       17.62       6,431       31%         Daily       91%       39.84       36.14       13,189       63%         Total       54%       35.45       19.14       6,986       100%         (21,009)**       *       *       Dollars Per Day and Per Year include days without heroin use.       **         Sum of three groups in column "Per Year;" base for percentages above.       **	Heroin User Typology	Daily A Percent of All Days W	Amounts and A by tage Person ith Some Consumption	Annualiz Heroin <u>Mean Do</u> Per He	ed Dolla User T Dollars c	ars of ypolog of Her Per	f Heroin Jy oin Used Per	Consumed % of Heroi Each	1 \$ Amoun n Usad Categor
Inregular       10%       23.02       3.00       1,309       0%         Regular       55%       32.30       17.62       6,431       31%         Daily       91%       39.84       36.14       13,189       63%         Total       54%       35.45       19.14       6,986       100%         (21,009)**         *       Dollars Per Day and Per Year include days without heroin use.         **       Sum of three groups in column "Per Year;" base for percentages above.	Innorula	neronn	1 Eq	22 (	<u>1y _ </u>	3 90	1 30		Ga Ca
Negular       03%       02.00       11.02       0,451       01%         Daily       91%       39.84       36.14       13,189       63%         Total       54%       35.45       19.14       6,986       100%       (21,009)**         *       Dollars Per Day and Per Year include days without heroin use.       **       Sum of three groups in column "Per Year;" base for percentages above.	Pogular	, (	5.59	32 1	יב אח ז	7.62	6 43		31 <i>9</i>
Total 54% 35.45 19.14 6,986 100% (21,009)** * Dollars Per Day and Per Year include days without heroin use. ** Sum of three groups in column "Per Year;" base for percentages above.	Dailv		91%	39.8	34 3	36.14	13,18	9	63%
Total 54% 35.45 19.14 6,986 100% (21,009)** * Dollars Per Day and Per Year include days without heroin use. ** Sum of three groups in column "Per Year;" base for percentages above.	burry						10310	-	
<ul> <li>Dollars Per Day and Per Year include days without heroin use.</li> <li>** Sum of three groups in column "Per Year;" base for percentages above.</li> </ul>		Total	54%	35.4	15 1	9.14	6,98	6 1 (21,	00% 009)**
	* Dollars	Per Da hree gi	y and Per Ye roups in col	ar inclu umn "Pei	ide days 'Year;'	s with ' base	out hero for per	in use. centages	above.
	** Sum of t								
	** Sum of t								
	** Sum of t								
	** Sum of t								

A Heroin User Typology -45-What the Heroin User Typology Does Not Reveal.

This Heroin User Typology is simply a classification of subjects based upon one dimension of their behavior--the frequency of their heroin use during the reporting period. While the daily heroin user may be considered as an "addict" by many people, evidence in previous papers (Johnson, et al., 1979; Johnson, 1981a) from this project as well as forthcoming papers, raise central questions about the phenomena of addiction in the lives of even the daily heroin user, much less the regular and irregular heroin user. Johnson (et al., 1979; Johnson, 1981a) showed that even daily heroin users consumed widely different amounts of heroin on consecutive days, frequently consuming other drugs as well. The regular and irregular heroin users generally have many days without heroin use. Nevertheless, on other dimensions, most of these subjects will be shown to be opiate dependent in that they use either heroin and/or methadone (licit or illicit). At this point, the reader is advised that the phenomenon of heroin and opiate addiction is more complex than previously documented. This report and later papers will document this complexity in much more detail.

Moreover, this Heroin User Typology conceals extensive variation in patterns of heroin and illicit drug consumption by individual subjects. For example, some respondents were daily heroin users for two or three weeks of reporting, but then ceased heroin consumption abruptly during the remainder of the reporting period. Such persons would be classified on the Heroin User Typology as regular heroin users when they may really have had episodes of daily use and virtual abstinence (see vignette D below; also Goldstein, 1982b).

Several of our subjects were enrolled in methadone maintenance treatment programs; a majority of these were classified as irregular or regular heroin users; very few were daily heroin users. Thus, many subjects were daily consumers of a legally provided opiate (methadone) but were classified as irregular heroin users on the Heroin User Typology. Chapter XIV provides more information about the kinds of subjects avoiding and in methadone treatment.



### A Heroin User Typology -46-The Heroin User Typology and Its Purposes

Nevertheless, this classification of these 201 street opiate users will prove useful in two different ways. First, in Chapters V - XII, the Heroin User Typology will be employed as an independent variable and associated with various measures of crime, drug use, income, expenditures, and economic value. The central question: Among these street opiate users, were the three types of heroin users significantly different from each other? If so, how different? Moreover, does the Heroin User Typology reveal a positive linear relationship with a given dependent variable (i.e., robbery)?

Most importantly, as the data are presented in the next several chapters, new insights about the differing lifestyles of daily, regular, and irregular users will emerge. The accumulating evidence will show that a very disproportionate share of the crime, economic value, and social impact was concentrated among daily heroin users who committed robbery (see Chapter XV).

information:

- - VII) and specific drug(s) (Chapter V).

Kinds of Information Provided in Chapters II - VII In each Chapter Y - XII, the analysis will present three major kinds of

1) The main text will present major questions and findings from this research; findings from previous research may be included. This narrative will intentionally be kept short; only 2 - 6 major findings and relationships to the Heroin User Typology are highlighted. This will aid the reader's ability to grasp the central findings quickly. These conclusions are generally based upon detailed tabulations.

2) The next section of the narrative briefly highlights major findings about the frequency and amount of respondent involvement in specific dependent variables; such as particular types of crime (Chapter VI,

### A Heroin User Typology -47-

- 3) Vignettes or brief descriptions of persons and some typical events are included at various points during the narrative and enclosed in boxes. These vignettes bring ethnographic richness to the statistical data and exemplify persons who engage in the focal behavior and describe some of their typical events. Subjects were featured in these vignettes who were relatively typical of those most active in the focal behavior, but were not necessarily the most successful. Since these vignettes emphasize ethnographic richness, cases were also chosen in which the interviewer wrote interesting stories or quoted the respondent about a particular focal behavior. We were generally able to select from among 3 - 10 subjects.
- 4) At the end of each chapter, a series of detailed tables have been provided. The data have been presented in all the detail collected in the original interviews. The Heroin User Typology will be the major independent variable; hence, the level of activity by a specific heroin user type for a specific activity (i.e., shoplifting for resale) can be examined in depth by analysts interested in such behaviors.

Instructions about how data in these tables were standardized and other information about how to read the tables are presented at the end of Chapter VI).

The analysis can now turn to a major topic of interest; patterns of drug use and purchase among these street heroin users.



## Drug Use and Purchase -48-CHAPTER V DRUG USE AND PURCHASE AMONG HEROIN USER TYPES

By definition, the street opiate users recruited for this study are likely to use heroin and or methadone. Although they were widely believed to be regular and heavy users of heroin, previous research (Robins, et al., 1973; 1974, 1975, 1979; O'Donnell, et al., 1976; Clayton and Voss, 1981; Brunswick, 1979ab) has found that heroin users were also among the heaviest users of many other drugs. This chapter provides additional documentation of this fact by providing detailed data about the drug use of a drug abusing sample.

Previous studies have generally been based upon self-reports of drug consumption spanning a month, year, or lifetime; thus, the detailed patterns of drug consumption and drug purchases documented in this chapter have not been widely available. While much information was available about their frequency of drug use, respondents in previous studies were rarely asked to report about the dollar amounts of drugs consumed in the recent past. The data below provide the first systematic information about the typical dollar amounts of drugs used by street opiate users, as well as their drug purchases. At the earliest stages of this research, pilot efforts quickly revealed an important initial finding which was generally ignored in the professional literature and prior research. This study shows that two common beliefs were seriously wrong: 1) drug users purchased their drugs (or even most of their drugs), and 2) drug consumption implied the cash purchases of drugs.

In this chapter the terms "use" or "consumption" of (drug name) will refer to the actual injestion (whether by swallowing, snorting, or injecting) of that drug. The "amount used" or "consumed" will refer to the standard retail value (measured in dollars) actually injested by respondent regardless of how the drug was obtained. The term "purchase" refers to the number of dollars (i.e. cash) used to buy (drug name). Other techniques for obtaining drugs without cash payments will be Respondents can clearly have "use-days" which do not involve heroin purchase. Likewise, "purchase-days" can occur without heroin use. Even on days with both use and purchase, the

described in Chapter VII. dollar amounts may differ greatly.

## 

### Drug Use and Purchase -49-

While the purchase of drugs, especially expensive drugs like heroin and cocaine, accounted for the majority of the dollar volume of these drugs consumed (see below), evidence in this chapter will show that a sizable proportion of the dollar value of cocaine or heroin was obtained without cash purchase. Specifically, future chapters will show that many subjects obtain drugs on a substantial proportion of person-days by: a) working in various drug distribution roles and receiving payments in drugs instead of cash, b) being given drug(s) as a gift or sharing them with a friend, c) by obtaining drugs by theft from or robbery of other drug users/dealers, or d) other means.

This chapter will provide detailed information about frequency and dollar values of consumption and purchase for 11 types of substances among the three heroin user types. Approximations of the proportion of all drug(s) consumed resulting from cash purchase will also be presented.

### A. MAJOR FINDINGS

### Did drug consumption and drug purchase vary by heroin involvement?

With the exception of heroin, cocaine, alcohol, and illicit methadone, no or little association was shown between the Heroin User Typology and the use or purchase of other drugs. For these four major drugs, however, important variation by the Heroin User Typology occurred.

### How did the drug use and purchase of daily heroin users differ from less regular heroin users?

1. The daily heroin users and less regular heroin users were equally likely to use any specific substance, but varied proportions of all subjects were involved with a given drug.

Table V. 1 showed that all (100%) of these street opiate users used or purchased one or more drugs. Almost all (90% or more) reported cocaine and alcohol use and 85% or more of all subjects purchased these drugs. About 50%



or more of the respondents used or purchased illicit methadone or marijuana. About a third of the subjects used or purchased tranquilizers, less than 10% used or purchased other opiates (morphine, opium, etc.), amphetamines, barbiturates, or psychedelics. While variation by the Heroin User Typology was not significant, daily heroin users seemed somewhat less likely to use or purchase marijuana, barbiturates, tranquilizers, amphetamines, or psychedelics.

Heroin User Typology.

Table V. 2 shows that daily heroin users used cocaine on 36% of the person-days compared with 25% and 21% of the person-days among regular and irregular heroin users, respectively. Likewise, daily heroin users used alcohol on 63% of the person-days compared to less than half the days among the less regular heroin users. Conversely, daily heroin users used illicit methadone on only 4% of the person-days, compared with 12% of the person-days among the less regular heroin users. While the proportion of person-days with purchases of cocaine, alcohol, and illicit methadone was lower than the use of these substances, the same relationship with the Heroin User Typology was evident. The proportion of person-days using marijuana, other opiates, barbiturates, tranquilizers, amphetamines, and psychedelics was very similar among the heroin user types.

frequency of heroin use.

Table V. 3 shows that the dollar amounts of cocaine consumed (\$7) or purchased (\$5) per day was about the same among the three heroin user types. Daily heroin users consumed and purchased somewhat greater amounts of alcohol, but somewhat lower amounts of illicit methadone, than the less regular heroin users. The dollar amounts spant on all other drugs was not associated with the Heroin User Typology.

Drug Use and Purchase -50-

2. Daily heroin users had more person-days with cocaine and alcohol use or purchase, but fewer days with illicit methadone use or purchase, than the regular and irregular heroin users. The proportion of person-days with use and purchase of other substances was not associated with the

3. The dollar value of alcohol consumed or purchased was positively associated and illicit methdone was negatively associated with the frequency of heroin use. For all other drugs, including cocaine, no association emerged between the amounts used or purchased and the

### Drug Use and Purchase -51-

4. The frequency of heroin use was positively associated with the dollar amount used per day of use or expended on days with a purchase for illicit methadone, and alcohol, but was negatively associated with cocaine and most other drugs.

Table V. 4 shows that daily heroin users consumed almost \$40 per heroin use-day, \* compared with \$32 for regular and \$24 for irregular heroin users. Likewise, daily heroin users purchased \$34 of heroin per purchase-day compared with \$29 for regular heroin users and \$23 for irregular heroin users. The amount of illicit methadone used per day of use or or purchased on days of purchase did not vary by the frequency of heroin use.

Irregular heroin users consumed more cocaine (\$33) than the regular (\$24) or daily (521) heroin users per cocaine use-day. Weak negative associations between the Heroin User Typology and consumption per use-day (and amount purchased per purchase-day) were found for marijuana, amphetamines, barbiturates, tranquilizers, psychedelics, and other drugs.

### During one street year, what dollar value is consumed or expended for which drugs?

5. Heroin and cocaine accounted for almost 90% or more of the dollar value of all drugs used and all drugs purchased. The amount used or expended for cocaine was not directly related to the frequency of heroin use.

Table V. 5 and Figure V. 1 showed that the daily heroin users consumed an annual average of \$17,283 in drugs, compared with \$9,847 by the regular and \$5.186 by the irregular heroin users. The daily heroin users annually consume over \$13,000 worth of heroin, compared to \$6,400 for heroin by regular heroin users and about \$1,400 by irregular heroin users. All three groups spend about \$2,500 (+ \$300) per person upon cocaine. The average daily heroin user consumed \$741 worth of alcohol per year, which was somewhat greater than the less regular heroin users who consumed less than \$500 worth. Consumption of all other substances added to less than \$500 per year, with the daily heroin users least likely to consume these other drugs.

Use-days exclude days without use of that drug from the denominator: purchase-days exclude days without purchase of that drug from the denominator.











### Drug Use and Purchase -53-

No.

Although the annualized dollar amount purchased was less than the value of drugs consumed, there was almost no association between the frequency of heroin use and the purchase of specific drugs.

### Approximately how much of the drug(s) consumed were accounted for by purchases of such drug(s)?

6. The purchase of drug(s) accounted for approximately 55%-90% of the drug(s) consumed, depending upon the specific drug and measure of purchase or consumption.

Among daily heroin users, heroin purchase occurred on two-thirds of the person-days with heroin consumption (61%/91% -- Table V. 2). The annualized dollar purchases by daily heroin users accounted for only 58% of the dollar volume of heroin consumed (\$7,601/\$13,189 -- Table V. 5). The ratio of cash purchases to dollar value consumed was higher among the regular (62% -\$4,019/\$6,432) and irregular (71% -- \$986/\$1,389) heroin users.

The percentage ratio of person-days with cash purchases of cocaine to days with cocaine consumption was: 72% (26%/36%) among daily heroin users, 68% (17%/25%) among regular heroin users, and 81% (17%/21%) among irregular heroin users (Table V. 2).

These percentage ratios may be misleading. They were computed from means for groups of subjects and considerable variation existed between days with drug purchase and drug consumption on different days by a given subject, and much variation between subjects. Moreover, some persons purchased cocaine or heroin to sell, thus elevating the mean purchases artificially. The complex relationships between drug use and purchase must await future analyses.

The important point was that many subjects, but especially the daily herein users, have developed other strategies (instead of cash purchase) for obtaining the heroin they consumed. Chapter VII will demonstrate that involvement in drug distribution activities provided a sizable share of their heroin.



All subjects used drugs, and almost all used heroin during their reporting period. Mainly due to the definition of the Heroin User Typology (see Chapter IV), daily heroin users consume and purchase about twice as much heroin as the regular heroin users on an annual basis, and eight times more heroin than irregular heroin users. In short, heroin is the main drug of choice for both the daily and regular heroin users, but a second choice (after cocaine) among irregular heroin users. In vignette A, respondents have been selected because their mean daily heroin consumption was virtually identical to the mean heroin consumption by daily (\$36/day), regular (\$18/day), and irregular (\$4/day) heroin users.

Kitty D. (black female, age 24) was a representative daily heroin user who also lengaged in prositution on a daily basis. She was relatively unique among daily heroin users in that she had almost no crimes other than prositution. She used heroin  $\gamma n$  each of her 33 reporting days at an average of \$36/day. The following week was representative of her heroin using behavior.

1/2/82		used	\$50	of	heroi
		her t	by a	fri	end.
1/3/82		used	\$50	of	heroir
		value	e by	"co	pping
1/4/82		used	\$15	of	heroin
1/5/82	-	used	\$25	of	heroin
1/6/82	-	used	\$25	of	heroin
1/7/82		used	\$25	of	heroin
1/8/82		used	<b>\$</b> 50	of	heroin

Kitty also used \$2-\$15/day of alcohol every day, about a \$1/ day of tranguilizers, and \$10 - \$30 of cocaine on about half of her reporting days. She did not report any illicit methadone or marijuana use. During one week, she reported purchasing and using Darvon for 50¢.

Drug Use and Purchase -54-B. HEROIN USERS AND THEIR CONSUMPTION OR PURCHASE OF SPECIFIC DRUGS

### HEROIN

Vignette A - Heroin and Opiate Use and Purchase Patterns Among Daily, Regular, and Irregular Heroin Users

n, but purchased \$20; \$30 of heroin was given to

which she purchased for \$40, thus, she got a \$10 short."

purchased for \$10 and \$5 by copping short. purchased for 520 and 55 by copping short. without purchase; given heroin by friend. without purchase; given heroin by friend. purchased for \$30, was given \$20 by friend.

Five subjects did not report any heroin use during the reporting period, but were receiving legal methadone; they have been arbitrarily included among the irregular heroin users for analytic purposes.

Drug Use and Purchase -55-

Vignette A (Continued) - Heroin and Opiate Use and Purchase Patterns Among Daily, Regular, and Irregular Heroin Users

Poet B (black male, age 45) was a regular heroin user and occasional thief. He graduated from high school and claimed to work as a self-employed lecturer and writer, mainly of poems which he sold on street corners; he reported no such income during the reporting period. Poet consumed an average of \$17 of heroin per day during his 33 person-days. A representative week revealed complex opiate consumption.

2-16-82 - used \$10 heroin from a \$10 purchase.

- 2-17-82 used \$25 heroin, but had no heroin purchases. He copped a quarter (a \$50 bag) for a friend who shared it evenly with him.
- 2-18-82 same as previous day.
- 2-19-82 no heroin use or purchase. Bought 50 mgs of illicit methadone for \$10, used half (25 mgs) of it.
- 2-20-82 no heroin use or purchase; used 25 mgs of illicit methadone from yesterday.
- 2-21-82 no heroin use or purchase, nor illicit methadone use. Even though this Sunday had no opiate use, Poet helped sell a quarter of heroin but did not receive cash or drugs as payment.
- 2-22-82 no heroin use or purchase; used 30 mgs of methadone given to him by a friend.

Poet also consumed \$1/day of alcohol during this week. Poet died in September, 1983 with high blood pressure, gastritis, and other alcoholrelated complications.

Barry D. (Hispanic male, age 37) was an irregular heroin user who never committed nondrug crimes (except evading subway fares) and rarely helped sell drugs (see chapters VI - VII). Barry consumed an average of \$4/day of heroin during his 33 reporting days. During the initial interview, he commented, "The reason I don't do too much crime, I'm trying to cut down on my drugs." During his reporting period, he had one week with no heroin use, a week with two days, another with three days, and one week with four days of heroin use. On only two days, did he use \$25; most heroin use days involved only

\$10-13. The following five days were representative.

- 10-8-81 used \$13 of heroin, but did not purchase it. He helped sell \$18 of heroin, for which he earned \$10. He spent \$10 on cocaine and shared it (used \$5) with a friend. In addition, a friend gave him a "capful" (the top of a bottle) of illicit methadone.
- 10-9-81 no heroin use or purchase. Purchased 60 mgs of illicit methadone for \$10, used 20 mgs on this day.
- 10-10-81- no heroin use or purchase; used 20 mgs. of illicit methadone purchased on previous day.
- 10-11-81- no heroin use or purchase; used 20 mgs. of illicit methadone purchased two days earlier.
- 11-12-81- purchased and used \$10 of heroin, but no illicit methadone.

During each of these days, Barry also used \$1 of marijuana and \$1 of alcohol. He also received \$20 in cash for helping in a restaurant on 10-10-81.



Almost all respondents (about 90%) reported cocaine use or purchase, with no variation by the frequency of heroin use. Although daily heroin users had more days (36%) with cocaine use than regular and irregular heroin users (25% and 21% respectively), the daily users consumed less cocaine (\$22) than regular (524) and irregular (533) heroin users per day of cocaine use. (A similar negative association holds for cocaine purchases per purchase-day.) Thus, no relationship existed between the frequency of heroin use and the dollar amounts of cocaine used (or purchased) per day or per year.

Nevertheless, cocaine approaches heroin in economic value. Subjects consumed about \$25 per cocaine use-day and purchased \$25 of cocaine on days when they purchased cocaine. Cocaine was clearly the drug of preference among irregular heroin users; they used cocaine on 21% of the days, consumed 532 per day of cocaine use, and annually consumed about twice as much cocaine (\$2,512) as heroin (\$1,389); the same held for cocaine purchase. Thus, subjects purchased and consumed about two and a half "dime" bags of cocaine on days of cocaine use. When days without cocaine use were included, subjects consumed less than one dime bag per day, and purchased only about half a bag per day (Table V. 3).

### Vignette B - Cocaine Use by Street Opiate Users

Nick T. (Hispanic male, age 29) was an irregular heroin user mainly because he was on a methadone treatment program. But he used an average of \$57/day of cocaine during his 33 reporting days. While he had two weeks during which he used each day, the following week indicated more variability. 5/30/81 - used \$225 of cocaine from a \$180 purchase; he copped short. 5/31/81 - used 570 of cocaine from a \$70 purchase. 6/1-3/82- used no cocaine or heroin. 6/ 4/81 - used \$30 of cocaine from a \$30 purchase. 6/ 5/81 - used \$120 of cocaine from a \$110 purchase; he copped short. Kitty D (see Vignette A) also used cocaine each day during the week; she used

\$20 worth on 1/2/81, then \$10, \$5, \$20, \$10, \$10, and \$10 on successive days. It was guite common for daily and regular heroin users to purchase both cocaine and heroin (if sufficient funds were available) and inject them together as a "speedball." If cocaine was used alone, it was as likely to be injected as snorted by our subjects.

### Drug Use and Purchase -56-COCAINE

## Drug Use and Purchase -57-ALCOHOL

WARE A DE LE CALLER

Alcohol emerged as the third most important drug among these street opiate users. Almost 90% of respondents used and purchased alcohol with little variation by frequency of heroin use. Surprisingly, alcohol was consumed on 63% of the person-days by daily heroin users, which was considerably higher than days with alcohol use among regular (48%) and irregular (43%) heroin users. (Similar findings emerged for alcohol purchases.) The actual dollar amounts consumed per day of use (about \$3) does not constitute a major purchase -- like heroin or cocaine expenditures.

Nonetheless, three dollars of alcohol per day is a considerable amount of alcohol to consume--equal to approximately a pint of hard liquor, or a quart and half of sweet wine, or three quarts of beer. This converts to approximately 6 ounces of pure alcohol per alcohol-use day. The most alcoholic subjects would typically begin drinking when they awakened in the morning with the shakes. They would drink steadily throughout the day. A pint of sweet wines (mainly Thunderbird or Wild Irish Rose) in a paper bag was constantly being consumed by individuals or a drinking group. Sometimes they would drink vodka or other hard liquors straight. Even when nonalcohol use days were included, the average subject consumed between two-three ounces of pure alcohol per day. The daily heroin users consumed about \$300 more alcohol and purchased over \$200 more alcohol per year than regular and irregular heroin users.

### Vignette C - Alcoholic Heroin Users

Kat E. (black female, age 35) was a daily heroin user and prostitute (also see Vignette K) who was seriously alcoholic. On 6/28/81, Kat reported being in the hospital for cirrhosis and stated: "I have cirrhosis of the liver. I can drink no more vodka. I drink wine." But on both the week before and after this hospitalization, she reported drinking \$5/day of alcohol.

Many other subjects drank staggering amounts of alcohol. [Kitty drank an average of \$6/day during the week featured in Vignette A. Nick T. drank \$8/day of alcohol during the week featured in Vignette B.]



Drug Use and Purchase -58-ILLICIT METHADONE

Illicit methadone was the only substance negatively associated with the frequency of heroin use. Daily heroin users were the least apt to use or purchase illicit methadone. They consumed illicit methadone on only 4% of the person-days compared to 12% of the person-days among regular and irregular heroin users. Nevertheless, on days of illicit methadone use or purchase, daily heroin users consumed dollar amounts (\$9-\$10) that were equal to the less regular heroin users.

On an annual basis, however, relatively small dollar amounts of illicit methadone were consumed or purchased (\$435 or less) by each heroin user type. The relationship between licit and illicit methadone was complex and will need to be analyzed in subsequent reports.

Less than 10% of the respondent reported any use of opiates (other than heroin or methadone) such as morphine, demerol, opium, or medicinal opiates. These were consumed on less than 0.5% of the person-days, and involved \$6 or less on days of use. Less than \$10 per year was expended for these drugs regardless of heroin use type.

Almost three-quarters of these street opiate users consumed marijuana or hashish (mainly marijuana) during the reporting period. Marijuana use was not significantly related to the frequency of heroin use, although daily heroin users seemed somewhat less involved than other groups. Marijuana was used on about a quarter of the person-days, and purchased about half as often. The average amount consumed per use-day was about \$3 or less. Because many subjects also engaged in marijuana sales, the mean amount (\$5) spent on marijuana per purchase-day was higher. Nevertheless, "loose joints" can be purchased for \$1 on the streets, and many subjects buy "trey" (\$3) and "nickel" (\$5) bags of marijuana which allows them to make 4 to 8 joints.

### OTHER OPIATES

### MARIJUANA

### Drug Use and Purchase -59-

Constant of the second s

The more enterprising buy an street "ounce" (generally considerably less than a standard ounce), roll about 25-50 joints which they sell to others. Typically, however, marijuana did not involve large annual amounts being used or purchased (generally under \$400 per year).

### TRANOUILIZERS

About two-fifths of the respondents used tranquilizers and such use occurred on about 6% and purchase occurred on 4% of the person-days. On days with tranquilizer use, about \$2 worth was consumed. Some of these tranquilizers were obtained from physicians, but most of them were purchased on the streets. Since the typical street price was \$1 per tranquilizer, subjects consumed about two pills per day of tranquilizer use. Nevertheless, such use involved less than \$50 annually. Tranquilizer use and purchase were not related to the frequency of heroin use.

### OTHER DRUGS

### (Amphetamines, barbiturates, psychedelics, and various pills)

The levels of use of these drugs was very low among street opiate users when compared, for example, with the proportion of school age youth reporting use of these drugs. Generally less than 10% of the respondents reported using any of these substances. Such use occurred on less than 0.5% of the person-days and involved modest amounts per day of use, and seldom accounted for over \$20 per year per substance. Purchases of these substances were even lower than use. Neither the use nor purchase of these substances was related to the frequency of heroin use.

### ANY DRUG USE

These subjects were, however, consistent drug users. All subjects used and purchased drugs, and on an average of 85% of their person-days. On a typical day, they used \$35 and purchased \$23 of drugs. Daily heroin users consumed considerably more drugs (\$49) than regular heroin users (\$32) and irregular heroin users (\$19) per use-day. Daily heroin users purchased twice as much drugs per drug purchase-day (\$30 vs. \$15) than irregular heroin users.



Drug Use and Purchase -60-The annual value of drugs consumed was over \$17,000 among daily heroin users, almost \$10,000 among regular heroin users, and over \$5,000 among irregular heroin users. Most of the variation in drug consumption and purchase was due to the differing levels of heroin involvement. Because daily heroin users had more days with and dollar expenditures for heroin, their overall involvement with any drugs was substantially higher than among the less regular heroin users.

This analysis of specific drugs does not address the very complex patterns of multiple drug use by these subjects. Heroin use without the use of other substances on the same day was relatively uncommon; analysis of the many and complex patterns of heroin and other drug use by these respondents will be detailed in future papers. In the vignette below, Norton S. was selected because his average heroin use (\$20), cocaine use (\$5), and alcohol use (\$1)during the 33 days closely approximated the grand means (\$19-heroin; \$7-cocaine; \$1.5-alcohol) for all respondents. His wide variability in the kinds of drugs used, the dollar values used and purchased day by day was a typical pattern among almost all respondents. Likewise the complex techniques for obtaining drugs by other than cash purchases was common among many respondents. These are delineated in more detail in Chapter VII.

Norton S. (hispanic male, age 37) was a regular heroin user. A common criminal activity was burglarizing abandoned buildings and stripping them of copper and fixtures which were sold to a junk yard. Even though his consumption of heroin, cocaine, and alcohol during 33 days were just about equivalent to the grand mean, the following information shows that such mean figures may be misleading.

Norton was actually a daily heroin user and alcoholic; during the 33 Norton also worked regularly at a quasi-legal "hustle," He regularly

days, he was hospitalized for alcohol detoxification, and then followed treatment plans which curbed his alcohol use -- but not his other drug abuse. bought cigarette lighters cheap (a box of 48 for \$12) and sold them on the street for twice as much making \$12 a box. On one day, he also sold (for \$2 each) six coke (cocaine) spoons on chains; he bought them for 50¢ making \$9. In selling these lighters, he told potiental customers that they were "hot" and worth \$5-6 each, thus conning them into believing that cheap lighters were actually worth more.

Vignette D - Daily Patterns of Substance Use Among A Regular Heroin User

### Drug Use and Purchase -61-

For each day described below, the number before the slash refers to the dollar value <u>used</u> while the number after the slash refers to value <u>purchased</u> for each specific drug:

- 8/10/81 heroin 55/30; cocaine 10/0; marijuana 5/2; alcohol 6/2; cigarettes 2/0. He received \$25 of heroin for copping drugs, \$10 of cocaine and \$3 of marijuana were recieved from friends. He purchased 2 pints of wine, a friend gave him a pint, and he received 3 pints of wine and two packs of cigarettes on credit from a store.
- 8/11/81 heroin 60/40; cocaine 10/0; marijuana 3/0; alcohol 6/4; cigarettes 2/0. He received \$33 of heroin, cocaine, and marijuana for copping drugs and turning on with friends. He received one pint of wine from friends. When Norton was asked about money that was not accounted for, he replied, "I don't know. Sometimes I black out."
- 8/12/81 heroin 40/20; cocaine 10/0; marijuana 3/1; alcohol 7/5; cigarettes 2/0. He received \$32 of heroin, cocaine, and marijuana for copping drugs and turning on with friends. He received one pint of wine from friends.
- 8/13/81 heroin 40/30; cocaine 10/0; marijuana 2/1; alcohol 9/6; cigarettes 2/0. He received \$21 of heroin, cocaine, and marijuana for copping short and turning on with friends. He received one pint of wine from friends. "I blacked out for about 10 hours. I don't know what happened to those 10 hours."
- 8/14/81 heroin 40/40; cocaine 20/0; marijuana 3/3; alcohol 5/5; cigarettes 2/0. He received \$20 of cocaine from a friend.
- 8/15/81 heroin 50/0; cocaine 10/0; marijuana 0/0; alcohol 5/5; cigarettes 2/0. He obtained \$60 of cocaine and heroin for helping sell drugs.
- 8/16-20/81 could not remember what drugs he had used because he missed his weekly interview (was interviewed on 8/28).
- 8/21-28/81 entered a hospital for one week of alcohol detoxification and treatment. He came out on 8/28; no drug or alcohol use during week.
- 8/29/81 marijuana 4/4. No alcohol or other drugs.
- 8/30/81 marijuana 3/3. No alcohol or other drugs.
- 8/31/81 heroin 13/20; marijuana 2/2; no alcohol. He turned on a friend, thus consumed less heroin than purchased.
- 9/ 1/81 heroin 30/30; marijuana 2/2; no alcohol.
- 9/2/81 heroin 68/40; cocaine 15/0; marijuana 2/2; no alcohol; cigarettes 1/0. Reported to his alcohol treatment program for group therapy and individual counseling.
- 9/3/81 heroin 10/0; cocaine 30/10; marijuana 2/2; no alcohol; cigarettes 1/0. Got heroin and cocaine from copping and shorts.
- 9/4/81 heroin 10/0; cocaine 30/10; marijuana 2/2; no alcohol; cigarettes 1/0. Got 30 in cocaine and heroin from friend. Reported to his alcohol treatment program for group therapy and individual counseling.

This description of Norton's drug use over a 26 day period simplifies the complexity of his behavior because he was of obtaining drugs via steering, touting, copping, copping short, and obtaining alcohol and cigarettes on credit or from one or more friends.

Norton's behavior was relatively unique in that he entered and followed a treatment schedule for alcohol detoxification. But this had little impact upon heroin consumption and cocaine use. He evidenced no interest in methadone treatment. As the following chapters unfold, we shall see that his drug using and criminal behaviors were relatively common among these street heroin users.



	-			-		-						
Type of Drug	Heroin Irregulai	<u>user Typ</u> Regular	ology Daily	Total	pof		<u>Heroin</u> Irregula	<u>User Typol</u> ar Regular	ogy Daily	Total	pof	
(Number of Subjects)	(61)	(78)	(62)	(201)	۲	r	(61)	(78)	(62)	(201)	<b></b>	<u>r</u>
	A. 1	Percentage	of <u>Resp</u>	ondents	<u>Using</u> :		Β.	Percent of	Respond	ents Pu	irchasi	ng:
Any Drug Use	100.0	100.0	100.0	100.0			100.0	100.0	100.0	100.0		
Heroin	90.2	100.0	100.0	97.0	.001	.23	88.5	98.7	100.0	96.0	.001	.23
Other Opiates	8.2	1.3	8.1	5.5	.12	.00	3.3	1.3	1.6	2.0	.69	05
Illicit Methadone	55.7	61.5	48.4	55.7	.30	06	52.5	52.6	43.5	49.8	.51	07
Cocaine	91.8	92.3	95.2	93.0	.73	.05	88.5	83.3	90.3	87.1	.44	.02
Alcohol	90.2	87.2	90.3	89.1	.80	.00	85.2	84.6	88.7	86.1	.77	.04
Marijuana	80.3	70.5	67.7	72.6	.26	11	73.8	66.7	58.1	66.2	.18	13
Amphetamines	11.5	7.7	11.3	10.0	.70	.00	8.2	5.1	6.5	6.5	.77	03
Barbiturates	8.2	6.4	8.1	7.5	.90	.00	6.6	7.7	1.6	5.5	.27	09
Tranquilizers	41.0	42.3	40.3	41.3	.97	.01	37.7	32.1	32.3	33.8	.75	04
Psychedelics	3.3	1.3	3.2	2.5	.69	.00	3.3	1.3	3.2	2.5	.69	.00
Other Drugs	34.4	25.6	33.9	30.8	.45	.00	29.5	19,2	19.4	22.4	.28	10
Legal Methadone*	42.6	25.6	11.3	26.4	.000	28	n ann ann ann Ann Ann a'					

A A

Î

(FA

Table V. 1 -- Percentage of <u>Respondents</u> Using and Purchasing Drugs by Heroin User Typology

Drug Use and Purchase -62-

\*Legal Methadone is not included in the "any drug" total. (See instructions for reading tables at the end of Chapter VI.)

Ð

1



Tune of Duus	Heroin	User Typ	ology	-			Heroin I	Jser Type	logy			
(Number of Subjects)	irregular (61)	Regular (78)	Daily (62)	Total (201)	p of F	<u>r</u>	Irregula (61)	r Regular (78)	Daily (62)	Total (201)	p of F	r
	A. P	ercentage	of <u>Pers</u>	on-Days Us	sing:		B. Pe	ercent of	Person-	Days Pur	chasing	<u>j</u> :
Any Drug Use	73.5	84.9	96.1	84.9	.000	.50	58.2	54.9	80.3	67.6	.000	.3
Heroin	16.0	54.6	90.7	54.0	.000	.94	11.8	37.7	61.1	37.0	.000	.7
Other Opiates	0.2	0.1	0.6	0.3	.19	.08	0.1	0.0	0.1	0.1	.54	.0
Illicit Methadone	11.5	12.1	4.4	9.5	.005	19	7.5	6.4	2.6	5.6	.014	1
Cocaine	20.9	24.9	36.3	27.2	.001	.24	16.9	17.2	26.2	19.9	.016	.1
A1coho1	42.3	47.7	62.7	50.7	.003	.23	31.2	33.9	52.7	38.9	.000	.2
Marijuana	29.9	23.3	25.4	25.9	.40	06	15.4	12.0	11.5	12.9	.41	0
Amphetamines	0.7	0.2	0.5	0.4	.22	05	0.4	0.1	0.4	0.2	.30	.0
Barbiturates	0.4	0.2	0.3	0.3	.57	02	0.3	0.1	0.1	0.2	.71	0
Tranquilizers	5.9	4.6	7.1	5.8	.59	.03	4.3	3.1	5.1	4.1	.60	.0
Psychedelics	0.1	0.1	0.2	0.1	.73	.05	0.1	0.1	0.2	0.1	.79	.0
Other Drugs	3.2	2.2	2.2	2.5	.56	07	2.2	1.5	1.3	1.6	.53	0
.egal Methadone*	30.0	13.1	4.4	15.5	.000	33						

.

5

Drug Use and Purchase -63-

(See instructions for reading tables at the end of Chapter VI.)

\$ 1

NA.

THE ALL THE PARTY OF THE

	Heroin	User Ty	/pology				Heroin U	ser Typ	ology			
Type of Drug (Number of Subjects)	Irregular (61)	Regulai (78)	r Daily (62)	Total (201)	p of F	<u> </u>	Irregular (61)	Regula (78)	r Daily (62)	Total (201)	p of F	r
	A. <u>Dol</u>	lar Valı	<u>ie of Drug</u>	g(s) <u>Used</u>	Per Da	iy:	B. <u>Dollar</u>	Value	of Drug(	s) <u>Purch</u>	<u>ased</u> Pe	er Day:
Any Drug Use	14.20	26.98	47.35	29.39	.000	.57	11.13	17,51	29.15	19.16	.000	.46
Heroin	3.80	17.62	36.14	19.14	.000	.66	2.70	11.01	20.83	11.52	.000	.62
Other Opiates	0.01	*	0.02	0.01	.13	.10	0.01	*	0.00	*	.13	13
Illicit Methadone	1.02	1.19	0.45	0.91	.014	15	0.93	1.03	0.37	0.79	.019	15
Cocaine	6.88	5.99	7,85	6.83	.47	.04	5.51	3.81	5.67	4.90	.22	.01
Alcohol	1.23	1.31	2.03	1.51	.003	.22	0.88	0.94	1.59	1.12	.002	.22
Marijuana	0.98	0.64	0.71	0.77	.18	10	0.87	0.53	0.60	0.65	.29	08
Amphetamines	0.05	*	0.01	0.02	.011	17	0.04	*	*	0.02	.01	18
Barbiturates	0.02	0.05	*	0.03	.58	03	0.01	*	0.00	0.01	.15	13
Tranquilizers	0.14	0.11	0.12	0.12	.88	02	0.13	0.15	0.08	0.12	.62	05
Psychedelics	0.01	*	*	*	.65	07	0.01	*	*	*	.55	08
Other Drugs	0.06	0.04	0.03	0.05	.37	10	0.04	0.03	0.02	0.03	.38	10
Legal Methadone Use*	Number 16.62	of mil 6.23	ligrams u 1.48	sed per da 7.92	.000	35						

· · · ·

Drug Use and Purchase -64-Table V. 3 -- <u>Dollar Value</u> of Drug(s) <u>Used</u> or <u>Purchased</u> Per Day by Heroin User Typology

\*-Less than \$0.01, but greater than zero. \*\*Legal Methadone is not included in the "any drug" total. (See instructions for reading tables at the end of Chapter VI.)

-

b and a second

by

**7** 

		naumeu	rer Day of	ruse or	value	Purchased	<u>d</u> Per <u>Day of F</u>	urchase	<u>e</u> by Heroi	n User T	ypolo	gy
Type of Drug (Number of Subjects)	Heroin Irregular (61)	User T Regula (78)	ypology r Daily (62)	Total (201)	p of F	r	Heroin Irregula (61)	User Ty Ir Regu (78)	/pology lar Daily (62)	Total (201)	p_of	'n
Any Drug@	A. Do 19.32	11ar Va 31.76	lue Consum 49.26	ned per <u>D</u> 34.60	ay of l .000	lse of: .48	B. <u>Dollar</u> 15.13	Value 20.51	Purchased 30.32	per Day 22.56	of Pi	urchase
Heroin	23.82	32.30	39.84	35.45	.006	.23	22.89	29,19	34,11	31 08	024	20
Other Opiates	3.49	5.33	3.92	3.94	.96	.02	5.52	3.00	**	2.31	.12	.20
Illicit Methadone	8.90	9,90	10.15	9.57	.44	.11	12.30	15.93	14.33	14.22	. 01 9	19
Cocaine	32.87	24.04	21.61	25.10	.000	29	32,53	22.07	21.59	24.57	.000	28
Al cohol	2.91	2.75	3.24	2.98	.16	.10	2.82	2.78	3.01	2.88	.68	.05
Marijuana	3.29	2.76	2.78	2.95	.12	15	5.62	4.39	5.24	5.07	.60	- 03
Amphetamines	6.99	2.80	1.65	4.50	.06	52	11.92	4.85	0.94	6.14	.28	47
Barbiturates	5.64	32.09	0.93	9.86	.033	09	4.99	2.62	0.00	3.03	023	- 78
Tranquilizers	2.31	2.33	1.66	2.07	.016	27	3.10	4.71	1.60	3.00	.043	- 16
Psychedelics	6.67	5.00	0.51	3,15	.06	94	7.47	5.00	0.32	3.35	.044	- 97
Other Drugs	2.01	1.96	1.51	1,86	.36	16	2.05	2.31	1.42	1.99	.67	- 08
egal Methadone Use***	Number 55.36	of mill 47.55	igrams use 33.63	ed per da 50.92	y of u .033	se 35					•••	•••

n.

Drug Use and Purchase -65-Table V. 4 -- Dollar Values Consumed Per Day of Use or Value Purchas

\*Dollar amount based upon 10 days or less with use/ purchase . \*\*Zero days of use/purchase, mean amount and Pearsonian r cannot be computed. \*\*\*Legal Methadone is not included in the "any drug" total. (See instructions for reading tables at the end of Chapter VI.)

		Heroin Use	er Typology				Н	leroin User	Typology			
Type of Drug (Number of Subjects)	Irregula	r Regular (78)	Daily (62)	Total (201)	p of F	r	Irregular (61)	Regular	Daily (62)	- Total (201)	p of F	r
	A. D	ollar Valu	ie of Drug(:	s) Used Per	· Year:		B. Dolla	r Value of	Drug(s) Pu	rchased Pe	r Year	;
Any Drug Use	5,185.83	9,846.75	17,283.07	10,726.03	.000	.57	4,061.15	6,389.78	10,639.43	6,993.92	.000	.46
Heroin	1,388.64	6,431.45	13,189.42	6,985.59	.000	.66	985.75	4,018.87	7,601.23	4,203.38	.000	.62
Other Opiates	2.92	1.23	8.54	4.00	.13	.10	2.08	0.16	0.00	0.69	.13	13
Illicit Methadone	373,56	435.92	162.88	332.78	.01	15	338.16	374.94	133.73	289.37	.019	15
Cocaine	2,511.82	2,187.46	2,864.09	2,494.61	.47	.04	2,010.30	1,389.73	2,068.16	1,787.33	.22	.01
Al coho l	448.69	479.26	740.94	550.70	.003	.22	320.71	343.49	578.82	409.17	.002	.22
Marijuana	358.82	234.81	257.87	279.56	.18	10	316.22	192.64	219.25	238.35	.29	08
Amphetamines	18.45	1.79	2.97	7.21	.011	17	15.49	1.08	1.27	5.51	.010	18
Barbiturates	7.48	17.83	1.04	9.51	.58	03	4, 82	1.29	0.00	1.96	.15	13
Tranquilizers	49.68	39.44	42.87	43.61	.88	02	48.49	53.76	30.11	44.87	.62	05
Psychedelics	2.12	1.53	0.35	1.35	.65	07	2,38	1.53	0.20	1.38	.55	08
Other Drugs	23.65	16.03	12.10	17.13	.37	10	16.75	12.30	6.67	11.91	.38	10
Legal Methadone Use	Numb 6,067.37	er of mill 2,272.69	igrams use 539.68	d per year 2,889.75	.000	35						

. . . . . 1 - 1

Drug Use and Purchase -66-Table V. 5 -- <u>Dollar Value</u> of Drug(s) <u>Used</u> or <u>Expended</u> Per <u>Year</u> by Heroin User Typology

\*\* Dollar Amounts Used Per Year = Mean Amount Used per Day times 365. (See instructions for reading tables at the end of Chapter VI.)

2

5

TA.



## Nondrug Crime -67-CHAPTER VI NONDRUG CRIME AMONG HEROIN USER TYPES

How much crime do street opiate users commit? How much income to they obtain from crime? How is crime linked to a lifestyle of daily heroin use? These questions still continue to be a topic upon which there is little consensus -- other than the brief answer of "a lot." In this and the following two chapters, we will address these apparently simple questions. Our answers, however, cannot be brief because the criminal lifestyles reported by our subjects were more complex than we had initially believed.

As reported in Appendix A, we discovered that the phenomena of "drug dealing" was vastly more complex and frequent than we had anticipated; thus, a whole chapter (VII) is devoted to describing the complex roles and economic values associated with the "drug business." Other forms of crime are referred to in this report as "nondrug" crimes because they do not include the illegal exchange of drugs. As used in this chapter, "nondrug crimes" refers to robbery, burglary, shoplifting and other larcenies, forgery, conning, prostitution/pimping, fencing, and other crimes." After describing both nondrug and drug business crimes, this information will be combined in Chapter VIII to provide more direct answers to the questions listed above.

Thus, the current chapter will provide critical new information about different measures of nondrug crime among these street opiate users and briefly compare our data with levels of self-report criminality in similar studies among criminals or drug users who were at high risk of criminal activity. This chapter will not attempt to provide a theoretical understanding of the subject's criminal behavior nor develop a typology of criminal lifestyles (see Chapter XIII and XV). Nevertheless, new insights about the nondrug crime rates and dollar returns will be presented.

\*Ihis chapter considers only nondrug crimes having significant economic values; crimes such as homicide, assault, rape, etc. were not routinely asked.

1 .... 1 ... 12 10 Panin Ca i Farang G kan e 1277 Sec. . hanna t 1250.0 in a second ineres in the -..... ..... . . . . . **199**4/01/11/11 Search Street 05:046.08 general en - water 

1\_\_\_\_\_

1.54

------

The literature on criminal behavior of deviant groups and the general population has recently undergone an important shift in methodological emphasis. Prior to 1975, almost all research on criminal behavior consisted of either the analysis of the arrest and conviction histories of persons who were in contact (through jail, prison, or arrest) with the criminal justice system (such as the FBI's Uniform Crime Reports), or self-reported criminal behavior of juvenile populations, usually in a school system or a reformatory. The late 1970's saw several major studies attempt to ascertain levels of self-reported crime in special populations at high risk for committing crime. Five studies of adults at high risk of criminal behavior have reported results that are quite similar to those presented below. All of these studies obtained detailed data in which respondents recalled their criminal behavior

during the recent past (one month, year, or more, etc.).

McGlothlin, Anglin, and Wilson's (1977) retrospective analysis of persons committed to California Civil Addict Program showed that over 50% of all 1970 admissions were criminally active and almost half dealt drugs; they committed about 100 crimes per year, from which they derived a mean criminal income of \$7600 per year. During periods of daily narcotics use when not incarcerated between 1970-1975, they found that respondents had a total monthly income of \$1744, of which 55% came from nondrug crime (mainly burglary and theft) and 28% came from drug dealing; the remainder came from jobs, welfare and others. In a similar study, Ball, et al. (1979; 1981) studied the impact of heroin addiction upon criminality among 243 Baltimore addicts. The mean number of crime-days per addict was almost 2,000 spanning a decade or more at risk. Moreover, during period of addiction, these respondents reported almost 250

crime days annually; that is, they were criminally active two-thirds of the time when addicted to heroin.

In a study similar to the present effort, Inciardi (1980; 1981) obtained information about the drug use and criminal behavior of 166 youth recruited

## Nondrug Crime -68-Prior Research On Self-Reported Criminality Among High Risk Groups

Nondrug Crime -69-

from Miami streets in 1978. The 70 heroin users reported 24,670 offenses in the past twelve months, or a mean of 352 offenses per subject per year. Further, 96 nonheroin drug users committed 29,982 offenses or a mean of 312 per subject per year. Thus, differences in criminal behavior between heroin and nonheroin drug users were not particularly pronounced although heroin users were more likely to engage in drug distribution, shoplifting, theft from vehicles, and handling stolen goods than the nonheroin users sampled. This study pointed to the complexity of the relationship between persons who were criminally active and their frequent involvement in a variety of crimes.

Peterson, Braiker, and Polich (1980) provided data on the criminal behavior of 624 incarcerated male felons in five California prisons in 1976. They concluded (p. 149): "differences between offenders who use drugs and those who do not are not as great as those associated with other offender characteristics." They estimated the offense rates of street offenders from their data among those involved in Part I crimes. As street offenders, their respondents had a self reported <u>annual</u> offense rate of 1.97 for armed robbery, 2.38 for assault, 136 for drug sales, 7.23 for burglary, 7.56 for con games, 4.35 for forgery, and 3.48 for auto theft. Shoplifting, prostitution, and other offenses by these felons were not reported. These researchers did not study the economic value of the crimes committed by their respondents.

Chaiken and Chaiken (1982ab) studied prison and jail inmates in Texas, California, and Michigan via a self-administered questionnaire. They calculated the approximate number of days per street-year on which specific crimes were committed by heroin addicts, nonaddicted heroin users, nonheroin drug users and nondrug users. They show that high cost heroin users have considerably higher levels of criminal involvement than nonheroin drug abusers. A direct comparison between their data and the current economic behavior project respondents is reported in Table VI. 8 below.

<u>.</u> - Bath Mar -----144000 يغير ويرورون C. TANKER ..... . Newsalaw n Melanas di 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 manipulation ------C. Seran 2512.77 -And Party of Control o 1.5 1.5 March and a state WARKS ..... -Section 1 المستح والدوسانين 100000000 Sec. . ۲۳۳۳۹ (۲۹۲۹) 

All of the studies reviewed above rely extensively upon retrospective information by their respondents. In all cases, the respondents, most of whom were obviously active in crime, were asked to recall the extent of their involvement for periods of one year (Inciardi), three years (Peterson, etal.), or segments of their entire life since initiation to drug use (Ball et al., McGlothlin, et al.). While such retrospective recall provided information that was considerably better than further analysis of arrest or other institutional data, this current research effort isolated several reasons why respondent self-reports of criminal behavior and drug use over a long time period (month, year, decade) may have inaccuracies.

Nondrug Crime -70-

Specifically, our data demonstrated wide variability for each subject's criminal behavior in whether a particular offense was committed at all during a given day, in the number of offenses committed per day, and in the dollar returns from crime during a given period of time. When asked to recall offenses across lengthy time periods, however, the respondent must necessarily average such irregular activity; he/she appeared likely to forget involvement in minor offenses which he committed numerous times. Stealing meat or clothes with a low value per item seemed so normal to some respondents that they were not be considered crimes nor would specifics about numerous minor crimes be recalled when averaged across long time periods.

In addition, Goldstein (1979, Goldstein, et al., 1982b) suggested that an opiate user's estimate of heroin habit size was frequently larger than actual consumption, in large measure due to the respondent's self image. When there is a discrepancy between self image and actual heroin-using behavior, the heroin user appeared to give information congruent with his self-image. Peterson, et al (1980,78) likewise reported that a respondent's criminal self-identity as "robber," "burglar," or "addict," was strongly related to the extent of criminal activity. Thus, a criminal self-image may also lead to inflated self-reports of criminal activity and income. Nondrug Crime -69-

from Miami streets in 1978. The 70 heroin users reported 24,670 offenses in the past twelve months, or a mean of 352 offenses per subject per year. Further, 96 nonheroin drug users committed 29,982 offenses or a mean of 312 per subject per year. Thus, differences in criminal behavior between heroin and nonheroin drug users were not particularly pronounced although heroin users were more likely to engage in drug distribution, shoplifting, theft from vehicles, and handling stolen goods than the nonheroin users sampled. This study pointed to the complexity of the relationship between persons who were criminally active and their frequent involvement in a variety of crimes.

Peterson, Braiker, and Polich (1980) provided data on the criminal behavior of 624 incarcerated male felons in five California prisons in 1976. They concluded (p. 149): "differences between offenders who use drugs and those who do not are not as great as those associated with other offender characteristics." They estimated the offense rates of street offenders from their data among those involved in Part I crimes. As street offenders, their respondents had a self reported <u>annual</u> offense rate of 1.97 for armed robbery, 2.38 for assault, 136 for drug sales, 7.23 for burglary, 7.56 for con games, 4.35 for forgery, and 3.48 for auto theft. Shoplifting, prostitution, and other offenses by these felons were not reported. These researchers did not study the economic value of the crimes committed by their respondents.

Chaiken and Chaiken (1982ab) studied prison and jail inmates in Texas, California, and Michigan via a self-administered questionnaire. They calculated the approximate number of days per street-year on which specific crimes were committed by heroin addicts, nonaddicted heroin users, nonheroin drug users and nondrug users. They show that high cost heroin users have considerably higher levels of criminal involvement than nonheroin drug abusers. A direct comparison between their data and the current economic behavior project respondents is reported in Table Vi. 8 below.

-1.000 • . الأسبي No. -الأتري مراحلتهم ~~~? -120 1.00.00 ې يېسې کې -------- series When some it n Maria da An T. (2) (1) (2) -Barren af Sentitive ----fotos de S Asse: 7.77 Rept ------Jan 14 W - 3 Service The service of the service o week-open ?" \_\_\_\_\_\_ \_\_\_\_\_

All of the studies reviewed above rely extensively upon retrospective information by their respondents. In all cases, the respondents, most of whom were obviously active in crime, were asked to recall the extent of their involvement for periods of one year (Inciardi), three years (Peterson, etal.), or segments of their entire life since initiation to drug use (Ba<sup>11</sup> et al., McGlothlin, et al.). While such retrospective recall provided information that was considerably better than further analysis of arrest or other institutional data, this current research effort isolated several reasons why respondent self-reports of criminal behavior and drug use over a long time period (month, year, decade) may have inaccuracies.

Nondrug Crime -70-

Specifically, our data demonstrated wide variability for each subject's criminal behavior in whether a particular offense was committed at all during a given day, in the number of offenses committed per day, and in the dollar returns from crime during a given period of time. When asked to recall offenses across lengthy time periods, however, the respondent must necessarily average such irregular activity; he/she appeared likely to forget involvement in minor offenses which he committed numerous times. Stealing meat or clothes with a low value per item seemed so normal to some respondents that they were not be considered crimes nor would specifics about numerous minor crimes be recalled when averaged across long time periods.

In addition, Goldstein (1979, Goldstein, et al., 1982b) suggested that an opiate user's estimate of heroin habit size was frequently larger than actual consumption, in large measure due to the respondent's self image. When there is a discrepancy between self image and actual heroin-using behavior, the heroin user appeared to give information congruent with his self-image. Peterson, et al (1980,78) likewise reported that a respondent's criminal self-identity as "robber," "burglar," or "addict," was strongly related to the extent of criminal activity. Thus, a criminal self-image may also lead to inflated self-reports of criminal activity and income.



While these issues of underreporting cannot be resolved here, estimates of offense probabilities based upon self-reported data involving lengthy recall periods may not be very precise, especially for minor crimes which occur frequently but with minor cash income for the street opiate users. Such self-reported crimes by groups at high risk for criminality, however, still provided much better information about their actual criminal behavior than the alternatives (FBI offense rates; juvenile arrests).

The detailed data about nondrug crimes have been presented in Tables VI. 1 - 7 at the end of this chapter. The central findings have been briefly summarized in a narrative which answer major questions stated in the text. The major findings will be presented as one sentence summaries; the evidence: (from Tables VI, 1-7) will be briefly described in a paragraph or two. Next findings about each major offense type (robbery, burglary, shoplifting for resale, other larcenies, and other crimes) will be provided in short paragraphs. This chapter also includes "vignettes" of persons who do particular kinds of crimes and provides a brief summary of their behavior from the data and from notes which interviewers made about specific crimes they reported. Such vignettes bring rich flavor and better understanding about the lifestyles of one or two subjects and the kinds of crimes they commit.

The chapter concludes with definitions and examples of the offense classes, and an introduction to the statistical data. This includes an explanation of how the data were standardized and the significance tests employed, a summary of what each table presents, and then the tables themselves.



Daily heroin users were criminally active on more days than regular or irregular heroin users, but such differences were important mainly for four kinds of crime -- robbery, burglary, shoplifting, and other larcenies. Daily heroin users did not commit more crimes per day or gain larger cash returns per crime, rather they were had more days with crime than the less regular heroin users. As a result, daily heroin users have more nondrug offenses and criminal cash income than regular or irregular heroin users. The frequency of heroin use was weakly or unrelated to other kinds of nondrug criminal behavior (forgery, con games, prostitution/pimping, other illegal acts, and shoplifting for own use).

### In what ways did the criminal behavior of daily heroin users differ from less regular heroin users?

Tables VI. 1 - 7 exhibit several important findings:

than regular or irregular heroin users.

Table VI. 1 shows that about 25% more daily heroin users than irregular heroin users reported a robbery, burglary, or larceny (other than shoplifting). While daily heroin users had somewhat higher proportions reporting shoplifting for resale, forgery, con games, and other illegal acts, and somewhat lower proportions engaging in prostitution/pimping, they were not significantly different from the less regular heroin users.

day.

Table VI. 3B shows that the number of offenses per criminally active day (x 100) was relatively similar among daily and irregular heroin users, respectively, for robbery (113 vs. 109), burglary (117 vs. 109), other

Nondrug Crime -72-A. MAJOR FINDINGS

### Does nondrug criminal behavior vary by heroin involvement?

1. A larger percentage of daily heroin users engaged in robbery, burglary, and other larcenies -- but not other types of crimes --

2. Daily heroin users did not commit more crimes per criminally active

### Nondrug Crime -73-

larcenies (124 vs. 109). Daily heroin users did not have the highest number of offenses per criminally active day  $(x \ 100)$  for shoplifting, congames, pimping, and other illegal acts, although they were somewhat higher for forgery and prostitution.

3. Daily heroin users were not more successful in obtaining a higher cash return per crime or per criminally active day from different offense classes or from the average nondrug crime.

Table VI. 5 shows that daily heroin users had lower cash returns per offense than the regular heroin users for robbery (\$77 vs. \$91) and forgery (538 vs. 5256), and other illegal acts (532 vs. 547). Daily heroin users had somewhat higher returns per offense for burglary (\$86 vs. \$78). shoplifting (532 vs. 529) and other larcenies (542 vs. 529) than regular heroin users. No consistent positive linear association was observed between the frequency of heroin use and cash income per specific offense.

Moreover, daily heroin users did not have a significantly higher cash return from their average nondrug crime (\$41) than regular (\$35) and irregular heroin users (525); the modest linear trend (r=.14) was not substantial. The association between the frequency of heroin use (i.e., the Heroin User Typology) and cash income per criminally active day was even less strong. (Table VI. 6).

4) Daily heroin users have more criminal offenses mainly because they commit crimes on more days.

Table VI. 2 shows that daily heroin users committed a nondrug crime about every third day (33% of the person-days), regular heroin users did so every fourth day (23%), and irregular heroin users did so every seventh day (14%). Table VI. 7B shows that daily heroin users, when compared with irregular heroin users, annually committed about twice as many nondrug crimes (209 vs. 116).



# more days than less regular heroin users.

Table VI. 2 also shows daily heroin users committed robbery (2.8%) and burglary (8.0%) on about twice as many person-days as regular heroin users (1.1% - robbery; 3.9% - burglary), and four times as many days as the irregular heroin users (0.6% - robbery; 1.1% - burglary). Likewise, daily heroin users committed shoplifting (13.6% vs. 4.8%) and other larcenies (4.3% vs. 4.8%)vs. 1.4%) on about three times as many days as the irregular heroin users. All other offenses (forgery, congames, prostitution, pimping, other illegal acts, and shoplifting own use) did not exhibit variation in the proportion or person-days active by the frequency of heroin use.

Table VI. 7B shows that daily heroin users, when compared with regular heroin users, annually committed twice as many robberies (11.8 vs. 4.1) and burglaries (33.9 vs. 15.1), and about one and a half times as many episodes of shoplifting (71.9 vs. 46.3).

had the highest criminal incomes.

Table VI. 7C shows that daily heroin users had an annualized cash income (\$8,540) from nondrug crime which was about one and a half times higher than that of regular heroin users (\$5,719), and almost three times higher than irregular heroin users (\$2,885). This higher criminal income came mainly as higher cash income from robbery, burglary, shoplifting, and other larcenies. Daily heroin users either had equivalent or lower cash income from other crimes when compared to their less regular heroin using counterparts.

Table VI. 4 shows that the typical respondent obtained \$16 per day in cash income from nondrug crime. Daily heroin users obtained \$23 on the average day, compared with \$16 for regular heroin users and \$8 for irregular heroin users. Most of this differential came from their higher cash incomes from robbery, burglary, shoplifting, and other larcenies. \* - These figures conceal considerable variation between respondents and for

Nondrug Crime -74-

5) Daily heroin users committed robbery, burglary, shoplifting, and other larcenies, but not other nondrug offenses, on significantly

6) Because they had more days of criminal activity, daily heroin users

7) The cash income per day<sup>\*</sup> from nondrug crime was relatively modest.

### Nondrug Crime -75-

Table VI. 8 -- Number of offenses per street year involved in specific crimes: Comparison of heroin users (this study) with prison and jail inmates in California, Michigan, and Texas.

	New Yo	ork +	Prison a	nd Jail Inm	ates Who Were
_	Heroin U	ser Type	Calif.	Michigan	Texas
Type of Nondrug crime (Number of Subjects)	Irregular (61)**	Regular (78)**	(94)*	(82)*	(48)*
	AMON( Numbe	G HEROIN USE er of Offens	RS WHO WEI	RE PROBABLY reet Year Co	NOT ADDICTED, ommitting:
Robbery	2	4	13	9	4
Burglary	5	15	31	35	29
Thefts	41 (sum)	67 (sum)	25	24	43
Shoplifting (Resale	e) 35	46			
Other Larcenies	6	21			
Forgery	1	1	13	14	7
Con Games (Fraud)	36	34	11	15	12

Turne of Nordaux Custors	New Stree Heroin Us	(ork et <u>er Type</u>	Prison and "Heroin Addi Calif. N	Jail Inma Iction, Al Michigan	tes Who Were 1 Months" in: 
(Number of Subjects)	Ually Hero (62)*	th Users	(204)*	(94)*	(59)*
	AM Nu	ONG HEROIN Imber of Of	USERS PROBAG fenses per St	BLY ADDICT creet Year	ED, Committing:
Robbery	12		34	17	5
Burglary	34		68	26	35
Thefts	92 (sum)		66	50	108
Shoplifting (Resal	e)	72			
Other Larcenies		20			
Forgery	3		18	7	20
Con Games (Fraud)	29		18	18	24

\* Source: Chaiken and Chaiken (1982b,161; Chaiken, 1983) \*\*Source: Table VI. 7B.



# How similar was the criminal behavior of these street opiate users when compared to heroin users in other high crime populations?

The level of criminality reported by Economic Behavior subjects was similar to, or generally higher than, comparable groups of subjects in other studies of addicts or career criminals.

In the introduction section of this chapter, rates of criminal involvement from other studies were briefly presented. When compared with the data in Tables VI. 1 - 7, Economic Behavior subjects exhibit rates of involvement in nondrug crimes that equal or exceed those reported by McGlothlin et al. (1977), Inciardi (1980; 1981), Ball, et al. (1979, 1981), and Peterson, et al. (1980), or Collins, et al. (1982abc). [Table VI. 8 about here]

A recent study by the Rand Corporation (Chaiken and Chaiken, 1982ab) of inmates in jails and prisoners in California, Michigan, and Texas provided data on the approximate number of offenses per year during which they committed a variety of nondrug crimes while at liberty on the street. Data from Chaiken (1983) were closely comparable to data in this study. Table VI. 8 shows that the number of offenses per year was roughly similar for most offense types, with considerable variation from state to state.

Specifically, among "heroin users who were probably not addicted," the annual robbery rate by New York regular heroin users (4) was similar to their counterparts in Michigan (9) and Texas (4), but lower than in California (13). The New York irregular (4) and regular (15) heroin users reported about half as many burglaries per year as their counterparts in these three states (31, 35, 29). On the other hand, New York regular heroin users (67) and irregular heroin users (41) had equal or higher theft rates than career criminals in other states (25,24,43). While forgery rates were lowest in New York (1), they had higher rates of fraud/con games (about 30) than their counterparts in the other three states.

Nondrug Crime -76-

### Nondrug Crime -77-

Likewise, among "heroin users who were probably addicted," the robbery rate among New York daily heroin users (12) was intermediate between Texas (5) and Michigan (17), but less than in California (34). The burglary rate among New York daily heroin users (34) was intermediate between (26) Michigan and (35) Texas but lower than California (68). The number of thefts per year by New York daily heroin users (92) was between California (66) and Texas (108). On the other hand, New York daily heroin users had lower involvements in forgery but were higher in fraud than their counterparts in other states.

These studies measured the frequency of criminal behavior in different ways (Chaiken and Chaiken by self-administered questionnaires; this study by daily or weekly interviews) and defined heroin use in somewhat different ways (Chaiken and Chaiken by the respondent's self-reported heroin addiction during a two year period while at liberty, this study by daily reports). Chaiken and Chaiken systematically sampled inmates in prisons and jails in the three states, while subjects were recuited from the streets of two communities of Manhattan. Nevertheless, the resulting annualized offense rates by heroin users appears remarkably similar in all four states. Heroin users in no state were systematically higher than other states on all offense classes. When compared to the Chaiken and Chaiken data, New York street heroin users appear less active in burglary and forgery, but more active in theft and fraud.

In all states, the group of subjects classified as most active in heroin generally appear to be much more criminally active (although not for every crime type) than their less active heroin using counterparts. The same finding we document here.



## B. HEROIN USER I How many subjects

Over a quarter of all subjects reported a robbery; almost half of the daily heroin users did so. A robbery occurred on about two percent of the person-days. The annualized rate among all subjects was 5.9 robberies per year with daily heroin users (11.8) twice as active. For all subjects, robbery seldom occurred more than once per day.

Robbery was an effective crime for raising relatively large amounts of money. The average robbery resulted in \$80 of cash income; the daily and regular heroin users had slightly (but not significantly) higher cash returns from robbery than irregular heroin users per robbery or per robbery day. Because they had more days with and acts of robbery, the annualized robbery income of daily heroin users (\$906) was twice as high as that of regular heroin users (\$377) and over five times as high as that of irregular heroin users (\$158). The following vignette provides a brief descriptions.

## Vignette E -- A Street Robber Doing Robberies

Geraldo N (Hispanic male, age 25) was a daily heroin user and one of our most active robbers (He was also equally active in burglary.) He was interviewed for 117 days, during which he reported 22 different robberies on 17% of his person-days; his annualized robbery rate was 69. His returns from robbery were relatively modest at \$43 per robbery or \$48 per robbery day. On an annual basis, he would have obtained \$2,964 in cash from his robberies. As the following interviewer representative notes about his robbery episodes show, he had many small scores and only an occasional large (over \$100) score from robbery.

2/25/81	**	"I mugged this young
7/10/81		"I snatched lady
• •		pocketbook. She ha
		food coupons to a st
6/11/81	-	"Respondent and par
		cassette player, wa
		different individual
11/7/81	-	"With partner, stuck
2/26/82	-	"Mugged a guy for \$3
		for self (with knife
2/28/82	÷	"Mugged a guy for a
3/5/82	-	"Geraldo and partner
		knife on them. They

## Nondrug Crime -78-B. HEROIN USER INVOLVEMENT IN SPECIFIC MAJOR CRIMES

## How many subjects become involved, with what frequency, and how much money do they gain from specific types of crime?

### ROBBERY

g guy. All he had was \$11." in the street by the neck and took her id \$40 cash and \$40 in food stamps. Sold the tore owner for \$23 cash." rtner snatched a man in the street and took atch, and ring. They sold these articles to ls for \$85 cash." k up a grocery store; each made \$158." 37 cash, l watch (sold for \$13), l coat - kept e)." radio, sold for \$20 (with knife)." r grabbed a man and lady in subway and pulled

## Nondrug Crime -79-BURGLARY

Over two-fifths of respondents reported a burglary; over half of the daily heroin users did so. A burglary occurred on about four percent of the person-days; daily heroin users were active 8 percent of the days. The annualized rate among all subjects was 17.7 burglaries per year, with daily heroin users about twice as active (33.9 burglaries per year).

Burglary was as rewarding per offense as robbery. The average burglary resulted in \$81 of cash income; days with one or more burglaries provided \$91 in cash income. The annualized burglary income of daily heroin users (52,906) was almost three times higher than that of regular heroin users (\$1,174) and over ten times higher that of irregular heroin users (\$256). For daily heroin users, burglary was the single crime providing them with the most cash income. The following describes burglars and their burglaries.

### Vignette F - Burglars Doing Burglaries

Kit G. (hispanic male, age 26) also did robberies, but did burglaries on 9% of his person days. He reported a relatively "sophisticated" burglary.

8/7/80 - Kit went to Brooklyn to get \$1500 from his sister, but his sister, who works, was not able to get to the bank. Kit came upon the annex of an appliance store, broke in, and, using a handtruck, stole 3 refrigerators. 'I just picked the locks in the store. I know how to do that good.' He sold each refrigerator for \$100 to three people he knew: one to a store owner, one to a social club, and one to a friend in Brooklyn who just got married.

Nirgil N (Hispanic, male, age 34) used heroin daily. Interviewed for 117 days, he committed a burglary on 52% of his person-days, averaging \$32 per burglary. On an annual basis, he would have committed 193 burglaries and earned \$6,071 form them. On over 50 occasions, Virgil and one or two other partners burglarized abandoned buildings to get brass pipes and copper and sold these items to junk yards; Virgil earned from \$10 to \$40 for these crimes. However, Virgil's burglaries were not limited to these small scores: 11/29/80 - With a friend, burglarized a grocery store for \$310 cash.

12/3/80 - With 2 partners, burglarized an apartment and got a TV and some jewelry. Items were sold on the street for \$300; Virgil got \$70, 12/18/80 - Burglarized a downtown apartment by himself, got a camera, a cassette player, and a diamond ring. Sold on the street for \$400. 12/22/80 - Burglarized a grocery store with a partner; split \$227. 7/11/81 - 'Lookout-man' for two friends who burglarized a grocery store.

Took an assortment of merchandise and cash; he received \$98. 11/20/81 - Broke into a grocery store; took about \$200 worth of groceries; sold these goods to persons on the street for \$140.

- and the second second	·
n geologia Maria	
	an a gran
Contract of Sales	A DAY OF THE OWNER OF
ve "1"" <sup>(A,2)</sup>	200 m. ( <b>9</b> 1
	9941 (VI)
i de marca d'An 1	lines <del>yn</del> r
	No. of Concession, Name
	ana ita ingga
	Ne con comp
and the second se	Provide Contraction
. Januar Britt	Ngan an a
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	THE REPORT OF
. and all	Kartur I <del>Aya</del>
and the second sec	1865° w 1
- e- watering	4999
C. Water	and the second sec
	atter (S. 1997) - 1
	1400 C
	Second Contemporate
د معند معند معند معند معند معند معند معن	Normal Contract
and the second se	ala a
्र <b>्भ हेई</b> ।	908 (* 1. j. 17) 19 - J
and the second sec	alaha Kata ang kanalaha kata ang kanala
	yr result o'r Mae
and and a second se	Annes Carlos
	description (1777) 
and the second sec	
	##************************************

Three-fifths of respondents reported shoplifting. Shoplifting occurred on over eight percent of the person days; daily heroin users were active on about 14 percent of the days and irregular heroin users on 5% of the days. Unlike robbery and burglary, shoplifting was likely to involve 1.6 shoplifting events on active days. The annualized rate of shoplifting among all subjects was 50.7; daily heroin users were more active with 71.9 per year.

Because it was committed so often, shoplifting raised relatively large amounts of cash income. But the returns per shoplifting offense (about \$30) was less than half that of robbery and burglary. Due to multiple shoplifts per day, however, these street opiate users obtained about \$49 per shoplifting The annualized shoplifting income of daily heroin users (\$2,334) was day. about twice as high as that of regular heroin users (\$1,342) and about three times greater than irregular heroin users (\$887). Shoplifting was the single crime providing all subjects, excepting daily heroin users, with the highest amount of cash income because of the large number of shoplifting events.

W.J. (Black male, age 37) was a daily heroin user who specialized in "boosting" from stores; he "retail fences" (sells) these items to people in the street, bars, or other neighborhood locations. He was interviewed for 33 days, and committed shoplifting on 55% of his person-days. He earns anywhere from \$20 to \$150 per incident, averaging \$36. On an annual basis, he would have committed 376 such offenses, earning \$13,483. Some examples: 11/12/81 - Stole 1 coat, 2 pairs of shoes, and cosmetics from 4 different stores; got \$100. 11/13/81 - Stole 5 pairs of adult jeans (worth \$25-\$30 each), 6 pairs of childrens jeans (worth \$12-\$15 each), 8 ski masks, 2 shirts, a belt, and some socks. Kept the belt, sold all the other items for a total of \$114. 11/14/81 - Stole 2 coats from a department store; got \$150. 11/16/81 - Stole a coat and a pair of jeans from 2 stores, sold the items to 2 persons and made \$80. Klip N (Black male, age 30) used heroin daily, was a low-level thief

specializing in 'cattle-rustling,' or stealing meat from grocery stores and selling it to neighborhood people who bought whenever it was available. He was interviewed for 33 days and shoplifted on 28 of these days (85% of his person-days). He committed 42 shoplifting incidents, 27 (68%) of which involved stealing and reselling meat; Klip got an average \$15 per theft.

Nondrug Crime -80-SHOPLIFTING FOR RESALE (see definitions below)

### Vignette G - "Boosters" and "Cattle Rustlers"

## Nondrug Crime -81-OTHER LARCENIES (see definitions and examples below)

Half of all respondents reported other larcenies, with daily heroin users most active. Such larcenies occurred on about 3% of the person days. The regular heroin users were as or more active than daily heroin users in this crime per day or per larceny-day, although daily heroin users obtained slightly more money per larceny. The annualized rate among all subjects was 16 other larcenies; irregular heroin users were considerably less active -with 5 larcenies per year. The cash returns per other larceny (\$36) were similar to shoplifting, and since multiple larcenies occur per day, these The annualized larceny street opiate users obtained \$46 per larceny-day. income of daily heroin users (\$818) was similar to that of regular heroin users (\$730) and larger than among irregular heroin users (\$142).

Very few subjects specialize in committing other larcenies. Rather persons who also committed in robbery, burglary, and shoplifting, committed other larcenies when the opportunity was present.

### Vignette H -- Multiple Offenders Doing Other Larcenies

Dino D. (black male, age 42) was a regular heroin user. He specialized in 'popping shorts' or 'car popping', ie., stealing merchandise or parts from automobiles. He also burglarized apartments and committed other types of larcenies. Dino was interviewed for 61 days, committing larcenies on 33% of his person-days with an average of \$39 per larceny offense. Annually, he would have committed 144 such offenses, making \$5,625. Some examples of Dino's 'car popping' and nonshoplifting larcenies:

9/18/81 - Stole 2 tires from a car - sold them for \$20. 9/21/81 - Broke into a car and got 3 tool boxes - sold them for \$45. 1/13/82 - Stole box of thermal underwear off truck sold to fence for \$80. 10/2/81 - Broke into a car and stole a portable radio, sold it for \$40. 10/12/81 - Stole TV from car, sold for \$45; also, with partner stole bike, sold for \$30, which was split. 11/17/82 - Broke into car, stole AM/FM cassette player, got \$40. 11/18/82 - Stole tire from van, sold to individual in the street for \$30.



### Nondrug Crime -82-MISCELLANEOUS CRIMES (Forgery, Con Games, Prostitution/Pimping and Other Illegal Acts -- see definitions and examples below)

Generally less than a third of respondents engaged in each of these behaviors, with the exception of "other illegal acts." Likewise, excepting prostitution, these crimes occurred on less than 2% of the person-days. On days that they occurred, con games appeared to be most frequent -- with almost five events per day. Such con games were mainly three-card monte players, shills, or lookouts conducting several games. Prostitution occurred about twice per day of involvement.

While the cash returns per active day were substantial (\$50 or more) for forgery, con games, prostitution, and pimping, they generate relatively modest cash returns on an annual basis (forgery -\$162; con games - \$406; prostitution - \$742; pimping - \$100, other illegal activities - \$339) with little variation by the frequency of heroin use.

Nefertiti K. (black female, age 32) was a regular heroin user who engaged in burglary, shoplifting, and prostitution, but was involved in forgery on a greater percentage of her person-days (9%) than any other crime. Annually, she would have committed 33 forgeries per year, making \$20,031. She had a friend who worked in the payroll section of a bank. They obtained a false ID and applied for loans. One time she applied for a \$2,000 loan but only got approved for \$1,500; she gave her friend \$500. Another time a loan for \$1,250 was approved, and they split this amount evenly. She also stole checks from mailboxes; on one occasion she stole a \$256 check, took it to a fence, and received \$186.

### Vignette J -- Con Games and Three Card Monte

Gabby E. (Hispanic male, age 25) was a regular heroin user. He engaged in a variety of crimes, but specialized in cons, running a 3-card monte game on a particular street corner of mid-town Manhattan. He was involved in these cons on 33% of the 124 days he was interviewed, making \$84 per day and \$12 per offense. Annually, he would have committed 820 cons (games) and made \$10,096. Typically Gabby worked with 2 or 3 partners; he deals and they work as shills and lookouts, although on occasion roles were switched. (Once, while he was the lookout-man, Gabby picked the wallet of a man who was watching the game.) Usually, as the dealer, he took a greater share of the earnings. 11/3/80 - 9 games-made \$125; \$50 for self, \$50 to shill, \$25 to lookout. 11/4/80 - 13 games-made \$220; \$100 for self, \$70 to shill, \$50 to lookout. 11/5/80 - 7 games-made \$75; \$40 for self, \$20 to shill, \$15 to lookout. Chased by police. 11/7/80 - 12 games-made \$350; \$150 for self, \$125 to shill, \$75 to lookout. 11/8/80 - 8 games-made \$200; \$75 for self, \$75 to shill, \$50 to lookout.

### Vignette I -- Forgery Events

Nondrug Crime -83-

### Vignette K -- Prostitutes and Pimps

Kat E. (black female, age 35) was a daily heroin user and engaged in prostitution on 61% of her person-days. She averaged \$85 per day and \$21 per incident. Annually, she would have engaged in 873 such incidents and made \$18,692. She charged \$20 per session, and tried to average 5 customers a day; 'Johns' must also pay for the hotel room. On one day she earned \$150 but when our interviewer asked, "How many tricks was that from?" Kat replied, "I don't know. I was on my back all day, I couldn't keep count." Kat also engaged in various other crimes. For example, on several ocassions she shoplifted food from grocery stores. One time she and a partner (who had a pistol) 'took off' an ice cream truck for \$250, which was split evenly.

Sleepy D. (black male, age 37) was a robber and dealer who used heroin regularly. He engaged in a wide variety of criminal activities, including shoplifting for resale, forgery, and robbery. On 24% of his person-days he was involved in pimping, earning \$64 per incident. Annually he would have engaged in pimping on 88 days, making over \$5,600 from this activity. Sleepy reported 8 pimping incidents to our interviewers. One time his girl friend 'turned a trick', earning \$30, which was split evenly. The following week he reported 4 pimping incidents, earning \$320. On one of these days Sleepy watched the back' of another friend while she turned 2 tricks; she gave him \$14 for doing this.

### Vignette L -- "Prostitutes" as Lures for "Chump" Robberies and Larcenies

Several female subjects poised as prostitutes to bring "tricks" to settings where a male partner would rob them -- they referred to this as a "chump robberv." In addition. they would also steal money from men in bars, or pick pockets of potiential customers ("chump larcenies"). Rarely, however. would they actually have sex with a "John" for money.

Sue S. (Hispanic female, age 34) was a low-level robber and a daily heroin user. During her life-history interview she informed our interviewer that for 10 months (but 'just on weekends') in 1972 she 'set up' about 12 tricks a night. She posed as a prostitute, pretending to take them home, she either robbed them herself or with a partner. During her weekly interviews she reported 4 such incidents:

12/11/81 - Set up 2 tricks. A friend robbed them of \$230. Money was split. 12/15/81 - Took home a drunk from bar and robbed him on the way. 12/17/81 - Robbed drunk from bar on way home for \$93 12/27/81 -With partner got \$130 in a con game making believe she was a prostitute. Split the money evenly with partner.



Kit G. (Hispanic Male, age 26) was a regular heroin user. On one occasion, Kit accepted an offer to earn \$225 from an older man who lived in his neighborhood to beat up a man who had 'stolen' his young wife. Kit received a \$100 down payment and beat the man up on 8/5. Also, the older man told Kit of a 'good score' worth \$10,000 that Kit could do shortly. Kit told our interviewer, "I'll tell you about it when I do it." On 8/6 Kit received the rest of the money from the older man. He was supposed to receive \$125 but Kit conned him out of an extra \$60 by telling him that a friend helped on the job and the extra payment was for his friend.

BDJ;8258A;0742A;bj

Nondrug Crime -84-

Vignette M -- Other Illegal Acts

## Nondrug Crime -85-DEFINITIONS OF NONDRUG CRIMES

The Uniform Crime Reports definitions were followed as closely as possible. Brief definitions of crimes and examples of the more complex crimes have been provided below (also see Vignettes for other examples.)

- Robbery -- was the taking of money or property from another person by force or threat of force.
- Burglary-- was breaking into or forcible entry into a building/apartment; this usually includes taking of money or property, but there was no personal confrontation with the victim. Included unlawful entry and attempted breaking and entry with the intent of stealing.
- Shoplifting was the taking of merchandise from a business or store and (Resale) -- reselling it for cash income. Sometimes subjects shoplifted merchandise which they traded for drugs having a standard economic value which was considered as the equivalent of cash income.
- Shoplifting was taking merchandise from a business or store, but keeping (Own Use) -- it for own use or to give to another person as a gift (i.e., no cash income was obtained).

Special Note Regarding Shoplifting Events and Amounts.

Respondents occasionally obtained several items of merchandise from a given shoplifting event, they sold most of it for cash (i.e. resale), but kept one or two items for personal use or as a gift (i.e. own use). This event was coded as one offense of "shoplifting (resale)" but not as a "shoplifting (own use)" event. However, the value was entered in both places. Thus, income from the sale of the merchandise was entered as "cash income from shoplifting-resale" and the approximate dollar value of the kept merchandise was entered as "value of shoplifting own use."

Since "shoplifting (own use)" may involve a dollar value, but not be an event, and since the value of merchandise kept for own use may be different than what would be obtained from an actual sale, the tables below present "shoplifting (own use)" as a separate category which is not added into the totals for nondrug crimes.

Other

involved almost all other forms of theft and includes stealing Larcenies-- from trucks or vehicles ("car popping" to our subjects), or stealing/taking parts from vehicles, pocket picking or purse snatching, taking merchandise from a delivery truck, stealing from family or friends, ordering and eating a meal then walking away without paying, and a variety of other similar behaviors. Also included in this category was auto theft; although the Uniform Crime Reports included Vehicle Theft as separate from larceny, only three auto thefts were committed by our low income subjects -- few of whom have learned to drive and/or steal cars.



	slips in order stolen checks o
ne s	involved obtai delivered, or involved the re game. Thus, n little income.
tutio	n- involved fema customer. Ma were frequent backs" (protec or other crimi was a quasi-emp
g	involved the woman who had subjects were for them. Rat (and usually t tricks" and the
1	involved a wide committed frequ being a dista recommending a woman's name; assist him or involved dire accomplice, bu stolen merchan fencing).
	By far, th respondents en

ne most common offense in which a majority of ngaged on a relatively routine basis was "theft of services," especially not paying subway or bus fare. But such fare evasion has not been included as a crime in these tables because it does not result in cash income (i.e., it is an avoided expenditure -- See Chapter IX).

### Nondrug Crime -86-

involved falsely signing names to checks or credit card purchase to obtain goods or money. They typically forged or credit card slips.

> ning money by promising something that cannot be other such fraud. The most common such event espondent as a leader or shill in a three-card monte numerous events (i.e., games) may have resulted in

> ale respondents having sex for money from a paying les engaged in such acts on rare occasions. Males ly hired by female prostitutes to "watch their ct them from being assaulted or robbed by customers nals). This differed from pimping in that the male ployee and nonsexual friend (usually) of the female.

> respondent receiving money from a prostitute or a obtained funds from her sexual activity. Male seldom classic "pimps" with several girls working her, pimping occurred when their spouse or current temporary) girl friend was sent out to "turn a few en gave the respondent money.

> le range of offenses which typically may be uently but had low dollar returns. This included ant accomplice to a crime (i.e., being paid for female who will forge a stolen credit card having a helping a burglar find another person who will identifying rich victims -- the respondent is not ectly in the crime as a participant or direct ut is paid for his referral activity). Selling idise which the respondent did not steal (a form of

Nondrug Crime -87-D. INSTRUCTIONS FOR UNDERSTANDING TABLES VI. 1 - 7

The following tables present a variety of measures for examining criminal behavior and the dollar returns from crime. Only the most important findings have been highlighted in the narrative above. Crimes which few subjects committed, that occurred infrequently, and/or raised little money have been included in these detailed tabulations.

Standardization of Respondent Criminal Activity and Cash Returns

This study was unique in that it obtained data on a <u>daily basis</u> about the number of criminal acts and the dollar returns from such activity. In fact, there were really <u>four</u> major units of analysis which can and will be used in meaningful ways: the individual <u>respondent</u>, the day or <u>person-day</u> (a 24 hour period for one respondent), the criminal <u>offense</u>, and the <u>dollar value</u>. Measurement of the dollar value was even more complex than anticipated because the returns from crime were in the form of: a) cash income (actual dollars received in a robbery or received from the stolen merchandise), or b) the approximate dollar value of the merchandise kept or drugs obtained when a direct conversion to cash did not occur.

In the data presented below, the actual reports (raw data) by a given respondent for specific person-days have been statistically standardized. That is, each of the 201 subjects contributed at least 33 days of data, but several subjects were interviewed for 60 or more days. Thus, the information for each respondent was standardized by dividing the number of days involved in a given behavior (i.e. robbery) by his total number of reporting days to provide an average per day.



Thus, even though a given respondent may be much more active than others, all data have been adjusted so that he contributes an equal number of person-days to the mean amounts presented in the tables below. After computing for each subject, the mean number of days, mean number of events, or mean dollar amounts per person-day, each individual subject's means have been summed and averaged across the 201 subjects (or heroin user subgroups) have been computed and presented in Tables VI. 1-7.

The standard deviations have <u>not</u> been presented because the tables were already complex. Moreover, only <u>between</u> subject variation was included here. The daily (day-by-day) variation within subjects has not been included but was very extensive. Nevertheless, for almost all means presented, the inter-subject standard deviation around a given mean was generally about as large as the mean itself. The reason for this substantial variation was that for any given behavior, such as robbery, a very sizable number of subjects are zero (they have done no robbery), but a small number have been quite active (many robberies and high robbery income). Thus, the noninvolved subjects kept the grand mean relatively small while the very active subjects greatly increased the standard deviation.

Nevertheless, an Analysis of Variance (ANOVA) has been undertaken to show whether the differences in the means of the daily, regular, and irregular heroin users were significantly different from each other. This ANOVA has been performed for each type of offense and each measure of activity (i.e. offense rates per day). These results are presented in the next to last column headed "p of F"; this is the significance level (p) of the "F" test in the one way analysis of variance of differences between group means.

Likewise, since the independent variable (Heroin User Typology) might be hypothesized to be positively linked to almost all forms of criminality, the Pearsonian correlation was computed by the ANOVA program, and has been presented in the last column headed "r." This measures the amount of slope

Nondrug Crime -88-

### Nondrug Crime -89-

for the best fitting straight line that can be predicted by the Heroin User Typology for a given dependent variable; the r value documents whether the expected positive linear trend was present or not. An r of .10 or less denotes a very weak linear trend; an r of .25 or over generally suggested a strong association in this study.

In tables where the denominator was the number of days or number of offenses (Tables VI. 3B, 5, 6), the base Ns for percentages were based upon the standardized 100 person-days per subject, but which have been adjusted by the number of subjects. This standardization created a "synthetic" number of days or offenses wherein each respondent's mean per day was multiplied by 100 in both the numerator and denominator. This greatly inflated the number of cases (i.e., number of person-days or number of offenses) upon which a given percentage was based and greatly increased the probability that the "p of F" would be highly significant,. To prevent this, the sum of squares of the between group means has been adjusted by the ratio of the number of subjects to the number of active days or offenses. Thus, the "p of F" given has been adjusted by the number of subjects. The "r" value will not be affected and the actual figures (means) presented are an accurate reflection of the data and respondent activity.

### Brief Definitions of Dependent Variables

Tables VI. 1 - 7 have a standard format. The independent variable, the Heroin User Typology, appeared at at the top, along with the totals for all respondents; measures of association discussed above ("p of F" and "r") have been presented in the last two columns. The row variables contain the same list of specific offenses, with a total for "any nondrug crime" in the first row (shoplifting-own use was excluded for reasons given above). The major change from table to table was the unit of analysis (subjects, person-days, offenses, or dollars) in the numerator and denominator. The following briefly defines the relevant measures in each table.

	Table	Number	Description of
	VI.	1	Percentage of each offense t
	۷I.	2	Among persons Typology, the one or more of
	VI.	3A	(Upper half) T every responde standardizatio
	VI.	3B	(Lower half) <u>active day</u> (x were committed
	VI.	4	The mean <u>cash</u> nondrug crime; the denominato
	VI.	5	The mean <u>cash</u> shows much cas
Standard and Standard an	۷1.	6	The mean <u>cash</u> Days without c
	VI.	7A	(top third) <u>N</u> types of crime of person-days
	VI.	7B	(middle third) This annualize extremely clos measure of the (See the longe
	VI.	7C	(bottom third) daily cash inc types would t street time ( zero dollars).
5			
	BDJ:82	258A:074	2A;bj

### Nondrug Crime -90-

Measure

respondents who reported committing one or more of ype during their reporting period.

classified in each category of the Heroin User e percentage of <u>person-days</u> in which they committed ifenses of a given type.

The <u>number of nondrug offenses</u> per day (x 100) with ent contributing the same number of days (due to n).

The number of <u>nondrug</u> offenses per criminally 100) per subject. This showed how many offenses on the days when that crime occured.

n income per day per subject from each type of ; days without criminal income were included in or.

h returns per nondrug offense per subject. This income criminals make from the average crime.

returns per criminally active day per subject. rime were excluded from the denominator.

lumber of days per year per subject that various e were committed. This annualizes the percentage s active.

) Number of nondrug offenses per year per subject. es the number of offenses per day. This is also se to a perfect measure of lambda, a statistical e crime rate per year while not institutionalized. er discussion of lambda in Chapter VIII below).

) <u>Cash income per year</u> per subject annualized the come. This shows how much heroin users of various cypically earn in cash income during 365 days of with short term jail stays included as days with Nondrug Crime -91-

	Heroin	User Typol	ogy			
Type of Nondrug Crime (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Total (201)	p of F	r r
	Durin Respo	g Report ndents Com	ting Day mitting:	ys, Per	centag	e of
Any Nondrug Crime	87	94	97	93	.10	.15
Robbery	20	23	44	28	.005	.21
Burglary	31	42	56	43	.017	.20
Shoplifting (Resale)	51	62	68	60	.15	.14
Other Larcenies	36	53	60	50	.03	.18
Forgery	10	9	13	10	.74	.04
Con Games	26	27	34	29	.58	.07
Prostitution	20	19	13	17	.53	07
Pimping	10	16	6	7	.70	05
Other Illegal Acts	38	40	42	40	.89	.03
Shoplifting (Own Use)*	31	35	32	33	90	01

Table VI. 1 -- Percentage of <u>Respondents</u> Committing Nondrug Crimes by Heroin User Typology

\* Shoplifting for Own Use is not included in "Any Nondrug Crime" total.

Type of Nondrug Crime (Number of Subjects)

Any Nondrug Crime

Robbery

Burglary

Shoplifting (Resale)

Other Larcenies

Forgery

Con Games

Prostitution

Pimping

Other Illegal Acts

Shoplifting (Own Use)\*

Nondrug Crime -92-

Table VI. 2 -- Percentage of <u>Person-Days</u> on Which Nondrug Crimes Were Committed by Heroin User Typology

Heroin	User Typolo	ogy			
Irregular	Regular	Daily	Total (201)	p of	r
(61)	(78)	(02)	(201)	<u> </u>	
Percer	ntage of <u>Per</u>	son Days	Committing	:	
13.8	23.2	33.5	23.5	.000	.37
0.6	1.1	2.8	1.5	.002	.23
1.1	3.9	8.0	4.3	.000	.29
4.8	7.3	13.6	8.5	.003	.23
1.4	4.3	4.3	3.4	.012	.18
0.3	0.3	0.4	0.3	.77	.05
1.8	2.2	1.6	1.9	.83	01
3.5	3.7	3.9	3.7	.98	.01
0.4	0.5	0.4	0.4	.98	.00
1.3	2.0	1.5	1.7	.57	.02
1.4	1.3	1.2	1.3	.84	04

\*Shoplifting for Own Use is not included in "Any Nondrug Crime" total.

	Nondrug	Crime -93	-			
Table VI. 3 Nondr Crimi	ug Crime: nally Active	Offendi e Day by He	ng Rates eroin User	Per <u>Da</u> Typology	<u>iy</u> and	Per
	Heroin l	Jser Typol	ogy		_	
Type of Nondrug Crime (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Total (201)	p of F	٣
	A. Nun	nber of <u>No</u>	ndrug Offer	ises per	Day (x	100)
Any Nondrug Crime	31.8	44.5	57.2	44.5	.06	.17
Robbery	0.6	1.1	3.2	1.6	.002	.23
Burglary	1.2	4.1	9.3	4.8	.000	.29
Shoplifting (Resale)	9.5	12.7	19.7	13.9	.12	.14
Other Larcenies	1.6	5.9	5.3	4.4	.006	.17
Forgery	0.3	0.3	0.7	0.4	.43	.08
Con Games	9.8	9.3	8.0	9.1	.97	02
Prostitution	6.3	7.4	8.7	7.5	.89	.04
Pimping	0.5	0.5	0.4	0.5	.97	02
Other Illegal Acts	1.9	3.2	1.7	2.4	,26	01
Shoplifting (Own Use)*	1.8	1.9	1.5	1.7	.82	03
	B. Nur wi	nber of No th One or I	ndrug Offen Nore Offens	nses per Ses**: (	Active X 100)	Day
Any Nondrug Crime	230	192	171	189	.20	13
Robbery	109	108	113	111	.20	.12
Burglary	109	106	117	112	.20	.15
Shoplifting (Resale)	197	174	145	164	.10	18
Other Larcenies	109	137	124	128	.20	.02
Forgery	100	107	161	127	.20	.35
Con Games	562	429	498	485	.20	03
Prostitution	181	201	223	203	.20	.17
Pimping	120	103	100	107	.20	15
Other Illegal Acts	144	157	114	142	.20	08
Shoplifting (Own Use)*	127	142	127	133	.20	.00

\* - Shoplifting for Own Use is not included in "Any Nondrug Crime" total.
 \*\*- The Base N for active days (3B) can be computed for each corresponding cell by multiplying the number of subjects in heroin use group (Ns= 61,78,62) by the percent of days active (3A) in the corresponding cells.



# Type of Nondrug Crime (Number of Subjects) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping Other Illegal Acts

Shoplifting (Own Use)\*

### Nondrug Crime -94-

Table VI. 4 -- Cash Returns from Nondrug Crime: <u>Mean Cash Income Per</u> Day by Heroin User Typology

ŀ	leroin	User Typolog	JY				
Irre (6	egular 51)	Regular (78)	Daily (62)	Total (201)	p of F	r	
	Mean	Cash Income	per Day	Per Subjec	t from	1:	
	7.90	15.66	23.40	15.70	.000	.34	
	0.43	1.03	2.48	1.30	.004	.23	
	0.70	3.22	7.96	3.91	.000	.27	
	2.43	3.68	6.39	4.14	.034	.18	
	0.38	2.00	2.24	1.59	.004	.21	
	0.15	0,82	0.25	0.44	.55	.01	
	1.12	1.12	1.10	1.11	.99	00	
	2.02	1.98	2.12	2.03	.99	.00	
	0.11	0.31	0.27	0.24	.75	.04	
	0.54	1.51	0.58	0.93	.46	.00	
	0.27	0.22	0.16	0.21	.47	09	

\* Shoplifting for Own Use is not included in "Any Nondrug Crime" total.

Type of Nondrug Crime (Number of Subjects)	Heroin U Irregular (61)	Jser Typolo Regular (78)	ogy Daily (62)	Total (201)	p of F	r
	Mean (	Cash per No	ondrug Off	ense from:		
Any Nondrug Crime	24.88	35.21	40.93	35.24	.20	.14
Robbery	69.27	91.20	77.02	79.96	.20	01
Burglary	57.19	77.63	85.60	80.78	.20	.06
Shoplifting (Resale)	25.62	28.98	32.47	29.81	.20	.12
Other Larcenies	24.56	34.11	41.89	35,98	.20	.14
Forgery	57.93	256.06	37.62	107.41	.05	19
Con Games	11.43	11.98	13.65	12.2	.20	.05
Prostitution	32.06	26.84	24.10	27.38	.20	22
Pimping	22.44	53,13	66.43	46.76	.05	.63
Other Illegal Acts	27.83	47.06	31.78	38.84	.20	.03
Shoplifting (Own Use)*	12.95	10.29	10.69	11.24	.20	06

\* Shoplifting for Own Use is not included in "Any Nondrug Crime" total.



Type of Nondrug Crime (Number of Subjects)

Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping Other Illegal Acts Shoplifting (Own Use)\*

## Nondrug Crime -95-Table VI. 5 -- Cash Returns Per Nondrug Offense by Heroin Use Typology

MARKED BAR

Nondrug Crime -96-

Table VI. 6 -- Cash Returns Per <u>Criminally Active Day</u> by Heroin User Typology

Heroin Us	ser Typolo	gy			
Irregular	Regular	Daily	Total	p of	
(61)	(78)	(62)	(201)	F	r
Mean <u>Cash</u>	Income pe	er Crimina	lly Active	Day	from:
57.25	67.63	69.86	66.76	.20	.08
75.79	98.21	87.30	-88.98	.20	.01
62.24	82.24	100.10	90.82	.20	.09
50.49	50.30	47.15	48.78	.20	05
26.83	46.65	52.11	46.22	.20	.16
57.93	274.00	60.51	136.06	.05	07
64.19	51.72	68.12	59.63	.204	.04
58.00	54.06	53.78	55.10	.20	.08
26.93	65.44	66.43	54.58	.05	.50
39.99	73.95	38.19	55.63	.20	03
16.47	14.67	13.59	14.94	.20	.07

\* Shoplifting for Own Use is not included in "Any Nondrug Crime" total.

Nondrug Crime -97-

Table VI. 7 - Annualized Crime Days, Offense Rates, and Cash Income from Crime(s) by Heroin User Typology

	Heroin l	lser Typol	ogy			
Type of Nondrug Crime	Irregular	Regular	Daily	Total	p_of	
(Number of Subjects)	(61)	(78)	(62)	(201)	<u>۲</u>	<u>r</u>
A Number of D		Deve Veeve I	Jaw Cubia	. + Committi		
A. Number of P	erson-Days	Per fear	rer Subjec		000	37
Any wondrug Urnne	20.4	20	122.5	53.0	000	.07
Robbery	<b>2.</b>   <b>1</b>	14 2	20.0	15 7	000	.20
Shoplifying (Decelo)	4.1	14.5	29.0	21 0	.000	.23
Shopiitting (Resale)	1/.0	20.7	49.0	12 5	.003	.23
Uther Larcentes	5.I	15.0	15./	12.5	.012	.10
Forgery	1.0	1.1	1.5	1.2	.//	.05
Lon Games	6.4	7.9	5.9	0.8	.83	01
Prostitution	12.7	13.4	14.4	13.5	.98	.01
Pimping	1.5	1.7	1.5	1.6	.98	.00
Other Illegal Acts	4.9	7.5	5.5	6.1	.57	.02
Shoplifting (Own Use)*	5.2	4.9	4.2	4.8	.84	04
			D V	Dave Cubica		<u></u>
B. Number (Lamda)	of Nondrug	g Offenses	Per Year	Per Subject	t of:	17
B. Number (Lamda) Any Nondrug Crime	of <u>Nondrug</u> 116.0	g Offenses 162.3	Per Year 208.7	Per Subject 162.6	t of: .06	.17
B. Number (Lamda) Any Nondrug Crime Robbery	of <u>Nondrug</u> 116.0 2.3	g Offenses 152.3 4.1	Per Year 208.7 11.8	Per Subject 162.6 5.9	t of: .06 .002	.17 .23
B. Number (Lamda) Any Nondrug Crime Robbery Burglary	of <u>Nondrug</u> 11 <u>6.0</u> 2.3 4.5	g Offenses 162.3 4.1 15.1	Per Year 208.7 11.8 33.9	Per Subject 162.6 5.9 17.7	t of: 06 002 .000	.17 .23 .29
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale)	of <u>Nondrug</u> 11 <u>6.0</u> 2.3 4.5 34.6	g Offenses 152.3 4.1 15.1 46.3	Per Year 208.7 11.8 33.9 71.9	Per Subject 162.6 5.9 17.7 50.7	t of: .06 .002 .000 .12	.17 .23 .29 .14
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies	of <u>Nondrug</u> 11 <u>6.0</u> 2.3 4.5 34.6 5.8	g Offenses 162.3 4.1 15.1 46.3 21.4	Per Year 208.7 11.8 33.9 71.9 19.5	Per Subject 162.6 5.9 17.7 50.7 16.1	t of: .06 .002 .000 .12 .006	.17 .23 .29 .14 .17
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery	of <u>Nondrug</u> 116.0 2.3 4.5 34.6 5.8 .96	g Offenses 152.3 4.1 15.1 46.3 21.4 1.2	Per Year 208.7 11.8 33.9 71.9 19.5 2.5	Per Subject 162.6 5.9 17.7 50.7 16.1 1.5	t of: .06 .002 .000 .12 .006 .43	.17 .23 .29 .14 .17 .08
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games	of <u>Nondrug</u> 116.0 2.3 4.5 34.6 5.8 .96 35.9	g Offenses 162.3 4.1 15.1 46.3 21.4 1.2 33.8	Per Year 208.7 11.8 33.9 71.9 19.5 2.5 29.2	Per Subject 162.6 5.9 17.7 50.7 16.1 1.5 33.0	t of: .06 .002 .000 .12 .006 .43 .97	.17 .23 .29 .14 .17 .08 01
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution	of <u>Nondrug</u> 116.0 2.3 4.5 34.6 5.8 .96 35.9 23.0	g Offenses 162.3 4.1 15.1 46.3 21.4 1.2 33.8 26.9	Per Year 208.7 11.8 33.9 71.9 19.5 2.5 29.2 32.1	Per Subject 162.6 5.9 17.7 50.7 16.1 1.5 33.0 27.3	t of: .06 .002 .000 .12 .006 .43 .97 .89	.17 .23 .29 .14 .17 .08 01 .04
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping	of <u>Nondrug</u> 116.0 2.3 4.5 34.6 5.8 .96 35.9 23.0 1.8	g Offenses 152.3 4.1 15.1 46.3 21.4 1.2 33.8 26.9 1.8	Per Year 208.7 11.8 33.9 71.9 19.5 2.5 29.2 32.1 1.5	Per Subject 162.6 5.9 17.7 50.7 16.1 1.5 33.0 27.3 1.7	t of: .06 .002 .000 .12 .006 .43 .97 .89 .97	.17 .23 .29 .14 .17 .08 01 .04 01
B. Number (Lamda) Any Nondrug Crime Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping Other Illegal	of <u>Nondrug</u> 116.0 2.3 4.5 34.6 5.8 .96 35.9 23.0 1.8 7.0	g Offenses 162.3 4.1 15.1 46.3 21.4 1.2 33.8 26.9 1.8 11.7	Per Year 208.7 11.8 33.9 71.9 19.5 2.5 29.2 32.1 1.5 6.3	Per Subject 162.6 5.9 17.7 50.7 16.1 1.5 33.0 27.3 1.7 8.6	t of: .06 .002 .000 .12 .006 .43 .97 .89 .97 .26	.17 .23 .29 .14 .17 .08 01 .04 01 01

C. Cash Income Per Year Per Subject from:

Any	Nondrug Crime	2,885	5,719	8,540	5,729	.000	.34	Since A
-	Robbery	158	377	905	474	.004	.23	
	Burglary	256	1.174	2.906	1,429	.000	.27	هير.
	Shoplifting (Resale)	887	1,342	2,334	1,510	.034	.18	
	Other Larcenies	142	730	818	579	.003	.21	<b>1</b> (14)
	Forgery	56	300	93	162	.55	.01	
	Con Games	410	408	400	406	.99	.00	
	Prostitution	739	722	773	743	.99	.00	Maga arris
	Pimping	41	114	100	87	.75	.04	
	Other Illegal Acts	197	552	212	339	.46	.00	
								<b>1</b>
Sho	plifting (Own Use)*	97	81	57	79	.49	09	

\*Shoplifting for Own Use is not included in "Any Nondrug Crime" total. All figures given are the amounts per day multiplied by 365 days.

BDJ;8258A;0742A;bj

Drug selling, especially of heroin, opiates, and cocaine, is one of the most serious crimes in American society. During the early years of this research, under the Rockefeller Drug Law of 1973, imprisoned persons convicted of selling large amounts of these drugs were mandated to be sentenced to life in prison or lifetime supervision after several years in prison. Moreover, the public ranks the sale of narcotics as an offense twice as severe as robbery (Collins, et al., 1982b; Wolfgang and Figlio, 1982); it is widely considered one of society's most serious offenses, exceeded only homicide and aggravated assault with serious injury.

Despite such legal statutes and public opinion, however, drug selling appears to be one of the most frequently committed crimes among prisoners (Chaiken and Chaiken, 1982ab), heroin addicts on the streets (Inciardi, 1980, 1981; Ball, et al., 1981; Ball, 1982), drug abusers in treatment (Collins, et al., 1982abc; Sells, et al., 1976). The public stereotype of a drug seller, as someone selling drugs which he does not use or uses seldom, was rare among research subjects. Most research evidence showed that persons tend to sell drugs they used quite frequently, and such dealing earnings were generally used to obtain the drugs they consumed (Johnson, 1973; Clayton and Voss, 1981).

When researchers have inquired about drug selling activity among drug using populations, they generally obtained extensive information about drug selling because so many drug users do it and do so at high rates. Nevertheless, few researchers have studied the many different kinds of activities which might fall under the rubric of "drug selling." In the following discussion the terms "drug business" or "drug distribution" will be employed to include a variety of ways in which these street opiate users engaged in illegal drug transfers or helped to facilitate such transfers.

## Drug Business Crimes -98-CHAPTER VII DRUG BUSINESS CRIMES AMONG HEROIN USER TYPES

Drug Business Crimes -99-

### Drug Sellers/Dealers

Prior research has generally assumed that positive responses to questions about drug selling involves face-to-face transactions between a buyer (who obtains drugs) and a seller (who obtains cash income). In the pilot phases of the Economic Behavior Project, however, staff quickly discovered much greater complexity in the drug distribution system than reported in the existing research literature. Research subjects clearly and routinely differentiated between "drug dealing" and a variety of other distribution roles. In an earlier paper from this project, Goldstein (1981) described a variety of ways of "getting by" or "getting over" by which street opiate users obtained drugs with no or little cash expense. Many of these activities involved the respondent in various distribution activities involving illegal drugs.

Face-to-face transfers of drugs from a seller to a buyer was called "dealing" or "selling" on the street. Even among "dealers," our subjects differentiated "nouse connections" from "jugglers." (See vignette N).

Vignette N -- Dealers, Jugglers, and House Connections

[Gaston I. (black male, age 50) was daily heroin user who also committed several thefts. Drug sales were his major activity, but this involved low level "juggling" in which he rebagged a larger amount into smaller bags and resold them. He reported cash income from dealing on 76% of his person-days. In one week he indicated the following dealing activity:

- 9/18/81 bought one quarter for \$40, made it into 10 \$10 bags, sold 6, used 4.
- 9/19/81 did the same as yesterday but used 8 and sold 2.
- 9/21/81 got 20 Tuinals on consignment, sold for \$4 each, made \$80, gave person back \$40.
- 9/22/81 bought a quarter, made it into 10 bags, sold 7 and used 3.
- 9/23/81 Same as yesterday, made the quarter into 8 bags used 3 and sold 5.

China A. (Hispanic female, age 29) was a regular heroin user who functioned for a couple of weeks as a "house connection" in which she performed several roles. A nonusing dealer would come to her apartment with large amounts of heroin and cocaine. She would inject some of the drugs and tell the dealer how much it could be cut. She would then "cut" and "bag" it behind a locked and fortified door which contained a small peep hole. Five to six runners or "cop men" would locate customers and bring their money to the door. With special code knocks, they would indicate how many bags of cocaine or heroin were wanted.

They would put the money through the peep hole; China would count the money and give them the drugs through the peep hole. China would keep rough records of the amount of business each runner did, and placed the money in a box. Her boss would come later, count the money, and bring new supplies of drugs. He paid her in drugs or cash or both. For example, on 7/15/81, she reported 70 heroin transactions, and 200-300 cocaine transactions for which she received payments of \$50 cash. \$50 heroin, and \$30 cocaine. She also paid the "runners "



### Steering, Touting, and Copping (STC)

In addition, these street opiate users performed a variety of other activities which they did not consider as "drug selling or dealing." These standard drug distribution roles included: "steer," "tout," "cop," "hold," "test," "lend works," "pick-up," and "run a shooting gallery." Steering, touting, and copping (henceforth abbreviated in this report as STC) involved overlapping role relationships between actors in drug distribution events. "Steering" involved the subject directing a potential customer to a dealer who would make the sale. "Touting" involved a subject finding customers for a particular dealer. "Copping" (person was called a "cop man") involved the respondent as an intermediary who transported money and drugs between a dealer and buyer who never met. A given respondent may engage in one and usually more such activities during a given day. Any given transaction might involve some elements in which a respondent acts as a steerer, as a tout, and then as a cop man.

### Vignette 0 -- Steerers, Touts, and Cop Men

Ephraim	S (Hispanic male
wide var	iety of low-level
period h	e indicated the f
7/6/81	- copped about
	neighborhood p
	and \$40 worth
7/7/81	- helped a hous
	street and tal
	and cop from t
	and \$5 worth
	\$65 worth of h
7/8/81	- bought 3 half
	bags of heroin
7/9/81	- helped the de
	Rico for 3 we
	other for 4 ye

Blackie D (Hispanic male, age 35) was a daily heroin user who also committed some thefts. He worked mainly as a steerer and/or tout for one particular heroin dealer. He reported drug income obtained from S/T/C on 31% of his person days. For every 5 persons that he sends to the dealer, he gets a 1/2 quarter of heroin if they bought 1/2 quarters; he receives a full quarter (\$50 value) if 5 buyers bought quarters. He also cons people in his drug business activity. One day, two people from out of town came to New York to buy dope they gave Blackie D. 5200 to cop for them. He took the money, went 'over the roof, and did not return with drugs.

Drug Business Crimes -100-

age 30), was a daily heroin user and engaged in a distribution activities. For example, in a four day pllowing: 200 worth of coke and \$200 worth of heroin for 2

eople and 3 white guys from New Jersey, earned \$15 of heroin.

e connection. It was a slow day. I stayed in the ked to drug purchasers. They gave me money to go up he house dealer. Got \$2 in cash, \$5 worth of heroin of coke from purchasers. I also got \$80 in cash and eroin from the dealer.

quarters for \$20 - sold them for \$25. Bought 3 \$10 for \$7 each and sold them for \$10 each.

ller again - dealer's regular steerer is in Puerto ks; dealer called me to help him - we've known each irs.

### Drug Business Crimes -101-

From the point of view of a person who wished to buy the best quality drugs for the standard street retail price, it might be difficult to distinguish a steerer (believed to be more neutral) from a tout (usually employed by the dealer to locate customers). If a potiential buyer asked the subject to "cop" some drugs for him, a subject who was steering would Since these three forms of drug typically convert to a "cop man." distribution occur frequently, and were almost impossible to keep separate, they have been combined in the data presented below into a category of "steering, touting, copping."

### Other Drug Business

Likewise, subjects reported parallel drug distribution activities which will be referred to as "other drug business" (see Goldstein (1981) for longer discussion and examples) and included acts such as:

- a) "lend or rent works" to a friend, acquaintance, or paying customer who does not have a needle and/or syringe;
- b) "run a shooting gallery" in which others pay the respondent for a place (usually an apartment or abandoned building) to inject their drugs;
- c) "hit" another person by injecting drugs into veins which they cannot reach;

### Vignette P -- Shooting Galleries

Mack C. (Hispanic male, age 29) was a regular heroin user who ran a "shooting gallery" and "rented works" - for \$1 or \$1.50 -"it depends. And if a person wants me to hit him, I'll charge him a dollar. Some give me cash, some a taste.' Its according to what they get, if they get enough. If they are in a good mood they give me a taste, and if they see I'm sick they'll give me a taste. if I ask." He works in his cousin's shooting gallery. The following is one weeks activity:

- 7/7/81 rented works to 30-40 people, got \$30-\$40. Also copped for 4-5 people, about \$150 of heroin - got about \$75 worth, gave 1/2 to cousin. 7/9/81 - rented works, made \$35; copped for 4 people, got \$80 worth of heroin, gave 1/2 to cousin.
- 7/11/81- rented works, got \$35; also copped for 4 people, got \$90 of heroin, gave 1/2 to cousin.



- it can be cut:
- and the transaction is complete.]
- locale to another.
- theft, or assault.
- things, and do general tasks as requested.

Neville E. (black male, age 32) was a daily heroin user working for a dealer. On one day he "held" the dealer's heroin and when the dealer got the correct amount of money; Neville handed the drugs to the buyer. On other occasions, he made sure the money was "correct." For example, a buyer who wanted a quarter of heroin will give \$50 to a cop man, who in turn gave it to Neville, who counted the money to be sure it was correct. He gave the money to his dealer, who gave him the quarter. Neville gave the drugs to the cop man, who returned it to the buyer. He did 10-40 such transactions per day.

Neville was paid \$50/day by his dealer. But in addition, he stole some of the money he received for the dealer (\$20-50/day). In addition, he "tapped the bags" before returning it to the cop man and stole \$20-\$50 worth of drugs which he consumed. He also bought drugs with cash from this dealer, paying \$35 per quarter (or has "avoided expenditures" for heroin of \$15). If a buyer approached him directly, Neville would also serve as a "cop man." Neville reported cash income from these activities on 67% of his person-days.

Tom S. (Hispanic male, age 30) was a daily heroin user. He used his apartment as a shooting gallery and rented works for \$2 per person. He also copped drugs for people, getting \$2 per person, and he rented works to all these customers. Oftentimes, instead of getting money from the people he copped for, they gave him 'a taste' (shared their drugs) from what they bought in drugs. For example, on 11/27/81, all 10 people that he copped for 'took him to the cooker.' Tom S. also sold heroin and cocaine for a dealer. He got \$4 for every \$10 he sells, but 'I always get it fucked up because I'm shooting drugs all day long.' He said that this made the 'big dealer' angry with him because the money is "always messed up". Tom's friend told our interviewer that 'his memory was not very good and he does not speak very good English'.

Darwin B (black male, age 18) was a daily heroin user who committed almost all forms of crime. During a two week period, Darwin worked in his friend's 'candy' store - " he sells cigarettes, candy, marijuana, heroin, and coke. I'm sort of like a watch man. I make sure the money is correct and no trouble starts. For this I receive \$75 worth of heroin and \$25 worth of coke [a day]. I I sell some of the heroin. I spend maybe 12-16 hours a day there. I sleep upstairs, they have a little office I stay in."

Drug Business Crimes -102-

d) "test" drugs for dealers to determine the purity and recommend how many times

e) "hold" drugs for a dealer [usually the respondent physically possesses ("holds") the dealer's drugs. The dealer makes the sale, takes the money, and directs the customer to the holder. The dealer indicates via hand signals the amount of drugs to dispense. The holder transfers the drugs to the buyer

f) "run" drugs for a dealer, transporting relatively large quantities from one

g) "guard" or "watch the back" of a drug seller to protect him/her from robbery,

h) "lookout" for police, undercover agent, known enemy, etc.

i) "serve" a drug dealer by cleaning apartment, cooking, shopping, building

## Vignette Q -- Role Diversity in the Drug Business
## Drug Business Crimes -103-

## Drug Thefts

A relatively rare, but very rewarding way of obtaining large amounts of cash or drugs, was to rob, burglarize, or steal from a drug dealer. When a subject reported such an event, the offense and the cash income obtained was classified under the appropriate nondrug crime category (i.e., as a robbery, burglary, or theft). When the respondent obtained drugs during such an event, the standard retail value of the (stolen) drugs was coded as a "drug theft;" such drug thefts constituted another way of obtaining drugs without cash payments.

Vignette R -- The Big 'Score' -- Robberies and Burglaries of Dealers

Ben Q (Hispanic male, 41 years old) was a robber-dealer and a daily heroin user. He committed the most lucrative crime reported during the study. With a partner, Ben burglarized a drug dealers apartment and got over \$16,000 cash; his share was \$8,235. He stated that he "wasn't interested in drugs - we had hit him for drugs before!" [see below].

When our interviewer commented that burglarizing a dealer's apartment was very risky, Ben replied, "Everything we do is risky". Ben gave \$4,000 to his wife. She was angry that he committed this crime, and stated that he better straighten out or she would leave him, however, "she took the money anyway." Ben planned to spend some of the money on a computer course, would like to send his wife to Puerto Rico, and maybe put a down payment on a house or a business. (He never did so).

Our interviewer asked Ben why, after this big score, he was out on the streets the next day selling loose joints -- his usual activity. Ben replied, "What am I going to do? Stay in my apartment all day? I like to hang out, and when I hang out I sell loose joints. I don't like to be bored."

Shortly after this crime, an individual from the neighborhood gave the dealer Ben's description; Ben then decided to enroll in an alcohol detox program at a hospital. The dealer was subsequently arrested for selling heroin to minors.

Four days before this event, he burglarized the same dealer who was also a fence. He and a partner stole jewelry and 20 quarters of heroin. They split the heroin, and sold the jewelry to another fence for \$1,000. Ben Q. made \$500 and 10 quarters of heroin (a \$500 value).

Kyle T. (black male, age 31) was an irregular heroin user who committed robberies and sold drugs. On 4/13/81 he and a partner stuck up 3 drug dealers for \$500 in cash and \$630 worth of heroin. They split the money and drugs in half. Kyle used, sold, and gave away his share of the drugs.



# "Drug Income" and "Payments in Drugs"

Persons involved in illicit drug distribution were typically offered payments in drugs or cash or some combination of both. In the data presented at the end of this chapter, income from drug distribution activities have been separated into "cash income" and "drug income." "Cash income" referred to the number of dollars (in direct cash payments) obtained; this included cash earnings from "dealing" (direct sales), "STC" (steering, touting, and copping), and "other drug business" as defined above.

"Drug income" referred to the standard retail street value (dollar equivalent) of drugs that have been obtained without the cash purchase of such drugs. This included drug thefts, "payments in drugs," and avoided expenditures for drugs.

- defined above).
- when he had drugs but the current giver did not.

Such exchanges involved the illegal transfer of drug(s) (i.e. a crime event) and an economic value (the approximate dollar value of the drugs) obtained without cost.

Drug Business Crimes -104-

1. "Drug Thefts" involved the dollar value of drugs obtained by robbing, burglarizing, or stealing drugs from another drug dealer or user (as

2. "Payments in drugs" were usually direct reimbursements for some labor intensive activity by the respondent that was directly analogous to money paid to salesmen in the regular job market. Unlike regular employment, however, the employer and employee relationship was very ambiguous, frequently temporary, and changed rapidly in the drug business. Some common examples have been given in vignettes N - S above.

3. "Avoided expenditures-drugs" involved respondents obtaining some of their drugs for "free," in that they did not pay cash, nor perform work, nor steal the drug(s). The main "cost" to the respondent was a powerful, but informal, expectation that he would reciprocate on a future occasion Drug Business Crimes -105-

## Vignette S -- Getting Drugs for "Nothing"

- A respondent obtained two "dime" bags (value of \$20) of marijuana as a birthday present.
- A subject "got down" (injected with) a friend and consumed half of his friend's bag of heroin (value of \$5).
- A respondent shared a joint with a friend who did not expect immediate repayment (although an obligation to reciprocate at a later date was implicit).
- A respondent "copped short." That is, he consumed five bags of heroin which he purchased for \$45. He obtained a \$50 value, while the \$5 difference was considered as drug income.
- While staying with a girl friend, a respondent drank a six-pack of beer in her refrigerator and smoked a pack of her cigarettes.
- A subject shared a pint of wine which he did not purchased with two bottle gang members. Another subject "borrowed" cigarettes from someone else when he needed it. The value of such alcohol or cigarettes was recorded as drug income -- avoided expenditures.
- Another subject obtained gifts of money for purchasing drugs. "I ran into an old girl friend who just got out of the army. She lives over on the West Side. I stayed with her all week. She gave me about \$80 a day to cop my drugs - she has a whole lot of money, I don't know how much but alot."



# Was drug business activity related to the frequency of heroin use?

Some aspects of the drug business were related to heroin use. While involvement in steering, touting, and copping to obtain drug income was generally related to the frequency of heroin use, drug selling was not. Moreover, daily heroin users were significantly more likely than less regular heroin users to obtain avoided expenditures-drugs (but not alcohol or cigarettes).

1. Daily heroin users and less regular heroin users were about equally likely to obtain cash income and drug income from drug business activity. All subjects engaged in some forms of drug distribution. Table VII. 1 shows that about three-quarters of all respondents obtained cash income from the drug business. Less than half obtained cash income from direct drug sales with the irregular heroin users slightly more involved in sales than the daily and regular heroin users. Two-thirds of the subjects gained cash from steering, touting, and copping (STC) with the daily heroin users somewhat more involved in these activities.

All respondents obtained some drug income (i.e., drugs without cash purchases). About a fifth of the subjects received drug income from drug sales, with no differences according to the Heroin User Typology. Irregular heroin users were less apt (62%) to obtain payments in drugs from STC than the regular and daily heroin users (over 80%). Daily heroin users were more apt to have some involvement in drug thefts. Since virtually all subjects did so. no variation by the Heroin User Typology was evident in the proportion of subjects receiving drugs, alcohol, or cigarettes for "free" (avoided expenditures).

## Drug Business Crimes -106-FINDINGS

## How does the drug business activity of daily heroin users differ from less regular users?

## Drug Business Crimes -107-

2. Daily heroin users were active on more person-days in gaining cash income and drug income from STC than the less regular heroin users. The percentage of days with drug sales for cash or drug income was not related to the frequency of heroin use.

Table VII. 2 shows that respondents engaged in drug sales on only about 9% (7.3% + 1.2%) of their person-days, most of these days with drug deals resulted in cash income (7.3% of person-days) rather than drug income; however, this did not vary by the Heroin User Typology. Steering, touting, and copping occurred more frequently (both with cash income and drug income); daily heroin users were active on about twice as many person days as regular heroin users, and were about three times more active than irregular heroin users, especially for STC with drug income.

Likewise, daily heroin users gained drugs through avoided expenditures on almost three times as many days as the irregular heroin users. But no variation by the frequency of heroin use existed in the percentage of person-days with avoided expenditures for cigarettes and alcohol, nor for drug thefts.

Nevertheless, almost half of the person-days of the irregular heroin users, almost two-thirds of the person-days of regular heroin users, and 80% of the person-days of daily heroin users involved some kind of drug business activity. Thus, on well over half of the days, respondents had some drug business activity.

3. Respondents obtained more drug income than cash income from their drug business activity.

"Drug business income" refers to the sum of cash income from drug transactions plus the sum of drug income from payments in drugs, drug thefts, and avoided expenditures for drugs, alcohol, and cigarettes.

Table VII. 3 shows that the annual drug business income (\$6,140) came almost equally in three major forms: as cash income (\$1,868), as avoided expenditures (\$2,083), and as payments in drugs and from drug thefts (\$2,189).



4. Daily heroin users obtained about twice as much cash income and drug income from drug business activity as the less regular heroin users. Daily heroin users obtained more dollars from almost all types of drug business activity than less regular heroin users. They obtained about twice (\$2,641) as much cash income from drug sales and STC than the less regular heroin users (under \$1,500).

The daily heroin users also obtained \$2,949 as payments in drugs for STC which was about twice as much as was received by the regular heroin users and over five times as much as was obtained by the irregular heroin users. Daily heroin users also obtained \$3,403 worth of avoided expenditures for drugs, alcohol, and cigarettes, or almost twice that received by regular heroin users and three times that received by irregular heroin users. Daily heroin users also obtained slightly more drug income from drug sales and thefts of drugs than their counterparts although the dollar amounts averaged under \$600 per year.

Overall, daily heroin users obtained about twice as much total drug business income (\$10,170) as the regular (\$5,284) and three times as much as the irregular heroin users (\$3,138).

Table VII. 4A shows that daily heroin users each have 880 drug transactions per year, only a third of which involved drug sales. Irregular heroin users committed 245 transactions per year while the regular heroin users have 823 transactions per year. The daily heroin users had more STC transactions than their less involved counterparts. These estimated numbers of transactions do not include days with avoided expenditures-drugs (in which illegal drug transactions occurred as respondents shared drugs with someone else, obtained them as gifts, found drugs, stole them, etc.).

Drug Business Crimes -108-

5. Respondents committed an average of 665 drug transactions per year, of which somewhat over half involved direct drug sales; the remainder involved steering, touting, and copping.

## Drug Business Crimes -109-

6. These respondents distributed, via drug sales or STC, drugs having an approximate value of almost \$15,000 per year; the daily heroin users distributed the largest dollar amount of drugs.

Table VII. 4B shows that the average daily heroin user distributed drugs having a value of almost \$20,000 per year via STC and another \$6,000 in direct sales. The regular heroin users distributed less than half this dollar volume (under \$9,000) via STC, but about the same amount (\$4,626) in direct sales. The irregular heroin users distributed less than half as much as the regular heroin users.

 About a third or more of the value of all drugs distributed via sales and STC involved payments for the sales work involved.

Respondents obtained cash or drug payments equivalent to 33% [(\$1,112 + \$269)/\$4,185] of the value of all drugs distributed via direct sales. Likewise, 22% [(\$671 + \$1,589)/\$10,411] of the value of all drugs distributed via STC resulted in cash or drug income for the respondent. In short, the illicit distribution of drugs involves relatively high costs for sales work, although this was not always received in cash.

# B. HEROIN USER INVOLVEMENT IN SPECIFIC FORMS OF DRUG BUSINESS ACTIVITY. How many subjects became involved, with what frequency, and about how much income do they gain from various drug business activities?

## DIRECT DRUG SALES FOR CASH INCOME

Although the public steroetype of a drug dealer was someone selling drugs for cash, among these street opiate users, these data showed that cash payments for drug distribution activity was relatively less common than payments in drugs. Less than half the respondents sold drugs for cash during their reporting periods. About 8% of the person-days involved cash sales of drugs. Approximately 364 such transactions per year per subject occurred. Such sales netted only \$1,112 per subject in annual cash income per year.



ſ

Moreover, none of these measures was significantly associated with the frequency of heroin use. In short, drug sales resulting in a cash income was one of the less common forms of drug distribution and it did not result in a particularly impressive amounts of cash income. CASH INCOME FROM STEERING, TOUTING, COPPING

More subjects engage in STC for cash income than direct sales, but they were active in both activities on the same proportion (8%) of person-days. The cash income from STC were about half (5670/year) as large as from direct sales. Daily heroin users were active on about twice as many days, and had twice as much cash income from STC than the regular heroin users. CASH INCOME FROM OTHER DRUG BUSINESS

Less than 10% of subjects engaged in other drug business. Regular and daily heroin users had a somewhat larger proportions involved and were active on more person-days than the irregular heroin users, but the dollar values from this source for all three groups were small (\$110 or less per year). DRUG PAYMENTS (INCOME) FROM DRUG SALES

Only about a fifth of the respondents obtained drugs as payments for their drug sales efforts and this occurred on 1% of the person-days. The value of the drug income was modest (an average of less than less than \$300/year). It was slightly higher among the daily heroin users. DRUG INCOME FROM STEERING, TOUTING, COPPING

Steering, touting, and copping drugs emerged as a major type of drug distribution among these street opiate users. Three-quarters of the subjects obtained payments in drugs for STC and did so on 12% of the person-days. Daily heroin users were active on twice as many days as the regular heroin users, and on four times as many days as the irregular heroin users. The daily heroin users obtained almost \$3,000 in drug income from STC, which was twice as much as the regular and six times as much as the irregular heroin users.

Drug Business Crimes -110-

## Drug Business Crimes -111-DRUG INCOME FROM DRUG THEFTS

Almost a fifth of these respondents reported one or more thefts (including robberies and burglaries) of drugs. Such events were rare, however, and occurred on less than 1% of the person-days and raised an average of \$331/year. Daily heroin users were slightly more involved than the less regular heroin users.

## OBTAINING DRUGS VIA AVOIDED EXPENDITURES

These street opiate users proved remarkably adept at obtaining drugs without cash purchase and without working in the drug distribution system (i.e., selling or STC). That is, they frequently found a friend who shared drug(s) at no direct cost to the respondent, or received drugs as a gift, or otherwise obtained drugs. Virtually all respondents received such "free" drugs and did so on two-fifths of their person-days. The value of such drugs (\$1,851) was equivalent to or exceeded that of drug income from STC, or from drug sales. Daily heroin users received twice as much in dollar value of "free drugs" than regular and four times as much as irregular heroin users.

## OBTAINING ALCOHOL AND CIGARETTES VIA AVOIDED EXPENDITURES

These respondents also obtained alcohol and cigarettes for "free." Three-quarters did so, on 12% or more of the person-days. The dollar value of such contributions, however, was modest; less than \$250/year in alcohol or cigarettes was obtained by respondents. Such "free" alcohol and cigarettes did not vary by the frequency of heroin use.

## SUMMARY

Among these street opiate users, drug distribution activity was more apt to involve STC with payments in drugs, and free gifts of drugs ("avoided expenditures") than drug sales with cash income. Almost half of the value of all drugs distributed went as drug or cash income to respondents for their labor in sales aspects of the drug distribution system.

Having now described in detail both nondrug crime and drug business crime, we can now directly address the questions: how many crimes do street a users commit? and how much income do the obtain?



Drug Business Crimes -112-Table VII. 1 -- Percent Distributi

## Type of Activity (Number of Subjects)

Total with Cash Income Any Drug Sale Any Steer, Tout, Cop\* Other Drug Business\*

Total with Drug Income (no cash)

Any Drug Sale

Any Steer, Tout, Cop\*

Any Drug Thefts\*

Avoided Expenditures Total

Drugs

Al cohol

Cigarettes

Total Proportion of Rs with Drug Business Income (Cash + Drug)

\*See text for definition of these terms.

<u>م</u>	9 04317	leas of filles					
0 i o	f <u>Re</u> n Acti	<u>spondents</u> vities by He	Reporting eroin User	Involveme Typology	ent	in	Drug
Ir	Heroi regula	n User Typo r Regular	logy Daily	Total	p of		
	(61)	(78)	(62)	(201)	F	r ie of	7
	Res	pondents Rep	porting:	1043, 1010	cneag		
		A. Any Cash	Earnings	from:			
	80.3	70.5	77.4	75.6	.38	03	
	55.7	43,6	43.5	47.3	.29	10	
	57.4	66.7	71.0	65.2	.27	.11	
	4.9	10.3	12.9	9.5	.31	.11	
	Β.	Any Drug In	come Obtai	ned From:			
	100.0	100.0	100.0	100.0			
	16.4	23.1	21.0	20.4	.62	.04	
	62.3	82.1	80.6	75.6	.014	.17	
	11.5	19.2	27.4	19.4	.08	.16	
:	100.0	100.0	100.0	100.0			
	98.4	100.0	100.0	99.5	.32	.09	
	80.3	78.2	79.0	79.1	.95	01	
	70.5	79.5	72.6	74.6	.44	.02	
	100.0	100.0	100.0	100.0			_

· \_ ·

Distributio	n Activit	les by Her	oin User	Typology		-
Type of Activity (Number of Subjects)	Heroin Us Irregular (61) Percen	ser Typolo Regular (78) Itage of <u>P</u>	gy Daily (62) erson Day	Total (201) <u>/s</u> Reportin	p of F	<u> </u>
	Α.	Cash Inco	me from:			
Total Cash Income	14.3	13.4	20.8	16.0	.04	.14
Any Drug Sale	9.1	5.7	7.6	7.3	.29	05
Any Steer, Tout, Cop*	5.5	7.4	12.3	8.3	.011	.20
Other Drug Business*	0.4	1.1	1.8	1.1	.27	.11
	B. Va	lue of Dru	g Income	Obtained f	'rom:	
Total with Drug Income (no cash)	42.9	61.0	79.7	61.2	.000	.60
Any Drug Sale	0.4	1.1	2.0	1.2	.11	.15
Any Steer, Tout, Cop*	5.2	11.6	20.0	12.2	.000	.37
Any Drug Thefts*	0.3	0.8	1.4	0.9	.16	.14
Avoided Expenditures Total:	39.9	54.7	68.8	54.5	.000	.47
Drugs	23.4	39.4	61.4	41.3	.000	.59
Al cohel	15.6	19.3	16.4	17.3	.48	.02
Cigarettes	11.9	15.0	9.5	12.3	.15	06
TOTAL DAYS WITH DRUG BUSINESS INCOME (Cash + Drug)	48.9	63.7	82.1	64.9	.000	.58

Drug Business Crimes -113-

Table VII. 2 -- Percent of Person Days Reporting Involvement in Drug

\*See text for definition of these terms.



Drug Business Crimes -114-

Table VII. 3 Annua from	alized Do Drug Dist	llar Amour ribution A	nt of <u>Cash</u> ctivities by	Income and Heroin Use	Drug r Typol	Income ogy
Type of Activity (Number of Subjects)	He Irreg (61	roin User Jular Reg ) (7	Typology ular Daily 8) (62)	7 Total (201)	p of F	r
	Α.	Dollar Am	ount of <u>Cast</u>	Income Per	Year f	rom:
Total Cash	1,566	1,402	2,752	1,868 \	.03	.15
Any Drug Sale	1,200	744	1,488	1,112	.22	.05
Any Steer, Tout, Cop*	ુ • 278	594	1,153	671	.014	.20
Other Drug Business*	· · 88	64	110	86	.78	.02
	B	. Dollar Va	lue of <u>Drug</u>	Income per	Year fr	om:
Total Drug Income (no cash)	1,572	3,881	7,418	4,272	.000	.50
Any Drug Sale	6 : 59	213	548	269 <sup>\</sup>	.15	.14
Any Steer, Tout, Cop*	476	1,378	2,949	1,589 t	.000	.30
Any Drug Thefts*	70	385	519	331	.17	.13
Avoided Expenditures Total:	968	1,906	3,403	2,083	.000	.33
Drugs	-752	1,663	3,168	1,851	.000	.34
Al cohol	149	161	185	165	.66	.06
Cigarettes	NG 67	82	50	67	.13	07
TOTAL DRUG BUSINESS INCOME (Cash + Drug)	3,138	5,284	10,170	6,140	.000	.41

Table VII. 3 A	nnualized Dol rom Drug Distr	lar Amount ibution Acti	of Cash I ivities by	<u>ncome</u> and Heroin User	Drug Typol	Income ogy
ype of Activity (Number of Subject	Her Irregu s) (61)	oin User Ty lar Regula (78)	pology ar Daily (62)	Tota] (201)	p of F	r
	Α.	Dollar Amoun	nt of <u>Cash</u>	Income Per	Year f	rom:
otal Cash	1,566	1,402	2,752	1,868 🗸	.03	.15
Any Drug Sale	· · 1,200	744	1,488	1,112	.22	.05
Any Steer, Tout, C	op* 🤄 🖕 278	594	1,153	671	.014	.20
Other Drug Busines	s* 🔧 🗾 88	64	110	86	.78	.02
	В.	Dollar Valu	e of <u>Drug I</u>	ncome per )	lear fr	om:
otal Drug Income (no cash)	1,572	3,881	7,418	4,272	.000	.50
Any Drug Sale	6 : 59	213	548	269 <sup>\</sup>	.15	.14
Any Steer, Tout, C	op* - 476	1,378	2,949	1,589 ;	.000	.30
Any Drug Thefts*	70	385	519	331	.17	.13
Avoided Expenditur Total:	es 968	1,906	3,403	2,083	.000	.33
Drugs	-752	1,663	3,168	1,851	.000	.34
Alcohol	149	161	185	165	.66	.06
Cigarettes	5 67	82	50	67	.13	07
OTAL DRUG BUSINESS INCOME (Cash + Dru	ig) 3,138	5,284	10,170	6,140	.000	.41

\*See text for definition of these terms.

Program Correct S Es Star 

Table VII. 4 Annuali Drugs D	zed <u>Numi</u> istribut	ber of Drug ed by Heroin	Transact User Typ	tions and ology	Dollar	Value
Type of Activity (Number of Subjects)	Her Irregu (61)	oin User Typo lar Regular (78)	Daily (62)	Total (201)	p of F	<u>r</u>
	Α.	Number of Dr	ug Trans	<u>actions</u> per	Year fr	om:
Total Drug Transactions	245	823	880	665	.15	.12
Drug Sale	123	591	316	364	.35	.04
Steer, Tout, Cop	122	233	564	301	.00	.26
	B. Dr	Dollar Valu ug via:	e of <u>Drug</u>	g(s) Distri	<u>buted</u> fo	r Any
Total Value of Drugs Distributed:	4,925	13,260	25,790	14,596	.000	.28
Drug Sales	1,561	4,626	6,212	4,185	.28	.11
Steer, Tout, Cop*	3,364	8,635	19,578	10,411	.00	.28

Drug Business Crimes -115-

\*See text for definition of these terms.

of

Street opiate users committed a large number of crimes of all types. The current chapter will provide detailed estimates of the number of crimes and the criminal income received by our subjects as well as compare them with subjects in other studies to place their criminal activity in a national context.

## Criminal Offense Rate

Criminologists have devoted considerable effort to describing and analyzing a measure of individual criminal offending (Greenberg, 1975; Blumstein and Cohen, 1973; Blumstein, et al., 1978; Moore, et al., 1981). This measure is called lambda (  $\lambda$  ) and is the rate of offending per unit of time (usually a year) within a specified offending population. This measure is computed by dividing the number of offenses by the amount of time during which the offender is "active" or "free" (or on the "street") to commit these offenses. Criminologists must resort to a variety of statistical manipulations to compute both the numerator and denominator of lambda using arrest and criminal justice data (Blumstein, et al, 1978; Moore, et al., 1981) due to the unavailability of good techniques for measuring individual criminal behavior per unit of time. These problems arise regardless of whether arrest data, conviction data, self report criminality, or other kinds of data are employed. Lambda is usually described as a the mean\* "number of offenses per year of street time for the entire sample or subgroups within it."

\*The use of the mean, however, is problematic because all research has documented that criminal activity is badly skewed. For example, Chaiken and Chaiken, (1982b,46) show that a substantial percentage (typically a majority) of the prison and jail inmate population in three states did not commit a given crime (i.e. burglary), several committed the crime only 1-10 times per year, but the 10% most active (90th percentile) committed burglary over 232 times per year. Thus, standard measures of central tendency may be misleading since the mode will almost always be zero or near zero, the median will be low (due to large proportion of noninvolved, while the mean will be somewhat larger (due to the contributions of the most seriously involved). (Continued next page.)

BDJ;8259A;0743A;bj

# Total Crimes and Criminal Income -116-CHAPTER VIII HOW MUCH CRIME AND CRIMINAL INCOME AMONG STREET OPIATE USERS?

Total Crimes and Criminal Income -117-Previous Studies of Offense Rates

A few previous research studies provided data about lambda (even though it was not actually calculated nor specifically called such) which will be compared with data from the street opiate users here. This brief review included only studies based upon self-reported criminal involvement by high risk samples (prisoners, street drug users, drug treatment populations) which may be similar to the subjects in this study. Offense rates based upon arrest or conviction data were not included because data about the arrests of our street opiate users were not available.

Peterson, et al. (1980, 28) studied 624 California prisoners and estimated the average number of crimes per year of street time among offenders who commit that crime; their data are summarized in Table VIII. 1.

Inciardi (1980; 1981) asked 166 youths who were active in the street drug scene to report their use of drugs and criminal involvement. The 70 heroin users had long histories of substance abuse, while the 96 nonheroin drug abusers had shorter histories. Almost all subjects reported criminal involvement in the preceding year. Moreover, the heroin users had an average of 352 offenses -- almost one crime per day per person. The nonheroin drug users were not a great deal less active -- with 312 crimes/year. Less than 2% of the crimes were violent (robbery or assault), while drug sales were 30-40% of all total offenses. Annual rates of offending for specific crimes were computed from Inciardi's data and given in Table VIII. 1.

Nevertheless, the mean more accurately reflected the average number of offenses than the median, and could be easily computed given the number of offenses per year and the number of subjects. Moreover, in the current study, a careful examination of frequency distribution of respondent offense rates showed the largest proportion clustered at zero offenses, and at one offense during the reporting period for any given type of crime. While the distributions were skewed, for the more active subjects, the distribution was relatively flat, with the most criminally active subject somewhat, but not many times, more involved in a given crime than the next most active subject.



In a large sample of clients admitted to a variety of treatment programs in five cities during 1979, Collins, et al. (1982b) provided data about the year before drug treatment intake, and presents the mean number of offenses per year (corrected for time at risk -- excluding jail time).

The recent study of career criminals by Chaiken and Chaiken (1982ab), provides carefully computed annual rates of criminal offending among prison and jail inmates in three states, California, Michigan, and Texas. These data show high rates of criminal offending, especially among prisoners in California. Moreover, respondents who reported heroin use, especially self-reported addiction to heroin, were among the highest rate offenders (see Table VIII. 1).

The McGlothlin, et al. (1977) study of the California Civil Addict Program did not present annualized offense rates during the period, but did present important data about the criminal income of daily heroin users (see section below on criminal income).

The research of John Ball, et al. (1979, 1981; Ball, 1981) and Nurco, et al., (1981abc) presented data about criminal behavior of 243 male heroin abusers in Baltimore, but did not show the annual rates of offending for specific crimes. They do, however, provide information about the total number of "crime-days" per year at risk. (see below). Most of these offenses involved thefts or drug selling.

Please describe the data used to support the statement that 100 untreated heroin addicts could commit between 50,000 to 100,000 crimes annually. Please indicate whether addicts involved had a history of criminal activity prior to heroin use ... "

(1981), provided the following response.

The statement that 100 untreated heroin addicts could commit between 50,000 to 100,000 crimes annually was based on the partially NIDA-funded contemporary studies of John Ball and his colleagues who extensively researched the behavioral outcomes of a cohort of 243 addicts from Baltimore for the past 11 years. Data from another NIDA-funded researcher, James Inciardi (1979), support the Ball findings.

Total Crimes and Criminal Income -118-

In 1981, Congressman Henry Waxman wrote to NIDA and asked:

The Director of the National Institute on Drug Abuse, William Pollin

## Total Crimes and Criminal Income -119-

Dr. Ball documented that drug abusers, while actively addicted to opiates, participated in 248 crime days per year; whereas, when not addicted. the subjects were involved in crime only 40.8 days per year. The level of crime necessary to maintain an active opiate habit was estimated by Masschusetts treatment personnel to be 2.5 crimes per day.

I asked my staff to verify this estimate with Dr. Ball....The statement in my presentation was based upon the following equation.

100 untreated active addicts x 248 crime days per year x 2.5 crimes per day = 62,000 crimes per year.

...Ball concludes: "One of the major findings of this study was that heroin addicts commit a staggering amount of crime and that this continues fairly much on a daily; basis for years and decades."

In a major literature review regarding the deterrence and incapacative

effects of imprisonment, Cohen (1978, 228) concluded:

A recurrent theme in this review has been the inadequacy of current estimates of individual crime rates.... Such estimates of (lambda) will require better data on criminal careers than are presently available... self reports by acknowledged criminals are probably the best source.

At the beginning of the Economic Behavior project, there was a clear recognition of the need to develop improved measures of the number of criminal offenses during a limited time period among subjects at high risk of such behavior.

## Measuring Criminal Offense Rates (Lambda):

This project carefully developed a methodology wherein respondents reported their criminal acts on a day-by-day basis. On every day that they reported one or more illegal acts, the interviewer asked them how many such events occurred during the specific calender day (as well as the dollar returns or drug income on that day from such crimes). As indicated in earlier chapters (II and III, also see Appendix A and B), the 201 subjects provided such information about their criminal and drug using behavior for at least 33 days, and an average of 57 person-days.

Thus, the data from this study provided an almost perfect measure of lambda (  $\lambda$  ) or annual offense rate per year of street time. Lambda was computed for specific offenses and related to the Heroin User Typology.



Specifically, for each respondent, the sum of number of crimes reported during all days of reporting was divided by the number of days about which he was interviewed, generating a mean number of offenses per day. This figure was then multiplied by 365 days to standardize and annualize the number of offenses.\* This calculation was performed for each category of crime. Such rates have been presented in chapters VI and VII.

The mean number of offenses per year for individual subjects were then combined to provide mean offenses for all subjects and for each heroin user type. These means included respondents with zero crimes/year, as well as those with high offense rates for a specific crime. The distribution of offense rates for each offense type were carefully examined. Extremely high values for offenders were carefully checked to assure that no apparent errors in respondent reports or data handling accounted for these high rates. Thus, the number of offenses per year of street time reported below are an accurate reflection of the volume of crime by these respondents and the heroin user subgroups.

Table VIII. 1 showed an extremely high volume of criminal activity by these street heroin users -- generally higher than the studies cited earlier -- although not for every specific crime category. These street opiate users were considerably more criminal than respondents in studies by Peterson, et al. (1980), Collins, et al. (1980), and Inciardi (1980). As discussed in Chapter VI, our subjects exhibited similar offenses rates to those in the Chaiken and Chaiken study.

\*In one minor way, these rates were not a pure lambda because a few days are included during which respondents were arrested and in jail for 1 - 6 days, and then released during the interview cycle. Thus, the estimates of lambda reported below are somewhat conservative since zero crimes were added into the numerator but the days in jail were included in the denominator. Removing days of jail time for the numerator and denominator would not increase the offense rates given in Table X1. 1 by more than 1-5 percent, depending upon the offense type.

Total Crimes and Criminal Income -120-

## FINDINGS ABOUT OFFENSE RATES

Total Crimes and Criminal Income -121-

Table VIII. 1 -- Annualized Offense Rates (Lambda) for Robbery, Burglary, Thefts, and Drug Sales Among Studies of Criminal Offending.

Author(s)/Year Description	Annu Robbery	alized Offen Burglary	se Rate (Lam Theft(s)	bda) for: Drug Sales
Peterson, et al. (1980) Calif. Prisoners as: Street Offenders Incoming Prisoners	2 4	7 15	na na	136 155
<u>Inciardi, (1980)</u> a Miami Heroin Users Miami Nonheroin Drug Users	4 5	8 8	b57 b63	137 114
Collins, et al. (1982) 1979 TOPS cohorte Corrected for Time at Ris	k 1	2	11	21
<u>Chaiken and Chaiken (1982b)</u> Prison and jail inmates: Median <sup>C</sup> 90th Percentile <sup>d</sup>	5 87	5 232	8 485	100 3251
Inmates who reported Heroin use, but not addicted all months in: California Texasi Michiganf	13 4 9	31 29 35	25 42 24	114 94 115
Heroin addiction, all months in: California Texas <sup>f</sup> Michigan <sup>f</sup>	34 5 17	68 35 26	66 108 50	158 150 124
This Report <sup>g</sup> New York street opiate users All subjects:	6	18	67	364
by Heroin User Typology: Irregular Regular Daily	2 4 12	5 15 34	40 68 91	123 591 316

a - Computed from Tables 6.2-3 by dividing the number of offenses during 12 months by the number of subjects.

b - The sum of shoplifting + theft from vehicle + pickpocketing + other theft.

- c Mean offenses (lambda) are not presented. Rather the median (50% above and 50% below) is given (Table 2.16, p. 48) .
- d Rates by the 10% of respondents who are most active in the given offense. e - Treatment Outcome Prospective Study (TOPS) of a 1979 cohort of persons
- admitted to treatment programs in 5 cities.
- f Data for California come from Chaiken and Chaiken (1982b, 161). Michigan and Texas data come from Chaiken (1983). All offense rates have been truncated so that no individual contributes more than 365 offenses per year to a given type of crime.
- g Source: Table VIII.2 (below). Theft is the sum of "shoplifting (resale)" plus "other larcenies." Drug sales do not include STC transactions. na - not available



## Total Crimes and Criminal Income -123-

The theft rates seemed lower than those implied by Ball, et al. (1981; Ball, 1982) and Pollin (1981) when they suggested that addicted subjects committed crimes (mainly thefts) on 248 days/year at risk. Our data suggested that while theft was an important crime, drug selling contributed many more offenses per year at risk.

When examining the total volume of "major crimes" (all nondrug plus drug distribution crimes), a very high level of criminal activity was recorded. The average respondent committed over 825 major offenses per year or almost 2.2 per day. In addition, the daily heroin users committed 1,089 major crimes per year (almost 3 per day) -- of which a half involve steering, touting, and copping. The regular heroin users were almost equally active and committed almost 1,000 crimes per year, of which 60% involved drug sales.

The irregular heroin users here seem as active or more active in most crimes as respondents in other studies (see Table VIII. 1). While their offending rates (360/year) were low in comparison with the regular and daily heroin users, they were considerably higher than those reported by the TOPS clients (Collins, et al., 1982b) and only somewhat lower than Inciardi's (1980) heroin users.

When minor crimes were added to the major crimes, the daily heroin user probably committed 1,400 per year or almost 3.8 per day on the average. The regular heroin users committed almost 1,200 and the irregular heroin users over 500 per year.



				7	[o ta	1 C
	The	data	provi	ded	Ь у	tł
to	the	ques	tions	to	whi	ch
yea	r on	the	street	,	the	fo

## 100 average subjects

82,790		MAJOR C
1 6 7 36 30	590 ,770 ,680 ,210 ,390 ,140	Robber Burgla Larcer Other prost Drug S Steer
	۲LUS	5
22,030	MINOR	CRIMES
104,820	TOTAL	CRIMINA

In short, our daily heroin users committed about twice as many crimes as were estimated by Pollin (1981) by using Ball's crime days for addicted heroin users. But almost all of this high volume of crime were drug sales, steering touting, copping, and minor criminal activity. Our daily heroin users committed only about 137 serious felonies (robbery, burglary, larcenies) per year per subject, a figure somewhat lower than Ball's estimated number of crime days (he assumed it to be theft without providing detailed breaks).

By counting systematically the minor crimes and offenses which occurred several times per day of involvement, the overall volume of criminal activity was extremely high among these street opiate users, higher than previously recorded in most other studies.

Crimes and Criminal Income -124nese street opiate users allow more precise answers Pollin (1981) was responding. During the average llowing number of crimes would be committed by:

## 100 daily heroin users

CRIMES

## 108,870

ries	1,180
aries	3,390
nies	9,140
Nondrug Crimes (forgery, congames,	7,160
itution/pimp, other crimes)	
Sales	31,570
ing, Touting, and Copping Transactions	56,420

(shoplifting for own use, fare evasion, illegal drug transfers)	31,110
L OFFENSES	139,980

## Total Crimes and Criminal Income -125-FINDINGS ABOUT CRIMINAL INCOME

Information about the criminal income of street heroin users was almost nonexistent. Robins (1979:327) writes "we know remarkably little about how much heroin addicts actually do consume or would consume if heroin [were less expensive]." Moore (1977:91) suggests that "heroin users appear as people who struggle to make \$10,000 in the periods they are not institutionalized.... Property offenses finance a much smaller proportion of the total heroin consumption than is usually assumed." Goldman's (1981) review of the economic literature further documents that little evidence is available regarding the criminal or other income of heroin users, especially of street opiate users while at liberty.

The most detailed data regarding the criminal income of heroin abusers were found in an evaluation of the California Civil Addict program (McGlothlin, et al., 1977, 73-74). These data showed that during the period 1970-1975. daily heroin users had a "mean income from crime/year" (excluding dealing/gambling, etc.) of \$10,900, nondaily heroin users had \$1,300 annually, and those using heroin less than monthly had \$300 annually. During person-months of daily heroin use, these subjects had monthly (here multiplied by 12) criminal incomes from robbery (\$396), burglary (\$5,172), theft (\$5,976), and drug dealing (\$5,904).

Data in Table VIII. 3 present various sources and annualized dollar amounts of criminal income by these street opiate users. The data show that daily heroin users have a criminal cash income (\$11,292) that is about one and a half times as high as that by regular heroin users (\$7,121) and two and a half times higher than the irregular heroin users (\$4,451). When their drug income was included as well as the economic value of minor crimes (fare evasion and shoplifting for own use), daily heroin users have a total criminal income of over \$18,500 compared to about \$11,000 for regular heroin users, and about \$6,000 for irregular heroin users.



Total Crimes and Criminal Income -126-Thus, the total dollar amounts of criminal income (both cash and drug) among daily heroin users appeared to be about twice as high\* as that recorded by McGlothlin, et al (1977). Nevertheless, the dollar returns from robbery for our daily heroin users (\$906) was higher than McGlothlin's figure (\$336), but lower for burglary (\$2,906), theft (\$2,334), and drug dealing (\$1,488)than the annualized amounts given in McGlothlin, et al. (1977, 74) and cited above. Thus, like Moore (1977:91), this study found that the criminal income of heroin users relied less on property crimes (burglary, theft, cons, etc.) than was usually assumed -- even though these street opiate users have offending rates that were as high or higher than those in previous studies.

Indeed, the cash income from drug sales and other drug distribution crimes were not substantial even among daily heroin users (\$2,752). Rather, much of the criminal income of daily heroin users, an amount somewhat smaller in magnitude than their cash income from nondrug crime, comes in the form of drug income. That is, daily heroin users obtain over \$7,000 annually as payment in drugs for their drug distribution activity and their success in obtaining "free" drugs (i.e. avoided expenditures - drugs).

Thus, assuming that most heroin or drug consumption was financed by cash income from property crime or direct drug sales would be misleading. These street opiate users obtained almost a third of their criminal income from activities in drug distribution that did not involve cash income.

## Summary

The street opiate users in this study committed criminal activities at rates which were among the highest ever recorded in a research study. This conclusion was true for specific crimes such as robbery, burglary, and theft, and especially true for drug sales.

\*McGlothlin's figures have not been adjusted to account for inflation, so they may be twice as high as given here.

## Total Crimes and Criminal Income -127-

Moreover, the systematic collection of data about steering, touting, or copping (STC) reveals a patterned activity that was committed by most respondents, on a relatively frequent basis, and which results in substantial "drug income" (see Chapter VII). Clearly, STC comprised three crimes frequently committed which have been overlooked in previous research. While legal statutes may implicitly include such activities as involving criminal facilitation, conspiracy, or accomplice, arrests or even official awareness of these roles appeared non-existent.

Although data about these subjects' rates of criminal offending in previous years were not obtained, our staff's impression was that these respondents were involved in such patterned criminal behavior for several years. Most of the subjects were now in their thirties and report many prior years of such activity (except when imprisoned).

This chapter also supported many central findings of and the reported high annual offense rates given by Chaiken and Chaiken (1982ab) and by Ball, et al. (1981). Moreover, careful measurement of the number of offenses on specific days during different interviews at intervals of differing length, lends credibility to the belief that the high crime rates previously documented by self-reported data based on long recall periods may not involve much exaggeration of criminal events in the distant past. Instead, these data suggest that previous studies may have underestimated the number of relatively minor crimes that were so routine and common that respondents would not recall them or researchers would fail to even ask about them.

The conclusions reached by Ball, et al. (1981, 62) and Pollin (1981) seemed conservative. One hundred of our typical subjects would commit over 100,000 crimes annually, and 100 daily heroin users would commit 140,000 crimes per year. Even our irregular heroin users committed as many crimes as were documented in several other studies. The number of drug distribution crimes was extensive on an annual basis, although not when compared with what would be needed to make a comfortable profit from drug dealing.



Total Crimes and Criminal Income -128-While the annual offense rates were high among these street opiate users, the cash income from nondrug crimes were relatively modest. While burglary and theft raised the largest amounts of cash, each of these offenses generated less than \$3,000 annually among daily heroin users, and less than half that amount among the regular and irregular heroin users. Likewise the cash returns from drug sales and STC and other drug crimes were relatively modest compared with the drug incomes (as payments for drug distribution activity and avoided expenditures). That is, approximately a third of the total criminal income of daily and regular heroin users was received as "drug income" and another 16% was cash income from drug distribution activity.

The cash returns from many nondrug crimes and drug distribution crimes were relatively small considering the possible penalities which respondents faced if captured. Perhaps because of the risks involved, many street heroin users preferred to work at the lowest levels of the drug distribution system where they obtained the drugs they wanted without having cash income.

Nevertheless, the data in this chapter showed that these street opiate users have crime rates for almost all offense types that were as high or higher than previous research. They generally had criminal incomes equivalent to or higher than amounts previously recorded, although not for each offense type. Their actual cash returns from crime were relatively modest compared to their total criminal income. Thus, crime provided a very significant portion of their income.

In the next chapter, data is presented regarding their ability to generate income from noncriminal sources.

BDJ:8259A;0743A;bj

ίσται	Crimes and	Criminal I	ncome -12	.9-		
Table VIII.2 -	Annualized Heroin User	Criminal Typology	Offending	Rate (Lam	bda) by	,
	Heroin	User Typo	loav			
Type of Nondrug Crime (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Total (201)	p of F	r
Number (L	ambda) of <u>Of</u>	fenses Per	Year Per	Subject of:		
<sup>a</sup> Any Nondrug Crime	116.0	162.3	208.7	162.6	.06	.17
Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping Other Illegal	2.3 4.5 34.6 5.8 1.0 35.9 23.0 1.8 7.0	4.1 15.1 46.3 21.4 1.2 33.8 26.9 1.8 11.7	11.8 33.9 71.9 19.5 2.5 29.2 32.1 1.5 6.3	5.9 17.7 50.7 16.1 1.5 33.0 27.3 1.7 8.6	.002 .000 .12 .006 .43 .97 .89 .97 .26	.23 .29 .14 .17 .08 01 .04 01 01
<sup>b</sup> Drug Business Crime (Drug Transactions)	244.8	823.5	880.0	665.3	.15	.12
Drug Sales Steer, Tout, Cop	122.8 122.0	590.8 232.7	315.7 564.2	363.9 301.4	.35 .00	.04 .26
ALL MAJOR CRIMES (Nondrug + Drug)	360.7	985.9	1,088.7	827.9	.09	.14
<sup>a</sup> Shoplifting (Own Use)*	6.6	6.9	5.4	6.3	.82	03
<sup>C</sup> Theft of Services (Fare evasion) <sup>d</sup> Obtain Illegal Drugs via	52.3	47.5	70.1	55.9	.49	.06
Other Drug Business Drug Thefts Avoided Expenditures -Drugs	1.5 1.1 85.4	4.0 2.9 144.3	6.6 5.1 224.1	4.0 3.3 150.7	.27 .16 ,000	.11 .14 .59
ALL MINOR CRIMES	147.0	205.6	311.1	220.3	.000	.40
TOTAL CRIMINAL OFFENSES (Major + Minor)	507.7	1,191.4	1,399.8	1,048.2	.04	.17

- THE REAL PROVINC

<sup>d</sup>The percent of active person days times 3.65. This assumes that one offense occurred on an active day; Source Table VII. 2.



Total Crimes and Criminal Income -130-Table VIII. 3 - Annualized Criminal Income for Each Source by the Heroin User Typology

Type of Nondrug Crime (Number of Subjects)

<sup>a</sup>Any Nondrug Crime

Robbery Burglary Shoplifting (Resale) Other Larcenies Forgery Con Games Prostitution Pimping Other Illegal Acts

<sup>b</sup>Total Drug Business Cash Income from:

> Drug Sale Steer, Tout, Cop Other Drug Business

TOTAL CRIMINAL CASH INCOME 4,451

<sup>b</sup>Any Drug Sale <sup>b</sup>Any Steer, Tout, Cop <sup>b</sup>Any Drug Thefts

<sup>b</sup>Avoided Expenditures -Drugs

TOTAL DRUG INCOME

<sup>C</sup>Theft of Services (Fare Evasion)

<sup>a</sup>Shoplifting (Own Use)

TOTAL <u>MINOR</u> CRIME INCOME TOTAL CRIMINAL INCOME [Cash + Drug + Minor)

a - Source: Table VI.7C

Heroin User Typology	
Irregular Regular Daily Total p of	
(6Ĭ) (Ť8) (62) (201) F	r
nnual Cash Income Per Subject from:	
	2
2,885 5,719 8,540 5,729 .000	.34
158 377 906 474 .004	.23
256 1,177 2,906 1,429 ,000	.27
887 1,342 2,334 1,510 .034	.18
142 730 818 579 .003	.21
56 300 93 162 .55	.01
410 408 400 406 .99	.00
739 722 773 743 .99	.00
41 114 100 100 .75	.04
197 552 212 339 .46	.00
1,566 1,402 2,752 1,868 .03	.15
	٥r
278 594 1 153 671 014	.00
88 64 110 86 .78	.02

	Dollar	Value of	Drug Income	per Year	from:	
)	59 476 70	213 1,378 385	548 2,949 519	269 1,589 331	.15 .000 .17	.14 .30 .13
5	752	1,663	3,168	1,851	.000	.34
	1,357	3,639	7,184	4,040	.000	.48
	39	36	53	42	.49	.06
	97	80	57	79	.49	09
	136	117	110	120	.76	05
	5,943	10,877	18,585	11,757	.000	.49

11,292

7,597

.000

.36

7,121

b - Table VII.3 C - Table IX.3

Noncriminal Income -131-CHAPTER IX

NONCRIMINAL INCOME AMONG HEROIN USER TYPES

Relatively little information is available about the income which heroin users or other drug users obtain. Most of the data used for economic analyses (Brown and Silverman, 1974; White Paper on Drug Abuse, 1975; Panel on Drug Use and Crime, 1976; Goldman, 1976; Research Triangle Institute, 1976; Silverman and Spruill, 1977) rely heavily upon reports of income given at admission to treatment, or intensive retrospective interviews in follow-up studies. Reliance upon such information, however, may suffer from admittees reporting their self-image rather than their actual behavior (Goldstein, 1979, 1982b). Moreover, street opiate user "income" tends to come in two forms, cash payments and drug payments. Reports to interviewers may combine such forms of income but researchers seem to have assumed all such income was cash.

One of the major surprises to emerge from the pilot years of research is the widespread nature of income other than cash among heroin users. Goldstein (1981) shows that many street respondents can "get by" or "get over" (obtain and use drugs) without ever having cash. Although the dollar amounts of drugs consumed without cash purchases was seldom large, a sizable proportion of respondents obtained drugs on several days without paying cash for them. Goldstein's (1981) analysis of various ways of "getting over" used ethnographic materials from this research and supplemented the quantitative economic data presented in this report.



Noncriminal Income -132-This chapter provides relatively detailed information about all income from sources other than crime -- which was presented in prior chapters. Because employment income was rare among these respondents, any cash income from legitimate work, even though tinged with illegitimacy, was considered as employment income (i.e., the respondent received payments from passengers as a a gypsy cab driver while using a stolen car; doing housework (such as buying groceries and cleaning and cooking for a drug dealer; doing carpentry work to fortify a dealer's apartment and similar quasi-illegal activities have been coded here as work or job). Of course, cash payments from work in a regular job (porter, cab driver, florist, carpenter, etc.) were also counted as work income.

Respondents avoided paying income tax on their cash income. Such nontaxed income was considered here as cash income from whatever source (criminal or legal). Most subjects had little or no contact with the Internal Revenue Service.

So pervasive was the phenomenon of "getting by" (Goldstein, 1981) among our respondents, and so typical was bartering of one type of good for another, that the very notion of "income" must be carefully reexamined. Of course, cash or money earnings were a part of income. In addition, a variety of other types of income must also be included. The following rule was developed: if a respondent received objects or goods having a relatively standard economic value for which money was typically paid, that value should be included (to the extent possible) as "income" to the respondent. Such noncash income predominately came in two major forms: a) payment in drugs, and b) avoided expenditures. Drug payments have been discussed in chapter VII, and will not be further discussed here.

## Noncriminal Income -133-

Many subjects frequently received extensive support from mothers, spouses, girlfriends, relatives, or other friends. Such persons frequently supplied these subjects with one or two meals per day (usually breakfast and/or supper) and a place to sleep during the night. These respondents did not typically provide their host with significant payments for such services, although they occasionally contributed a few dollars (see Chapter X) when they had unexpended funds -- which was seldom. Being able to eat and sleep without major cash expenditures was a substantial contribution to respondent income. But a problem arose, how much were such "free" meals and lodging worth? The subjects were unwilling to place a value upon them, or did not perceive that a meal from their mother, spouse, or friend was a form of income. Even though all of our subjects were over age 18, and the majority were over age 30, and were emancipated from their parents, many continued to live with them. Moreover, while some subjects lived with their wives, common-law relationships were more likely. Many subjects moved from girl friend to girl friend. Others resided in the parks or abandoned buildings.

When subjects received goods or services having an economic value without paying cash, it will be referred to as "avoided expenditures." These values, however, were a real cost to someone in society and permitted the respondent to spend his cash income for other purposes, mainly drugs. In order to place a value on "free" meals and shelter, the investigators obtained information from the Welfare Department. A person on home relief (the form of income assistance available to poor persons without employment in New York City) received a monthly allowances of \$194 (\$94 for food, \$100 for shelter) plus food stamps. In the statistical data below, person-days in which meals or shelter costs were paid by someone else, a conservative value of \$3 for meals (breakfast and supper), and \$3 for shelter were assigned as the value of noncash income received by the respondent. In addition, subjects occasionally reported gifts of clothes, attending a movie, etc. for which they did not pay. This was counted as an "other" avoided expenditure.



A frequent form of crime committed by many respondents was fare evasion. They jumped subway turnstiles, sneaked in the back of buses, or tossed a few nickels in the coin box of a bus. While these were minor crimes, they still received no cash income from them but avoid necessary costs -- so this was included here as an "avoided expenditure - transportation."

In addition, respondents frequently received free alcoholic beverages (a can of beer, wine, etc.) from friends. Several respondents were also good at "bumming" cigarettes from other persons, frequently the equivalent of a pack or more per day, or they "borrowed" someone else's cigarettes with no intention of repaying. Alcohol and cigarettes obtained without cash purchases were not included in this chapter; they were included as "avoided expenditures - drugs" in Chapter VII on drug business crimes.

Sly C. (black male, age 31) was an irregular heroin user. He paid minimal 'rent' by staying with his cousin and paying him 'whatever I can afford." He stayed with his cousin on all 33 reporting days, paying him an average of \$17 per week. Also, Sly gets almost all his food for free. On 23 of his person days he took 3 meals a day at a nearby Youth Center - "I bring the food home and cook it myself". On 5 days he took all his meals at his cousins and he ate for 4 days with his mothers; on one day he did not eat at all.

Sylvio X. (black male, age 41) was a daily heroin user who lived with his sister even though he had a wife and five children -- to whom he claimed to contributed \$100 about 3 times per year. Although he claimed to play the rent for his sister's apartment in his life history interview, he never once reported a payment for rent during 117 days of interviews across three calendar years. His sister provided him with a place to sleep every night and cooked most of his meals for him. He typically spent \$3/day on food (mainly snacks) and rarely expended more than \$5/day on food.

On 12/17/80, he gave \$50 to his sister and reported the next week that "I didn't get nothing for Christmas. I didn't give nothing." On 9/2/81, Sylvio obtained \$250 from a burglary and spent \$55 on heroin. The following day, he gave \$55 to his sister and bought \$118 of clothes for his kids. Significantly, he had six meals with his wife during that week. He also gave his sister \$10 on 11/27/81, \$5 on 12/3/81, \$25 on 3/3/82. and \$20 on 3/17/82. No further contributions to his wife or children were reported. But on all 117 days he reported lodging at his sister's apartment, and eating meals (supplied by her) on three quarters of the days. Thus, his sister provided him with far more "free" shelter and meals than he provided her with money.

Noncriminal Income -134-

## Vignette T -- Avoiding Expenditures

Noncriminal Income -135-FINDINGS

## Was noncriminal income related to the frequency of heroin use?

For every source of noncriminal income, no significant differences in noncriminal income emerged among heroin user types. All three heroin user types have very low incomes from any noncriminal source.

## What were the main sources of noncriminal income among the three heroin user types?

Graph IX. 1 and Tables IX. 1 - 3 show that the most striking finding was the relative absence of legitimate sources of cash income. About half the respondents reported some employment, 45% reported welfare support, while a third reported other public support, mainly food stamps. Only 3% had unemployment benefits.

Subjects reported employment income on only 6% of their person-days. Although the statistical data do not indicate it, most of this "work" involved short term odd jobs. Typical work activities included: helping a friend or relative move and receiving \$30 for the day; helping a delivery man watch his truck; working on a friend's car; painting friends' apartments; taking thrown out mattresses to a mattress factory; washing car windows, etc. Such employment provided about \$1,000 per year per subject in cash income. The irregular heroin users had somewhat higher income from work.

Likewise, welfare, food stamps, and unemployment benefits provided support for about half the subjects, but the amounts of cash were limited to about \$1,250 per subject annually. The regular heroin users appeared somewhat more likely to receive such payments, but the differences were small.



Noncriminal Income -136-

## Noncriminal Income -137-

Family, spouse, and friends provided relatively modest amounts of cash income to these respondents (under \$500 each source). While respondent payments would have averaged about \$700 per year had they been interviewed for 52 weeks (none were), all subjects received the same \$10 for a weekly interview. Such income would not normally be available to them. All subjects received respondent payments, which was not true for other sources of income. Panhandling was least rewarding, gambling was only slightly more so.

Did these subjects avoid necessary expenditures?

Almost all subjects avoided expenditures which would be major expenditure items for the average person. As the next chapter will show, these respondents spent very little (under \$2,000) for food and shelter. Nevertheless, most of the subjects appeared to be relatively healthy, and normally fed in comparison with others in their low income neighborhood.

Table IX. 1 shows that virtually all subjects systematically engaged in expense avoidance--95% of the respondents received one or more free meals, 82% received free shelter, and almost half received some other commodity or service during the reporting period. Moreover, "free" meals and shelter were obtained on over half of all person-days (Table IX. 2). When these avoided expenses are calculated at three dollars per day, the subjects received an annual value of over 5600 in food and almost 5600 in shelter (Table IX. 3). These avoided expenses were largely "paid" by the welfare system because the subject's relative (mother, spouse/paramour, sister, other relative, or friend) had a welfare grant (or other welfare income) which paid for the apartment in which the subject slept. Likewise, the host's food stamps and the welfare grant paid for the food which the respondent consumed.

The data on avoided expenditures also shows that about about half the respondents engaged in fare evasion ("avoided expenditures transportation"), although they do so on less than 10% of their days. The dollar value of such fare evasion was not large (about \$40 per year). Again such avoided expenditures did not vary by the frequency of heroin use



Clearly, these respondents live at or below the poverty level. They probably have an annual noncriminal cash income of less than \$5,000, with almost no variation by type of heroin user. (See Graph IX. 1)

On the other hand, their annualized criminal income was equal in dollar amount to (among the irregular heroin users) or considerably greater than their noncriminal income (among the regular and daily heroin users). If their criminal income was included as a form of cash income, their total cash income doubled among irregular heroin users or triples among daily heroin users above the levels provided by their noncriminal income.

## Summary

The central finding of this chapter, however, was that our respondent's noncriminal income did not vary by the frequency of heroin use. All three types showed little ability to raise much money from work, welfare, or other public support. They depended heavily upon family and friends to provide them with the minimum essentials for living such as food and shelter. In 1981, the official poverty level was \$4,729 for a single person aged 15-64 (Beeghley, 1983); our respondents obtained just about \$2,000 in legitimate cash income (work, welfare, unemployment benefits), so they were seriously impoverished by comparison. In these study communities, however, these respondents could probably live a comfortable low income existence on their average annual income of about \$12,000 (cash income from crime plus nondrug crime cash income). But most of these street opiate users lived considerably below the poverty line -- because they spent so much on drugs.

than drugs.

BDJ;8260A;0744A;bj

## Noncriminal Income -138-How did criminal income compare with noncriminal income?

In the next chapter, we examine their expenditures for purposes other

## Noncriminal Income -139-

Table IX. ] --

ŗ

Percentage of <u>Respondents</u> Reporting Cash Income and Avoided Expenditures from Various Sources by Heroin User Typology

	Heroin	User Typol	ogy			
Type of Income (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Tota (201	] po ) F	f r
	Durin <u>Respo</u>	ig Report ndents Rep	ting Da orting Ca	ays, Pe ash Incom	rcenta e from	age of :
A. Any Noncriminal Source	98.4	98.7	98.4	98.5	.98	.00
Working	50.8	51.3	43.5	48.8	.62	06
Welfare	37.7	55.1	40.3	45.3	.08	.02
Unemployment	4.9	2.6	3.2	3.5	.75	04
Other Public Support	23.0	37.2	33.9	31.8	.19	.09
Family	72.1	66.7	50.0	63.2	.03	18
Spouse/Paramour	57.4	52.6	50.0	53.2	.71	06
Friends	59.0	66.7	64.5	63.7	.64	.04
Panhandling	23.0	29.5	25.8	26.4	.68	.03
Gambling	27.9	25.6	33.9	28.9	.55	.05
Respondent Payments**	100.0	100.0	100.0	100.0		
Other	50.8	57.7	61.3	56.7	.50	.08
B. Avoided Expenditures Total	98.4	96.2	98.4	97.5	.62	.00
Room (Shelter)	80.3	79.5	87.1	82.1	.47	.07
Meals	96.7	93.6	95.1	95.0	.70	03
Transportation*	40.1	50.0	53.2	43.3	.37	.10
Other	39.3	48.7	45.2	44.8	.55	.05
TOTAL NONCRIMINAL INCOME (A+B)	100.0	100.0	100.0	100.0	~~~	
C. Criminal Income	93.4	96.2	98.4	96.0	.38	.10
Drug Business	78.7	70.5	77.4	75.1	.48	01
Nondrug Crime	85.2	91.0	96.8	91.0	.08	.16
Total Cash Income (A + C)	100.0	100.0	100.0	100.0	19 10 10 10 10 10 10	till och dat get den tan
TOTAL INCOME (A + B + C)	100.0	100.0	100.0	100.0		

\*\* Not included in the totals.



# Type of Income (Number of Subjects) A. Any Noncriminal Sources Working

Welfare

Unemployment

Other Public Support

Family

Spouse/Paramour

Friends

Panhandling

Gambling

Respondent Payments\*\*

Other

B. Avoided Expenditures Total

Room (Shelter)

Meals

Transportation\*

Other

TOTAL NONCRIMINAL INCOME (A+B)

C. Criminal Income

Drug Business

Nondrug Crime \_\_\_\_

Total with Cash Income (A + C)

\* Jump Subway Turnstile mainly \*\* Not included in the totals.

Noncriminal Income -140-

Table IX. 2 -- Percentage of <u>Person Days</u> Reporting Cash Income and Avoided Expenditures from Various Sources by Heroin User Typology

	Heroin	User Typolo	ogy	<b>-</b>		
	Irregular (61)	Regular (78)	Daily (62)	(201)	p of F	r
	Perce	ntage of <u>Da</u>	<u>ays</u> Report	ting Cash	Incom	e from:
	26.9	28.0	25.6	26.9	.72	03
	6.7	5,7	5,3	5,9	.75	05
	1.6	2.6	1.9	2.1	.08	.04
	0.4	0.1	0.2	0.2	.41	05
	0.7	1.1	0.9	0.9	.41	.06
	6.7	4.9	4.0	5.2	.15	13
	8.3	6.3	4.2	6.3	.12	15
	2.6	5.4	5.4	4.6	.03	.16
	0.8	2.7	3.6	2.4	.09	.15
	1.3	0.9	1.2	1.1	.54	02
	17.6	17.9	18.2	17.9	.64	.07
	2.4	3.6	3.0	3.0	.37	.05
	67.0	64.1	67.5	66.0	.81	.01
	53.8	46.9	58.4	52.5	.22	.05
	57.5	56.3	56.2	56.6	.97	02
	8.1	7.0	10.4	8.4	.48	.05
	1.6	2.3	1.6	1.9	.31	.00
-						
	83.4	84.2	87.8	85.1	.33	,10
	24.7	31.9	50.4	35.4	.000	.43
	13.9	13.1	20.7	15.7	.030	.15
	13.5	22.6	33.3	23.1	.000	.38
-	44.6	52.6	65.7	54.2	.00	.41

## Noncriminal Income -141-

Table IX. 3 --Mean Dollar Amounts of Cash Income Per Year and Annual Value of Avoided Expenditures from Various Sources by Heroin User Typology

Heroin User Typology								
(Number of Subjects)	(61)	(78)	(62)	(201)	<u> </u>			
A. Any Noncriminal	4,551	4,556	4,209	4,447	.8204			
Income: Working	1,546	815	852	1,048	.1711			
Welfare	584	1,177	757	868	.031 .05			
Unemployment	153	50	71	91	.5806			
Other Public Support	244	366	256	295	.45 .01			
Family	303	235	251	260	.7304			
Spouse/Paramour	443	462	477	461	.98 .01			
Friends	76	167	118	125	.030 .07			
Panhandling	11	59	50	41	.27 .08			
Gambling	160	95	301	178	.20 .08			
Respondent Payments**	r 733	760	770	755	.47 .08			
Other	299	360	307	325	.93 .00			
B. Avoided Expenditures Total	1 ,401	1 ,341	1,398	1,377	.90 .00			
Room (Shelter)*	588	518	639	577	.25 .05			
Meals*	649	618	615	626	.8504			
Transportation	39	36	53	42	.48 .06			
Other	125	169	91	132	.6803			
TOTAL NONCRIMINAL INCOME (A+B)	5,952	5,897	5,607	5,824				
C. Criadnal Income	4,404	7,010	11,252	7,528	.000 .36			
Drug Business	1,580	1,349	2,714	1,840	.030 .14			
Nondrug Crime	2,824	5,661	8,541	5,687	.000 .35			
Total Cash Income (A + C)	8,955	11,566	15,461	11,975	.000 .32			
TOTAL INCOME (A+B+C)	10,355	12,906	16,860	13,352	.000 .32			

\*\*Resnandant navments



Remarkably little is known about cash expenditures of street opiate users and street hustlers for purposes other than drug purchase. The treatment outcome literature and the social costs of drug abuse literature was almost devoid of references to, much less data regarding, the nondrug expenditures of heroin users or other drug users. Occasional ethnographies (Preble and Casey, 1969; Gould, et al., 1974; Hughes, et al., 1977) make passing references to techniques used by addicts for living with minimal cash expenditures. Almost no dollar figures were located about how much money street addicts spend for basic living costs or other purposes. The data presented in this chapter and ethnographic accounts reported by Goldstein (1981; Goldstein and Duchaine, 1979) constituted some of the first available quantitative data on patterns of cash expenditures by street opiate users.

Previous researchers have had a good reasons for ignoring nondrug expenditures by street opiate users: they expended as little as possible on basic needs. Many subjects frequently reported that they "get by" or "get over" (Goldstein, 1981); that is, they obtained drugs or made a living but did not pay cash. Many of the mechanisms by which they avoid expenditures (see Chapter IX above) permit minimal cash expenditures for basic survival.

These subjects also paid minimal amounts for shelter and food. They occasionally contributed to a household in which they were residing, but had relatively few cash expenditures for any other purpose. Even their "savings," recreation, gambling, and alcohol/cigarette consumption was tinged with illegality.

# Nondrug Expenditures -142-CHAPTER X NONDRUG EXPENDITURES AMONG HEROIN USE TYPES

Nondrug Expenditures -143-FINDINGS

Were nondrug expenditures related to the frequency of heroin use? For almost all kinds of nondrug expenditures, as well as totals, few significant differences emerged among heroin user types. All three groups exhibited very low levels of cash expenditures for nondrug purchases.

## What were the main kinds of nondrug expenditures among the three heroin user types?

Tables X. 1 - 3 and Graph X. 1 show that although almost all respondents only about half of these respondents had any cash purchased food, expenditures for any shelter costs (i.e., they lived with someone else). About two-thirds of the subjects provided cash to their families and purchased While two-thirds also reported some "savings," this was some clothes. typically money set aside for the following day. Three-quarters spent money for transportation and to purchase "other" goods (such as gifts for someone else, hardware, etc.). Half of the subjects reported recreation (movies, video games, etc.) expenses. Almost all respondents (94%) were cigarette smokers and purchased this substance. Excepting shelter costs, the daily heroin users were not significantly different than less regular heroin users in these expenditure patterns (Table X. 1).

Even though a sizable proportion of respondents engaged in cash expenditures for the various purposes, such expenditures were not regular activities. Table X. 2 showed that only on about one day a month (3% of person-days) did they pay for shelter costs. On only half of the person-days do they pay for any food. Daily heroin users paid for food on about 12% more person-days than less regular heroin users. Subjects paid for transportation on 15% of the person-days. Cash expenditures to family, gambling, and other purposes occurred on about 7% of the person-days. Clothes, savings, recreation involved cash expenditures about 3% of the time. Legal fees were rarely paid by respondents because they were rarely arrested and were almost



CARGE OF LOUGH ST.

\$4,645

Shelter	420	
Food	1,183	
Famil y	731	
Clothes	260	
Trans-	177	
Savings	327	
Gambling	115	
Other Nonsub – stance Expenses	462	
Alcohol	578	
Cigar - ettes	291	
[Ilicit Drug Purchase (see Table V.5)	10,060	

Nondrug Expenditures -144-

Graph X.1

Nondrug Expenditures by Heroin User Types



HEROIN USER TYPE

## Nondrug Expenditures -145-

These respondents had marginal cash expenditures for nondrug purposes. An annualized estimate of all nondrug expenditures was \$4,256 per year, with the daily heroin users having slightly lower expenditures (under \$3,800/yr) than the regular and irregular heroin users. About \$1,100 was spent on food, and about \$700 was spent for shelter or and a similar amount was given to family (sometimes as partial payment for shelter and meals eaten there). Clothes, transportation, recreation, and gambling received under \$300/year in expenditures each. "Savings" and other expenditures accounted for less than \$600 in each category.

Generally, no significant variation by the frequency of heroin use emerged. Daily heroin users, however, expended only \$420/year for shelter and hence were significantly less likely to pay for shelter then the regular heroin users.

The central finding was that respondents spend very little cash for nondrug purchases. Food and shelter were important components of the average household's expenditures, but not for these respondents (see next section). They expendied cash upon drugs, alcohol, or cigarettes on 78% of their days. versus 60% of days having cash purchases of nondrug items (including food). In fact, they had a larger percentage of days with purchases of drugs (56%) and cigarettes (50%) than with purchases of food (48%). They had more days with alcohol purchases(39%) than any other nondrug, nonfood item (all under 10% of their person-days (Table X. 2). Thus, their typical days were more often filled with cash purchases of drugs, alcohol, and cigarettes, than with expenditures for other purposes excepting food. Moreover, the dollar amounts expended on substances (excepting the irregular heroin users) per year was considerably higher than their nondrug expenses.



While the typical low income person generally spends a 25% to 35% of his or her money upon shelter, these respondents spent about 15% of their nondrug income, and about 6% of their total cash expenditures, for shelter. Moreover, the dollar amount paid for shelter per year, under \$1,000 by any heroin user type, was very low even when compared with low income persons receiving home relief (who receive about \$1,200/year for shelter). The daily heroin users expended about half that amount on shelter. All three groups depended heavily upon others to provide them with free lodging (about \$600 worth per year) on over half their days (Table IX. 2). Moreover, these figures for shelter do not include person-days when the respondents slept in an abandoned building, in the park, or did not sleep -- such days have not been assigned an economic cost although they constitute an "avoided expenditure."

## Vignette U -- "Getting by" Shelter Costs

Tato A. (Hispanic male, age 25) was a regular heroin user. Tato was separated from his wife "because she is against my drug use". He maintained his own apartment but he can't go there because "I owe several months rent (\$80/month)." So he slept wherever he could - abandoned buildings, park benches, or at friends houses. On one occasion he reported: "a friend on welfare whose house I slept at Sunday night doesn't like my drug use. It was very late when I got home, too late to wake him up, so I slept in the park."

Ouinn N. (Hispanic male, age 32) was a daily heroin user. In July, 1980 he lived in an empty room in an apartment that he keeps clean for the building superintendent; he gives 'the super' \$15 twice a month for the room. "When somebody comes to look at the apartment I have to leave."

reported in Vignette T.

Nondrug Expenditures -146-HEROIN USERS AND NONDRUG EXPENDITURES FOR SPECIFIC PURPOSES

## SHELTER

Sylvio X. also avoided almost all shelter costs by living with his sister as

## Nondrug Expenditures -147-

Although the data were not presented here, parents, spouses/paramours, or other extended family provided much of the free shelter received by these respondents. In some cases, the subject's cash expenditures to "family" provided a few dollars which may have paid for some share of the rent, some food purchases, or otherwise offset some of the other household costs. Typically, large payments for rent occurred mainly among respondents who received welfare payments or had some employment income; occasionally criminal income was used to pay rent. Another way of paying rent was to provide the parent/family member with \$10-\$50 once or twice per week, with the other person using such cash for whatever was needed.

## FOOD

While expenditures for food occurred on half the days, and involved the largest single item of nondrug purchase (51,100) per year, the dollar amount (\$6.30) per day with food purchase was relatively modest. This figure included both small purchases of soda, candy, slices of pizza, hot dogs, sandwiches, as well as purchases of larger meals and shopping trips for the family. Snacks were the most frequent way of spending money upon food. Full meals were typically provided by parent/spouse/paramour/family and were described as "avoided expenditures" in Chapter IX. Another common way of obtaining food was by shoplifting donuts, fruit, canned goods, sodas, etc. for personal consumption; this was included in "shoplifting (own use)" ---Chapter VI.

## FAMILY

Cash expenditures or money given to the "family" generally involved the respondent providing small amounts (\$10-\$20 or less) to the woman of the household where he was currently residing to "buy some groceries" or "help with the rent." Frequently, such sums were given by the respondent due to a direct request for the funds. The respondent frequently had to be located on



Nondrug Expenditures -148the street with cash in his pocket or be threatened with being placed out of household to obtain such funds. Except for several female respondents with children, most of these subjects exhibited little obligation to supporting or helping support in a systematic way their family or household expenses -- even though most respondents were males in their thirties.

To these respondents, and the way the interview schedule was designed, "savings" meant having a few dollars (generally less than \$10) in their possession at the end of the day or reporting period. Almost none of these respondents had a bank account, in which unexpended cash was deposited and saved. These cash "savings" typically disappeared early the next day. Only on rare occasions when the respondent had a "big score," and had several hundred dollars, would money be given to a spouse or other person with an account for deposit.

These subjects expended about \$275 per year on clothes. In addition, some respondents obtained clothing from shoplifting. Payments for transportation typically involved taking a taxi (or car service in this neighborhood) to carry stolen merchandise to a fence, or to make a "connection" to buy drugs. Subway turnstiles were routinely jumped and this expenditure avoided. Subjects seldom reported any legal fees.

About half of these subjects played the numbers (or other form of gambling), and did so on about 8% of the person-days. The dollar amounts lost (513] annually), however, was not substantial.

## SAVINGS

## CLOTHES, TRANSPORTATION, AND LEGAL FEES

## GAMBLING

## Nondrug Expenditures -149-OTHER NONSUBSTANCE EXPENDITURES

This category included a wide variety of purchases which did not fit within the other categories. Purchases of hardware for the household, buying parts for a car which was being fixed for a customer, money or gifts given to friends, etc. are included here. Such expenditures involved less than \$600 per year.

## COMPARISONS WITH EXPENDITURES FOR SUBSTANCES

These respondents had many more days with expenditures for substances than for nondrug expenses, excepting food. Moreover, the dollar amounts expended upon substances was generally higher. Only among irregular heroin users did the average annual amount expended for nondrug purposes (\$4,635) exceed annual expenditures (\$4,275) for drugs, alcohol, and cigarettes. On an annual basis, regular heroin users spent one and a half times more on substances than for nondrug purposes (\$6,605 vs. \$4,344). Daily heroin users spent about three times more on substances than for other nondrug purposes (\$10,859 vs. \$3,776).

## Summary

Street opiate users were quite successful in limiting their cash expenditures for nondrug purposes. They paid very little for shelter and food in comparison with their low income counterparts. The dollar amounts expended for other purposes were even lower. This permitted them to maximize the amount of cash expended upon drugs, alcohol, and cigarettes. There was no significant variation, however, by the frequency of heroin use.

In the next chapter, their total income and expenditures is compared.



Table X. 1 --Percentage of Respondents Reporting Cash Expendituresfor Various Purposes by Heroin User Typology

## Type of Expenditure (Number of Subjects)

Any Nondrug Expenditures Shelter Food

Family

Clothes

Transportation

Savings

Recreation

Gambling

Legal Fees

\_\_\_\_\_

Other Nonsubstance Expenditures

Substance Use Total

Drugs

Al cohol

Cigarettes

Total with Cash Expenditur

Nondrug Expenditures -150-

	Hero	in User Typolo	gy			
T	rregul	ar Regular	Daily	Total	p of	r
<u>.</u>	<u>(01)</u> Pe	rcentage of	Respon	idents Repo	rting	Cash
	Ex	penditures for				
	100	99	100	100	.46	.00
	56	54	40	50	.17	12
	98	97	100	99	.46	.05
	77	62	68	68	.15	07
	69	64	53	62	.19	13
	80	86	77	82	.42	03
	69	58	65	63	.39	03
	51	53	45	50	.67	04
	51	53	48	51	.89	02
	2	4	0	2	.27	05
	90	81	90	87	.16	.00
	100	100	100	100		
	100	100	100	100		- ç-
	82	85	89	85	.57	.07
	95	94	92	94	.78	05
es	100	100	100	100		

Nondrug Expenditures -151-

A NEW YORK THE AND A DESCRIPTION OF A DE

Table X. 2 -- Percentage of <u>Person Days</u> Reporting Cash Expenditures from Various Sources by Heroin User Typology

	Heroin U	ser Typolo	gу			
ype of Expenditure (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Total (201)	p or F	r
	Percen	itage of	Days	Reporti	Cash	
Any Nondrug Expenditures	Expend 56.8	litures for 56.8	66.2	59.7	.03	.16
Shelter	3.1	2.7	1.8	2.6	.11	14
Food	42.2	44.5	57.5	47.8	.001	.24
Family	7.2	5.8	5.9	6.2	.63	06
Clothes	3.7	3.2	3.4	3.4	.82	02
Transportation	16.2	15.3	13.2	14.9	.57	07
Savings	3.8	3.3	3.3	3.5	.71	05
Recreation	3.9	3.1	3.4	3.4	.78	03
Gambling	6.8	7.3	8.3	7.5	.81	.05
Legal Fees	*	*	0.0	*	.17	03
Other Nonsubstance Expenditures	8.3	7.4	7.8	7.8	.77	03
Substance Use Total	71.4	75.7	86.3	77.6	.000	.27
Drugs	43.5	53.9	70.3	55.8	.000	.46
Alcohol	30.9	33.6	52.5	38.6	.000	.26
Cigarettes	46.1	47.6	57.9	50.3	.07	.15
Total with Cash Expenditures	80.7	83.6	91.3	85.1	.001	.25

\* Less than 0.05, but greater than zero.



Type of Expenditure \_\_\_\_(Number of Subjects)

Any Nondrug Expenditures Shelter Food Family Clothes Transportation Savings Recreation Gambling Legal Fees Other Nonsubstance Expenditures Substance Use Total Drugs Al cohol Cigarettes \_\_\_\_\_ TOTAL CASH EXPENDITURES (Nondrug + Drug)

Ho Irre (6	eroin User 1 gular Regu 1) (78	Typology Ilar Daily 3) (62)	7 Total (201)	p of F	r
[	Dollar <u>Expe</u> r	nditures Per	<u>Year</u> Per S	ubject for:	
4,635	4,344	3,776	4,257	.58	07
641	886	420	668	.03	08
1,099	1,038	1,183	1,101	.70	.03
664	675	731	689	.96	.02
328	247	260	275	.59	06
133	127	177	144	.53	.06
872	548	327	578	.39	10
116	66	102	92	.42	02
125	149	115	131	.77	01
2	2	0	1	.48	06
654	606	462	576	.65	06
4,275	6,605	10,859	7,210	.000	.45
3,699	6,019	9,990	6,540	.000	.45
336	341	578	413	.003	.20
240	245	291	258	.29	.10
8,910	10,949	14,635	11,467	.000	.27

Nondrug Expenditures -152-Table X. 3 -- Mean Dollar <u>Expenditures</u> Per Year for Various Purposes by Heroin User Typology.





- 0

# Total Income and Expenditures -153-CHAPTER XI

# TOTAL INCOME AND EXPENDITURES BY HEROIN USER TYPES

In 1975, several experts undertook an analysis of the existing drugs and crime literature and reported findings to the National Institute on Drug Abuse. The report of the Panel on Drug Use and Criminal Behavior (1976) raised several issues and questions that can now be addressed by data from this project.

- a. How much income do street opiate users have and from what sources?
- b. How much cash do street opiate users expend on various items (food, shelter, heroin, other drugs)?

Previous chapters have presented detailed data regarding all major components of income and expenditures. These data will be combined in various ways in this chapter. Total income from all sources has been considered first, followed by cash expenditures.

## TOTAL INCOME

The income of these respondents was more complex than previously reported. This study found that four analytically different kinds of "income" were received by street opiate users. Cash income from both drug distribution and nondrug crime was the major form of income for heroin abusers. While cash income from noncriminal sources was expected to be relatively substantial, little information was previously available about the amounts obtained by active street opiate users.

Two other major forms of income did not involve the direct receipt of money (cash). "Drug Payments" were obtained when the respondent exchanged his labor in the drug business for the drugs he used. That is, he received "payments in drugs" instead of cash for his labor in helping to sell or distribute drugs.

9 9



"Avoided Expenditures" were a form of noncash income in that respondents did not expend cash to pay for other services. That is, someone else paid for food or shelter costs so the street opiate user avoided such necessary cash expenditures. Likewise, drugs which they received as gifts or by "copping short" (see Chapter VII), and alcohol or cigarettes which others give them (but involved no cash expenditures) were a form of "income" to which a relatively standard economic value was assigned.

The important point was that such noncash income constituted real economic value to the street opiate users that was directly equivalent to cosh income. That is, respondents received payment for their labor (i.e, sales work) instead of cash income or someone else paid cash income for the food or shelter which they consumed, or drug distributors lost some economic value when respondents avoided cash payments for the drugs consumed.

Table XI. 1 combined categories of data presented in other chapters, as indicated in the footnotes. For example, theft includes shoplifting for resale and other larcenies; "other property crime" is the sum of forgery + con games (fraud) + other crimes in Table VI. 7C.

## How much income do street opiate users have and from what sources?

The data in Graph XI. 1 and Table XI. 1 showed that street opiate users had a total <u>cash income</u> of almost \$15,000 per year of street time (assuming no incarceration) compared to over \$8,000 per year among the irregular heroin users. Moreover, the daily heroin users obtained a total income (from all four types of income) of almost \$23,600 per year, compared to \$16,300 among the regular heroin users, and about \$11,400 among the irregular heroin users.

Total Income and Expenditures -154-



Total Income and Expenditures -156-Daily heroin users had a cash income from noncriminal sources (under \$4,000/year) that was equal to the less regular heroin users (as shown in more detail in Chapter IX). The daily heroin users, however, obtained about twice as much income as irregular heroin users from criminal cash income (\$11,292 vs. \$4,451), and avoided expenditures (\$4,857 vs. \$2,484) and about nine times (54,016 vs. about \$605) as much in drug payments. The regular heroin users were intermediate, but generally had amounts of income that were more similar to irregular heroin users than to that of daily heroin users from each type of income.

Cash income of all types (criminal + noncriminal) constituted only two-thirds (\$11,289/\$17,017) of the total income received by the average subject. Daily heroin users obtained only 62% of their total income as cash income, while 67% of the regular heroin users, and 72% of the irregular heroin users total income came in the form of cash.

Thus, daily heroin users had the same low noncriminal cash income as the less regular heroin users, but they obtained higher income from almost every other category. In short, daily heroin users had a much higher total income because their cash returns from crime were substantially higher, as was their ability to gain drug payments and to avoid expenditures. All three types of respondents had a total income (projected) for a street-year which was quite high in comparison with the cash income of welfare families and even the low income workers in the study community. Nevertheless, these respondent's living standard was generally far below the poverty level -- because they spent so much upon drugs.

various items (food, shelter, heroin, other drugs)? Table XI. 2 presented a similar analysis of the mean cash expenditures by these street opiate users is presented. Data are recombined from tables in earlier chapters (see footnotes). Graph XI. 2 presents this information in visual form.

How much cash do street opiate users expend on

## Total Income and Expenditures -157-

## Graph XI.2





(57,250/\$4,255) as upon food, shelter, and other nondrug expenditures.

Daily heroin users had the same or slightly lower cash expenditures for nondrug purposes than the less regular heroin users. On the other hand, daily heroin users (\$10,930) expended close to three times as much on drugs as the irregular heroin users (\$4,301) and about twice as much as the regular heroin users (\$6,635). This difference was entirely due to the dollar amount of heroin purchased (this was largely due, of course, to the definition of the Heroin User Typology). For no other drugs or nondrug expenditures were the daily heroin users substantially different from the less regular heroin users.

irregular heroin users. was true of all three heroin user types.

Total Income and Expenditures -158-The average respondent had somewhat over \$11,500 in cash expenditures per year, of which over a third (\$4,203) involved cash purchases of heroin. Over half (\$5,990) of these cash expenditures were for heroin or cocaine. The average respondent spent over one and a half times as much upon drugs

Comparison of Income, Cash Expenditures, and Substance Use The data in Table XI. 1 and 2 can be briefly compared to see how closely cash income and cash expenditures matched. These values have been summed from a variety of different component parts (as indicated in the Tables). Table XI. 2 showed that cash expenditures were approximately equal to cash income among daily and regular heroin users (less than \$30/year difference), but cash expenditures were approximately \$700 greater than income among This showed that the their cash income and expenditures were approximately the same. The additional income available as respondent payments may accounts for the differences which emerge here.

Tables XI. 1 and 2 also show that cash expenditures for nondrug purposes (54,255) slightly exceeded noncriminal cash income (\$3,691) (although including respondent payments in the cash income would equalize these two figures. Thus, these subject received approximately \$4,000 in cash income from noncriminal sources and expended almost \$4,000 for nondrug purposes. This

# Total Income and Expenditures -159-

The last line of Table XI. 1 showed the "available criminal income" (sum of criminal cash income plus drug payments plus avoided expenditures for drugs and alcohol/cigarettes). In a similar fashion, the last line of Table XI.2 showed the dollar amount of substances which these respondents used during the Comparing these two lines showed that income year (from Table V. 5). available for supporting drug use was generally about \$1,000/year more than the actual dollar value consumed by each heroin user type.

Thus, at the aggregate level, their noncriminal income about equaled their nondrug expenditures. Likewise, their criminal income (including drug payments and avoided expenditures for drugs and alcohol) was just about equal to their actual annual consumption of drugs.

## Summary

This chapter showed that daily heroin users had a total income which was twice as high as that of the irregular heroin users. The greater total income of the daily heroin users was entirely due to greater cash income from crime, greater income from drug payments, and from avoided expenditures.

While daily heroin users had significantly higher cash expenditures than less regular heroin users; this difference was due entirely to the amount spent upon heroin. The cash expenditure for all other drugs and for nondrug purposes was equal among all three heroin user types. Thus, the cash income and cash expenditures closely balance each other. The daily heroin users greater criminal income was the main way in which their higher purchases of heroin was financed.

Moreover, within each heroin user type, the dollar amounts of criminal income (nondrug, drug, and avoided expenditures) equaled the dollar amount of drugs consumed. At the aggregate level, it seemed that criminal income was Likewise, noncriminal income almost matched mainly expended upon drugs. nondrug expenditures.

In the following chapter, we explore the economic values for society associated with these street opiate user's lifestyle.

Type of Income (Number of Subject A. CASH INCOME FROM

<sup>a</sup>Any Nondrug Crime

Robbery Burglary Theftb Other Property Prostituting &

CAny Drug Business: (Drug Sales + STC Other Drug Bus

-----

B. CASH INCOME FROM NONCRIMINAL SOURC

Work Public Support<sup>e</sup> All Other Legal Sc TOTAL CASH INCOME (

C. CTOTAL DRUG PAYME Drug Sales+STC \_\_\_\_

D. AVOIDED EXPENDITU

CDrugs CAlcohol & Cigare dMeals-Shelter dTransport & Other **9Shoplifting** (Own

TOTAL INCOME (A+B+C

Available Criminal

a-Table VI. 7C. <sup>b</sup> Theft included shoplifting (resale) + other larcenies; other property crimes includes forgery + con games + other illegal acts. C- Source: Table VII. 3. d- Source: Table IX. 3 e- Included welfare + unemployment + other public support (food stamps mainly). f- Included cash income from family, spouse/paramour, frineds, panhanding. gambling, and other sources; respondent payments were not included here since subjects would not receive them in normal circumstances. g- Source: Table VII. 7C; although shoplifting for own use was a crime, no dollars were obtained. Thus, when stolen goods were used by the subject. this included above as an avoided expenditure h- Sum of Line A + C + Avoided expenditures for Drugs + Alcohol/Cigarettes. BDJ;8260A;0744A;bj

Total Income and Expenditures -160-

Table XI. 1 - Total Income per Year of Street Opiate Users by Heroin User Typology

Heroin User Typology								
te)	Irregular (61)	Regular	Daily	Total (201)				
CRIME	4,451	7,121	11,292	7,597				
	2,885	5,719	8,540	5,729				
Crimes <sup>b</sup> Pimping	158 256 1,029 663 780	377 1,174 2,072 1,260 836	906 2,906 3,152 705 873	474 1,429 2,089 907 830				
C +	1,566	1,402	2,752	1,868				
iness)								
ESq	3,818	3,796	3,440	3,691				
ourcesf	1,546 981 1,292	815 1,603 1,378	852 1,084 1,504	1,048 1,254 1,390				
A+B )	8,269	10,968	14,732	11,289				
NTS: +Drug The	605 efts	1,976	4,016	2,189				
JRES	2,484	3 ,328	4,857	3,539				
ttes r Use)	752 216 1,237 164 97	1,663 243 1,136 205 81	3,168 235 1,254 143 57	1,851 232 1,203 174 79				
+D )	11,358	16,221	23,605	17,017				
Incomeh	6,016	11,054	18,711	11,869				

## Total Income and Expenditures -161-

## Table VI. 2 - Total Cash Expenditures Per Year for Drugs and Nondrug Purposes by Heroin User Typology

	Heroin (	Jser Typol		
Type of Income (Number of Subjects)	Irregular (61)	Regular (78)	Daily (62)	Total (201)
A. CASH EXPENDITURES FOR DRUGSa (i.e. Purchases)	4,301	6,635	10,930	7,250
Heroin	986	4,019	7,601	4,203
Cocaine	2,010	1,390	2,068	1,787
Alcohol	321	343	579	409
Illicit Methadone	338	375	134	289
Marijuana	316	193	219	238
All Other Drugs <sup>b</sup>	90	70	38	66
Cigarettes <sup>c</sup>	240	245	291	258
B. CASH EXPENDITURES FOR NONDRUG PURPOSESd	4,635	4,344	3,777	4,255
Shelter	641	886	420	668
Food	1,099	1,038	1,183	1,101
All Others Purposes <sup>e</sup>	2,894	2,420	2,177	2,486
TOTAL CASH EXPENDITURES (A+B)	8,936	10,979	14,707	11,505
Amount of Drugs Usedf	5,186	9,847	17,283	10,726

a-Source: Table V. 5.

b All other drugs included amphetamines, barbiturates, tranquilizers, psychedelics, other opiates (not heroin or methadone), and other drugs.

С Source: Table X. 3; note that cigarettes are not included in data regarding drug purchases (Table V. 5) from which the remainder of the drug expenditure data are taken; thus, the cash expenditures for drugs are about \$250 higher than those given in Table V. 5.

d-Source: Table X, 3.

e All other purposes includes cash expenditures upon family, clothes, transportation, savings, recreation, gambling, legal fees, other nonsubstance expenditures.

f Source Table V.5A: The dollar value of drugs consumed per year. this question has been addressed in a number of ways, this chapter will provide new information about those economic values which have been most problematic in prior research -- those associated with the criminal income and drug distribution by these respondents. Most of the previous research in this area has dealt with "social costs" of drug abusers. Most such research has made estimates of several different components (i.e. police, judicial, corrections, treatment, forgone productivity, private security costs, etc.) of these social costs. The biggest component of many social costs estimates, however, depended upon a formula that was both controversial and may contain assumptions which may not be accurate. For example, DSAS (1983) estimated the dollar value of thefts alone committed by an active heroin user in New York to be \$26,800 per year (an amount remarkably close to our estimates given below). The formula for computing this amount assumed that an addict used heroin on 255 days per year costing \$70 per day of active use, acquiring half of his/her money through theft, and that the fence factor was three. Such estimates were then multiplied by the estimated number of addicts to determine the social cost of heroin user theft.

Many economists (see Research Triangle Institute, 1976) hold, however, that the merchandise stolen by heroin users does not constitute a "social cost." That is, the economic value of an stolen article did not change or vanish when the thief forcibly transferred it from (i.e. stole from) one person (the victim) to another person (the buyer) (Casey and Preble, 1974). Most ordinary citizens and government officials, however, were very concerned about the economic losses suffered by victims, and were unwilling to accept the premise that the society did not suffer by this transfer. Thus, this chapter will specifically avoid reference to the term "social costs."

## Economic Values -162-CHAPTER XII

## ECONOMIC VALUES AMONG STREET OPIATE USERS

What overall effect did these street opiate user have upon society? While

Economic Values -163-

Rather the term, "economic values" will be employed as a more value neutral concept in order to assign a dollar value to their behaviors without making a judgment as to whether a social cost occured. Economic values will be defined as the dollar amounts (or their best approximation) associated with the illegal behavior of our street opiate users regardless of whether individual victims or society has been affected. Implicit in the "economic value" concept, and the focus of this chapter, however, was the assumption that our street opiate users obtained some dollar amount (or economic value) because they engaged in an act which criminally victimized someone else, obtained goods or services without paying for them (i.e. avoided expenditures), or their behavior negatively affects the broader society by withholding tax revenue or enriching the illegal drug distribution system.

This chapter will not consider the many other components of social costs such as those associated with governmental efforts to prevent opiate user crime (prevention/and intervention programs and patrol), to arrest and process criminals (police, prosecutors, defense attorney, judges, other court personnel), to jail, imprison, or supervise those convicted (prison and jail officers and management, parole and probation officers), or to treat and intervene in their lives (drug and alcohol treatment, medical and hospital

In a similar fashion, the economic value of nongovernmental costs associated with preventing crime (private security guards, locks and secure doors, higher prices to cover theft losses, etc.) will <u>not</u> be included here. Lifewise, this chapter will not consider the noneconomic costs of fear of crime experienced by citizens, nor the abuse and neglect of family and children by these respondents, or other similar factors. Such social costs have been more carefully estimated by others and would be beyond the scope of our data (Research Triangle Institute, 1976; Moore, 1970; Casey and Preble, 1974; Hopkins, 1975; Gray, 1979; DSAS, 1983).

The second major problem with estimating theft losses associated with street opiate users lie in the assumptions behind the formula. First, the formula assumed that most active heroin users engaged in the taking of merchandise (i.e. theft, burglary, robbery) -- but other forms of criminality (drug business crime, con games, forgery, etc.) were ignored. Second, an arbitrary number (three) was assumed to be a standard "fence factor" for all types of theft, when the fence factor may be quite different for robbery, burglary, and larceny. Third, it assumed a high amount of heroin (\$70) consumed per day of use. Unlike previous estimates, however, this study can make new estimates Thus, we will construct careful estimates of economic values among our

about economic values by building from specific types of crimes, assigning different fence factors to the various crimes, and including economic values that have not been previously included in such social cost estimates. Rather than assuming a standard average daily heroin consumption, we can ignore the dollar amounts actually consumed and concentrate upon how income was obtained. street opiate users by including 33 different components, each of which has a differing set of assumptions, specified fence factors, and unique dollar amounts. Moreover, these estimates of economic values can be demonstrated for daily, regular, and irregular heroin users, as well as all subjects.

In this analysis, the economic values imposed by our street opiate users can be conveniently classified into two major components: Direct and Indirect economic values. These are briefly defined below. (Definitions and the rational for each component follow Table XII.1

## Economic Values -164-

# DIRECT ECONOMIC VALUES

Direct economic values were those monetary values which other persons/ organizations provided either voluntarily or involuntarily to street opiate users. Thus, this chapter will consider and compute the approximate economic value of losses experienced by crime victims (they have real economic losses) without considering the corresponding economic gain of those purchasing stolen goods. The major component of such direct economic values are the economic losses associated with the nondrug crimes of street opiate users.

Economic Values -165-

Moreover, street opiate users also obtained income (both as cash and avoided expenditures) from friends, family, and other sources who provide these on a "voluntary" basis. In return, these street opiate users also reciprocated by providing some cash (expenditures) or services to family, friends, and others which resulted in a net gain (see Chapter XI) to the opiate users. In this chapter, however, only the income received by opiate users from others were included below in estimates of direct social costs. These were not reduced by their economic contributions to others. INDIRECT ECONOMIC VALUES.

Indirect economic values were those associated with activities which were illegal by statute (mainly drug sales/STC, income tax evasion) or those which contributed to the economic vitality of the illicit drug distribution system. Such indirect economic values did not involve direct economic losses to victims or costs to specific persons. Rather, indirect economic values contributed substantially to the economic success of the illegal economy and revenue loss to the government. In some ways, indirect economic values measured the economic productivity of street opiate users to the illegal economy. Indirect economic values involved three analytically distinct measures: 1) the economic values of cash plus drug income from drug selling, steering, touting, coping, other drug business, and drug theft; 2) the absence of income tax paid on income (from both drug and nondrug crime as well as most work income); and 3) the "value added" to the illicit drug business by the labor of these street opiate users via their drug crimes.

Like retail salesmen who receive a commission proportional to the economic value "added" by their sales work, street opiate users received payments (in either drugs or cash) which were roughly proportional to the gross anount of drugs which their activity helped sell. That is, in addition to obtaining drug and cash income from drug sales or STC, the labor of these subjects contributed a substantial economic "value added" to maintaining the entire illegal drug distribution system which society desired to eliminate. The main findings about the social cast.

• •

The main findings about the social costs of street opiate users have been summarized in Graph XII. 1. Detailed data about each of the components included in these costs has been provided in Table XII. 1. The concluding section of this chapter explains each social cost component in Table XII. 1, and provides the rationale for multiplying each specific component by its own unique "fence factor." (see below).

# Economic Values -166-



a a

Graph XII. 1 (above the line) shows that the average street opiate user had direct economic values of almost \$19,500 per year. Over two-thirds (\$13,952) of this cost was the dollar value of the goods obtained or cash raised from nondrug crime. Sixty percent (\$5,718 + \$3,775 + \$1,736 + \$569) of the direct economic values involved goods or money taken during burglary, shoplifting for resale, larcenies, and robbery. Only \$2,645 of the direct economic values\* came as cash income from noncriminal sources. An equivalent value was contributed by others to street opiate users in the form of avoided expenditures for shelter, meals, transportation, drugs, and alcohol or cigarettes.

Moreover, direct economic values among daily heroin users were three times greater than among irregular heroin users (\$29,587 vs. 9,645). This was due to the fact that the nondrug crimes committed by daily heroin users (\$22,844) have four times more economic impact than those by the irregular heroin users (\$5,592) and about twice as much impact as the regular heroin users (\$13,422). This difference was also due to the fact that daily heroin users were able to obtain more economic value from avoided expenditures, mainly as drugs.

The indirect social costs imposed by the average street opiate user was also substantial (\$14,441). About 28% of this cost (\$4,057) was derived from drug or cash income from the respondent's drug distribution activity, and 60% (\$8,764) was the value added to the illicit drug distribution system by the respondent's labor in that market. Income tax evasion (\$1,620) provided a relatively modest contribution to the indirect social costs. The daily heroin users have an indirect social cost (\$25,423) that was about five times greater than irregular heroin users (\$5,100) and almost twice as high as the regular heroin users (\$13,205).

## Economic Values -168-FINDINGS

The economic value of legal employment was not included as an economic value contributing negatively to society.

# Economic Values -169-

When both the direct and indirect costs were summed, the total economic values associated with an average street opiate user was almost \$34,000 per year (assuming no time off the street). Moreover, daily heroin users imposed economic values of \$55,000 per year. This was a substantial amount of economic value to support the consumption of approximately \$13,000 worth of heroin and \$4,000 worth of other drugs (mainly cocaine). Although the social cost was substantially less (\$14,745) among the irregular heroin users, this was still significant to support \$5,000 worth of drug use.

5

34

Moreover, none of the components included in these economic values accounted for the failure of almost all subjects to contribute positively to the national economy in a conventional way. Most of these respondents are males or females in their 30s, out of school for many years, and are in a period of life which was generally productive for their blue and white collar counterparts of the same age. Assuming that the average income of the lowest paid service worker was \$7,280 (minimum wage), these subjects earned only about \$1,100 in work related income. In short, the average street opiate user imposed about \$6,000 per year in foregone legitimate productivity, plus any economic value which such labor may add ("value added") to the national economy of a legitimate nature.

The DSAS (1983) estimate of \$26,800 annually due to theft by active heroin users seemed about right. Our daily heroin users had direct economic values of almost \$30,000 while the regular heroin users had about \$19,000.

The DSAS (1983) estimate was based upon some correct and less accurate assumptions. They assumed that subjects obtained about half their income for heroin from theft; the percentage was 60% among our street opiate users -a remarkably accurate estimate of total direct economic values from theft, burglary, and robbery. Likewise, they assumed a fence factor of three, while these four crimes overall had a combined fence factor of 2.4 (13,952/ 5,729) -- see Table VI. 7C). They assumed \$70 per day of heroin use versus about \$35 per heroin-use day among our subjects -- Chapter IV).

ويترجعه والمربة وبالمستحاف والتخاط

Economic Values -170-On the other hand, the DSAS estimates may be misleadingly low as well because they do not include economic values associated with drug business crimes, and especially the value added to the illicit drug distribution When these were included, the average street opiate user had system. almost \$34,000 in economic values, about \$7,000 higher than the DSAS estimate. Among our daily heroin users, their economic values were twice (\$55,000) that given by DSAS (1983). Summary

-..

• •

The data presented in this chapter clearly showed that these street opiate users had substantial economic values. These figures were very similar to or somewhat higher than those calculated by the National Institute on Drug Abuse (Pollin, 1981; Research Triangle Institute, 1976), and by the New York State Division of Substance Abuse Service (1983). Moreover, the data developed here had clear definitions and specified assumptions about 33 different possible economic values which were related to the frequency of heroin use.

These respondents had direct economic values from nondrug crime of almost \$14,000 per year, and another \$5,000 in social costs to family, friends, government transfer payments, and other drug users who also provide them with cash or help them avoid many important expenditures.

Moreover, the daily heroin users had the highest direct economic values especially for nondrug crime and drug crime-related values. The economic values due to the four major crimes (burglary, shoplifting for resale, larceny, and robbery) accounted for much of their criminal cash income and high economic values. Daily heroin were substantially more active in obtaining drug plus cash income from their involvements in drug sales and STC and responsible for much of the "value added" to the drug business.

These social costs did not include many other ways in which they impose further costs upon society such as their nonparticipation in the legitimate economic system, and the costs of incarcerating, supervising, and treating street opiate users which were included in previous estimates of social costs.

. .

## Economic Values -171 -

Table XII. 1 -- Direct and Indirect Economic Values per Year among Street Opiate Users by the Heroin User Typology.

		Hei	roin Use	er Typol	ogy		Her	oin User	Typolog	y	A North Market	
		irre	g. Reg.	. Daily	Total	M137+4	Irreg.	Reg.	Daily	Total		
		I	ncome Pe	er Year		pliera	Econo	mic Valu	es Per Y	par		
A.	CASH VALUE OF					·	DIRECT	SOCIAL	COSTS			
• ••	NONDRUG CRIME						5 502	12 400	22 044	12 073		
٦	Dabbarra			*			J-J-J-J-Z	13,422	22,844	13,973		
2.	KODDery Burglary	158	377	906	474	x1.2	190	453	1,087	i 569 -		
3.	Shoplift (Resale)	887	1,342	2,900	1,429	x4.0	1,023	4,694	11,624	5,718/		
4.	Other Larcenies	142	730	818	579	x2.5	426	3,350	5,834	3,//5.		
5.	Forgery	56	300	93	162	x1.0	56	300	2,400	162		
7.	Prostitution	410	408	400	406	x1.0	410	408	400	406		
8.	Pimping	41	114	100	87	x1.0	/39	722	773	743 -		
9.	Other Illegal Acts	197	552	212	339	x2.0	394	1.104	425	87 ° 598 :		
10	.Snopinft(Own Use)*	97	81	57	79	x1.0	97	81	57	79		
Β.	CASH INCOME FROM										8	
	NONCRIMINAL SOURC	ES					2.272	2 981	2 581	2 645		
11	Working 1	EAC	010				- <b>j</b> -/ -	4 <b>,</b> 501	100ء 2	2,040		
12	. Welfare	,240 584	815	852 757	1,048	x0	0	0	0	0		
13.	. Unemployment	153	60	71	91	x1.0 x1.0	584 153	1,177	757	868	254	
14,	. Other Public \$	244	366	256	295	x1.0	244	366	256	295		
15.	family Spouse/Paramour	303	235	251	260	x1.0	303	235	251	260	α. 	
17.	Friends	76	402 167	4//	461	x1.0	443	462	477	461		
18.	Panhandling	11	59	50	41	x1.0	/0	10/	118	125()	390	
19.	Gambling	160	95	301	178	x1.0	160	95	301	178		
20.	Pavments**	733	760	770	765		•	,				
21.	Other	299	360	307	/55	XU VI 0	200	0	0	0	141 (28	
	"我会我们的我们的我你能能的你们吗?					V . I . 	433 	300	307	325		
6.	AVOIDED											
	LAFENDITURES						1,781	2,729	4,162	2,883	-	
22.	Room (Shelter)*	588	518	639	577	x1.0	588	51.0	620			
23.	Meals*	649	618	615	626	x1.0	649	618	615	5//		
24.	Other	39	36	53	42	x1.0	39	36	53	42 13	71	
26.	Drugs	752	1.663	3.168	1 851	x1.0	125	169	91	132		
27.	Alcohol/Cigarette	216	243	235	232	x1.0	216	1,003	3,168	1,851 🦛	10 H	•
ວ	DIRFCT FCONOMIC				***			A. 1. A	200 			
	VALUES/PERSON						0 EAC	10.100			5 2 P	
	(A + B + C)						y,045	19,132	29,587	19,501		
-	a - Multiplier -	A + 20	+08 H.C.	1 to -	1 4 4 4 1 4						the second second	

Multiplier - A factor used to multiply the values presented in each line. The line number corresponds to the text's discussion of the source of these figures and the rationale for each multiplier.

T

E. DRUG + CASH INCOME

28. Drug Sales 1,2 29. Steer, Tout, Cop 7

- 30. Other Drug Business
- 31. Any Drug Thefts
- F. INCOME TAX EVASION\*\*
- G. "VALUE ADDED" by Subject Drug Distribution Activi
- 32. Drug Sales\*
- 33. Steer, Tout, Cop\* 2,6
- H. INDIRECT ECONOMIC
- VALUES/PERSON (E+F+
- TOTAL (DIRECT + IND
- ECONOMIC COST PER PI (D + H)

8273A

1 28 6

# Economic Values - 172-

Table XII. 1 -- (Continued) Direct and Indirect Economic Values per Year among Street Opiate Users by the Heroin User Typology.

Her	oin Use	er Typol	)gy		Her	<u>oin User</u>	Typolog	IJ
rreg	. Reg.	Daily	Total		Irreg.	Reg.	Daily	 Total
				Multi-				
In	come Pe	r Year		plier <sup>a</sup>	Econo	mic Value	es Per Y	<u>ear</u>
					2,170	3,378	6,768	4,057
259 754	957 1,972	2,036 4,102	1,381 2,260	x1.0 x1.0	1 <b>,25</b> 9 754	957 1,972	2,036 4,102	1,381 2,260
88 70	64 385	110 519	86 331	1.0 1.0	88 70	64 385	110 519	86 331
r				- <b>u</b>	600	1,430	2,934	1,620
					2,330	8,265	15,721	8,764
ty 02 10	3,669 6,663	4,176 15,476	2,804 8,151	x0.8 x0.8	242 2 ,088	2935 5,330	3,340 12,381	2,243 6,521
G)	*====				5,100	13,073	25,423	14,441
IREC ERSO	T N				14,745	32,205	55,010	33,941
								1. S

\* - Value of drugs distributed minus respondent's actual drug + cash income. \*\*- Total taxable income includes nondrug cash crime income (lines 1-10) plus work

income (line 11), plus drug and cash income from the drug business (lines 28-31). The standard deduction (\$1,000) is subtracted and the federal income tax for 1982 was used to compute the amounts given here. a - Multiplier - A factor used to multiply the values presented in each line. The

line number corresponds to the text's discussion of the source of these figures and the rationale for each multiplier.

Source: "Income per Year" comes from Tables VI.7C, VII.3, and IX. 3. The "Economic values" come from the income per year times the multiplier.
### Economic Values -173-DETAILED RATIONALES AND MULTIPLIERS IN ESTIMATING ECONOMIC VALUES AMONG STREET OPIATE USERS

This section provides detailed information about the data included in Table XII. 1. Specifically, each line will be mentioned in order, and a rationale for the calculation given. This table included, for nondrug crimes, a best estimate of the "fence factor" for specific crimes -- that is how much more than the respondent's cash income the stolen merchandise was worth at retail value. This fence factor has been expressed in Table XII. 1 as a "multiplier." When a multiplication factor other than one (1) is employed, a rationale has been presented.

# A. CASH VALUE OF CRIME (all data on left side come from Table VI. 7C)

- 1. Robbery was the most serious of the crimes included in this study. Most robberies were brief crimes, the perpetrator threatened or otherwise obtained the victim's wallet or purse. Money (cash income), in the experience of our field staff, was the only thing taken in about 75% of the robberies. Other goods (watches, clothing, etc.) taken in robbery have little street resale value. On the other hand, jewelry and chains obtained in robbery probably gained the robber a cash amount of approximately a third of its value. Credit cards sold on the street for about \$50 each, identification documents (driver's license, etc.) were also worth about \$25 each. Overall, given that products other than money were estimated to occur in only about 25% of the robberies, a multiplication factor of 1.2 was applied to respondent's cash income from robbery.
- 2. Burglary was one of the most economically rewarding crimes for street opiate users, and having the highest economic values because stolen property/goods must be resold ("fenced"), generally for significantly less than their actual worth. While cash may be taken as well, most burglaries involved property removal of a variety of items whose actual retail value were unknown and difficult to ascertain.

There were four major ways in which street opiate users sold stolen merchandise: to community residents approached on the street or bars ("street fencing"), to persons who specialized in paying cash for stolen goods ("professional fences"), and to drug dealers who accepted merchandise for drugs ("dealer fences"). Some merchandise (gold, jewelry, coin collections, antiques, etc.) may be sold to legitimate merchants who did not realize (many do) they were buying stolen goods. Street fencing of stolen goods brought a cash return of approximately a third of the actual value, but obtaining this return, expecially for an expensive item, involved several hours of locating a buyer with enough cash. On the other hand, professional fences provided instant cash for merchandise, but seldom better than a fifth of the fair retail value of merchandise and for as little as a twentieth of the retail value.

· · ·

- multiplier of one was used.

- seemed most appropriate.

### Economic Values -174-

Likewise, dealer-fences will seldom provide an amount of drugs equivalent to a third of the retail value. In addition, the drugs provided have been marked up twice to three times, so the dealer-fence's actual cash expenditures was probably a sixth to a tenth of the retail value of the stolen merchandise. On the other hand, gold chains/jewelry, coin collections, silverware, and antiques, could be sold to legitimate or quasi-legitimate business for their approximate retail value (minus the dealer's mark up).

Our field staff's impression was that the most frequent burglars were most likely to sell to professional fences, unless they could street fence some goods rapidly, or go to a legitimate dealer. Overall, across all burglars and types of merchandise, a conservative estimate would be that the street opiate user obtained a fourth (or less) of the value of the merchandise stolen. Thus, a multiplier of four (4) was used.

3. Shoplifting for resale -- was the most common nondrug crimes, but not always the most rewarding. Shoplifting typically involved two different items which were commonly stolen: meat and clothing. Meat, cigarettes, and alcohol were typically stolen from stores with retail prices marked; they were sold to community members for half the retail price marked. New clothing with price tag attached could be street fenced for a third of the value. Clothing typically had a higher economic value per unit than meat. Almost all other shoplifted merchandise was sold for a third or less of the retail value. Systematic shoplifters, however, were more likely to use professional fences or dealer-fences where the economic returns were less but cash or drugs were provided directly. Staff estimated that approximately half of the shoplifting events

involved the theft of meats or cigarettes (fence factor of 2), and half involved clothing or other merchandise (fence factor of 3 or more). A conservative multiplier of 2.5 was used.

4. Other larcenies -- Thefts of bikes, parts of autos or contents from autos, office thefts, etc., were generally fenced for a third or less of their actual value, with much variation by kind of item and the approximate age of the goods. A multiplier of 3 was employed.

5. Forgery - involved obtaining cash by using false signatures, so a multiplier of one was appropriate.

6. Con Games (Fraud) - generally obtained cash. The main offense by these street opiate users was three-card monte, which obtained cash. A

7. Prostitution - involved sex for money, cash was almost always obtained. A multiplier of one was appropriate.

8. Pimping - involved obtaining money from prostitutes or women. A multiplier of one was appropriate.

9. Other illegal acts - included fencing of goods (which the respondent did not steal), helping criminals find victims, etc. A fence factor of 2

10. Shoplifting for own use - was stealing a commodity and keeping it. Where the value of the item was given, this was counted as the dollar value of merchandise consumed. A multiplier of one was appropriate.

### Economic Values -175-

- B. CASH INCOME FROM NONCRIMINAL SOURCES (all data on left side from Table IX. 3)
- 11. Working was employment in jobs which were productive in the conventional economic world and constituted a positive contribution which was not harmful to society. Thus, a multiplier of zero (0) was used.
- 12. Family, Spouse, Friends, Welfare, Unemployment, Other Transfers (mainly
- food -stamps), Panhandling, Gambling, and Other Cash Income were sources 21 of cash income, so a multiplier of I was appropriate.
- 20. Respondent payments were cash income drived from participating in this study (remember respondents did not obtain such payment for a full year, only a few weeks -- the figures here were annualized). Since subjects would not normally have obtained this income, the multiplier of zero (0)
- C. AVOIDED EXPENDITURES (all data on left side from Table IX. 3)
- 22. Room (shelter) were lodging expenses for which the respondent did not have to pay. While providing a bed or couch for the respondent may be a burden on some household, it generally did not involve additional cash expenditures by the apartment owner. A multiplier of zero (0) was used.
- 23. Meals were food which the respondent consumed, but for which he typically did not pay cash on that day. The person (s) providing such food, however, had to pay cash for the food consumed. The data assumed that \$3 per day of food was eaten by the respondent. A multiplier of one (1) was used.
- 24. Transportation involved not paying fares by jumping subway turnstiles or sneaking into the back of buses. Occasionally, subjects also jumped out of a taxi without paying. Such theft of services involved provision of services with a real economic value. A multiplier of one (1) was used.
- 25. Other (nondrug) avoided expenditures involved respondents obtaining clothing, presents, newspapers, etc., for which someone else paid. A multiplier of one (1) was used.
- 26. Drugs were frequently obtained without cash expenditures by "copping short," as gifts, and part of reciprocal obligations to share drugs. The value of such drugs obtained without cash payment by the subject actually costs someone else real economic amounts. A multiplier of one (1) was
- 27. Alcohol and cigarettes were frequently obtained by respondents at someone else's expense. This meant that someone paid for these substances. A multiplier of one (1) was used.
- E. DRUG PLUS CASH INCOME FROM DRUG DISTRIBUTION (data on left taken from

These street opiate users devoted considerable energy to working in the illegal drug distribution system, generally at the lowest levels. They were working mainly to earn the drugs they wished to use and earned both cash income and drug income. While no single other person or "victim" was directly paying for such costs, the value of this illegal labor was certainly an indirect economic value by street opiate users.



28. Drug Sales (Dealing) - involved the direct transfer of money for illegal drugs. The cash plus drug income received by the subject were a direct payment for the respondent's labor. A multiplier of one (1) was used.

- was used.

- F. INCOME TAX EVASION

Respondents seldom obtained jobs from which income tax was routinely withheld. Rather, they obtained odd jobs on an occasional day for which they were typically paid in cash. Almost all cash from jobs was "off the books." None of the cash income from crime, friends, or family, was reported by these subjects. Most of the respondents have never filled out an income tax form. Except for a few dealers, almost none of these street opiate users worried about income tax evasion. Their total cash plus drug income has been calculated (see footnote), from which the standard deduction (\$1,000) has been subtracted. The remainder was considered as taxable income; the approprate 1982 tax on that amount was entered in the social cost columns.

3 and 4)

When a person has been employed in the regular job market, his/her labor produces materials, goods, and information which typically generated two to three dollars for every dollar which the person received as payment for labor. The economic value of their output minus the labor costs may be considered as the "value added" by such labor.

In the same way, the labor which street opiate users devoted to drug sales and steering, touting, and copping contributed a "value added" to the economic worth of illegal drug distribution system, which was in addition to the relatively direct payments (drugs + cash) they received for their time and work. Since society defined such transactions as illegal, and strictly penalized such behavior, the economic value added to the drug distribution system was included as an indirect social cost.

### Economic Values -176-

29. Steer, Tout, Cop - involved the respondent in helping one or more dealers sell their illegal drugs. They were generally paid in drugs (although sometimes in cash) for their assistance in finding customers. Such drug or cash income were payments for illegal labor. A multiplier of one (1)

30. Other Drug Business - involved the respondent in various other drug roles (i.e. lending works, running a shooting gallery, holding or transporting drugs, testing drugs, etc.) for which he was paid in cash or drugs. A multiplier of one (1) was used.

31. Drug Thefts - involved the respondent in obtaining drugs via robbery, burglary, or theft from other drug users or dealers. The drugs obtained had a real economic value to those from whom they were taken, equivalent to the amount taken. A multiplier of one (1) was used.

G. "VALUE ADDED" TO THE DRUG DISTRIBUTION SYSTEM (data taken from Tables VII.

3] Drug Sales and Steer, Tout, Cop: The potential maximum of this "value - added" was the total dollar value of drugs distributed minus the cash + 32 drug income received as payments for their labor. This calculation was given on the left side of Table XII. 1.

The potential for double counting of money or economic value. however, was high. The money which paid for the drugs sold by respondent A may already have been counted as criminal income by respondent B. Or at least, the same kinds of persons who spent criminal income (already computed above) to purchase heroin (for example) were part of the value added by the labor of other street opiate users who dealt on street during

# Economic Values -177-

Yet the probability of such double counting of the same dollar (due to exchanges between two respondents) were considered by staff to be relatively low; such double counting may have occured on less than two transactions out of 10. Thus, a multiplier of 0.8 was used below. The "value added" was calculated as 80% of the difference between the dollar amount of drugs distributed minus direct payments for labor in the drug

The above information provided the rationale for including certain components having economic value and the reasons for using specific multipliers with specific components. The main components of economic value were combined in logical ways to provide the data generated in Graph XII. ] and Table XII. 1.

also commit nondrug crimes.

A study of career criminals admitted to prisons and jails in California, Michigan, and Texas has recently been completed. Chaiken and Chaiken (1982ab, Chaiken, 1983) developed a new way to classify criminal offenders which were called "criminal varieties." By using information about self-reported criminal behavior indicated on a questionnaire, Chaiken and Chaiken (1982a) showed that "criminals can be categorized according to the combinations of crimes they commit" and that those:

their personal characteristics.

BDJ; 8273A;0745A;bj BDJ; 8227A; bj

# Criminal Offenders, Heroin Users, and Criminality -178-CHAPTER XIII

CRIMINAL OFFENDERS, HEROIN USERS, AND CRIME RATES

The search for a criminal typology to describe the offending behavior of criminals has frequently resulted in considerable frustration. Most typologies have classified subjects according to the maximum offense for which they have been arrested or convicted. Relatively little evidence of criminal specialization is exhibited in the arrest and self-reported patterns of

criminal behavior. That is, most persons who rob are also thieves and commit many more thefts than robberies; many burglars also rob; many drug sellers

who committed specific combinations of crimes were distinguishable ... by their crime commission rates, their persistence in committing crimes, and

One important result of the study is our ability to identify and characterize the most serious category of offenders. Criminals in this category reported committing robbery, assault, and drug deals during the one- to two-year measurement.... We found that these criminals, whom we call 'violent predators,' usually committed the defining crimes at high rates, and they often committed burglaries, thefts, and other property crimes at high rates too -- sometimes at higher rates than any other type of criminal, including those who specialized in those crimes.

... Ten types of offenders [were] defined in terms of the crimes they do or do not report committing. The offender types are arranged hierarchically, with the lower ones relatively less serious than the higher ones. ... offenders in the lower categories not only committed fewer serious crimes at lower rates, but their patterns of employment, drug use, and juvenile behavior were more socially acceptable than those of other [more serious] offenders. (Chaiken and Chaiken, 1982a, 2-3).

Criminal Offenders, Heroin Users, and Criminality -179-

A critical crime in the Chaiken's classification system was drug sales. Among persons reporting major offenses like robbery and burglary, those who also sold drugs had generally higher rates of offending behavior for several crime types than their counterparts who did not sell drugs. Among persons who robbed, Chaiken and Chaiken (1982a) showed that those who engaged in assault were more seriously involved in most crime types than those without assault.

The current chapter compares findings from our New York subjects with the Chaiken and Chaiken (1982b) findings and addresses the following questions:

- 1) Can respondents who were recruited as street opiate users in the Harlem and East Harlem section of New York City be meaningfully classified into criminal types that were similar to Chaiken and Chaiken's "criminal
- 2) How similar were criminal patterns among New York opiate users to the career criminals studied by Chaiken and Chaiken? That is, are their findings replicated among New York heroin abusers? In what way does the current data extend their findings.
- 3) How were heroin user types related to criminal offender types?
- 4) Within each criminal offender type, were increasing levels of heroin consumption related to higher crime rates and criminal incomes, and if
- 5) Do certain types of criminal offenders commit a disproportionate share



Criminal Offenders, Heroin Users, and Criminality -180-

idents	who v	were r	ecrui	ted	as	stre	et d	opiate	user	s in	the	Harlom
larlem	sect.	ion of	New	York	Ci	ty h	0 mc	aning	full.			nuriem
tynes	that	10000	cimi	1011		Cy Di		anny	Jully	Clas	<u>S1T1</u>	ea into
MPC3	cnac	were	51011	Idr	τ0	Cna	iken	and	Chai	ken's	"C1	riminal
-												

The present study clearly differed in significant respects from the Chaiken's study of career criminals. Specifically, our New York respondents were recruited because they had a history of heroin abuse and injection -drug abusers who were never heroin abusers were systematically excluded, this included heavy cocaine users and pill abusers, the more numerous marijuana users, and nondrug using criminals.

Moreover, all New York respondents were recruited and interviewed while at liberty on the streets (although a few had short term jail stays between their interviews). All of Chaiken and Chaiken's respondents were new inmates at prisons or jails when participating -- although they responded to questions about their criminal behavior when they had been at liberty on the street.

New York subjects were interviewed about their criminal behavior and drug use on a day-by-day basis (although the actual interview was conducted once per week). The data collected cover only the days about which they were interviewed. Between 33 days (a majority of subjects) to over 100 days (a few subjects) of data are available for analysis. Chaiken and Chaiken's subjects completed a self-administered questionnaire and answered questions about the frequency of committing specific types of crime during the last two years while they were at liberty on the street; hence they had a longer opportunity to commit the focal offenses. Moreover, since incarcerated inmates were generally the most seriously involved criminals, the percentage of subjects classified into the most serious criminal types should be somewhat larger than among the New York opiate users at liberty on the streets.

Criminal Offenders, Heroin Users, and Criminality -181-

The original objectives of the Economic Behavior project did not include a focus upon crimes which did not have economic value. Thus, the interview schedule did not systematically question respondents about whether they had engaged in assault or other noneconomic crimes (property destruction, arson, rape, etc.). The Chaiken and Chaiken questionnaire, however, included questions about assaultive crimes. On the other hand, the Chaikens asked only about "drug selling." Subjects were not asked about what this project calls "steering, touting, copping." Inmates who who engaged in such behaviors probably did not include such STC activities when asked about "drug selling or dealing." Because the Chaiken and Chaiken study involved only males, questions about prostitution/pimping were not asked. This study included 50 female subjects, many of whom reported activity in prostitution and a few male subjects reported pimping activity. Thus, separate information about these crimes have been reported below.

Thus, there were many differences in methodology (interviews vs. questionnaire), subjects (imprisoned male criminals vs. heroin abusers of both sexes at liberty on the street), and locale of research (a small section of New York City vs. jail and prison inmates in Michigan, California and Texas). Such differences would suggest that divergent findings might emerge.

Such was not the case, however, as the following discussion shows. Little difficulty was encountered in constructing the criminal offender types according to the Chaiken and Chaiken criteria. One minor change in definition was made in classifying the street opiate users. The majority of subjects reported about their behavior for only 33 days, while other subjects had 60 to 100 or more days of reporting. The latter subjects had more opportunity to commit a crime, like robbery, that occurred with a low frequency. In order to ensure that most subjects had a relatively equal chance of being classified as "active" in various crime types, the percentage of person-days commiting each specific offense was computed.

Table IX. 1 presents that this study's version of the Chaiken and Chaiken (1982a) criminal varieties (or Hierarchical Subgroups of Offenders). Since information about assault was not obtained from these street opiate users, the Chaiken and Chaiken categories with assault distinctions have been ignored. Thus, the Chaiken and Chaiken violent predator was included with the "robber-dealer" category while the "robber-assaulter" and "mere assaulter" types were included among "low level robber" category in Table IX. 1. In addition, two categories of lower level offenders were included here but not in the Chaiken and Chaiken (1982a) types. "Low level distributors" do not engage in robbery, burglary, property crimes, or drug sales, but do report steering, touting, and copping. Respondents who do not fit in the above crime types are included in the "none of above" category; almost all were prostitutes or pimps.

Table IX. 1 presents what will be called the Criminal Offender Typology which shows the definition for each type and the percentage of 201 respondents classified in each type. When classified according to the Chaiken and Chaiken classification criteria, these street opiate users were quite well distributed across the crime types. For comparison purposes, the distribution of the Chaiken and Chaiken sample of career criminals was also presented. A lower percentage of the New York street opiate abusers were classified in the two most serious categories than the Chaiken and Chaiken career criminals.\* Nevertheless, the hierarchical classification of respondents based upon combinations of offenses was useful and meaningful.

\*This is probably due to the incarceration of the most serious offenders in prison. The Chaiken and Chaiken (1982, 31) study also shows the percentage of respondents classified in the "robber-dealer" and "violent predator" categories was 21% in Michigan, 18% in Texas -- figures somewhat more similar to the 11% of the New York street opiate users, but considerably lower than in among California (33%) inmates.

Criminal Offenders, Heroin Users, and Criminality -182-Subjects who committed offenses on 2% or more of their person-days were classified as "active" in each crime type for the purposes of reproducing the Chaiken and Chaiken Criminal Variety Types.

Crimin - Table XIII. 1	al Offe - Defin So Cl	nders, nitions lassifi	Heroîn Users, a of the CRIMIN ed	and Crimin AL OFFENDE	ality - ER TYPOLI	183- DGY and	Percenta	ge of	Sample	
Respondent Criminal Offender	<b>.</b> .	Offen	se Types Includ Property Offens	led in Defi ea	inition o	or Not				
Typology	Rob- bery	Burg- lary	Theft/Forgery Cons/Other	Pimp/Pro- stitute	· Drug Sales	Drugs STC*	Percent Sample	Perce Male Caree	era	
Robber-Dealer	Yes	?	?	?	Yes	?	11	24	IId LS	
Low Level Robber	Yes	?	?	?	No	?	13	20		
Burglar-Dealer	No	Yes	?	?	Yes	?	8	10	- 122G	
Low Level Burglar	No	Yes	?	?	No	?	12	10	an a control (	
Theft-Dealers	No	No	Yes	?	Yes	,	12	0		٩
Low Level Thieves	No	No	Yes	?	No	•	10	6	and second and	
Drug Dealers	No	No	No	?	Yee	: 2	20	8		
Low Level Distributors	No	No	No	?	No	r Yes	6 8	6	н 	
None of Above	No	No	No	?	No	No	4			
Total							(Otl	lers) 18		
10041							100	00	e na secondada	
es - Group member co	ommits t	his cri	ime by 2.00% or	more of p	erson-da	ivs by	of initia	-		

not commit this crime on more than 1.99% of the person-days, by definition.

? - Group member may or may not commit this crime; most members do so.

a - Property Offenses include Shoplift (Resale), Other Larceny, Con Games, Forgery, and Other Crimes; does not include shoplift (own use).

\* - STC - Steering, Touting, Copping Drugs

b - Source: Chaiken and Chaiken (1982b, 27)

N.S.

How similar we career crimina findings replic the current da

2)

The central findings of the Chaiken and Chaiken (1982ab) report were replicated among the New York street opiate users in almost all essential respects. The data also extended the Chaiken and Chaiken (1982a, 1981b) findings in important ways by presenting previously unavailable information about the criminal income and about involvements in steering, touting, and copping.

Specifically, the data in Tables XIII. 2 - 5 show that respondents classified in the most serious categories of this hierarchy, when compared with subjects in the less serious categories:

- offenses. (Table XIII. 5).

The most serious category, the robber-dealers, committed robbery and drug sales at high rates. The robber-dealers also committed burglaries and STC at high rates too. The robber-dealers also had high incomes from robbery (\$2,433), drug sales (\$3,888), burglary (\$3,356), and STC (\$3,138). As a result, robber-dealers had the highest criminal income (\$16,666) of any criminal offender type. Specifically:

Criminal Offenders, Heroin Users, and Criminality -184-

re crimi	nal pati	erns a	imong l	New York	opiate	users	to the
<u>ls studi</u>	ed by C	haiken	and C	haiken?	That	is. are	their
cated am	iong New	York	heroin	abusers	? In	what wa	y does
ta exten	d their	finding	js?				<u> </u>

1. Exhibited greater breadth of criminal involvement (i.e., commit a larger number of different kinds of crime).

2. Were as likely or more likely to report involvement in any given specific offense. (Table XIII. 2).

3. Were apt to commit definitional crimes on a greater proportion of person-days, and to commit nondefinitional crimes on as many or more person-days (some exceptions exist, see below -- Table XIII. 3).

4. Committed as many or more crimes per year (Table XIII. 4).

5. Obtained a high annual cash income from specific offenses and from all

Criminal Offenders, Heroin Users, and Criminality -185-

- 1. Among the robber-dealers, 64% committed a burglary, 91% a property offense and 91% engaged in STC. These were among the highest percentage by (nondefinitional) offender types involved in these offenses. They were somewhat less involved in prostitution than other offender types (Table XIII. 2).
- 2. The robber-dealers had the greatest proportion of days (58%) in which they were active in any crime. They also had the highest or second highest percentage of person-days of involvement in robbery (8%) and drug dealing (18%), as well as STC (22%). While active on somewhat fewer person-days than other criminal offender types in property offenses (16%) and prostitution (2%), such differences were not substantial (Table XIII. 3).
- The robber-dealers also had the highest or second highest (after those who were defined as specializing in a given offense) number of offenses per year for robbery (33), burglary (36), drug sales (855), STC (381), and all offenses (1,411). They had somewhat fewer offenses than other types in property offending (101) and prostitution (8) (Table XIII. 4).
   The robber-dealers had the highest criminal income (\$16,666) which was over \$5,000 higher than the next most successful criminal types (burglar-dealers, thief-dealers, and low level robbers). This total criminal income was due to the robber-dealer obtaining among the highest annual incomes from from robbery (\$2,433), burglary (3,356) and drug selling (\$3,888), STC (\$3,187). While their income from property offenses and pimping/ prostitution was not high compared with other offender types, robber-dealers also obtained substantial incomes (property -\$3,274 and \$529-prostitution). (Table XIII. 5)

Criminal Offenders, Heroin Users, and Criminality -186-In comparison with the Chaiken and Chaiken (1981b) findings, street opiate users classified in specific criminal offender types appeared to have similar patterns of activity to their career criminals. For example, over half of the subjects in both studies who committed robbery (robber-dealers and low level robbers) also reported burglary and also committed thefts/property crimes.

Tables XIII. 2-5 also provided additional findings not reported by the Chaikens. The widespread nature of steering, touting, and copping was clearly evident. Almost all street opiate using respondents engage in this offense and did so between 12 - 20% of their person-days. Moreover, STC appears to be relatively independent of "drug dealing" in that criminal offender types who do not sell drugs appear about as likely engage in, and have similar offense rates and incomes from STC, as their similarly criminal counterparts who sell drugs. Likewise, involvement in prostitution/pimping was not clearly related to other criminal offender types.

While the criminal income of the robber-dealers was higher than other criminal types, the differences did not seem as striking as suggested by the much wider variation in the annual number of offenses committed. Clearly the relationships between crime commission and criminal income was quite complex. It will be considered in subsequent papers.

Nevertheless, the classification of respondents developed by Chaiken and Chaiken (1982ab) provided a meaningful distribution of respondents among the New York street opiate users. Moreover, the basic findings about criminal behaviors documented in their study was replicated in almost all major respects among these New York street heroin abusers.

Criminal Offenders, Heroin Users, and Criminality -187-3) How were heroin user types related to criminal offender types?

In Chapter IV, a Heroin User Typology was developed which classified respondents according to the frequency of their heroin consumption. Table XIII. 6 shows that these two typologies were strongly related to each other.

Especially important was the fact that 42% of the daily heroin users were classified as robbers, while 18% of the regular heroin users and 12% of the irregular heroin users were so classified (Table XIII. 6A). When the four most serious criminal types were combined 62% of the daily heroin users, 44% of the regular heroin users, and only 27% of the irregular heroin users were classified as robbers and burglars. The irregular heroin users were especially concentrated among those who were classified in the theft categories. Very few (7%) of the daily heroin users were classified as drug dealers or lower categories compared with 28% of the irregular heroin users. (Table XIII. 6A)

When the relationship was examined in the opposite direction (Table XIII. 6B), 64% of the robber-dealers, 46% of the low level robbers, 38% of the low level burglars, and 35% of the low level thieves were daily heroin users. With the exception of robber-dealers, relatively few (under 20%) of the burglar-dealers, and theft-dealers, and drug dealers were daily heroin users. These groups seem somewhat more likely to be irregular heroin users than their counterparts who do not sell drugs.

Given that the heaviest heroin users were disproportionately classified into the highest categories of criminal involvement, how do offending rates and criminal income vary within criminal offender types according to their frequency of heroin use?

# 4) so, for what crimes?

Tables XIII. 7 & 8 present data about the annualized offense rates for robbery, burglary, property crimes, drug sales, STC, and all major offenses (all of these including prostitution/pimping which was not presented separately). Within each crime category, subjects were classified on the Heroin User Typology. Because the number of cases were very small (under 5 cases) in many cells, the mean values given may fluxuate widely due to the contributions of one subject. The central finding from these tables was relatively clear: Even within criminal offenders of the same kind, those who were daily heroin users generally had higher offending rates than their regular or irregular heroin users. This was not true for every offender type or each major crime category. Several interesting relationships emerged.

Among robber-dealers, the irregular heroin users (N=3) committed fewer robberies (18/year) than their daily (34/yr) heroin using counterparts (Table XIII. 7). Further, daily heroin users commit many more burglaries (48/year) than the regular or irregular heroin users (19 and 6/year). Among low level robbers, no relationship occurs between the Heroin User Typology and robbery offenses, but a strong relationship occurs for burglary. Burglar-dealers do not exhibit heroin related differences in burglary. But among low level burglars, daily heroin users commit 86 burglaries/year compared to 31 among the irregular heroin users).

1 1

The property offense rates, however, did not vary systematically by the Heroin User Typology among the upper six categories of the Criminal Offender Typology. Nor were there systematic differences in property offense rates between the six criminal types. In short, virtually all six categories who are defined as possibly engaging in theft do so, regardless of how much they use heroin (Table XIII, 7).

Criminal Offenders, Heroin Users, and Criminality -188-Within each criminal offender type, were increasing levels of heroin consumption related to higher crime rates and criminal incomes, and if

Criminal Offenders, Heroin Users, and Criminality -189-The offense rates for drug sales varied by heroin user type among the robber-dealers and theft-dealers. That is, among robber-dealers, the daily heroin users committed over 1,000 drug sales/year compared with 766 among the regular heroin users, and 157 among irregular heroin users.

The annual rates of steering, touting, copping were generally higher among daily heroin users when compared to the the regular heroin users for each criminal offender type. But those classified in the most serious offender categories did not always have the highest rates of STC.

Tables XIII. 9 and 10 present data about the criminal income<sup>\*</sup> from these major crimes. For the most part, criminal income per subject per year from specific crimes (robbery, burglary, property, drug sales, and STC) did not vary systematically by the Heroin User Typology within specific criminal offender types. This absence of systematic variation occurred for property offenses, STC, and even robbery. On the other hand, the daily heroin users appeared to have higher burglary income among the robber-dealers and low level robbers. Income from drug dealing among robber-dealers did not vary by the Heroin User Typology. Drug dealers showed no heroin-related variation in dealing income.

Nevertheless, for most criminal types, daily heroin users had the highest cash income from all major crimes. Thus, while heroin use appears to be systematically related to the sum of the cash income from all crimes within each criminal offender type, this relationship does not hold true for specific offenses which make up the components on that criminal income.

\* - "Income" refers to cash income from nondrug crimes and cash plus drug income derived from the drug sales and STC transactions included in the Cash income from "other drug business," and drug income from "drug thefts," and "avoided expenditures-drugs" were not

# of the volume of crime?

Table XIII. 11 shows that various offense classes are not evenily committed by all criminal offender types. The right-most column contains the percentage distribution of our 201 subjects across the 9 Criminal Offender Typology categories. The other columns present the annualized number of offenses which these subjects would commit in a year, and the proportion of these offenses committed by subjects in each category.

The first row shows that 11% of our subjects were robber-dealers. Yet these few subjects committed 60% of the robberies, 22% of the burglaries, and 26% of the drug deals; they committed 19% of all offenses by these subjects. They committed about their proportionate share of property crimes and STC crimes, and a small proportion of the prostitution crimes.

Low level robbers constituted 13% of all subjects, but committed 40% of the robberies, 25% of the burglaries, 19% of the theft offenses, and 18% of the prostitution crimes. Because they were defined as nondealers, they contributed only 9% of the total volume of crime.

crimes.

Thief-dealers, 18% of all subjects, did not contribute a disproportionate share of the property offenses or drug sales (their definitional crimes), but did commit 30% of the STC offenses. Thief-dealers, however, did contribute about their proportionate share of all crimes (21%). Drug dealers, 6% of all subjects contributed 33% of all drug deals, but only 4% of the STC transactions.

All other groups contributed relatively lower proportions of crimes than their proportion in the sample. Even the low level distributors, defined by their exclusive involvement in STC activity, contributed only 8% of all STC transactions, exactly equivalent to their proportion in the study.

Criminal Offenders, Heroin Users, and Criminality -190-

5) Do certain types of criminal offenders commit a disproportionate share

Low level burglars were 12% of the subjects, but did 44% of the burglaries, and about their proportionate share of all other definitional

Criminal Offenders, Heroin Users, and Criminality -191-Summary

This chapter developed a criminal offender typology which was systematically related to the Heroin User Typology. Further, this Criminal Offender Typology replicated similar findings reported in a major study of career criminals conducted by Chaiken and Chaiken (1982ab). Respondents were classified according to combinations of offenses which they reported on two percent or more of their person-days.

The resulting Criminal Offender Typology was related to the percentage of respondents active in various (nondefinitional) crimes, the percent of person-days active, the annual offense rate, and the annual criminal income. These data replicated the Chaiken and Chaiken (1982ab; Chaiken, 1983) findings in most important respects. Moreover, the Criminal Offender Typology was systematically related to the Heroin User Typology in expected ways.

Daily heroin users were much more likely to be classified as serious offenders, the irregular heroin users were more likely to be thieves and low seriousness offenders. Conversely, the robber-dealers were especially likely to be daily heroin users, as were burglars and thieves who did <u>not</u> sell drugs.

Among criminal offender types, the frequency of heroin use was generally associated with total higher crime rates and criminal incomes, but not necessarily for each specific crime which entered into this total.

In the following chapter, we examine what kinds of persons did not enter methadone treatment during their reporting period.

### Criminal Offe Table XIII. 2 --

Respondent Criminal Behavior

Among Crime Typ Robber-Dealer \*\* Low Level Robber \*\* Burglar-Dealer Low Level Burglar Thief-Dealers Low Level Thieves Drug Dealers

Low Level Distributors

None of Above

Percent of Rs Active<sup>D</sup> in This Crime

\*0 - No group member commits this crime on 2% or more of their person-days, by definition; the cell is set equal to zero, although a few persons may this commit this offense occasionally. \*\*100 - All group members commits this crime, by definition. \* - STC - Steering, Touting, Copping Drugs a - Percentage of respondents active in any definitional crime; excludes occasional offenses in which they were defined as inactive (and set equal to zero). b - Percentage of respondents definitionally included in the column.

BDJ; 8273A;0745A;bj

uttenders, Heroi	n Users, and	Criminal	ity	-192	-		
Percentage of L Each Criminal Va	<u>Respondents</u> riety Type	Active <sup>b</sup>	in	Each	Crime	Туре	Among

lob- oery	Burg- lary	Offense Types Theft/Forgery Fraud,Con Gam	Pimp/Pro: e titute	s- Drug(s) Selling	Drugs STC*	Any Crime <sup>a</sup>	
pe,	Percenta	ge of Responde	nts Who Are	Active in	1 Above	Crime:	
100	64	91	18	**100	91	100	
100	62	77	12	*0	81	100	
*0	**100	100	29	**100	94	100	
*0	**100	75	13	*0	71	100	
*0	*0	**100	28	**100	81	100	
*0	*0	**100	23	*0	75	100	
*0	*0	*0	33	**100	58	100	
*0	*0	+0	10				
0		~0	12	**()	**100	100	
*0	*0	*0	43	*0	*0	43	
24	35	75	21	43	78	98	

Table XIII. 3 -	- Prop Each	ortion Crimin	of <u>Person Day</u> s al Variety Type	<u>s</u> Active <sup>b</sup>	in Each	Crime	Type Among
Respondent Criminal Behavior Typology	Rob- bery	Burg- lary	Offense Types Theft/Forgery Fraud,Con Game	Pimp/Pros- titute	Drug(s) Selling	Drugs STC*	Any of These <sup>a</sup>
Among The Cr	iminal	Types,	Proportion of Pe	erson Days I	Active In	Above	Crime:
Robber-Dealer	8	8	16	2	18	22	58
Low Level Robber	5	9	11	4	*0	21	44
Burglar-Dealer	*0	5	19	4	13	14	44
Low Level Burglar	*0	15	14	1	*0	13	38
Thief-Dealers	*0	*0	16	4	19	19	48
Low Level Thieves	*0	*0	21	5	*0	13	34
Drug Dealers	*0	*0	*0	3	17	16	32
Low Level Distributors	*0	*0	*0	3	*0	16	18
None of Above	*0	*0	*0	22	*0	*0	22
Percent of Days That Active <sup>b</sup> Rs Commit This Crime	6	10	16	4	7	16	40

Criminal Offenders, Heroin Users, and Criminality -193-

\*0 - No group member commits this crime on 2% or more of their person-days, by definition. The cell is set equal to zero, altough some days may be nonzero in reality.

\* - STC - Steering, Touting, Copping Drugs

- a Only active respondents whose crimes were not definitionally excluded. Thus, the burglaries, thefts, prositiution/pimping and STC by low level robbers were included here, but since they sold drugs less than 2% of their person days, their drug sales were excluded here.
- b Percentage of person-days active among subjects definitionally included in the column.

Table XIII. 4	al Offe Num Cri	nders, Iber of me Type	Heroin Users, an Offenses (Lambo Among Each Crim	d Crimina <u>da) Per `</u> inal Vari	lity ~194- <u>Year</u> of Sti ety Type	reet Ti	me in E
Respondent Criminal Behavior Typology	Rob- bery	Burg- lary	Offense Types Theft/Forgery Fraud,Con Game	Pimp/Pro titute	s- Drug(s) Selling	Drugs STC*	Sum of Crimes
Among The	e Crimi	nal Typ	e, Number of Off	enses Per	Year of Ab	ove Cri	me
Robber-Dealer	31	36	101	8	855	381	1,411
Low Level Robber	18	34	157	41	*0	356	606
Burglar-Dealer	*0	18	161	30	1,128	125	1,463
Low Level Burglar	*0	64	117	18	*0	249	448
Thief-Dealers	*0	*0	122	21	298	506	948
Low Level Thieves	*0	*0	139	31	*0	236	406
Drug Dealers	*0	*0	*0	14	2,003	181	2,198
Low Level Distributors	*0	*0	*0	27	*0	294	321
None of Above	*0	*0	*0	144	*0	*0	144
Total by Active <sup>b</sup> Rs in Column	24	39	132	29	836	312	824

number may be nonzero. \* STC - Steering, Touting, Copping Drugs a - Only Active respondents whose crimes were not definitionally excluded. See footnote (a) of Table VIII. 3. b- The mean number of offenses/year by persons definitionally included in this column.

Table XIII. (	5 <u>Ann</u> 0f1	ualized ender Ty	<u>Criminal Income</u> pe	for Each	ality - Offenso	195- e Among Ea	ch Crimina	1
Respondent Criminal Offender Typology	Rob- bery	Burg- lary	Offense Types Theft/Forgery Fraud,Con Game	Pimp/Pro titute	s- Drug( Selli	s) Drugs ng STC*	\$ From Thesea	
	Dolla	rs per Ye A	ear of Cash Inco mong Each Crimin	me From Al	bove Cri	ne		
Robber-Dealer	2,433	3,356	3,274	529	3,888	3 187	36.600	
Low Level Robber	1,469	3,702	1,971	916	*0	2 360	10,000	
Burglar-Dealer	*0	1,667	4,679	672	2 656	2,300	10,417	
Low Level Burglar	*0	3,530	3,889	364	- ,000 +0	1,2/9	11,252	
Thief-Dealers	*0	*0	2,997	742	2 27 2	1,565	9,348	
Low Level Thieves	*0	*0	4,171	950	3,313	3,403	10,455	
Drug Dealers	*0	*0	*0	052	*()	2,247	7,270	
Low Level	*0	*0	+0	049	2,154	1,793	4,596	
Distributors	-	v	~U	775	*0	1,426	2 ,201	
None of Above	*0	*0	*0 4	.209	*0	40		ł
otal Dollars from This Crime by Active <sup>b</sup> Rs	1,911	3,181	3,460	830	3,170	*U 2,340	4,209 9.166	
						-	- 1100	

Criminal Offenders, Heroin Users, and Criminality -195-

\*O - No group member commits this crime on 2% or more of their person-days,

by definition. Cell is set equal to zero by definition, although the actual \* - STC - Steering, Touting, Copping Drugs; also Drug Sales and STC included both

cash and drug income derived from the drug distribution offenses counted in

a - Only Active respondents whose crimes were not definitionally excluded. See b - The mean income/year by persons definitionally included in this column.

Criminal Offender Typology

Robber-Dealer Low Level Robber Burglar-Dealer Low Level Burglar Thief-Dealers Low Level Thieves Drug Dealers Low Level Distributors None of Above Total

Robber-Dealer Low Level Robber Burglar-Dealer Low Level Burglar Thief-Dealers Low Level Thieves Drug Dealers Low Level Distributors None of Above Total

C. Nu

Robber-Dealer Low Level Robber Burglar-Dealer Low Level Burglar Thief-Dealers Low Level Thieves Drug Dealers Low Level Distributors None of Above Total

by definition.

# Criminal Offenders, Heroin Users, and Criminality -196-Table XIII. 6 -- Percentage of Respondents in the Criminal Offender

Typology and Heroin User Typology

Heroin U	ser Typol	ogy		
frregular	Regular	Daily	Total	
(61)	(78)	(62)	(201)	
A. Vertical	Percenta	ges		
5	6	23	11	
7	13	19	13	
8	12	5	8	
7	14	15	12	
26	18	10	18	
20	18	23	20	
13	4	2	6	
12	10	3	8	
3	5	2	4	
101	100	102	100	
B. Horizontal	Percenta	iges		
13	23	64	100	
15	39	46	100	
29	53	18	100	
17	46	37	100	
44	39	17	100	
30	35	35	100	
67	25	8	100	
41	47	12	100	
29	57	14	100	
30	39	31	100	
umber of Subje	cts in Ea	ch Cella		
3	5	14	22	
4	10	12	26	
5	9	3	17	
4	11	9	24	
16	14	6	36	
12	14	14	40	
8	3	1	12	
7	8	2	17	
2	4	1	7	
61	78	62	201	

\*0 - No group member commits this crime on 2% or more of their person-days,

a - Number of cases upon which information in Tables XIII. 7 - 10 is based.

Crimi Table XIII. 7	inal O N Ri Va	ffende umber obbery ariety	ers, Hero of <u>Off</u> , Burg] Type and	in User enses ary, i Heroi	rs, and <u>(Lambo</u> and F n Use	d Crimina da) Per Troperty Type	ality - Year Crimes	l97– of St Among	reet Time Each Cr	e for iminal		
Respondent Criminal Behavior Typology		Robt	Dery		Bui	rglary	Prop	erty O	ffenses:			
Heroff User Type:	Irre	g. Re	g. Daily	/ Irre	g. Re	g. Dail	y Irred	1/Other	ery rs	1		i
<b>P</b>	A Nur	mong i nber o	he Crimi f Offense	nal and s Per	l Heroi Yeara	n User 1 Of Above	ypes,		J. Daily			
Kobber-Dealer	018	32	34	06	19		crime					
Low Level Robber	@17	15	21	63	26	то 51	0117	123	90	-		
Burglar-Dealer	*0	*0	*0	19	19	015	0319	47	196			
Low Level Burglar	*0	*0	*0	31	59	015	42	230	0151	3		Diginary - same
Thief-Dealers	*0	*0	*0	*0	+0	86	144	108	116			
Low Level Thieves	*0	*0	*0	*0	~U	*0	91	178	78			****
Drug Dealers	*0	*0	*0	*0	×0	*0	104	134	174	14 A	1782	<b>G</b> arder
Low Level Distributers			-	~0	*0	*0	*0	*0	*0		united to	
None of the	*0	*0	*0	*0	*0	*0	*0				ete d	-
Total to a set	*0	*0	*0	*0	*0	*0	*0	*U	*0	<b>g</b> 1.3	······································	
Rs in Column	17	21	28	16	33	55	116	*0 139	*0 139		or ().	

\*0 - No group member commits this crime on 2% or more of their person-days, by definition. The cell is set equal to zero by definition, although the actual number of offenses may be nonzero.

 a - Only Active respondents whose crimes were not definitionally excluded. See footnote (a) of Table VIII. 3. The mean number of offenses/year by persons definitionally included in this

Q - Number of offenses is calculated from less than 5 cases. (see Table

Crimina Table VIII o	al Off	enders,	Heroin U	lsers, and	l Crimi	nality	-198-		
ladie XIII. 8 -	- Num Sal Typ	ber of es, STC e and He	Offenses , and roin Use	<u>Per Yea</u> All Major Type	r (Lan Crime	<u>ibda)</u> of s Among	Street Each Ci	Time f riminal	or Drug Variety
Respondent Criminal Offenders		Dave Ce	1	Steering	, Tout	ing	All Ma Burgla	jor Crin ry + Pro	nes: <sup>a</sup> Robbe operty+ Pro
Heroin User Type:	Irreg	. Reg.	Daily	Irreg.	ng Reg	. Daily	tution Irreg.	+ Drug Reg.	Sale + STO Daily
		Among Number	The Crin of Offe	ninal and enses Per	Heroin Year o	User Ty f Above	pes, Crime		
lobber-Dealer	0157	766	1,036	0655	153	403	0954	1,111	1,617
ow Level Robber	*0	*0	*0	0110	233	541	@450	337	882
Burglar-Dealer	34	1,984	0385	54	174	0100	219	2,418	0669
ow Level Burglar	*0	*0	*0	039	160	452	0214	364	655
hief-Dealers	246	223	613	133	691	1,068	515	1,095	1,761
ow Level Thieves	*0	*0	*0	41	55	585	149	201	833
rug Dealers	350	020, 79	0166	152	0171	0442	515	@7,215	0608
ow Level Distributors	*0	*0	*0	111	97	@1 ,721	154	117	@1,721
lone of Above	*0	*0	*0	*0	*0	*0	00	0252	00
otal Offenses by Active <sup>b</sup> Rs in Column	24	1,480	813	126	245	573	370	980	1,105

- column.

----

 a - Only Active respondents whose crimes were not definitionally excluded. See footnote (a) of Table VIII. 3. b - The mean number of offenses/year by persons definitionally included in this Q - Number of offenses is calculated from less than 5 cases. (see Table XIII.6C).

Cri Table XIII.	minaî Of 9 <u>A</u> n	fenders, nualized	Heroin	Users, a	and Crim	inality -	199-													
Respondent Criminal Behavio	r	imes Anoi	ng Each (	l Income Criminal	e for Ro Variet	bbery Bu Y Type and	Heroin Use	ind Prop Type	Derty		Crimi Table XIII. 10	nal Offe D Ann Maj Tvp	nders, ualized or Crin	Heroin U Crimin Nes Amon	Jsers, and al Income g Each Cri	Crimin for [ minal \	ality - Drug Sal Variety	-200- es, STC, Type and H	and A eroin U:	11 se
Heroin User Typ	e: Irreg.	Reg.	Daily	Bu Irreg	rglary • Reg	. Daily	Proper Theft/I <u>Fraud</u> ( Irreg.	ty Offe Forgery Con Game Reg.	nses:	Rd Ci Ty	espondent riminal Behavior /pology	l	Drug Sa	les	Steering	, Touti	ng	All Major Burglary +	Crimes: Propert	Robbery + ty+ Prosti-
Robber-Dealer	De	ollars o	f Cash Ir	ncome Per	d Heroir r Yeara	User Type	s,		Juliy		Heroin User Type	Irreg.	Reg.	Daily	Irreg.	Reg.	Daily	Irreg.	rug Sale Reg	+ STC J. Daily
Low Level Robber	@{ ,134	3,130	2,462	@237	2,232	4.425	e crime					Dol	Among lars of	The Crim Cash In	inal and come Per	Heroin Year <sup>a</sup> f	User Typ	es,		
Burglar-Dealer	0947	1,296	1,787	@194	1,444	6.752	e7,/32	2,17	5 2,71							,		e or rilles		
Low Level Burglan	*0	*0	*0	735	2,070	@2.009	650	1,918	2,455		Robber-Dealer	02,525	3,756	4,227	03,136	929	4 004	014 765	12 247	10.050
Thief-Dealers	×U +0	*0	*0	@2,076	4,208	3,347	1,/04	5,667	@6 <b>,</b> 673	Lo	w Level Robber	*0	*0	*0	0516	2 195	3 113	62 216	7 252	18,259
Low Level Thieves	*U +0	*0	*0	*0	*0	*0	e4,070	4,208	3,420	Bu	rglar-Dealer	696	2,370	@6,784	343	2 343	1 346	4 007	/ ,353	15,6/1
Drug Dealers	~U +0	*0	*0	*0	*0	*0	2,000	4,139	4,069	Lo	w Level			-		- ,0 10	, 1 <sub>9</sub> 040	4,557	12,740	17,218
Low Level	~0	*0	*0	*0	*0	*0	2,088	3,352	6,755	B	urglar	*0	*0	*0	144	979	2,912	6,290	10,175	9,697
Distributors	*0	*0	*0	*0			~0	*0	*0	Th	ief-Dealers	3,050	1,886	7,345	1,144	5,159	5,330	7,322	11,324	16,780
None of Above	*0	*0	*0	*0	*0	*0	*0	*0	*()	Lov Tł	v Level Nieves	*0	*0	*0	352	502	E EQC	0 504		
Total Dollars			-	~0	*0	*0	*0	*0	*0	Dru	lg Dealers	2.053	2.356	a2 345	686 (	552 1 110 /	5,520	2,504	4,344	14,280
from This Crime by Active Rsb	1,027 1	,907 2	.151	842 0					• •	Low	Level	,	-,		000 (	21 <b>,</b> 410 (	911,780	3,354	@4,732	014,124
				042 2	,586	4,714	2,300	3,686	4.094	Di	stributors	*0	*0	*0	597	1,326	@4,727	1,987	1,757	@4,727
*O - No group member by definition	commits	this cri	Me on 29						1	Non	e of Above	*0	*0	*0	*0	*0	*0	*0	07 <b>,</b> 366	0
actual dollar am a - Only Active res footnote (a) of b - The mean income/ this column	Ine cel ounts may pondents Table VII year of	ll is se y be non whose I. 3. this cri	et equal zero. crimes y	or more to zer Vere not	of the o by d defin	ir person-c efinition, itionally c	although	the See		Tot f b	al Dollars rom This Crime y Active Rs <sub>a</sub>	2,384	2,374	5,248	779	2,079	4,168	4,943	8,256	14,865
e - Number of offense	s is cal	culated	from les	by pers s than 5	ons def cases.	initionall (see Tab	y included le XIII.6C	in ).		*() a -	<ul> <li>No group member by definition. amounts may be Only Active res of Table VIII.</li> </ul>	r commit The ce nonzero spondent 3.	s this 11 is s s whose	crime on et equa e crimes	1 2% or mo 1 to zero were not	re of t by def defini	heir per inition, tionally	son-days, although excluded.	the actu See for	ual dollar otnote (a)
										0 -	Ine mean incom column. Number of offen	ne/year uses is (	of thi alcula:	s crime ted from	type by less tha	person n 5 cas	s defin es. (se	itionally e Table XI	included	l in this
	··									8 <b></b>										
					•					0										

1

QEA.

# Criminal Offenders, Heroin Users, and Criminality -201-Table XIII. 11 -- Percentage of All Offenses of Each Crime Type Committed During

Year by Subjects in Each Criminal Variety Type

Criminal Behavior Typology	Rob- bery	Burg- lary	Property Theft/Forgen Fraud,Others	ry Pimp/Pro s titute	s- Drugs Sales	Drugs STC*	All Crimes	Percent of Ss.
		Percen Above (	t of All Off Crime Among E	enses Per Y Each Crimina	ear for al Type <sup>a</sup>			
Robber-Dealer	60	22	10	3	26	14	19	11
Low Level Robber	40	25	19	18	*0	15	9	13
Burglar-Dealer	*0	9	12	9	26	4	15	8
Low Level Burglar	*0	44	13	8	*0	10	6	12
Thief-Dealers	*0	*0	20	13	15	30	21	18
Low Level Thieves	*0	*0	26	21	*0	15	10	20
)rug Dealers	*0	*0	*0	3	33	4	16	6
.ow Level Distributors	*0	*0	*0	8	*0	8	3	ê 8
one of Above	*0	*0	*0	17	*0	*0	1	4
otal in Column <sup>a</sup>	100	100	100	100	100	100	100	100
nnual Number of Offenses by (1 Those in Column (Base N)	,151)(3	3,507)(2	1,817) (	5,837) (7	2,762) (	60,566) (165	5,640)	(201)

STC - Steering, Touting, Copping Drugs

Only Active respondents whose crimes were not definitionally excluded. See

This chapter will describe what types of criminals and heroin users did not have any methadone treatment during their reporting period. The absence of such treatment has important policy implications.

In making any assessment of these data regarding methadone, however, one must remember that recruitment patterns focused upon obtaining respondents for whom methadone treatment was likely to have marginal impact. The data presented here do not provide a basis upon which to "evaluate" the success of methadone maintenance. Specifically, our field workers were unlikely to recruit many types of methadone clients such as those with fulltime jobs, who spend most of their time with family, or who otherwise avoid the street scene. Moreover, our recruitment methods over-selected methadone clients who were continual abusers of drugs and alcohol, and the most troublesome for clinics to manage. Thus, when this study recruited street opiate users enrolled in methadone programs, such subjects would be among the least successful and most criminally active.

This chapter compares subjects with "some" methadone treatment to those with those none. Respondents were defined as having "some" treatment if they reported using legal methadone on two percent or more days during their reporting period. Among the 52 subjects who met this criteria of "some" methadone treatment, extensive variation was evidenced in the number of days they were enrolled, whether they took their medication as prescribed, distributed it to others, or otherwise missed medication days. Analyses of The critical finding, however, shows that street opiate users heavily treatment. Prior to demonstrating this conclusion, data will be presented

these complex personal differences must await further analyses of these data. involved in heroin and serious criminal behavior were rarely in methadone about background factors and enrollment patterns in methadone programs.

### Who Was Not in Methadone Treatment -202-CHAPTER XIV

# WHICH STREET OPIATE USERS WERE NOT IN METHADONE TREATMENT?

# Who Was Not In Methadone Treatment? -203-What factors prior to interview, if any, differentiate street opiate users without and with some methadone treatment?

Among our 201 subjects, 26% (N=52) were classified as having some methadone treatment and 74% were not in methadone treatment during their reporting period. Table XIV. 1 presents a variety of background variables by which these subjects might be anticipated to differ.

# [TABLE XIV. 1 about here]

The data here suggest that subjects with and without some methadone treatment during their reporting periods did not differ greatly by sex, ethnicity, and age. Data not presented also show no differences by methadone treatment for neighborhood of residence (East or Central Harlem), marital status, education, type of work, or the use during the prior year of illicit methadone, cocaine, or other drugs.

Although it might be anticipated that those without methadone treatment might be more deviant than subjects with such treatment, Table XIV. 1 shows no differences (with one exception) in the prior histories of our subjects. Both groups reported a similar number of prior arrests, years of incarceration, and years of heroin use. Both groups reported about the same frequency of heroin use and daily amounts of heroin used per day in the year prior to interview. Both groups reported similar percentages addicted to heroin, alcohol, and cocaine .

The one significant difference was in their self-reported major means of supporting drug use. Those without methadone treatment were more likely to report illegal support via theft, while those with some methadone treatment reported more involvement in the drug business.

Nevertheless, both groups had similar incomes from criminal and drug dealing sources, as well as similar frequencies in the previous year of shoplifting, burglary, robbery, and drug business activity.

In short, these street opiate users, regardless of whether in methadone treatment or not, exhibited almost the same background characteristics and patterns of drug use and criminality during the prior year.

	Methadone Treatment	Subjects	WITHOUT	and With So
	Any in	Methadon Reporting	e Treatmen Period?	t
Characterist	tic Number of Subjects:	None (149)	Some (52)	Significance Level of Chi
Sex:	Male Female	77% 23	71% 29	.56
Age:	under 26 26-30 31-35 36-40	16 26 22 13	04 29 29 21	.10
Ethnicity:	41 and older Black	19	17	
	Hispanic Other	56 44	52 48	.77
Number of Ar	rests:			
	None 1 - 5 6 or more	19 38 43	09 45 45	.31
Number of Yea	ars Incarcerated			
	1-4 5 or more	42 37 21	25 52 23	.11
Number of Yea	rs of Heroin Use	07	0.0	
	6-10 11-15 16 & over	26 25 22	20 34 23 23	.72
Frequency of Prior to I	Heroin Use in Year nterview			
	Daily Weekly Monthly	66 27 07	53 28 19	.07
Average Daily in Year Pri	Amount Heroin Used/Day or to Interview	27		
	\$21 - \$50 Over \$50	37 38 25	56 20 24	.05

	Who Was Not In	Methadon	e Troat	mon+2 205		Water State
Table XIV. 1 (	continued)	Characto Some Me	eristic thadone	s of Subject Treatment	s Without and Wit	:h
		Any	Methad	one Treatment		
Characteristic	Number of Su	bjects:	None (149)	ng Period? Some (52)	Significance	
Percent Self-Report	ting Addiation .	•	<u></u>		Level of Chi Squar	<u>e</u> –
No D	rugs	):	F	0		
Hero	in		5 75	8	.64	
Ille	gal Methadone		4	/ D 8	.99	
Coca	ine		9	10	.36	
AICO	101		20	14	.99	
Major Means of Supp	Orting Drug Nees				, 10	
Thef	t		50	~~		and and a second
Drug	Business		50 17	28	_	
Work	ing		16	37 14	.03	and the second
Other	r		16	21		
Estimated Criminal	Income in Dutan V			- • · ·		1.000 ga
\$1.00	0 & linder	ear		•		and the second s
\$1,00	01-5,000		44 25	41		a st
\$5,00	)1-10,000		25	23	.90	
over	\$10,000		14	10		
Estimated Drug Expon	ditume to be	•		10		
	0 & Under	Year				
\$1,00	1-5.000		38	47		- where
\$5,00	1-10,000		41 20	3/		
over	\$10 <b>,</b> 000 .		01	14	.46	
Shoplifting Frequence	a dia Districtiona		- •	02		مر میداده . مرب
Daily	y in Prior Year					
Weekly	<b>V</b>		11	05		
Month	, ly		22	36		A CONTRACTOR OF A CONTRACTOR O
None	, e		21	17	.25	
Punglaux C.			70	43		
Mary rrequency in	Prior Year					e dinanie e
Meek ly Monthi	a Daily		14	02		and the second se
None	У		19	13	06	
none			67	85		
Robbery Frequency in	Prior Year					and the second sec
Weekly	& Daily		7	0		· · · · · · · · · · · · · · · · · · ·
Monthl	у		16	0 20	0-	
None			77	72	.81	and a second
Drug Business Frequence	rv in Price Var			·		the second s
Daily	-J IN FRIOR Tear	÷	0			8
Weekly		2	.U .E	38		
Monthly	/	ן ד.	3	24	.34	in the second
None		5	3	12		
Monthly None	/	1 5	5 3 3	24 27 12	.34	

### Who Was Not In Methadone Treatment? -206-<u>What types of criminals and heroin abusers had</u> no methadone treatment during their reporting periods?

Table XIV. 2A examines what types of criminal offenders and heroin users had <u>no</u> methadone treatment during the period they were interviewed on a weekly and daily basis. The criminal offender typology (see Chapter XIII) was collapsed so that the dealer-nondealer distinction was ignored. Thus, respondents were classified in a hierarchical order of robbers, burglars, thieves, and "none of these" (drug dealers, low level distributors, and prostitutes). Among daily heroin users, 92% or more of the robbers, burglars, and thieves had no methadone treatment during their reporting periods. Among irregular heroin users, about three-fifths had no methadone treatment. Among the 16 robbers and burglars who were irregular heroin users about 44% had no methadone treatment. Methadone treatment was more likely to have an impact upon clients if they

nethadone treatment was more likely to have an impact upon clients if they continuously consume their medication on a daily basis, as directed. Nevertheless, less than half the subjects with some methadone treatment used methadone on 66% or or more of their reporting days.

Table XIV. 4B shows that of the subjects (N=7) who were daily heroin users with some methadone treatment, only one subject consumed legal methadone on more than two-tnirds of the reporting days. All other daily heroin users either dropped out or used their legal medication on less than 66% of their reporting days. Likewise, among the regular heroin users, while a quarter had some methadone treatment, 100% of the robbers and "none of these," 90% of the thieves, and 80% of the burglars consumed legal methadone on 66% or more of their days.

In short, respondents classified as daily heroin users (averaged heroin use on 6-7 days per week during entire period), most (90%) avoided methadone treatment completely, or if they obtained such treatment, they did not consume legal methadone for a substantial proportion of the reporting days. Although the proportions were higher among regular heroin users, similar findings also emerged.

### Who Was Not In Methadone Treatment? -207-

Among irregular heroin users, while over half of the 16 robbers and burglars had some methadone treatment, about a third of them consumed methadone over 2/3 of the time. Among irregular heroin users, about 60%having some methadone treatment consumed their legal methadone on two-thirds Thus, only among irregular heroin users were high proportions of the time. in treatment and likely to consume methadone as expected. Clearly, however, information about these respondent's behaviors on a daily basis are needed to disentangle complex patterns of admission and termination of methadone treatment, and how subjects used their medication.

### Among all respondents, how many subjects exhibited which patterns of criminality, heroin use, and methadone treatment?

Regardless of how methadone treatment influenced heroin use and criminality, a central finding was that the most criminally active and socially damaging subjects (see Chapter XI and Chapter XV) were not in methadone treatment.

### (Table XIV. 3 about here)

Table XIV. 3 shows the distribution of all respondents according to their heroin user type, offender type, and methadone treatment type. Subjects have been reclassified into daily/nondaily heroin users and into robbers/burglars and thieves/others (see detailed data at bottom of table).

Only 4% of all subjects were daily heroin users with some methadone treatment. By contrast, 27% (9%+18%) of all subjects were daily heroin users having no methadone treatment. Almost half (46%) were nondaily heroin users without methadone treatment. It was mainly among the nondaily heroin users that a sizable (22%) proportion of subjects had some methadone treatment.

In the next chapter, we will explore the association of criminal and heroin user lifestyles to various indicators of social impact. Data in this chapter demonstrated that the robber/burglar-daily heroin users were generally not in methadone treatment. Almost none of this key group used legal methadone on two-thirds or more of their reporting days.

treatment programs.

Three quarters of our subjects were not in methadone treatment during the interview period. On several occasions, such subjects would comment that they wanted to get off drugs or alcohol and needed help. "Maybe" they would go to treatment. One or two even planned to go to treatment. But at the next interview. they would invariably have forgotten about such intentions. When asked by the interviewer about their plans for treatment, they would state that something else came up, they forgot about it, or they were still thinking about it, or they had some other similar excuse. Although the clear purpose of this study was to investigate the actual

economic and criminal behavior of these subjects without making moral judgments or trying to rehabilitate them, the staff, in consultation with the Institutional Review Board, had agreed to aid subjects in securing admission to a nearby program.

That is, if a subject was ready to go to treatment, our staff was directed to refer them to any program of their choice and help them cut through 'red tape' to gain admission. With one exception (see below), expression of interest by subjects in treatment was never followed by commitment.

### Who Was Not In Methadone Treatment? -208-Were Those Without Methadone Treatment Avoiding It?

Although this project did not ask detailed questions about why respondents were not in methadone treatment, considerable evidence suggested that many persons not in treatment were avoiding entry into methadone or denied their need for it. A parallel research project, the Tristate Ethnographic Project (Hunt, et al., 1983), found that street opiate users not in methadone programs had a variety of folk beliefs about methadone and other reasons why they would not enter programs (although many had some previous enrollment history). Our field experience also documented this avoidance of methadone and other

Who Was Not In Methadone Treatment? -209-Even when staff offered to help subjects obtain admission, they would decline or avoid making a decision. Thus, among 149 subjects who were not in drug treatment during the study, they did not express a clear commitment to obtaining treatment, although many stated a wish to change their lives and "det off" heroin or other drugs.

The one exception occurred in 1979 when a 26 year old subject whom Preble had know for many years to be a severe alcoholic, passed out in the storefront from intoxication. Preble took him to a nearby detoxification program where he spent 8 days "drying out;" he resumed his drinking shortly afterwords.

A more typical respondent was 26 year old Kim T., who indicated that he had never previously entered a treatment program, although reporting several prison terms for assault, marijuana possession, and other charges. During his 33 reporting days, he reported three robberies, two burglaries, a contract assault, and several days with large drug sales. He also reported an arrest for burglary for which he was fined \$200 (and which he paid). He used between \$50-\$300 of heroin per day except for a period of 8 days when he "kicked" alone and abstained from heroin. After completing his reporting cycle, he returned for a visit with Tom Miller during which the following dialogue occurred:

Kim: I started shooting up again. I don't know why. I have to get on a program. Miller: But you don't like methadone. Kim: I think its shit. I mean a drug free program. Miller: You mean like Project Return. Kim: Yeah, How long do I have to stay there. Miller: About 18 months. Kim: Forget it. I'd rather be in jail.

Although most other subjects did not venture an opinion, this subject certainly speaks for many subjects not in treatment as preferring jail to either a methadone or drug free treatment program with a long term commitment. Thus, avoidance of methadone treatment programs was common among our daily heroin users.

A.3 ۹ Q

Summary

Among our street opiate users, subjects reporting some methadone treatment when compared with those lacking such treatment (during the interview periods) had similar backgrounds and self-reported prior drug abuse and criminal histories.

Moreover, most daily and regular heroin users avoided methadone treatment completely; if they obtained such treatment, they did not consume legal methadone for a substantial proportion of the reporting days. Only irregular heroin users had large proportions enrolled, and most of them used their methadone on a regular basis.

Thus, a central conclusion of this chapter was that respondents, especially those using heroin daily and involved in robbery/burglary were very unlikely to have any methadone treatment. Our experience with these heaviest heroin users and seriously criminal subjects was that almost all were avoiding methadone and other forms of treatment (see more extensive documentation by Hunt, et al., 1983); if for some reason they enrolled, they were at high risk of dropping out or not taking medication as directed.

society.

BDJ:8274A:0745A;bj

### Who Was Not In Methadone Treatment? -210-

In the next chapter, we develop a new typology of "Intensive criminality" and relate it to several measure of social impact, to determine what combinations of criminality and heroin use imposing the most damage upon Who Was Not In Methadone Treatment? -211-

Table XIV. 2-- Percentage of Criminal Offenders and Heroin Users Types with No Methadone Treatment During Their Reporting Period.

	Hen Irregular	roin User Typol * Regular	ogy Daily	Row Tota
Nondrug Criminal Offender Typology	Percent i User Grou	n Cell (i.e. O p) With <u>No</u> Meth	ffender/Drug nadone Treatment	
None of These	65	03	_	
Thief	<u></u>	33	50	75
Runglan	01	65	95	71
burylar	44	64	0.2	<i>,</i> 1
Robber	43	07	52	68
Total		87	93	79
·····	57	74	90	74

Percent of Group <u>Not Using</u> Methadone on 66% or More of Reporting Days

Nondrug Criminal Offender Typology

يىسىتىتى ئىيتىنى

None of These	82			
Thiof	02	100	75	89
nijej	79	89	100	00
Burglar	56	80		88
Robber	71		100	82
Total	1	100	100	93
	75	91	98	
			50	89

# Number of Cases upon Which Above Percentage:

None of these	/	a concages are Based					
Thief Burglar Robber	(17) (28) (9) (7)	(15) (28) (20)	(4) (20) (12)	(36) (76)			
Total	(62)	(15)	(26)	(58)			
	(02)	(78)	(61)	(201)			

Who Was	Not In Met	thadone Treat	non+2 _212_	
Table XIV. 3 - Percenta User Typ	age of All bes by Whet	Repondents A ther Having Sc	mong Criminal ( Dme Methadone T	Offender an reatment or
Any Methadone Treatment? Criminal Types		H M	leroin User Typ lot Daily	es Daily
Some Methadone Treatment				
Thieves + None of These	<b>e</b> .	(	12	Т.) <sup>.</sup>
Burglars/Robbers		22	10	· { 4
No Methadone Treatment		•		
Thieves + None of These	•	(	28 0	3
Burglars/Robbers		46		27
				Total
Nondrug Criminal	Her Her Percenta	s: Detailed So roin User Typo r Regular Ne of All Subj	ource for Above blogy Daily jects	Row
Nondrug Criminal Offender Typology	Her Irregular Percentag (Responde	s: Detailed So roin User Type r Regular ge of All Subj ents with Some	ource for Above blogy Daily jects Methadone Tre	atment)
Nondrug Criminal Offender Typology None of These	Her Irregular Percentag (Responde 3	s: Detailed So roin User Typo r Regular ge of All Subj ents with Some *	ource for Above blogy Daily iects Methadone Tre *	e Table Row atment) 4
Nondrug Criminal Offender Typology None of These Thief	Her Irregular Percentag (Responde 3 5	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4	ource for Above plogy Daily ects Methadone Tre * *	e Table Row atment) 4 10
Nondrug Criminal Offender Typology None of These Thief Burglary	Her Irregular Percentag (Responde 3 5 5	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3	ource for Above plogy Daily ects Methadone Tre * *	e Table Row atment) 4 10 6
Nondrug Criminal Offender Typology None of These Thief Burglary Robber	Her Irregular Percentag (Responde 3 5 3 2	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3 2	ource for Above plogy Daily iects Methadone Tre * * 2	r Table Row atment) 4 10 6 6
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal	Her Irregular Percentag (Responde 3 5 3 2 13	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3 2 10	ource for Above plogy Daily ects Methadone Tre * * 2 3	r Table Row atment) 4 10 6 26
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal	Her Irregular Percentag (Responde 3 5 3 2 13 (Res	s: Detailed So roin User Typo r Regular ge of All Subj ents with Some * 4 3 2 10 pondents With	ource for Above plogy Daily iects Methadone Tre * 2 3 out Methadone	r Table Row atment) 4 10 6 6 6 26 7reatment)
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal None of These	Her Irregular Percentag (Responde 3 5 3 2 13 (Res 5	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3 2 10 pondents With 7	ource for Above plogy Daily ects Methadone Tre * 2 3 out Methadone 1	e Table Row atment) 4 10 6 26 7reatment) 13
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal None of These Thief	Her Irregular Percentag (Responde 3 5 3 2 13 (Res 5 8	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3 2 10 pondents With 7 8	ource for Above plogy Daily ects Methadone Tre * 2 3 out Methadone 1 1 8	e Table Row atment) 4 10 6 26 Treatment) 13 24
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal None of These Thief Burglary	Her Irregular Percentag (Responde 3 5 3 2 13 (Res 5 8 2	s: Detailed So roin User Typo r Regular ge of All Subj ents with Some * 4 3 2 10 pondents With 7 8 6	ource for Above plogy Daily iects Methadone Tre * 2 3 out Methadone 1 1 8 5	e Table Row atment) 4 10 6 6 6 26 7reatment) 13 24 13
Nondrug Criminal Offender Typology None of These Thief Burglary Robber Subtotal None of These Thief Burglary Robber	Percentag (Responde 3 5 3 2 13 (Res 5 8 2 2 2 2	s: Detailed So roin User Type r Regular ge of All Subj ents with Some * 4 3 2 10 pondents With 7 8 6 8	ource for Above plogy Daily iects Methadone Tre * 2 3 out Methadone 1 1 8 5 13	e Table Row atment) 4 10 6 26 Treatment) 13 24 13 23

\*\*Does not add to 100% due to rounding.

### Social Impacts -213-CHAPTER XV

# THE SOCIAL IMPACT OF CRIMINAL AND HEROIN USER LIFESTYLES

This chapter will provide information about a central question which has seldom been previously addressed: What combinations of criminal and heroin user lifestyles have the greatest impact upon society? To answer this question, we developed a new Intensive Criminality Typology which combines the Heroin User Typology and a robber-nonrobber dimension (see below). This new typology (as the independent variable) will permit an assessment of whether heroin users or nondrug criminal type (robber-nonrobber) or both make significant contributions to seven measures of social impact.

The concept of "social impact" will involve four major dimensions: a) number of crimes committed, b) severity of nondrug criminal behavior, c) criminal income, and d) economic values from crime. To sharpen the focus of this analysis of social impact, careful distinctions will be maintained between social impact due to nondrug crimes and that coming from all sources (nondrug crime plus drug business crimes, income from others, dollar value added to drug distribution system, etc.). These indices for each dimension of social impact are defined below.

### Dimension A -- Criminal Offense Rates

In Chapter VIII, measures of the annualized criminal offense rate (lambda) were developed; two indices will be used below:

1) Annual Nondrug Criminal Offense Rate -- the number of nondrug crimes committed on an annual basis.

\*- In all the indices below, nondrug crimes includes the following offenses: robbery, burglary, shoplifting-resale, other larcenies, forgery, con games, prostitution/pimping, and other illegal offenses; shoplifting for own use and all drug business crimes were excluded from nondrug crimes (see Chapter VI for rationale).



# in Chapter VIII and Table VIII. 2). Dimension B -- Severity of Criminal Behavior

This dimension will employ as weights ratios developed by Wolfgang and Figlio (1982; Collins, et al., 1982). In a National Survey of Crime Severity, a representative sample of the general population were presented with over 200 scenarios in which the offense type, amount of personal injury, property value, and other factors were varied. By combining scores from these scenarios, a mean "severity ratio" was computed for each type of offense. The severity ratio for a simple theft was arbitrarily set equal to 5.0, and severity ratios of other crimes were assessed in magnitude to such thefts. For example, robbery had a severity score (11.2) which was more than twice as serious as a theft. The severity ratios for specific nondrug<sup>\*</sup> offenses will be employed:

Social Impacts -214-

2) Annual Total Criminal Offense Rate -- the annual number of offenses of all kinds (including nondrug crimes, drug business crimes, shoplifting own use, fare evasion, and other illegal transfers of drugs (see definitions

robbery--11.2; burglary--6.6; larcenies and shoplifting for resale--5.0; forgery/ fraud/ con games--5.8; prostitution/pimping--4.1; fencing/selling stolen property--7.7 (Collins, et al., 1982,61).

3) Nondrug Crime Severity Index -- is the annualized seriousness of criminal behavior employing the Wolfgang-Figlio weights. These weights were multiplied by the annual offense rate for a given offense class.

Respondent values on this index reflect only the weights given for the type of offense. Other weights in the Wolfgang-Figlio seriousness scale for the degree of injury, amount taken, etc. were not included here.

In interpreting the mean values given in the tables below, dividing by 5 would provide the approximate number of theft-equivalents committed annually.

\*- While the general population considers the sale of heroin and cocaine (severity score=24.7) as approximately twice as serious as robbery (severity score =11.4), among our respondents, such opinions were not shared. They engaged in drug sales, including heroin and cocaine, at very high rates. Thus, for the same reason that Collins, et al. (1982) eliminated drug sales from his analysis of criminal severity, the analysis in this chapter will not include weighting for drug sales.

## Social Impacts -215-

Dimension C--Criminal Income

In Chapter VIII, measures of the annualized criminal income (in dollars) were developed: two indices will be used below:

- Annual Nondrug Criminal Income -- the projected annual cash income from all nondrug crimes.
- 5) Annual Total Criminal Income -- the projected annual dollar value of criminal income from all sources (including nondrug crime, cash and drug income from drug business, fare evasion, and shoplifting-own use (see definitions in chapter VIII and Table VIII. 3).

Dimension D -- Economic Values of Criminal Lifestyles

In Chapter XII, the economic values of direct and indirect social costs (measured in dollars) among street opiate users were developed. Two indices of economic value will be used below.

- 6) Economic Values from Nondrug Crime -- the annualized dollar value of money, stolen merchandise, and illegal services obtained by street opiate users from the commission of nondrug crimes.
- 7) Total Economic Values -- the projected annual value of all social costs associated with the street opiate user lifestyle. This included the cash value of nondrug crime plus cash income from noncriminal sources, avoided expenditures, drug plus cash income from drug business, income tax evasion, and the "value added" to the drug distribution system (see Chapter XII and Table XII. 1).

Thus, these seven indices will measure the social impact which heroin using and criminal lifestyles have upon society. In the following section, a new typology of intensive criminality is developed.

# Intensive Criminality Typology In Chapter XIII, the inte Offender Typologies was pre-

In Chapter XIII, the interrelationship of the Heroin User and Criminal Offender Typologies was presented and analyzed. For the purposes of analyzing the relative contributions of criminal and heroin use lifestyles upon social impact measures, however, a simpler typology appeared necessary. After carefully examining the data in Chapter XIII, especially in Tables XIII. 6-11, subjects classified as daily heroin users <u>and</u> robbers were the most criminal and had the highest criminal incomes.

In order to develop a new typology and to investigate the relationship between robbers and heroin users, the Heroin User Typology was related to a dichotomy of robbers\*-nonrobbers; this distribution of respondents has been provided at the bottom of Table XV. 1. The data show that 44% of the daily heroin users were robbers, while 20% of the irregular heroin users and 23% of the regular heroin users were robbers. Moreover, almost half (47%) of the robbers were daily heroin users, while only 24% of the nonrobbers were daily heroin users. In short, a relatively strong association emerged between robbery and daily heroin use. Based upon the number of cases in these six cells, various classifications of respondents were examined.

A five category typology was found to effectively differentiate respondents, and at the same time assess the relative contributions of daily heroin use and robbery upon the social impact variables. Table XV. 1 shows how subjects were classified on the <u>Intensive Criminality Typology</u> as:

All subjects who did one or more robberies during their reporting periods were included as robbers here. Hence, this is not a collapsed version of the Criminal Offender Typology in Chapter XIII where a person had to have been active on 2% or more of their person-days to be a robber. Chapter XIII also showed that robbers committed all offenses at high rates.

A

Social Impacts -216blogy

### Social Impacts -217-

robbers and daily heroin users (N=27) -- as "intengives;" nonrobbers but daily heroin users (N=35) -- as "highs;" robbers and nondaily heroin users (N=30) -- as "actives;" nonrobbers and regular heroin users (N=60) -- as "inactives;" nonrobbers and irregular heroin users (N=49) -- as "lows."

This typology was essentially a four-fold classification among the robber-nonrobber and daily-nondaily heroin users dimensions, except that the large cell of nonrobbers/nondaily heroin users (54% of all subjects) were subclassified into nonrobbers who were regular and irregular heroin users.

This classification permitted an assessment of how much more social impact occurs: among robbers who were daily versus nondaily heroin users (intensives vs. actives); among daily heroin users who were robbers versus nonrobbers (intensives vs. highs); among nonrobbers, according to the regularity of heroin use (highs, inactives, and lows).

Because of the way that the typology was formed, a linear trend should not anticipated. There was no prior reason to believe that actives would necessarily have higher criminal income (for example) than the highs or inactives. One of the key issues is whether a robber lifestyle or daily heroin use was more important in affecting measures of social impact.

Before presenting the main findings, however, information about the background characteristics of subjects classified in this typology will be presented in Table XV. 2.

### How similar were the backgrounds and prior criminal and drug histories of intensive criminals when compared with less active criminals?

Table XV. 2 shows that this Intensive Criminality Typology was related to only a few background and prior criminality/drug use variables.

### Background Characteristics

The Intensive Criminality Typology showed no variation with the background variables of ethnicity, education, and marital status. Data not presented showed no variation by type of typical employment and parent's occupation.

Moreover, while robbers were likely to be male, there was little variation by sex. Little variation by locale of recruitment was evident; although the highs (nonrobbers-daily heroin users) were more likely to come from Central Harlem.

Age showed interesting variation. The active criminals were more likely to be young (under 31), while the intensive criminals were older and had an age distribution similar to the "lows." Overall, however, persons in each category of this typology were very similar to each other.

Although intensive criminals and high criminals were defined by their current daily heroin use, these two groups of subjects generally did not report more heroin involvement than their nondaily counterparts during the life history interviews. Specifically, the Intensive Criminality Typology showed little variation (in the year prior to interview) in the percentage who claimed addiction to heroin, used over \$50 per day, or had ten or more years of heroin use. Moreover, although over four-fifths of the intensive and high criminals claimed daily heroin use in the year prior to interview, well over half of the actives and inactives, and 40% of the lows claimed similar levels of use as well. In short, the robbers-nondaily heroin users (highs) were as or more likely than the intensives to report having been a daily heroin users during the previous year.

In a similar fashion, although the active and intensive criminals were defined by their current robbery involvement, only about half claimed to do robbery in the prior year, while several (under 20%) of the lows, inactives, and high criminals claimed to do robbery previously.

Prior involvement in burglary was highly correlated with the Intensive Criminality Typology; two-thirds of the intensive criminals reported burglary compared with only 7% of the lows in the prior year. But no such variation was evident in prior reports of shoplifting.

### Social Impacts -218-

### Drug and Criminal Histories Prior to Interviews

rà.

### Social Impacts -219-

Criminal income reported for the previous year was not significantly related to the Intensive Criminality Typology. Data not presented here also showed that the Intensive Criminality Typology was not associated during the prior year with: shoplifting earnings, drug business earnings and frequency of drug business activity, estimated drug dealing income, drug expenditure levels, principal means of supporting drug use, and reported use of or addiction to alcohol, cocaine, or other drugs.

Thus, these data suggest that in the year prior to interview, the intensive criminals were most likely to engage in robbery and burglary, but the active criminals were almost as likely to do so. In other ways, however, the intensive criminals were not greatly different than their less active counterparts in heroin use, other forms of criminality, and background characteristics. In short, during the year prior to interview, intensive criminals appeared roughly similar to their less active counterparts in most kinds of behaviors, excepting robbery and burglary.

### How much more social impact did the intensive criminals have when compared with their less criminal counterparts?

Table XV. 3 presents data showing the relationship between the Intensive Criminality Typology and the seven measures of social impact as defined earlier. Several important findings emerged.

While intensive criminals appeared to commit about over twice as many nondrug crimes (245/year) as the "lows" (103), these differences were significant only at the .10 level. Likewise, while the intensives had almost three times as many total crimes as the lows (1,649 vs. 456), this difference was significant at the .08 level. Thus, in terms of the number of nondrug crimes and all offenses, the intensive criminals did not appear to engage in a significantly greater number of crimes than their street opiate using counterparts, although they were consistently higher.

And the second sec

When the severity of the type of offenses was calculated (using Wolfgang and Figlo's severity ratios), the intensive criminals were about three times more seriously involved than the lows (a significant difference) and almost half again as high as their closest counterparts, the actives and highs, although such differences were not significant.

For the four measures of criminal income and economic value, however, the intensive criminals were significantly different from the lows, inactives, and actives. In short, robbers-daily heroin users had substantially greater criminal income and economic value from crime than all street opiate users who did not use heroin daily.

Specifically, intensive criminals had four times as much criminal income (both from nondrug crime and all crimes) as the "lows" and about twice as much criminal income as the inactives and actives. Moreover, the economic value of nondrug crimes was approximately six times greater among intensive criminals (530,038) than among the lows (54,856) and almost two and a half times greater than among the inactives (513,217) and actives (511,902). Significantly, the intensive's economic values from nondrug crime were about two times greater than among the highs (nonrobbing-daily heroin users).

Daily heroin users who did not commit robbery (highs) had greater total criminal incomes and economic impacts when compared with the lows.

The data showed that daily heroin use has a <u>greater social impact</u> than robbery, especially for criminal income and economic value measures. When compared with their most similar counterparts, the intensive criminals were significantly different on only one measure (economic values from nondrug crime), than daily heroin users who did not commit robbery (i.e., the highs). On the other hand, intensive criminals were significantly different than their counterparts who robbed but were nondaily heroin users (i.e., the actives) on all measures of criminal income and economic values.

A.

Social Impacts -220-

### Social Impacts -221-

Likewise, daily heroin users who did not rob (i.e., the highs) were significantly different from nonrobbing-nondaily heroin users (i.e., the lows and inactives) on total criminal income and from the lows on the two measures of economic value. In contrast, nondaily heroin users who robbed (i.e., the actives) were not significantly different than their nonrobbing nondaily heroin using counterparts (i.e., the lows and inactives) on any measure.

In short, criminal incomes and economic values among daily heroin users was significantly greater than among nondaily heroin users, regardless of whether robbery was committed or not. Robbery had significant impact upon only one of these measures of criminal impact when daily heroin use was controlled. Thus, daily heroin use contributed substantially more to these social impact measures than did robbery.

### What proportion of the social impact were intensive criminals responsible for on an annual basis?

Table XV. 4 shows that these 201 street opiate users committed annually almost 33,000 nondrug crimes and about 210,000 total crimes (including drug distribution activities). They obtained \$1.15 million in nondrug criminal income and \$2.4 million in total criminal income. The economic values of their nondrug crimes was \$2.8 million and the total economic value of their illegal activity (including the "valued added" to the illegal distribution system) was almost \$7 million.

For each of these measures of social impact, the intensive criminals were disproportionately responsible. Although only 13% of the 201 subjects were intensive criminals, they accounted for:

- 20% of all nondrug crimes
- 21% of total crimes
- 23% of the severity index scores (i.e. all crimes standardized to . theft-equivalents)
- 25% of the nondrug criminal income
- 24% of the total criminal income
- 29% of the economic values associated with nondrug crime
- 25% of the total economic values

And Southern State 

Moreover, those who were daily heroin users (intensives and highs) constituted 30% of all street opiate users, but contributed about 40% of the nondrug crimes, total crimes, and severity index scores. Because they gained higher dollar returns and economic values, however, these daily heroin users gained about 50% of the nondrug criminal income, total criminal income, and economic values associated with such crime.

The nonrobber-irregular heroin users ("lows") constituted about a quarter of the sample, but contributed disproportionately few criminal offenses (about 15%) and gained relatively little (about 10%) criminal income or economic values. The inactives and actives, however, contributed about their proportionate share of criminal offenses, income, and economic values.

Darwin B. (black male, age 20) was a daily heroin user and robber-dealer. He dropped out of high school in his junior year, and has been working on and off in flower shops for the last three years. Although single, he has three children, aged 2, 3, and 4. He spent four years in youth institutions and reports more arrests than he can remember, though he does recall arrests for grand larceny, homicide, prostitution, and burglary. He claimed to have begun selling heroin at age 9, and using heroin at age eleven. He reported supporting his drug use mainly by selling drugs. He has never been treated for his drug use, though he claims to have been addicted to heroin at age 13 and to alcohol at age 14. He was living at a male friend's apartment when first recruited.

Darwin committed 4 robberies (totaling \$342) and one burglary (\$150) during his first 4 days in the study. He reported only one other nondrug crime, a burglary (\$210) during the remaining 29 days that he was interviewed. How then was he able to support his daily heroin use (and he used heroin every day), which averaged \$71 per day, during this period of noncriminality? Darwin reports the following for his first weekly reporting period.

I ran into an old girl friend who just got home from the army... I stayed with her all week. She gave me about \$80 (per day) to cop my drugs. She has a whole lot of money.

(PA

### Social Impacts -222-

Vignette V - An Intensive Criminal -- Diversity of Criminality

Social Impacts -223-

Vignette V (continued) - An Intensive Criminal -- Diversity of Criminality

Darwin's life took a new turn the following week.

I started working in my friend's candy store. He sells cigarettes, candy, pot, heroin and cocaine. I'm sort of like a watchman. I make sure the money is correct and no trouble starts. For this I get \$75 worth of heroin and \$25 worth of cocaine [per day]. I sell some of the heroin and spend maybe 12 or 16 hours a day there.... I sleep upstairs. They have a little office that I stay in. The girl I was with last week went back to the army. She gave me 575 before she left.

Darwin continued working at the store during his third week of interviews. His "salary" was upped to \$100 worth of heroin per day, but he ceased getting any cocaine. He used \$75 worth of the heroin each day and sold the remaining 25 dollars worth. Darwin stopped working at the candy store on day 3 of the final weekly interview. On that day he received his welfare check (\$130) and purchased \$100 worth of heroin which he consumed over three days. On day 5, with his cash and heroin supply exhausted, he broke into an apartment and stole a color TB, stereo, blender, toaster and iron. He sold all of these items to people in the street for a total of \$210 and, on day 6, purchased \$150 worth of heroin that he consumed on days six and seven.

Darwin was one of the subjects who helped establish the forementioned relationship between robbery and heroin use. He was the sixth most active robber in our sample, committing 4 robberies in 33 days, or an average of one robbery every 8.2 days. Yet he had no contact with police for these crimes. He used alcohol on 80% of his days and reported a mean of \$6 per day of alcohol use.

Yet, the above 33 day slice of Darwin's life suggested much complexity and variation in his criminality which would be obscured by computing means and rates. Darwin clearly was a high quantity, daily heroin user and a high quantity, near daily alcohol user, and committed serious crimes (robbery and burglary) and was hired because he clearly risked being assaulted or having to use physical force to make the money "correct" for the dealer.

But his criminality was far from continuous. There was one 4 day cluster containing 4 robberies and 1 burglary at the beginning of his reporting period and one burglary near the end of his reporting period. For one week his heroin use was financed by a girl friend (no crime when money was available). for three weeks it was supported through employment at a drug distribution outlet, and for a few days it was financed through public support. He also regularly sold a portion of the heroin that he received for being a watchman. Darwin clearly availed himself of whatever resources or opportunities existed and appeared to opt for predatory crime only in the absence of alternatives.

In this respect, Darwin exhibited a common pattern also shared by other intensive criminals or robber-dealers. But his persistence in obtaining (criminal income with which to purchase large amounts of drugs every day of the 33 day period, made him distinctly different than most regular and irregular users who avoided resorting to predatory crime and preferred to avoid activities which might lead to physical force or assault.



Do the most criminal have more contacts with the criminal justice system? Having established that intensive criminals have considerably greater social impact than their less criminal counterparts, we will now consider whether the criminal justice system has been relatively successful in identifying these persons and punishing them. The emphasis here was not upon whether they were convicted and punished for specific crimes. Rather, we have examined their self-reported<sup>\*</sup> arrest and incarceration histories (from the life history interviews) to determine whether those who were intensive, high, and active criminals during their reporting periods have more arrests and/or incarcerations than their less active counterparts.

The data in Table XV. 5 presented the relationship between the Intensive Criminality Typology and the number of arrests and years of incarceration. The data showed that intensive criminals almost always reported being arrested (only 5% were not), and that three-fifths report six or more arrests. The active criminals were considerably less likely (38%) and the high criminals somewhat less likely (50%) to report six or more arrests in their criminal histories. Nevertheless, the mean number of arrests did not show significant variation by this typology. Thus, while intensive criminals almost always have prior arrests and most have 6 or more arrests, their average number of arrests did not differ from the less active subjects.

In a similar fashion, the number of years of incarceration (jail plus prison) shows no significant association with the Intensive Criminality Typology. That is, intensive criminals appeared to be incarcerated about the same amount of time as their less criminal counterparts. The average subject

\*- Their self-reported arrest and incarceration histories may not be equivalent to their official criminal arrest and incarceration histories (which were not obtained in this study). In a careful examination of external validity of self-reported information, however, Chaiken and Chaiken (1982, Appendix B) show that jail and prison inmates were generally quite accurate (about 75% or more agreement) in their self-reports of arrests and criminal convictions when compared with official "rap sheet" and prison records. Moreover, discrepancies in self-reported arrest histories had relatively little effect upon their self-reports of criminal offending.

Social Impacts -224-

### Social Impacts -225-

reports under three years of incarceration during ten or more years of heroin use and probably 15-20 years since reaching age 16. If their criminal behavior in previous years was similar to that reported during their reporting period, they have spent relatively little time during their criminal careers in jail or prison.

### Summary

In this chapter, we developed a new typology of criminality. This Intensive Criminality Typology permitted an assessment of whether involvement in serious crime (i.e., robber versus nonrobber) was more important than heroin use (daily vs. nondaily) on a variety of social impact measures.

The data here clearly showed that the Intensive Criminality Typology:

- a) was not strongly associated with background characteristics (except that robbers were likely to be males);
- b) was not systematically related to self-reported heroin and drug use in the year prior to interview;
- c) was related to self-reports of robbery and burglary in the previous year;
- d) was not strongly related to a variety of other measures of drug distribution and criminal income in the previous year;
- e) was not related to the total number of arrests in lifetime,
- f) was not related to the number of years of incarceration,

In short, intensive criminals were not significantly different than their less criminal counterparts on most major background characteristics or drug use and criminal income during the year previous to interview.

Nevertheless, on a variety of measures of social impact, intensive criminals (and sometimes the highs) had significantly higher values than their less criminal counterparts. Although intensive criminals had a somewhat higher number of nondrug crimes, the association was barely significant. For almost all other measures, however, intensive criminals had three to six times greater social impact than the lows and had about twice as much criminal income and economic value as their inactive and active counterparts. While intensive criminals always had the highest mean values on these social impact measures, their values did differed significantly from the "highs" on only one measure (economic values of nondrug crimes).

In short, daily heroin users had a significantly greater social impact than nondaily heroin users regardless of robbery activity. But among daily heroin users, involvement in robbery boosted the economic values of their nondrug crimes even higher. Robbery was not an important contributor to high values on social impact measures when daily heroin use was controlled. All of these findings, as well as those in previous chapters, have important policy implications which are considered in the following chapter.

### BDJ;8274A;0745A;bj

Social Impacts -226-

Table XV. 1 - 1	Social I The Intensive Cri leroin User Typolog	mpacts -227- minality Typo Ny by Robbery	ology As Deriv	ed from the
Robbery Type	Her Irregular	oin User Typo Regular	logy Daily	Total
Percentage D	istribution of Resp	oondents in In	tensive Crimina	l Typology
Nonrobbers	Lows 24%	Inactives 30%	Highs 17%	72%
Robbers	Activ 15%	/es	Intensives 13%	28%
Total				100%
Robbery Type	Hero Irregular	in User Typolo Regular	ogy Daily	Total
Number of Nonrobbers	Respondents upon 49	Which Above Pe 60	ercentages are   35	Based 144
Robbers	12	18	27	57
Total	61	78	62	201
Perc	ent of Heroin User 20%	Types Who Com 23%	mitted Robbery 44%	28%
Percent	of Robber Types Cla	ssified among	Heroin User Ty	pes
Nonrobbers	34	42	24	100%
Robber	21	32	47	100%

CI	/pology, naracterist	Percentage fics and Pr	s Exhibitin ior Drug/Cr	ig Certain iminal His	Background tories.		
Robbery Type: Heroin User Type: No. Subjects:*	Intens Lows Nonrobber Irregular (49)	ive Crimin <u>Inactives</u> Nonrobber Regular ( 60)	al Offender Actives Robber Nondaily ( 30)	Typology <u>Highs</u> Nonrobber Daily (35)	<u>Intensives</u> Robber Daily ( 27)	Total (201)	þ (
Backgound Characteristics						<del></del>	<u></u>
% Male	69	67	87	74	93	75	.(
% Black	53	57	50	61	48	55	۶.
% Age 30 &	6	10	20	14	-		-
	U	12	30	14	/	13	.(
a from Central Harlem	33	27	27	57	33	34	.0
% Married	25	21	27	25	19	23	.1
% High School Dropout	40	33	41	46	48	40	,7
In Year Prior to Interview: % Used over \$50 of Heroin per Day	18	23	26	30	37	25	.5
% Addiction to Heroin	76	70	93	67	74	75	.1
% with Over 10 Years of Heroin Use	39	52	52	44	50	49	.7
% Reported Daily Heroin	40	56	72	91	84	63	.00
% with Robbery	14	20	46	5	53	24	.00
% with Burglary	7	23	41	32	68	29	.00
% Shoplifting Weekly or More	25	38	36	45	35	35	.98
Mean Criminal	2,384	7,662	6,785	5,834	9,742	6,041	.07

Due to missing information, the number of subjects upon which percentages are based may be less than this figure. \*

Table XV. 3	3 - Indices Typolog	Soci of Soc y	al Impacts cial Impa	s -229- ct by	the Intens	ive Crimi	nality	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Table XV. 4 -	Percenta	Social ge_of_Annu	Impacts alized	-230- Volume of	Social 1	Impact	Committed
Robber Typ Heroin User Typ N Subjects	I Lows be: Nonrobb e: Irregula ( 49)	ntensive <u>Inacti</u> er Nonrob ar Regula ( 60)	Criminal ( <u>ves</u> <u>Active</u> ber <u>Robber</u> r <u>Nonda</u> ( 30)	Offender <u>es Highs</u> r Nonro ily Daily ( 35	Typology <u>Intens</u> bber Robber Daily ) ( 27)	<u>ives</u> Total (201)	p of F	r	Robber Type: Heroin User Type: N Subjects:	Dy Inose In Lows Nonrobbe Irregula ( 49)	Classifie tensive Cr <u>Inactive</u> r Nonrobbe r Regular ( 60)	d in the iminal O: <u>s Active</u> r Robber Nondai ( 30)	Intensive ffender Typ <u>s Highs</u> Nonrobbe ly Daily (35)	Criminal pology <u>Intensi</u> r Robber Daily ( 27)	ity Type <u>ves</u> Total	ology. Annual Amounts <sup>a</sup>
Social Impact I 1) Annual Nondrug Criminal Offense Rate	ndices 103	161	168	180	245	163	.10	.18	Percentage of all Subjects Pe	24% rcentage [	30% Distributio Committed	15% on of Ani by Crim	17% nualized S inal Types	13% ocial Imp Above:	100% act Ind	(20 ices
2) Annual Total Criminal Offense Rate	456	1,310	766	1 ,208	1,649	1,048	.08	. 14	1) Annual Nondrug Criminal Offense Rate	16	30	15	19	20	100	(32,67
3) Nondrug Crime Severity Index	530	853	1,033	934	a 1,506	903	.02	.22	2) Annual Total Criminal Offense Rate	11	37	11	20	21	100	(210,68
4) Annual Nondrug Criminal Income	\$2,595	5,552	5 ,392	6,765	abc 10,841	5 ,729	.000	.35	3) Nondrug Crime Severity Index	14	28	17	18	23	100	(181,46
5) Annual Total Criminal Income	\$5,372	a 10,953	10,403	at 16,737	c abc 20,981	11,575	.000	.50	4) Annual Nondrug Criminal Income	11	29	14	21	25	100	(\$1,151,53
) Economic Value of Nondrug Crime	\$4 ,856	13,217	11 ,902	a 17,294	abcd 30,038	13 ,951	.000	.38	5) Annual Total Criminal Income	11	27	13	25	24	100	(\$2,363,21
) Total Economic Values	\$13,221	31 ,454 <sup>a</sup>	29,037	a 48,194	abc 65,049	33 ,921	.000	.46	6) Economic Value of Nondrug Crime	8	28	13	22	29	100	( \$2 ,804 ,31
significant inactives;	the Sche ly differe c -the acti	ffe test ent at t ves; and	of dif the .10 d - the	ferences level fr highs.	of group om: a -	means; th the lows	iis me ; b	an is - the	7) Total Economic Values	10	28	13	24	25	100	(\$6,818,12

a- Sum of the annual amounts committed by all 201 street opiate users; base number upon which percentages in the row were calculated.

sa.

<b>-</b>		Social Imp	acts -23	- 1				
Table XV.5 - Amo Per Sys	ng Subje centages tem.	cts Class Having Se	sified i elf-Report	n the I ted Conta	ntensive cts with	Criminal The Crim	Typol iinal Ju	ogy, stice
Robbery Type: Heroin User Type: No. Subjects:*	In <u>Lows</u> Nonrobber Irregular ( 49)	tensive Cr <u>Inactives</u> Nonrobber Regular ( 60)	iminal Of Actives Robber Nondaily ( 30)	ffender Ty <u>Highs</u> Nonrobbe Daily (35)	pology Intensi r Robber Daily ( 27)	<u>ves</u> Total (201)	p	
Prior to Interviews: Self-reported Number of Arrests None 1-5 6 +	9 59 32	25 29 46	14 48 38	25 25 50	5 33 61	16 40 43	.03	
Number of Years of Incarceration: None 1-4 5 +	41 41 18	40 42 19	41 41 17	39 35 26	19 46 33	16 41 21	.77	
Mean Number of Arrest	s 6.6	8.8	7.1	9.1	10.3	0.0	60	

8.2 .60 Mean Years of Incarceration 2.4 2.3 1.8 3.4 4.5 2.7 .11

9.1

10.3

Number of subjects upon which percentages are based may be less due to miss information.

This chapter will not attempt to summarize the central findings from each previous chapter (this was done in the Executive Summary). Rather, the emphasis will focus upon findings which emerged in several chapters and have special relevance for social policy.

This chapter provides policy implications which are understood to be a set of goals or objectives flowing naturally from the research findings. Treatment or criminal justice practitioners may be able to design programs, practices, or techniques to accomplish such objectives. These implications, however, are not "policy recommendations" which suggest specific steps about how to implement changes in existing arrangements. "Recommendations" assume an analysis of relevant institutions and practices which have not been the focus of this research. This chapter will use evidence from this study and other recent research to document a central major policy implication for American society:

### The most criminally active street opiate users are "slipping between the cracks" of the criminal justice and treatment systems.

The most criminally active persons in this study, the "intensive criminals" were defined as robbers and daily heroin users. Similar findings were reported by Ball, et al. (1981); Chaiken and Chaiken (1982); Chaiken. (1983); Moore, et al. (1981); Johnson, et al. (1983a). This study and these sources demonstrated that additional information about daily and/or high cost heroin use may assist in identifying high risk persons who currently slip between the cracks of the criminal justice and treatment systems. Such intensive criminals were rarely in methadone treatment and other drug treatment was rare. In short, intensive criminals almost alway "slip away" from or avoid treatment.

### Policy Implications -232-CHAPTER XVI

### SUMMARY AND POLICY IMPLICATIONS

### Policy Implications -233-

The criminal justice system largely ignores the drug use patterns of arrestees in making prosecutorial and sentencing decisions. Moreover, the prior arrest and incarceration histories did not distinguish intensive criminals from their less criminal counterparts. Thus, they appear to "slip through" the criminal justice system with jail and prison sentences which are typically short and generally not more severe than their less criminal counterparts. These points are more fully developed below.

Policy Implication A -- Daily heroin users who committed robbery were the most criminally active. These "intensive criminals" had very high crime rates, criminal severity scores, and economic values and should be a major focus of criminal justice agencies and social policies to address criminality among street opiate users.

Data in Chapter XIII showed that subjects classified as robbers who were daily heroin users had the highest rates and the most criminal income for most crimes. Moreover, within each offender type, the more regular the heroin use, the greater the crime rate and criminal income. These findings were directly parallel to those reported by Chaiken and Chaiken (1982a) and Chaiken (1983) among jail and prison inmates in California, Texas, and Michigan. Parallel findings were reported by Ball, et al. (1981); Ball (1982); Johnson, et al., (1983); Moore, et al. (1981).

In Chapter XV, data showed that the daily heroin user-robbers ("intensive criminals") had considerably more social impact than their less seriously involved counterparts on measures of criminal incomes, economic values, and criminal severity scores. Although they constituted only 13% of our street opiate users, they were responsible for 20% of the nondrug crimes, 29% of the nondrug crime economic values, and 23% of the criminal severity. In short, even among street opiate users -- all of whom were routinely criminal -- the intensive criminals exhibited greater social impact.

Policy Implication B - Intensive criminals report prior arrests and incarcerations that did not differ greatly from their less criminal counterparts; thus they will be difficult to systematically identify from current information maintained by the criminal justice system.

The data in Chapter XV showed that self-reported prior arrests and incarceration histories did not differentiate the intensive criminals from the less criminally active. Moreover, during their reporting periods, the intensive criminals were unlikely to experience arrest.

While this project did not gain access to official criminal records (so the respondent's reported arrests/incarceration history cannot be verified), Chaiken and Chaiken (1982a) report almost identical findings among jail and prison inmates. Specifically, they found that self-reported criminal activity did not correlate systematically with official criminal histories and showed that criminal history information would be highly inaccurate in identifying the most serious criminals.

Their review also found that information about drug histories in official records was absent in Michigan and Texas or highly inaccurate in California. Although official criminal arrest records were not obtained, New York State criminal justice official records ("rap sheets" or court records) do not contain systematic information about drug use patterns or referrals to drug treatment programs among arrestees.

Thus, available information suggests that the criminal justice system does not systematically collect or utilize information about arrestee drug use in making official processing decisions. Official criminal histories provided few clues by which to accurately identify intensive criminals or other high rate offenders among those who come to their attention. Until improvements can be made in detecting intensive criminals, they will continue to "slip through" the criminal justice system, being processed in the same way and receiving similar punishments to that of their less criminal counterparts.

Policy Implications -234-

### Policy Implications -235-Findings Regarding Criminal Behavior

These street opiate users were among the most criminally active ever studied. In addition to the studies cited in Chapter XI (Table XI. 1), findings from two national studies also help establish how deviant our respondents were. A search for the most criminal among a random sample of the nation's youth, aged 14-20 in 1979, showed that 1.3% of these youth committed three or more (FBI) index offenses and used heroin/cocaine three or more times during the previous twelve months (Johnson, et al., 1983b). Unpublished data from this survey indicate that only 4 subjects or 0.26% of the nation's youth committed 12 or more index offense and used cocaine/heroin 12 or more times in the past 12 months. Virtually all the street opiate users recruited in this study, committed more than 12 index offenses and used heroin or cocaine several times per month. Thus, these street opiate users were considerably more deviant in both crime and heroin/cocaine use than the most criminal 0.3% of all American youth.

Blumstein (1983) studied a sample of Washington, D.C. offenders arrested for an index crime other than larceny. He estimated that these individuals committed 10-15 index crimes per year. Our average street opiate users committed 25 robberies or burglaries per year plus an additional 67 larcenies. This does not include their frequent nonindex offenses: 61/year for victimless crimes, 665 drug distribution crimes, and over 200 minor crimes (Table XI. 2). Thus, the criminality of these street opiate users were significantly higher than subjects in other highly criminal samples.

# 10 NO weithin granter

# their criminality.

inappropriate for four major reasons:

First, half of these street opiate users were nonrobbers and nondaily heroin users. Most of these were primarily thieves and small scale drug distributors who avoided serious crimes like robbery, assault, and burglary. Although they committed a substantial number of offenses (16-30 nondrug crime/year, over 500 total crimes/year), their cash criminal income was low (\$2,500 to \$6,000 from nondrug crime/year, \$5,000 to \$11,000 from all crimes/year).

Second, even among subjects who committed robbery, the nondaily heroin users had substantially lower crime rates and income than the daily heroin users. This suggests that even robbers would decrease their criminality by 50% or more if they could be shifted to nondaily heroin use.

Third, the vast bulk of the crimes committed by these street opiate users were of minor importance. The average dollar income per nondrug crime was \$35. Even if this were multiplied by a fence factor of 2.5, the value (\$88) of the stolen merchandise would be relatively unimportant (Table VI. 5) by current criminal justice standards. Moreover, most nondrug crimes provided cash income of less than \$35. Very few crimes involve cash returns of over \$200 (or a projected merchandise value of over \$500 -- detailed data not presented).

Policy Implications -236-

Policy Implication C -- Despite their very high crime levels, however, a social policy of incarcerating street opiate users does not appear to be a socially appropriate or economically reasonable solution for

The findings presented in Chapter XIII and XV show extensive variablity in criminal offending rates and income according to the subject's criminal and

heroin use lifestyles. A standard policy of incarceration appears highly

β.A.

### Policy Implications -237-

Fourth, daily heroin users had the same low returns per nondrug crime as the less regular heroin users (Table VI. 5). Thus, information about the type and value of property crimes provides few clues for identifying serious offenders.

Ultimately, however, a policy of incarceration would likely prove more expensive. In 1983, the annual cost of maintaining one inmate in New York City was estimated to be \$21,000, including capital construction costs. (DSAS, 1983). A recent study ( )claimed that the annual cost of incarcerating one prisoner in Minnesota (which is probably less expensive than New York, was almost \$35,000.

Our average street opiate user, however, had an annual criminal cash income of only \$7,600 (Table VI. 7C). When the value of drug income and minor crimes was added, the total criminal income was slightly under \$12,000. (Table VIII. 3). Daily heroin users had about 1.5 times more and irregular heroin users had about half as much criminal income as the overall average. Thus, only among daily heroin users would their total criminal income (519,000) approach the cost of incarcerating them in jail or prison for a year.

Even if the economic values were considered (i.e., the value of stolen merchandise was included), the average street opiate user had nondrug criminal economic values of under \$14,000. Daily heroin users (\$22,844) imposed four times more nondrug criminal social costs than irregular heroin users (55,500). (Table XII. 1).

Thus, if the objective of an incarceration policy were to incapacitate errestees like our subjects so that they could not commit crimes which economically harm others\*, the above comparisons suggest that the annual costs (about \$21,000/year) of incarceration would be greater than the dollar value of nondrug crime actually inflicted by the regular (under \$14,000) and

irregular (under \$6,000) heroin users. Only if the daily heroin users could Only if the objective of an incarceration policy were to prevent both the

be located and incarcerated would the annual nondrug criminal social costs (\$22,500) exceed the incarceration costs, but only by a small margin.\*\* social costs of nondrug crime plus the indirect social costs associated with drug distribution (both cash + drug income and the value added to the illegal drug system) would the overall social costs exceed the incarceration costs by a large margia for both regular (\$32,000) and daily (\$55,272) heroin users. Even then, irregular heroin users have lower total social costs (under \$15,000).

Clearly, a policy of incapacitation via imprisonment does not appear economically reasonable since it would generally cost society more than the crime it was designed to prevent. A policy of probation with treatment as a requirement (residential drug free @ \$9,000/year, methadone ambulatory @\$2,300/year) arpears more economically reasonable for the less than daily users. Probation with mandatory treatment and routine urinalysis to detect frequent hard drug use should be designed to systematically pressure daily heroin users to reduce their heroin consumption by 50% or more; this would be likely to reduce their criminality by a similar degree. This would be a much more cost effective social response than incarceration. To be effective, probation/treatment officials need to employ the threat of incarceration for violations of conditions of probation.

respondent's cash income).

Policy Implications -238-

This study did not ask about assault and other noneconomic but serious crimes, so such criteria can not be considered here.

<sup>\*\*</sup> From an overall societal perspective, the economic values associated with street opiate users upon the victims of property crimes would be largely offset by the economic gain accruing to the purchasers of the stolen property. From this perspective, the economic value of stolen merchandise would remain unchanged, but be forcibly transferred from the victim to the purchaser; the purchaser's cost would be his payment to the street opiate user for his labor and the risk of being caught (our

# Policy Implications -239-Findings Regarding Drug Use Patterns of Street Opiate Users

These street opiate users exhibit diverse patterns of drug consumption. All subjects used drugs; they did so on 85% of the days (or 6 out of 7 days) on the average. Heroin and alcohol were consumed on half of the days, with cocaine and marijuana used on a quarter of the days. All other substances were used on less than 10% of the days (Table V. 2). Nevertheless, annual expenditures for any given substance did not exceed \$500, except for heroin and cocaine.

Policy Implication D -- Despite widespread patterns of multiple drug use among street opiate users, the drugs imposing major economic problems were heroin (primarily) and cocaine. Social policies designed to reduce by half the regularity or dollar amount of heroin (and/or cocaine) consumed, especially by daily users, would have substantial benefits for both society and these street opiate users.

Daily heroin users consume over \$13,000 worth of heroin and \$2,500 of cocaine annually (Table V. 5). Daily heroin users constituted 31% of these respondents but consumed almost two-thirds of all the heroin (Table IV. 1-2).

Street opiate users lived at or below the poverty level and typically expend less than \$5,000/year on food, shelter, and other nondrug expenses, an amount roughly equal to their noncriminal income. Thus, at the aggregate level, street opiate users consumed a dollar value of drugs which was roughly equivalent to their entire criminal income (both cash and drug) (Tables XI. 1-2).

If daily heroin users could be induced to reduce their consumption by half (to about three days per week or less), overall demand for heroin would greatly diminish and the amount of crime involved would probably be decreased by half or more. Thus, a "demand reduction" policy especially directed at current daily heroin users would have important benefits in reduced criminal offending and income.

Methadone treatment has been available for 15 years in Central and East Harlem. In late 1983, however, all programs were filled to capacity by street opiate users voluntarily seeking methadone treatment; waiting lists currently existed at most clinics in these neighborhoods (Newman, 1983; DSAS, 1983).

Despite such treatment efforts, however, open drug sales of heroin and cocaine in the street and from apartments was widespread during the five years of this study. This street scene gives no evidence of diminishing and appeared to be expanding (DSAS, 1983).

By recruiting heavily from among street hustlers, this study found that three-quarters of 201 street opiate users had no methadone treatment (although many were former clients) during their reporting period. Unfortunately, this study was unable to estimate the size of the population of out-of-treatment street opiate users in this area, so it was not possible to ascertain whether the proportion out of treatment was as high as indicated by our data. Estimates of the narcotic abuser population in New York City (DSAS, 1983), however, also suggested that about three-quarters of all active narcotics abusers were not in methadone or other treatment.

Thus, regardless of the actual figures, a substantial proportion of street opiate users were not enrolled and appeared to be avoiding treatment for their drug abuse. A parallel research project (the Tristate Ethnographic Project. Douglas S. Lipton, Principal Investigator) has documented the various "methadone folkways" (incorrect beliefs about methadone) among street opiate

Policy Implications -240-

Policy Implication E -- While additional capacity is needed for all persons voluntarily seeking methadone treatment, new social policies need to be designed having the objective of effectively identifying criminally active street opiate users and pressuring and monitoring them to reduce their heroin, other drug abuse, and criminal patterns. Such policies must be especially directed towards street opiate users currently avoiding methadone or other treatment.

### Policy Implications -241-

users both in methadone programs and those not enrolled (Goldsmith, et al., 1983; Hunt, et al., 1982, 1983abc). The patterns of treatment avoidance which were not well documented by the Economic Behavior data (the focus was not upon treatment) were more fully documented in several papers from the TRISEP project (Hunt, et al., 1983abc).

Thus, a very large proportion of street opiate users not currently in treatment will probably not respond to various inducements to enter methadone programs or other forms of treatment. Even if such persons did enter, they would be at high risk for departing against medical advice or being discharged for reason. The standard program requirement that potential clients must "voluntarily" seek and remain in treatment has the unintentional outcome of effectively excluding intensively criminal street opiate users who continue to "slip away" from systematic involvement in the drug treatment system.

This study cannot definitely answer the difficult question, "what then can be done with these highly criminal street opiate users?" Suggested directions implied by the data, however, involve closer coordination and cooperation between the criminal justice and treatment systems. New social policies and institutional arrangements need to be carefully developed so that more systematic, probably daily, pressure is placed upon criminally active street opiate users to detoxify and enter long term drug abuse treatment programs. Once arrested and convicted of crimes, including minor offenses, they should face manditory routine monitoring for drugs by urinalysis, and be required to attend treatment programs daily and report for long counseling sessions. Release to pretrial custody should require legal employment and consistent evidence of sustained drug treatment program participation. Policy Implications -242-Also emerging are a variety of innovative techniques which are not popular with street opiate users because they may prevent the desired "high." Long acting agonist chemotherapy (such as long acting methadone), the use of antagonists and other innovative techniques need to be tried with populations of street opiate users on probation or parole to ensure that they remain in treatment and opiate free. Especially among those who are convicted of crimes, correctional settings with therapeutic communities for these persons, once incarcerated, should be attempted. Work release with Naltrexone (an antagonist) should be tried. But whatever solutions are developed, the high criminality levels among street opiate users demand to be addressed more directly than during the five year study period of the Economic Behavior Project.

0

Appendix A - A Methodological History -243-APPENDIX A A METHODOLOGICAL HISTORY OF THE ECONOMIC BEHAVIOR OF STREET OPIATE USERS PROJECT

### Introduction

The development of methodologies for collecting systematic data from street heroin users was a major accomplishment of this research. When the project first began in 1978, there was major concern that such data could not be obtained. Even today, many scientists and laymen have a hard time believing that street opiate users can be systematically studied. While a careful description of the methods developed is not of central analytic interest, such information will be central to the credibility with which the analytic results will be viewed.

Thus, this lengthy Appendix provides important information about the methodologies employed by the Economic Behavior of Street Opiate Users project. These methods were emergent and were frequently modified during the course of the research. This appendix is written with three main goals in mind: (1) to fully describe our methodology so that substantive findings can be adequately interpreted; (2) to describe some of the difficult problems which confronted our research staff; (3) to provide a thorough account of our experiences so that future researchers wishing to employ similar techniques may understand both the opportunities and the difficulties of the methods employed herein. These descriptions include both our successes and failures in overcoming methodological obstacles.

The initial project plans intended to collect data on the totality of respondents' economic interactions, with a primary emphasis on the relationship between drug use and crime. Previous research in this area was subject to many limitations. Most existent data about drug use and crime have been derived from official statistics or self-reports by offenders and/or addicts obtained at some point during arrest, incarceration, or treatment processes, or in follow-up studies of such populations. Subjects in these studies usually provided retrospective self-reports about their criminal lifestyles and drug using behaviors spanning one month to several decades. Such extended recollections, while valuable for life history and many kinds of analyses, may be affected by faulty or selective recall. Further, self-reports given in an institution of social control (e.g., jail, treatment facility) may be perceived by subjects as influencing how they will be dealt with by that institution.

Institutionalized subjects might lie to interviewers for several instrumental reasons. Individuals entering methadone treatment may exaggerate the amount of drugs used on the street in the hope that they will receive more medication in the program. They may be afraid that the program will not give them sufficient medication to avert painful withdrawal symptoms. They may hope that maximum medication will get them "high," or they may wish to sell "extra" medication.

Official statistics issued by police or treatment agencies may only depict the tip of the iceberg and fail to reflect the nature and scope of the vast majority of crimes that go undetected, unreported, and unrecorded. Official crime statistics may be better indicators of law enforcement activity than of actual criminal behavior of arrestees (Chaiken and Chaiken, 1982).

Appendix A - A Methodological History -244-
Appendix A - A Methodological History -243-APPENDIX A A METHODOLOGICAL HISTORY OF THE ECONOMIC BEHAVIOR OF STREET OPIATE USERS PROJECT

#### Introduction

The development of methodologies for collecting systematic data from street heroin users was a major accomplishment of this research. When the project first began in 1978, there was major concern that such data could not be obtained. Even today, many scientists and laymen have a hard time believing that street opiate users can be systematically studied. While a careful description of the methods developed is not of central analytic interest, such information will be central to the credibility with which the analytic results will be viewed.

Thus, this lengthy Appendix provides important information about the methodologies employed by the Economic Behavior of Street Opiate Users project. These methods were emergent and were frequently modified during the course of the research. This appendix is written with three main goals in mind: (1) to fully describe our methodology so that substantive findings can be adequately interpreted; (2) to describe some of the difficult problems which confronted our research staff; (3) to provide a thorough account of our experiences so that future researchers wishing to employ similar techniques may understand both the opportunities and the difficulties of the methods employed herein. These descriptions include both our successes and failures in overcoming methodological obstacles.

The initial project plans intended to collect data on the totality of respondents' economic interactions, with a primary emphasis on the relationship between drug use and crime. Previous research in this area was subject to many limitations. Most existent data about drug use and crime have been derived from official statistics or self-reports by offenders and/or addicts obtained at some point during arrest, incarceration, or treatment processes, or in follow-up studies of such populations. Subjects in these studies usually provided retrospective self-reports about their criminal lifestyles and drug using behaviors spanning one month to several decades. Such extended recollections, while valuable for life history and many kinds of analyses, may be affected by faulty or selective recall. Further, self-reports given in an institution of social control (e.g., jail, treatment facility) may be perceived by subjects as influencing how they will be dealt with by that institution.

Institutionalized subjects might lie to interviewers for several instrumental reasons. Individuals entering methadone treatment may exaggerate the amount of drugs used on the street in the hope that they will receive more medication in the program. They may be afraid that the program will not give them sufficient medication to avert painful withdrawal symptoms. They may hope that maximum medication will get them "high," or they may wish to sell "extra" medication.

Official statistics issued by police or treatment agencies may only depict the tip of the iceberg and fail to reflect the nature and scope of the vast majority of crimes that go undetected, unreported, and unrecorded. Official crime statistics may be better indicators of law enforcement activity than of actual criminal behavior of arrestees (Chaiken and Chaiken, 1982).

3ª1

Appendix A - A Methodological History -244-

## Appendix A - A Methodological History -245-

Individuals under the jurisdiction of the criminal justice system (e.g., police, courts, corrections) may also conceal information from interviewers for instrumental reasons. Such persons may exaggerate or understate their drug use in the streets, depending upon whether "copping" a plea and accepting "diversion" to a treatment program is perceived as better or worse than spending time in jail. Arrestees may also overestimate their drug use as a "technique of neutralization," and absolve themselves of personal guilt or responsibility for a criminal act. They may conceal numerous crimes for which they have not been apprehended for obvious reasons.

Thus, a major methodological task of the Economic Behavior project was to develop techniques of data collection that would be as free as possible from shortcomings of previous research. The question, "how can hard-to-reach heroin addicts be involved in qualitative and quantitative research?" was one which had not been previously addressed with great success (Panel on Drug Use and Criminal Behavior, 1976).

Edward Preble had previously been involved in three major research projects, all involving heroin abusers as subjects. The most recent study (1975-1977) investigated the lifestyles of Irish and Italian heroin users in this community. During these years, Preble pioneered the development of the storefront methodology (see below) which was central to conducting systematic research among this street opiate user population. Preble had also developed a "right guy" reputation among many street opiate users and career criminals. He was perceived as a fair and knowledgable "outsider" by the street grapevine and was not a cop. He had also developed a network of prior subjects who were also friends; they could be easily recruited again and would be willing to approach others about cooperating in the research. In all respects, this research builds upon his contacts, experiences, and skill as an ethnographer prior to 1978 (see Appendix C).

**a** *o* - -

1 10 1

activities.

During the earliest stage of the project, in the first half of 1978, the Principal Investigators (Johnson and Preble) hired staff and located a storefront. A qualified project director, Paul Goldstein, was hired and assigned responsibilities for quantitative instrument development, data collection, coding, computerization, and data analysis. The field team under Preble's direction began with a skilled and experienced interviewer (Tom Miller), but lacked a field worker who was well known and respected in the East Harlem study area, especially among opiate users and other street hustlers who were to be the central focus of this research. Efforts were made to locate an exaddict and/or former criminal offender who would have major responsibility for recruiting subjects, and entering social systems that would be initially closed to professionals. Finally in the summer of 1978, Preble located a person with the very background needed. This field worker had extensive involvement in heroin during the late 1960s and early 1970s, had a "right-guy" reputation among both the addict and nonaddict criminal subcultures, and several years of incarceration. He was no longer using heroin and not involved in crime. During the course of the five years of research, several other former heroin abusers were recruited to assist in locating respondents, and in some cases, to conduct interviews (see Appendix B for project experiences with such indigenous staff).

Appendix A - A Methodological History -246-

From 1954 (and until his death in 1982), Edward Preble was intensively involved in the East Harlem and Yorkville low income communities as a community center director, friend of many residents, godfather to the children of research subjects, and companion in many community leisure time

## A. Initiation of Field Work



Appendix A - A Methodological History -247-B. The Storefront as a Place for Routine Data Collection

In the original application of 1977, different potential techniques for obtaining routine economic behavior data from respondents were planned. These included ethnographic techniques, informal conversations, formal interviews in the street or apartments, and formal interviews in the field office. The formal interview at a field office--the "storefront"--proved to be an efficient way of obtaining high quality economic behavior data from respondents. Other techniques were tried but found difficult to implement.

The technique of following research respondents around and observing their behavior, has disadvantages. Field workers may become witnesses to crimes. Victims of these crimes may believe the field worker was a participant and situations of legal complexity are likely to arise. Subjects may be deterred from their daily routines by the presence of the field worker. One armed robber commented that he would be unable to concentrate sufficiently on his activity if he knew he was being observed and that such a failure of concentration could be fatal. The main disadvantage of direct field observation techniques, however, was that it took an enormous amount of field time. Most of the subject's time was "dead time" spent standing around and talking. Nevertheless, many respondents have been observed engaging in their typical activities (see validity and reliability discussion below).

Informal conversations between the field worker and the respondent also proved to be of little value. Such informal conversations have a tendency to be too unfocused, range over too wide a variety of topics, and to lack specificity. Crawford (1977) found "formal interviews with addicts on the street.... to be necessary..." for similar reasons. Moreover, the data to be collected required a level of specificity that made it impossible for field workers to memorize all the questions and answers.

Appendix A - A Methodological History -248-The formal interview in the street or neighborhood suffers from the fact that addicts are difficult to locate at any given time (also see Crawford, 1977). They may be around the neighborhood taking care of business, but yet be quite difficult to locate at any specific time. It is inefficient at best to have field workers waiting around on the streets for subjects in order to conduct formal interviews -- after which a private place to conduct the interview must be found.

The data collection technique that proved most effective was to locate the field staff in a field office that blended naturally into the study neighborhood. During five years of research, the project rented three storefronts in the midst of major copping blocks (see detailed descriptions in Appendix B). In short the subjects felt right at home in the storefront because it reflected their living situation in a ghetto neighborhood. The major difficulties with using a neighborhood storefront as an

interview site were: 1) the delapidated condition of such space (e.g., roaches, poor plumbing, rats), 2) congestion of respondents wanting to be interviewed during the morning hours, and 3) problems associated with staffing and supervision of the nonprofessional staff. These difficulties are specified in more detail in Appendix B. Nevertheless, the advantages of conducting a confidential and structured interview to obtain detailed information about drug use, crime, income, and expenditures in the project storefront vastly outweighed the disadvantages.

C. Recruitment of Subjects Locating research subjects was among the easiest tasks which this research faced, especially since respondent payments were available. As Preble (1980b) has indicated, among street opiate users the need for money is continuous and,

## Appendix A - A Methodological History -249-

hence, they are eager to cooperate in most forms of research. But what initially appeared to be a mean motive, the need for money, was transformed into a fruitful research relationship through skillful rapport building and interviewing. The respondents quickly provided high quality information about their activity in return for a modest payment (\$5 or \$10) for 30 minutes to 2 hours of their time. Given the quality of information obtained from these respondents, the interview payments were clearly a major bargain.

### 1. Drop-in Respondents

During the pilot years (1978-79), snowball selection techniques encouraged field workers to spread the word about our study in the street. Current subjects brought in friends. We accepted most respondents who met basic criteria such as having a history of heroin or methadone use. While such subjects were convenient and useful for pilot testing of instrumentation (see section E below), the staff came to believe that such drop-in respondents were a relatively poor reflection of the street opiate-using population. A certain type of subject began to predominate in the pilot study; namely, older Black methadone clients who were generally heavy consumers of alcohol. These drop-in subjects were relatively less active in heroin, other illicit drugs, and crime than persons specifically recruited for their activity. Drop-in subjects tended to be heavy alcohol consumers, street "touts" or "jugglers", and small item shoplifters. While these respondents claimed to be addicts, provided lengthy histories of heroin use, and were viewed by their neighbors and other street users as "addicts," their usual activities raised important questions about whether they were physically dependent on opiates and about the nature of their dependency.

harassing behavior.

cocaine, etc.).

Appendix A - A Methodological History -250-They were aptly described by Preble and Miller (1977) as living on "Methadone, Wine, and Welfare." These respondents were almost always unemployed and did little during the day except hang around and try to raise money for a few bottles of cheap wine which they drank continuously during the day. They would occasionally engage in minor crimes (shoplifting, selling loose joints, steering customers to dealers) and use heroin or cocaine on an occasional basis when they had money or had worked for a drug dealer. Thus, the interview fees constituted a real windfall for them. Since they typically shared bottles of wine, such subjects could bring in many "friends."

Besides having the potential for seriously biasing results, the influx of the older "methadone-wine-and-welfare" crowd led to additional difficulties. More active street hustlers and opiate users were disgusted to be seen with such types. Senior staff began to get feedback from field workers that some potential subjects were reluctant to come into the field office during 1979 because it had become a hang-out for the "lowest of the low." In addition, such methadone "winos" were undeterrable "moochers." They would come to the storefront consistently, perhaps five or six times in a single day, barged right into an interview in progress, and demanded a few dollars for "carfare" or something to eat. No amount of diplomacy was able to forestall this harassing behavior.

Once the field office was well established, persons appeared at the storefront almost every day asking to become research subjects. Many interruptions were endured and considerable time was expended in politely refusing such requests while thanking them for dropping by. Nevertheless, some of the drop-ins were accepted as research subjects because they exhibited some interesting characteristics (i.e., had some employment, used heroin and

### Appendix A - A Methodological History -251-

In sections H and I (below) we describe efforts which project staff made to develop a quasi-scientific sampling frame of street opiate users. For reasons outlined there, we were not able to develop a list of potential respondents from which to sample. Instead, a variety of other recruitment techniques were followed during the main study (1980-1982) and are described below. In the East Harlem and Central Harlem areas of Manhattan, however, street opiate users could be easily located and quickly involved in a routine course of research. They provided valuable and highly detailed information that was analyzed in previous chapters.

### 2. Selective Recruitment

A more valuable technique for recruiting respondents began with locating one or two exaddicts who were part of large social networks containing street opiate users and criminals. Exaddicts were sought who were respected by their peers and had a reputation as a "right guy." Such reformed drug abusers/ex-felons also helped the research effort by "watching the backs" of the professional staff in the storefront. That is, their presence in and around the storefront deterred others from "ripping off" staff and prevented violence in the storefront. Nevertheless, supervision of such exaddict staff was another problem (see Appendix B).

Several early subjects fit into this role and recruited most of the subjects for this study. But it took time and persistence to find such persons and to separate those who claimed to have good reputations and contacts from those who really have them. Such persons had prior associations with the relatively active opiate users and other criminals and could go to virtually any neighborhood to locate heroin users and to recruit them as subjects. Appendix A -They also physically walk interview and, if necessary, interviews. Such key field wo establishing good research re exhibiting diverse lifestyles. Another fruitful technique ask respondents, especially t friends or drug-using associ physically escort their friend

201

and the second s

During 1980-1982, the staff refused to accept many "drop-in" subjects for reasons given above. Instead, we relied heavily upon the judgments of exaddict/exoffender staff who were sent into the streets to locate subjects. These field workers sometimes approached unknown persons on the street, but more frequently found a previous acquaintance who introduced them to other potential respondents. In order to ensure a diversity of lifestyles among respondents, these staff were directed to go to different geographical areas in the study community with directions to avoid bringing in too many "low lifes," and to concentrate on finding subjects who did robbery, burglary, or frequent thefts. They were also expected to bring in about one female for every two or three males (our estimate of the probable sex ratio). Bernacki and Waldorf (1979) make similar use of such selective snowball sampling techniques. A substantial proportion of those approached (the percentage was not systematically obtained) ended up as research respondents. \*

\* - During the earlier part of 1980 when the staff tried to do a block census of heroin users, the names of potential subjects were identified and randomly selected; a few refused to meet the field worker or any research staff; their friends could not persuade them to come to the storefront. Yet if the researchers had persisted in pursuing such "hard-to-reach" subjects by standard tracking techniques, the street norms--that a person should not be pursued unless willing to be--would be violated.

Appendix A - A Methodological History -252-

They also physically walked respondents to the storefront for the first interview and, if necessary, located them at later times for additional interviews. Such key field workers were critical in locating, recruiting, and establishing good research relationships with desirable types of respondents exhibiting diverse lifestyles.

Another fruitful technique for recruiting heavily involved subjects was to ask respondents, especially the more active and successful, to refer their friends or drug-using associates. Typically such respondents needed to physically escort their friends to the storefront at which time the subject was given a small payment for his referral.

Appendix A - A Methodological History -253-3. Maintaining Contact with Respondents

In order to sustain initial and maintain long-term contact with respondents, the first week was crucial. At the beginning of the pilot study, staff decided that repeated interviews with respondents on consecutive days were critical for obtaining the necessary information (the rationale is given in section E below). Conducting short interviews with the respondent on five successive days was important for maintaining short-term and long-term cooperation in several respects.

First, respondents who were interested only in the money, who appeared to stretch the truth about their involvements, generally did not reappear after the first or second interviews. If their stories initially seem untrue, the fieldworker could check the validity of their reports with other users or other sources and report back. Interviewers carefully probed about questionnable incidents. In most cases, however, the initially deceptive respondent either departed voluntarily or began to give the kinds of information needed by the researchers with little evidence of exaggerating or understating.

Repeated interviews during the first week also provided time to build rapport and to gain knowledge about each subject. The interviewer and respondent came to know each other as individuals and developed a sense of trust, respect, and liking for one another. The respondent became convinced that this really was a serious research effort, that the interviewers and other staff were "right," that he would not be harmed by his involvement, and that he was being paid fairly for his time (and the same as other respondents). This rapport between the staff and the respondent grew rapidly during three to five interviews, and provided the basis for all subsequent data collection efforts. During this time, the respondent also learned about

Appendix A - A Methodological History -254the kinds of information being sought and paid somewhat more attention to the details of their income, expenditures, crime, and drug use for future interviews.

During the initial interviews, the respondent frequently indicated something about his typical behavior patterns and locales of activity. Such clues helped the field worker locate subjects at a later time (one week, several months, or even a year later). If the respondent did not reappear for a scheduled interview, the field worker would visit the respondent's last known hang out, block of residence, or known associates. He initiated conversations with persons in the neighborhood to locate individual subjects, if they were still in the neighborhood (most were) or to learn their whereabouts (dead, out-of-town, etc.) if not.

The trust established during the initial interviewing period frequently continued for long periods. Former respondents frequently dropped by the storefront just to say hello and to rap with staff about their activity (this was true even a year after interviewing ended) -- and to inquire whether we wanted them back on any new project in the near future. Thus, most of the persons interviewed for one week completed 33 consecutive days of interviews. and many were found at a later time for other interviews. Clearly, the initial effort at making contact, and establishing a firm research relationship paid major dividends (valid and reliable data, consistent future reporting, and ease of maintaining a long term relationship) with the street opiate users being studied in this project.

D. Protecting Confidentiality The Economic Behavior Project procedures for obtaining informed consent and protecting human rights was approved by the Institutional Review Board and carefully followed by staff. These procedures are briefly outlined below.

Appendix A - A Methodological History -255-1. The Informed Consent Procedure.

Upon walking into the storefront, the field worker would generally introduce the potiential respondent to a staff member/interviewer. The respondent was free to give no name or use a street name or other nickname. If the subject seemed uncomfortable at first, staff would briefly talk with him about other people they might know in common or about some other topic which seemed to be of interest.

Usually, the respondent would be invived to an interview cubicle in the back of the storefront. The interviewer would go through the informed consent procedure. This involved explaining that participation was voluntary, who the funding agency and sponsors of the research were, that data would be used only for research purposes, that the researchers were protected so information could not be used for legal purposes, and that staff would answer any other questions the respondent might have (this usually involved money). As the interviewer explained each element of the informed consent statement, he marked a check list of items covered. At the end of this explanation, the respondent was asked if he gave his informed and voluntary consent to participate in the research. All respondents agreed to participate. Afterwards the interviewer explained that each respondent was assigned a code number and code name. The interviewer then told the respondent his code name/number and asked him to remember it. In addition, the interviewer entered the code number and code name at the top and at the bottom of the informed consent check list. At the bottom of the informed consent check list was a tear off section on which the code number, code name, and the respondent's real name (his signature) was entered. At the end of the interview, this tear off sheet was detached from the informed consent check list and placed in a sealed envelope and forwarded to the research offices at the World Trade Center. The respondent's actual identity was kept in a locked drawer separate from the other research files.

Appendix A - A Methodological History -256-Immediately after completing the informed consent procedures, the interviewer began either the life history or the first daily interview (see section E. below). As interviewing got underway in 1978, however, the research staff found that subjects were much less restless with the daily interview schedule than with the life history schedule as a first interview. The informed consent procedure was first given, followed directly by the first daily interview. The subject returned for daily interviews on subsequent days. After two or three interviews and as rapport between the field worker and respondent developed, the lengthy life history interview could be scheduled for a convenient time. The result was better quality data and sustained respondent involvement in the research.

In order to ensure that the researchers were protected from any attempts by police or courts to obtain data from this research project, the Principal Investigators (Johnson and Preble) applied for and received a Federal Certificate of Confidentiality. This insured that all employees of the project and all project documents were protected from subpoena in civil or criminal court actions. The certificate also stated that researchers assumed a responsibility not to divulge confidential material. The protections accorded to staff members and project documents were explained to all subjects during their initial visit to the storefront. A copy of this certificate was on hand at the storefront to show respondents that such legal protection had actually been obtained.

### 2. Federal Certificate of Confidentiality

Appendix A - A Methodological History -257-3. Maintaining Confidentiality

Prospective subjects were assured that all information would be held strictly confidential. Most subjects were inclined to believe us because they knew and trusted the fieldworkers who had recruited them for the study. Moreover, the longer that the project was situated in the neighborhood, the more evident it became that nobody was getting arrested or into any other trouble because they had talked to us. Word of mouth on the street then gave credence to our assertions of confidentiality.

Staff members were trained not to discuss specific incidents reported by subjects with anyone, including other subjects. Respondents frequently gave accounts that not only could lead to their arrest but could lead to their getting assaulted or killed on the streets. Such accounts concerned "rip-offs" in drug transactions, burglaries of drug dealers' apartments, muggings of neighborhood residents, etc.

Each respondent was assigned a code name and code number by which he or she was known to project staff. Staff members were trained to think of subjects in terms of their code names, lest in discussing findings they accidently let slip enough information to connect a specific individual to a specific event. The process of selecting code names became increasingly difficult as the project exhausted the more common names. Field staff began to assign code names from the worlds of art (e.g., Modigliani), ancient history (e.g., Nostradamus), sports (e.g., Norton), and movies (e.g., Nocturna). Occasionally field staff manifested a perverse sense of humor, as when three consecutive subjects were dubbed Ubie, Dubie and Umie.

Confidentiality procedures created some problems for follow-up research. During the main data collection years (1980-1982 -- see section F below), respondents were initially interviewed for five consecutive days, then for

S States

However, some subjects were lost to contact after completing a phase of the research cycle. Much of the time field staff knew what had happened to these persons: some had been hospitalized or were in residential treatment programs; others were in prison/jail, had been killed, or left the area. Some subjects just disappeared and nobody on the street knew where they had gone or what had happened to them. In several cases (especially at the beginning of the study), however, our confidentiality procedures were so effective that field staff were no longer sure just whom they should be locating. After seeing many respondents, and thinking of them by fictitious (code) names, field workers forgot the real identities of a few subjects.

Staff members also consciously avoided any contact with law enforcement personnel in the study area for several reasons. We did not wish to give respondents any basis for believing that we might be collaborating with the police. Police officers are skilled at extracting information and there was fear that staff might unwittingly divulge something that could compromise a subject. Finally, staff members who had worked with law enforcement agencies on previous occasions believed that if we were to ask the police for any help (e.g., to locate a subject) that at some time in the future that favor might have to be reciprocated, an eventuality that we wished to avoid. One contact that staff had with police occurred when several staff members

and respondents were standing outside the storefront. A police car drove

Appendix A - A Methodological History -258four consecutive weeks, then "cut loose" for three or more months, then interviewed for another four consecutive weeks, "cut loose" for another three or more months, again interviewed for four consecutive weeks, and so on. Most subjects were back long before their three month hiatus expired. In fact, many subjects maintained a steady contact with the storefront staff regardless of whether they were being currently interviewed.

# Appendix A - A Methodological History -259-

slowly up the street and stopped in front of the storefront. One of the officers yelled out a question, "What's going on here?" One of the researchers yelled back, "Research." The officer said "Oh," and drove on. Several subjects snickered about how the police seemed afraid to get out of their vehicle. The researchers, by comparison, mingled well on the streets and enjoyed a very different sort of reputation among neighborhood residents.

Another issue concerned prior knowledge of a crime event. Researchers were committed to protecting the confidentiality of disclosures about past events. But a very different ethical question (discussed with the Institutional Review Board) would be presented if researchers gained knowledge about a crime not yet perpetrated, especially if that crime had potential for serious violence. [Staff were directed to ask only about completed crimes which had occurred in the past week or past day; they were instructed not to ask about future crimes.] Fortunately, this issue presented itself on only one occasion. The manner in which it happened illustrated the trust held by subjects toward field staff--as well as the respondent's determination to carry out such plans.

A respondent had completed his daily interview for the day. A few hours later he burst into the storefront in a state of hyperexcitement and, not even waiting to be acknowledged, yelled that he and some friends were on their way to commit an armed robbery, that they might be killed, that he wasn't sure if he would appear for his interview tomorrow because he might be dead or in jail. He wanted us to know so we wouldn't think he was irresponsible if he didn't show up tomorrow. With that, he ran out the door leaving everyone sitting there in a stunned silence. Field staff decided they had too little information to do anything even if they wanted to. Moreover, they decided they

probably should not do anything given the project's basic orientation toward protection of subjects. The subject returned the next day; the robbery attempt never materialized to the great relief of the researchers.

Thus, the Economic Behavior Project developed and followed procedures to protect the confidentiality of all respondents, even from each other. Following these procedures and acting consistently increased trust and rapport with subjects. To the knowledge of our staff, there has been no violation of any subject's identity nor did any harm come to them as a result of their participation in this project -- even though they reported many different crimes, some quite serious.

This trust made it possible to ask very detailed questions about the respondent's criminal and drug using activities and many other aspects of their economic lives, a topic to which we now turn.

E. Developing Instrumentation and Coding One of the greatest challenges which this research encountered was developing instruments (interview schedules and coding categories) by which the complexity of respondent behavior could be captured and measured in a systematic quantitative fashion. Despite some of the obstacles and problems outlined below, this research did develop instrumentation which can accurately assess much of the complexity of respondent lifestyles.

In any research about complex behaviors, the investigators must make crucial decisions about what to include and what not to include in data collection instruments. At the beginning of the research two fundamental decisions were made.

Appendix A - A Methodological History -260-

#### 1. Initial Decisions

# Appendix A - A Methodological History -261-

First, since this was a study about the economic behavior of street opiate users, we decided that data collection instruments would include only the kinds of crimes where an economic value (i.e., money, drugs, or goods) could be obtained or estimated. All other crimes which could not provide an economic benefit to the respondent were excluded. In short, respondents were not asked questions about assault (aggravated and simple), rape, homicide, arson, weapons possession, vandalism, disorderly conduct, property damage, littering, etc.

Given recent developments in the drug/crime literature showing that heroin users have among the highest rates of violence and assault (Chaiken and Chaiken, 1982; Johnson, et al., 1983), the omission of assault questions was particularly unfortunate since it would have allowed subjects to be classified on this important dimension of serious criminal behavior. In addition, these subjects can frequently be observed engaging in public nuisance crimes (defacing property, dropping wine bottles and soda containers on the sidewalk, throwing bottles into the street, conversing loudly at late hours of the night, etc.). If such minor offenses are considered as crimes, then the crime rates reported in earlier chapters should be considered very conservative.

Second, the investigators had originally hoped to interview respondents about where crimes occurred, types of goods involved, estimated retail or cash value, etc. As drafts of the daily interview instrument were developed, however, it became apparent that an instrument obtaining such details about all criminal or drug use episodes would be hopelessly long and complicated. Rather, a decision was made that the respondent would be asked to sum across all crime and drug-related episodes of a given kind for a single day, and report the total dollar income. This meant that the smallest

\* - This was clearly a correct decision; after completing data collection on the Economic Behavior Project, Bruce Johnson and Eric Wish have continued research at the storefront into the details about one crime event and specific details about drug and alcohol consumption shortly before and after such an event. This interview takes 1-3 hours to complete.

Appendix A - A Methodological History -262-During the course of five years of research, two major data collection

unit of analysis was the person-day (see below section F). This decision meant that details about crime events would not be systematically obtained for each event. Rather, to the extent possible, the interviewer would write a brief scenario of the crime on the back side (the 'flip side') of the interview form. These decisions, while limiting the research in some ways, helped maintain the emphasis upon the economic aspects of our subjects' lives. instruments have been developed. The experience with the daily data collection instrument during the pilot years (1978-9) led to the development of a weekly interview schedule. To provide information about each respondent's background prior to participation in this project, Life History Instruments (both long and short form) were developed. The following sections describes each of these instruments and what project staff learned during the development of each.

## 2. Daily Interview Instrument (1978-79)

and the second

- 1866

Interviews with respondents were focused upon their economic behavior during a particular 24 hour period, called the person-day. From the beginning, the person-day was a key unit of analysis. The central question was how to best obtain detailed quantitative data about the crimes, drug use/purchase/sale, income, and expendures during each reporting day. Interviews about economic activities on specific days were planned as the core of the quantitative data collection effort; they provide the data upon which this report is based.

During this project, the field workers arrived at the field office between 9 and 9:30 in the morning. The peak period for subjects to be interviewed was between 10:00 A.M. and 12:30 P.M. During 1978-79, subjects reported to the field office for 30 or more consecutive days after giving their informed consent. Respondents were interviewed Monday through Friday.

Appendix A - A Methodological History -263-On Monday, they were interviewed about their activities on Friday, Saturday, and Sunday for which they were paid \$10. On Tuesday through Friday the respondents reported on their behavior for the preceding day and were paid \$5 for the daily interview. That is, they were paid \$30 for a week of information.

The first draft of the daily interview form in early 1978 had room on the back side of each page for the interviewer to write comments or record respondent's accounts of criminal activity or drug use episodes. These written accounts on the "flip side" became critical to understanding complex events and for coding the dollar amounts involved. In fact, most respondents provided a description of crime events and drug use patterns to the interviewer. Based upon these descriptions, the interviewer summed and coded such data for each person-day. These written accounts raised many problems for coding as well (see section E. 4 below).

The earliest version of the daily interview form quickly encountered a major conceptual problem. This form included one line which asked for information about the number and dollar returns from "drug sales/dealing" among all other kinds of nondrug crimes. Our early respondents were confused because they did not directly sell drugs to customers for cash (which they defined as "dealing"). But they reported being active in distributing drugs and obtained drugs (mainly heroin and cocaine) for such efforts. In short, we uncovered several roles in the drug distribution system which have not been well-described previously nor well-documented in terms of numbers of persons involved, frequency, or economic returns.

Such respondents engaged in low level drug transactions (steering, touting, copping, juggling -- all defined in Chapter VII) that may be viewed as part of a quasi-barter economy (Goldstein, 1981). For such services, the drug seller or buyer pays the respondent in cash or drugs or both. Such "drug" payments may be shared by the respondent with other drug users.

In addition, there was no standard economic value for all drug transactions, so some dollar value must be assigned to drugs received. The complexity of drug transactions constantly challenged the best data recording and coding schemes.

A similar problem emerged when we tried to find how much respondents spent for food and shelter. Many subjects report no or very minimal expenditures for food and a place to live. But they did not look famished or exhausted. They described how they routinely slept at the apartment of a girlfriend, spouse, mother, or other friend who usually provided them with a meal for the day. Yet they paid little or nothing to such persons for their shelter and food. In addition, these subjects reported daily interactions involving the exchange of services or favors for which no cash or drugs were received; but a clear expectation existed that a similar favor would be returned in the future. Thus, assigning economic values to goods or services was difficult where bargaining or noncash exchanges occurred. Cash payments for drugs were easier to monitor.

After these problems emerged, the daily interview instrument was revised and used for collecting about 30 consecutive person days of data from 51 subjects (1530 days total). Several preliminary papers were generated from these data. (Johnson, 1979, 1981ab; Johnson and Schmiedler, 1981; Goldstein and Duchaine, 1979; Goldstein, 1979, 1981, 1982ab; Preble, 1979, 1980ab). Experience with the daily interview showed that many subjects who were not very active in criminality and/or drug use completed the daily interview rapidly (5-10 minutes). Moreover, the quality of data provided for three different days (Friday, Saturday, and Sunday) during an interview on Monday morning was about as good as the data collected day-by-day. Thus, during the last half of 1979, staff developed a weekly interview form which became the

major data collection instrument for the main study (1980-1982) and is described below.

Appendix A - A Methodological History -264-

## Appendix A - A Methodological History -265-3. The Weekly Interview Schedule

In order to reduce the costs associated with each daily interview. predominately respondent payments (\$30/week) and field worker time (about \$8-10/hour), as well as pressure by respondents to conduct many interviews rapidly, the Project Director (Goldstein) developed a weekly interview schedule that collected almost all the information given on the daily interview but obtained seven person-days of data in one interview. This weekly interview took from 20 to 40 minutes to complete, depending upon how active the respondent was and his ability to recall his economic behavior during the past seven days.

The respondents could usually provide economic data about each day so that their income and expenditures balanced fairly accurately. Since these respondents lived very much in the present, usually dealt in cash or drugs (almost never checks or other negotiable instruments), and kept no records to which they could refer, they might forget miscellaneous expenditures that involve relatively small sums of money (e.g., loans from friend; family; expenditures for alcohol, cigarettes, candy, soda, newspaper, etc.). Respondents also sometimes found it difficult to recall the economic value of drugs (especially marijuana and alcohol) which was shared with friends, as well as other in-kind income or expenditures.

For some high income and high spending respondents, the amount of cash flowing through their hands comes and goes so quickly that large sums may go unreported or could not be recalled -- even though they make a good faith effort to be accurate. Thus, while respondent memories do not provide a perfect balance of income and expenditures on each day, the economic data presented in this report are sufficiently accurate for a sociological analysis of crime and drug use. The recall problems involved in moving from daily to weekly interviews, were relatively small when compared with most previous studies that have asked respondents to recall behaviors spanning months, years, and even decades.

Appendix A - A Methodological History -266-The weekly interviews were a cost effective means of collecting valuable data and obtaining a good picture of the respondent's activity for the week. The interviewer, guiding the respondent to balance income and expenditures, caused the respondent to recall other events occuring in the same time period. He might initially recall committing three shoplifting episodes during the week, but could not recall which episode occurred on which day. Later questions about what drugs or other expenditures occurred on a specific day aided him in recalling which shoplifting episode occurred on a given day. Nevertheless, if the respondent did not recall on which day a specific activity occurred, the interviewer assigned that event and the dollar amount to a particular day after probing the subject to make a best estimate. Thus, a further source of inaccuracy in balancing income and expenditures was that the income or expenditures may have occurred at some time other than the date specified in the interview form, even though the event occurred at some time during the week.

Most respondents were now in their thirties and had over a decade of experience as street opiate users behind them. Although the main focus of the Economic Behavior Project was upon their current patterns of behavior, their backgrounds might also have an important influence upon their current behavior. The project staff developed a relatively lengthy open ended life history interview (LHI) schedule which obtained information about demographic characteristics (sex, ethnicity, age, marital status, education, etc.), self reported involvement in a variety of crimes and drug use (including age of initiation), and prior treatment and arrest/incarceration histories. An additional set of questions asked for estimates of prior year's income and expenditures.

### 4. The Life History Interview

- ------

\_\_\_\_\_

and the second

# Appendix A - A Methodological History -267-

At the beginning of the project, this instrument was the first interview administered. Some respondents found it too much like a standard welfare or treatment program admission interview. After a few trials, staff found that respondents were much happier beginning with the daily interviews. Nevertheless, at some point during the first 33 days of reporting, the interviewer scheduled the respondent for the life history interview (generally in the afternoon when few respondents were in the storefront). Life History Interviews were tape recorded and later transcribed. Each LHI session lasted an hour or more; most subjects required two or three sessions to complete the interview.

Certain problems were encountered in scheduling and doing life history interviews; these problems led to a substantial loss of data. The major problem was the one hour or more length of each LHI session as compared to the 20-40 minute duration of the typical daily or weekly interview. Despite efforts to schedule them differently, subjects tended to all arrive at the storefront during the same time period, between 10:30 A.M. and 12:30 P.M.

This was because most subjects would awaken, come to the storefront for the interview, collect the interview fee, and then go about their daily business. One or two interviewers would typically be present in the storefront at any given time. During the peak period, five or six subjects were awaiting interviews and some might have been accompanied by friends. Subjects were disgruntled about having to wait. Their annoyance would be increased if they had impatient friends waiting with them.

It was imposible for interviewers to do LHI's during the peak hours. The large number of subjects awaiting daily and weekly interviews, and the attendant noise in the storefront, precluded lengthy taped sessions. Interviewers thus scheduled LHI's for the afternoon, generally commencing about 2:00 P.M. Subjects were given specific appointments but these were seldom kept. Appendix A - A Methodological History -268-Some subjects became too involved in an activity (e.g., drug use, crime, hanging out) to return to the field office. Others claimed to have forgotten. Others were arrested, injured, or ill. Others complained about having to undergo an hour long interview. As the number of missed LHI appointments escalated, and new appointments were made, these appointments were being scheduled further and further into the future until interviewers were scheduling LHI sessions two or three weeks away. The inevitable outcome was that more subjects forgot their appointment.

1. I

A Construction of the second sec

•

Adding to the backlog of Life History Interviews were occasional problems with the tape recorders. On two occasions, tape recorders were stolen and it took some time to replace them. On other occasions the tape recorders developed mechanical problems. On still other occasions, the supply of blank cassettes was exhausted and there was no money to replace them.

For all of these reasons, a substantial number of Life History Interviews had not been done as the data collection entered 1982. Senior staff decided to develop a new short form LHI that could be administered quickly and without the use of a tape recorder. Fieldworkers were directed to locate subjects who still owed Life History Interviews and to inform them that the interviews were now much shorter. LHI short forms were administered to many subjects. However, some could not be located. Fortunately, interviewers had begun to obtain sex, ethnicity, and year of birth at the first interview, so these characteristics were available for all subjects. Many subjects did not complete a LHI, however, either the long or short form. This accounts for a substantial number of cases (25 - 40 subjects) in which data on other demographics, prior criminal and drug use histories, and other variables was unavailable.

# Appendix A - A Methodological History -269-

5. Coding and Classifying Responses

Even with clear instrumentation, however, the complexity of respondent behavior frequently made it difficult to code what crime occurred and what dollar amounts were obtained. In many events, subjects obtained drugs rather than cash. While interviewers frequently wrote down what the subject said, important additional information was often left out. This section gives just one example (many others could be presented) showing some of the difficulties faced by the ethnographic and coding staff in fitting cross validated respondent reports into appropriate codes for one subject. The following unedited information was written by the interviewer on the back side of an interview schedule:

Two Jersey faces-Paul X told them he knew where to get some good Qs (550 bags of heroin). They gave Paul X., G.S. and Frank A. 5250.00 and they went into the building over roof and beat them. After they bought 5 Qs all three holded up Friday, Saturday, Sunday in a girlfriend's house with two other girls. They shot all the drugs in the three days and they gave the girls some drugs (not much-maybe \$15-20 a day for all three of G.S. and Frank A. didn't know how much drugs they shot on the weekend each the rest. The girls were young (16-18 yrs.) and one girl never shot drugs before.

In this account two respondents (Paul X. and Frank A.) and a third person (G.S.) conned two Jersey boys out of \$250. With this money, they were able to buy 5 quarters. They shared some of this heroin with three girls who, in turn, supplied them with food, wine, cigarettes and a place to stay for three days. The value of the crime (con game) was split between the three perpetrators (\$83 each); each was estimated to consume about \$67 worth of heroin (after deducting \$50 given to the girls). The amount and value of the quantified. It was impossible to know if both subjects consumed precisely the same amount of heroin.

The important point is that research staff were able to classify many different forms of illegal behavior, but some may be misclassified while others may have a slightly inaccurate amount. Such problems were endemic when respondent lifestyles were so complex. Appendix A - A Methodological History -270-F. The East Harlem and Central Harlem Study Groups The locale of this research was in the East and Central Harlem areas of Manhattan. There were three major reasons for selecting these communities as the location for the research. First, these two communities have among the highest levels of opiate use in the country (see below); many heroin users reside there or come to these neighborhoods to obtain drugs. Second, many of these opiate users spend most of their time on the streets, especially on several blocks (called "copping communities" by Hughes, 1977). Third, Edward Preble had many years of experience with street opiate users in East Harlem and had been able to recruit them for previous research activities. Thus, gaining access to and the trust of heroin users in these communities was easily accomplished in a short time (a process that normally takes half a year or more).

East Harlem
 East Harlem, referred to as "Spanish Harlem" or "El Barrio," is the area
 from Fifth Avenue to the East River north of 96th Street. Demographically,
 the population is about 44% black (1977 data); the remainder is mainly of
 Hispanic origin, although a few whites also live in the area. East Harlem is
 generally high on indicators of social disorganization.

While this community has many problems, it also had important strengths which community members felt should not be neglected. Third Avenue and 116th Streets had many economically viable small businesses (and few empty storefronts). A Puerto Rican Superintendent of Schools (now the Chancellor of the NYC Board of Education) massively upgraded the quality of education in East Harlem in the 1970s. These schools currently import about 400 white students who voluntarily send their children to educational programs at schools in "El Barrio." About a quarter of the blocks contain well-managed housing projects, and two major middle income projects have been occupied since 1975.

## Appendix A - A Methodological History -271-

. Produ

Of most importance to this study, however, was that East Harlem had perhaps the largest (perhaps exceeded only by Central Harlem) number of street opiate users in the country. (Since then another largely Hispanic area, the Lower East Side, has surpassed East Harlem in terms of active street dealing. One of the better indicators of the large numbers of opiate users in the East Harlem community comes from the methadone Central Registry. Graph II. 1 showed the number of currently active methadone clients by the ZIP code of their last known residence. Southern East Harlem (ZIP 10029) had more currently active MMTP clients than any of the nearby ZIP code zones (10025, 10026, 10027, 10035) and these have many more MMTP clients than other ZIP codes. Almost 1,600 clients (or almost 5.7% of all MMTP clients in New York City) resided within the two East Harlem ZIP codes (10029, 10035).

Furthermore, at the end of 1979 the methadone registry suggested that one additional person had been enrolled between 1-1-74 and 9-30-79 for every person currently enrolled and the former enrollee's last known address was in one of these ZIP codes. In addition, there may be street opiate users who have never been in methadone treatment -- probably as many as are currently in treatment. Thus, the total active street opiate user population may number about 5,000 in East Harlem.

Street opiate users in East Harlem generally concentrate their activity on a few blocks (see section I below). But few of these blocks have attracted major media attention as have some blocks of Central Harlem (see below). In blocks where street opiate users hang out, their presence tends to be interwoven into the regular fabric of street life and neighborhood functioning. Opiate dealers do not aggressively approach passers-by in an attempt to solicit business. They are generally more circumspect. Dealers, touts, and cop men may approach persons whom they believe may be buyers, but many transactions are conducted in building or apartments. Our staff discovered that street opiate users frequently do not know about other street opiate users who reside on their block. Most opiate users generally go to nearby blocks to engage in illegal activities. Appendix A - A Methodological History -272-2. Central Harlem Central Harlem includes the area from 5th Avenue to Morningside Avenue and St. Nicholas (on the West) from 110th to 135th Street. Over 95% of the population is black. Central Harlem's reputation throughout the country is that of the heroin capital of the U.S.A., if not the world. Almost all problems of ghetto life affect Harlem, but widespread heroin distribution and use have been common in this community since World War II. Newspaper accounts have frequently mentioned major avenues and several cross streets as blocks where street dealing is blatant.

Drug dealers from all over the city and suburban areas are reputed to go to these blocks to buy heroin for personal use and/or large quantities for resale in their home community. Dealers on such blocks in Harlem are much more aggressive and pursue potential heroin sales (as well as cocaine and other drugs) with many passersby. So many dealers sell heroin that they now use "brand names" (like Tragic Magic, Black Death, Dynamite, Red Stripe) to build up and maintain repeat business (Goldstein, et al., 1982c).

Despite Harlem's notoriety as a drug distribution center, extensive media coverage, and continuous undercover work by both the Drug Enforcement Admininistration and the New York City police, and frequent sweeps of the area, heroin and cocaine dealing have remained relatively constant during the five years of research. While most attention focuses upon the street scene, relatively little information has been available about the typical economic behavior of street opiate users in Harlem.

In 1979, our staff believed that important differences might exist between East Harlem and Central Harlem street opiate users in terms of their cash income and/or drug income, the types of crimes typically committed, and the kinds of drug used. The data in Chapter III show that such differences between East Harlem and Central were not significant (other than ethnicity and age). Other data not presented show little or no variation in the kinds of crime, the frequency of offending, the dollar returns, or other measures

Appendix A - A Methodological History -273-3. Cross Sectional Study

The main emphasis in this research was upon street opiate users in East Harlem who were studied both for one cycle (see below) and also longitudinally. Nevertheless, subjects were recruited by field workers from many different blocks in these two study communities.

Similar to the research concept of a cross sectional survey, this study conducted nine different interviews with respondents covering 33 consecutive days (referred to as the first cycle). This provided just over one month of data from each respondent. After giving his informed consent, each respondent reported to the storefront for five consecutive days and was interviewed about his economic behavior during the previous day. The information was recorded as a "daily" interview on the weekly interview schedule described above. On the fifth day, he was directed to come to the storefront for four subsequent interviews which were seven days apart. The interviewer then conducted a "weekly" interview about his activities during each of the 7 days following the preceding interview. During this 33 day period, the life history interview was scheduled and conducted (see comments in section E. 4). The first cycle with East Harlem subjects was conducted mainly in 1980-81, while the Central Harlem interviews were conducted mainly in 1981-82.

Having 33 separate days of data was especially important for calculating the rates of criminal offending for each subject. That is, the number of different offenses committed could be divided by the number of days during which he could possibly have committed them. With 33 days of data, a relatively stable offense rate could be computed and annualized.

This first cycle was completed with 132 respondents who were recruited in East Harlem, and with 69 respondents recruited from Central Harlem. The data covering 33 days from these 201 subjects constitutes the minimum amount of information available for calculating participation in various crimes and drug use and the rates of offending reported in previous chapters.

4. The Longitudinal Study Among the East Harlem subjects, efforts were made to conduct additional cycles of data collection in order to approximate a longitudinal design. Among the subjects recruited in 1980, respondents were asked to return every three months for additional cycles of four interviews covering 28 consecutive days. Quarterly cycles were decided upon in order to analyze whether there was seasonal variation in subjects' criminal activity or drug-using behavior. During 1981, interviewing was very intensive at the storefront. So many subjects were reporting in the mornings (and not returning in the afternoons for which they were rescheduled) that it was difficult to followup on each subject. Moreover, turnover occurred among the field staff with

ranged from 33 days to 186 days.

22. 1

80000

14

in a star

August 1987

Withour S

00

, E.

C. Starte

**F**\_\_\_\_\_

مهيدي ،

an sange

E Sugar

Constant Section

N. Y. Starter

E There

2171,092

The analysis of this longitudinal data will address whether and how respondent's lifestyles changed over time. A variety of other analyses are planned in future papers. The analyses in this report treat each respondent's person-days equally, and does not address the effects of time across weeks or cycles.

G. Measuring Time Another major objective of the first two years of this study was to determine the most effective way for studying individual respondent behavior across time. The problem was to resolve the classic dilemma of trying to produce a quality research study at less than a prohibitive cost.

Appendix A - A Methodological History -274-

responsibility for recruiting and following subjects. New field staff did not know old subjects so found it hard to locate them. Despite the many reasons why subjects did not return, 87 East Harlem subjects (or 66%) returned and completed the second cycle; 53 completed three cycles; and 32 completed four or more cycles. Thus, the number of person-days of data for a given subject

## Appendix A - A Methodological History -275-

All respondents exhibited considerable variation in income and expenditures, either cash or in-kind, as well as in the amount of drugs consumed on a day-by-day basis. Random sampling of individual days would be inadequate to establish the range and pattern of variation peculiar to an individual subject. When subjects were interviewed about their behavior across several consecutive days, however, these ranges and patterns could be more readily established. Obtaining data about the past seven days in a weekly interview covered a time period that was recent enough so that most events could be accurately recalled by the subject and recorded by the interviewer. Moreover, if respondents have a large income one day which they expended during the next several days, their cash flow could be traced over several days.

Further, if the researchers have been out of contact with a respondent for a period of time, it would be difficult to locate the respondent on any particular day. He might have shifted his activities to a nearby area and not be in the immediate vicinity any longer. He might be engaged in different activities, or be in jail or the hospital.

Yet within two or three days the whereabouts of the subject could generally be ascertained by the field worker via the street grapevine. If the field worker told some of the local street hustlers (especially jugglers and touts) that we wanted to see an individual subject, the probability was high that the subject appeared at the field office during the week. If individual days, rather than a block of several consecutive days, were the sampling units, the probability was high that many subjects could not be located and interviewed during the sampled days. Thus, for the main longitudinal study (1980-1982), individual subjects were to be interviewed for four consecutive weeks (called "cycles"). Interviewing a respondent four times during a selected month provided quality information about 28 different days for \$40 in respondent payments and only about 4 hours of interviews.

----------Mary - 19 (many) Annes Are Jugar 10 

4522

- marga

The second second

in the second

The stand

C Torres

-

Carlos and

C. Cara

12 · · · ·

Appendix A - A Methodological History -276-Thus, while the systematic sampling of time might be feasible, the week or month was a much more cost effective means of obtaining the necessary information. Several cycles of monthly interviews were used in the longitudinal aspect of the research.

H. Techniques for Obtaining Valid and Reliable Economic Behavior Data A frequently asked question is: how can you trust what they (respondents) are telling you? This question addresses the longstanding problem of the validity and reliability of self-report data. A definitive answer to such problems will not be provided here. Other researchers have carefully assessed the validity and reliability of self reported criminality and drug use and concluded that self-reports were considerably better than any other source of information. (see Ball, 1967; Chaiken and Chaiken, 1982; Wish, et al., 1983; Hindelang, et al., 1982).

This project relies upon two major kinds of evidence to suggest that the data obtained were generally valid and reliable. First, by interviewing individuals on several different occasions during an extended time period, they re-reported similar types of crimes and drug consumption, although their accounts exhibited some variation about the details of the event and the dollar returns. Second, many respondents were observed by field staff while engaging in the very behaviors they reported in the interview. These points have been delineated briefly below.

Respondents were generally eager and active participants in the research. They answered the questions and frequently provided detailed elaborations. Several internal checks were built into the daily and weekly interviews instruments. For example, detailed data about each category of income and expenditures were first obtained and these were summed by the interviewer. Later in the interview, the respondent was asked to estimate how much income he had that day from crime, and the dollar amount of drugs consumed. By comparing the sum of the individual items and the respondent's overall

Appendix A - A Methodological History -277-

Such internal consistency checks quickly identified unreliable or exaggerated claims. Any discrepancies were politely noted, and the respondent was asked to correct or otherwise resolve the discrepancy. In most cases discrepancies between summated totals and respondent-estimated totals were small, in the magnitude of less than 5%. Such discrepancies were usually resolved when the subject thought carefully about what he did and how he expended his money.

The most suggestive evidence that reliable and valid data were obtained, however, was that respondents tended to report relatively consistent patterns of behavior during different days. That is, a person who reported shoplifting on one day would typically report it on subsequent days. While consistent patterns of behavior emerge, the precise locale and nature of their criminal activity and dollar amounts exhibited considerable variation. Thus, while the investigators have been unable to check the truthfulness of each specific crime and the precise dollar amounts actually obtained, the types of crimes committed and the general magnitude of returns showed considerable similarities.

This study also had other ways of assessing the truthfulness of respondent reports. Ex-addict field workers were familiar with many respondents' behavior from long periods of association. Such workers could generally tell whether reported crime or drug use was consistent with their knowledge. Respondents themselves provided information about other respondents. Especially when two respondents had committed crimes together, their separate reports of the dollar amounts and facts were usually very similar (Paul X and Frank A. reported the same facts given above; also see Preble (1980b) for another example.

1

٠

Append Interestingly, the a acquaintances tended to who were married to contradictory accounts of Each claimed credit for care of their spouse by lazy or incompetent. Research staff inte pilot phases) for a min These respondents got respondent, the intervi teaching the researcher and acceptance. Incre share information. Thi in their everyday life.

-

.....

-

-

100 THE

C. C. Standy

1 Same

R. Station

C. X Server

.

W. Contra

ETNICASE

C. Sec.

1000

Sec. 2

atter C

i Igense in d

C.M.

Field workers and professional staff have observed respondents engaging in behaviors that supported claims of criminal activity, such as the following. One respondent who consistently reported stealing parts from cars was observed walking down the street with a car bumper over his shoulder. Another subject claimed to have committed a burglary the previous day in which he obtained \$600 and bought \$200 worth of drugs; he shows the field worker \$300. Another subject who claimed to have robbed a gun-runner was observed carrying four 32 calibre pistols in a shopping bag. A respondent who reported serving as a tout on the streets was observed day after day talking to anybody who looks like an addict while trying to drum up business for a local dealer; he was observed making sales. Four persons who sell drugs from an apartment were observed in their apartment for several days; they made 15-30 tranactions per day in front of research staff. These observations could be extended many times.

Appendix A - A Methodological History -278-Interestingly, the accounts given by crime partners who were friends or acquaintances tended to be more consistent than those given by crime partners who were married to each other. Husbands and wives frequently gave contradictory accounts of who had masterminded or effectuated a "big score." Each claimed credit for the criminal event. Each claimed that they had taken care of their spouse by sharing their drugs. Each would refer to the other as

Research staff interviewed over 250 respondents (including over 50 in the pilot phases) for a minimum of thirty days during the course of this project. These respondents got to know and trust the field workers. For the respondent, the interview became not only a way to raise money, but a way of teaching the researcher about his life as well as obtaining some understanding and acceptance. Increased rapport led to more honesty and willingness to share information. This acceptance permitted the researchers to observe them is their everyday life.

## Appendix A - A Methodological History -279-

Such observations led field staff to feel comfortable that the data obtained and used in this report were sufficiently reliable and valid to provide new insights about the economic behavior of street opiate users. Of course, such evidence did not constitute statistically documented evidence of reliability and validity. In some ways, however, observational evidence may be a more convincing indicator of validity than formal statistical tests.

## I. Difficulties in Conducting a Census of Street Opiate Users in East Harlem

One objective stated in the 1979 application to NIDA was to estimate with as much scientific rigor as possible the number of street opiate users in East Harlem. A stratified sampling procedure was designed, but proved difficult to implement. For reasons given below, this research was not able to develop an adequate sampling frame, nor select respondents from the a sampling frame at a known probablity. Too many obstacles were encountered. This section, however, describes the effort which staff undertook in early 1980 to accomplish this objective. In the process, we also learned some important lessons about East Harlem and about the lack of interaction between street opiate users living on the same block.

Sampling strategies underwent a variety of modifications as the study progressed. During the pilot years, we relied mainly upon snowball selection techniques. Subjects were referred to the study by both field workers and other subjects and by those who dropped in and asked to be interviewed. This technique for selecting subjects was inadequate as the subjects were relatively inactive, older, heavy-drinking methadone clients.

The next strategy attempted was to plan for a census of opiate users on selected blocks in the study area. While data were available about the parameters of treatment populations and incarcerated populations, these might or might not be applicable to a general sample of street opiate users. There was no systematically collected data that provided a framework from which a representative street sample could be drawn.

C. Tr

7<sup>8</sup>1

1.00

1000

-----

-----

- States in the state

-3: ---- Ja.--

a.6.18

Concernance of the local data

\*\*\*\*\*

4443 ····

40-1-0

-049400

-----

-----

Address of the

·

-----

Frankers -

-----

A block survey was undertaken of the entire East Harlem Area, defined as extending from 96th Street to 145th Street and from 5th Avenue to the East River. The purpose of the block survey was twofold: (1) to identify specific locations where illicit drug business was being transacted; and (2) to record those blocks in which people resided and the predominate building types. The latter classification was intended to comprise the universe from which blocks would be sampled as part of major study. Each and every block in the study area was visited by the Project Director (Goldstein) and a field worker.

Appendix A - A Methodological History -280-1. A Block Survey in East Harlem.

A block survey was undertaken in early 1980. This survey was designed to provide for a subsequent stratified sample of blocks having high and low levels of street opiate user activity. This block survey resulted in some interesting findings wholly independent of the sampling issues.

A block was defined as two opposite sides of a street facing one another. This appeared more reasonable than other possible definitions, e.g., a four-sided figure. First, and most important, it corresponded to the everyday language of subjects and researchers (e.g., "First Avenue between 105th and 106th Street" has an immediate experiential referrent to all concerned). Secondly, there are various problems associated with defining a "block" as a four-sided unit. For example, due to the expansion of housing projects, many four-sided land units in East Harlem extend for the normal one block on their east to west boundaries, but may extend for 3 blocks on their north to south boundaries (e.g., a "single block" would extend from 106th to 109th Street between Second and Third Avenues, from 112th to 115th Street between Third and Lexington Avenues, etc.). Also, there tends to be greater homogeneity of building types between opposite sides of streets than there is between North, South, East and West sides of a four sided unit.

11

## Appendix A - A Methodological History -281-

and a second state of the second state of the

الارومين ب

10 10 10 10

-

The second second

Be wanted

- Stranger

and the same

----

72.4

-

Server 1

Salary in

444407-010

former of

Aller and

10 M

200 C

-----

مى ئىلى يېچىنى

diam'r.

. .

And the second

in the second second

- ----

10000

Stor marsh ?

Constant of

There were obvious socioeconomic differentials within the area. For example, Fifth Avenue facing Central Park (from 96th to 110th Street) was quite elegant, characterized by large well-kept apartment buildings, museums and hospitals. However, north of 110th Street, Fifth Avenue was similar to the rest of the study area. Scattered throughout the study area were several large, modern, hi-rise apartments (e.g., Taino Towers covered an entire block between 2nd and 3rd Avenues, from 122nd and 123rd Streets, and the District Council 1199 middle income housing, between First Avenue and the East River Drive, 100th to 104th Street). Pleasant Avenue (which runs from 114th to 120th Street east of First Avenue) was a diminishing Italian enclave in predominantly Black and Hispanic East Harlem.

Many housing projects have been constructed in the study area. These projects usually include parks, playgrounds and parking lots. Project buildings are often laid out in such a fashion (e.g., "catty-cornered" spanning several blocks) that classifying them as being on a specific block would be an exercise in futility.

Several streets have been zoned for storefronts and small shops; these were usually on the Avenue's and on selected cross streets (96th, 110, 116, and 125). Most of these commercial streets were doing poorly; many stores were closed and "for rent." Stores frequently were very run down and not prosperous. Many such "businesses" were actually numbers joints, chiba shops (stores selling marijuana), or fronts for other illegal businesses. Only along Third Avenue and along 116th Street were most of the stores in business and appeared to be prosperous.

Certain streets have a unique character: 116th Street is a commercial thoroughfare with many outdoor stalls selling clothes, wigs, etc., reminiscent of 14th Street. Park Avenue, north of 97th street has elevated railroad tracks running through the middle of the street. Under these tracks from 111th to 116th Street is La Marqueta, an enclosed market with individual

vendors selling an assortment of items (e.g. food, herbal remedies, clothing). An active prostitution scene was observed under the Park Avenue railroad tracks between 123rd and 125th Street. The block survey revealed the following information. A total of 500 blocks were visited. Of these about one-half (n=253) had some residences (other than housing projects) on them. About a third of the blocks had housing projects (117 on both sides, and 43 with both housing projects and other residences--generally on opposite sides of the street). About a sixth (n=87) of the blocks contained no residences at all. These blocks included totally commercial areas, empty lots, hospitals, schools, museums, abandoned buildings, etc.

About half the blocks were composed of tenements of "railroad" design (so called because one had to walk through two bedrooms to get from the kitchen to the living room). In general, the strongest visual impression of these blocks was the large number of abandoned properties or vacant lots. Few blocks were without at least one abandoned building. Many blocks were predominantly or completely abandoned. Some of these buildings were sealed with tin or concrete blocks. Others were simply open shells. These abandoned buildings were generally former residential buildings, but also included commercial properties, schools, etc. Many vacant lots were filled with rubble.

The basic thrust of our sampling effort at this point was sampling blocks according to the amount of street opiate users or activity. The Block Survey revealed 296 blocks that contained non-project residences. However, some of these blocks were eliminated prior to drawing a sample. There was little utility foreseen in including those blocks that were on a much higher SES

Appendix A - A Methodological History -282-

Appendix A - A Methodological History -283-

A STATE OF A

level than the rest of East Harlem (e.g., Fifth Avenue between 96th and 110th Street). It was unlikely that our street informants could identify any drug users living in these blocks or that such users would consent to become part of our systematic sample. A total of 30 such blocks were eliminated, leaving a universe of 266 blocks from which a block sample would be drawn.

Twenty-seven locations of drug activity were identified during the block survey. Table A.I reports the nature of this drug activity.

### TABLE A. 1

### NATURE OF DRUG ACTIVITY IN EAST HARLEM (by blocks)

	Number of Blocks	20
Street Action	9	33
House Connections	4	15
Chiba (Marijuana)Shops	3	11
Dealing in Parks	З	11
Social Clubs	3	11
Street Action and House Connections on Same Block	2	7
House Connections and Social Clubs	1	4
House Connections and Dealing in Parks	1	4
Dealing in Bars	1	4
Total	27	100%

1000 Z 4000 T Charles 127-9 12 ----41 4144 (inter a state 123 - 1944 Courses and

L'a Tripa - 1

Appendix A - A Methodological History -284-The most obvious street drug activity was centered on Lexington Avenue between 123rd and 125th Street with extensions of this activity going up the side streets (mainly 124th Street). On one occasion, around midday, 24 people were observed on Lexington Avenue between 124th and 125th Street and an additional 31 people were observed on the Southwest corner of Lexington Avenue and 124th Street. All of these people appeared to be involved in drug distribution.

Drugs were sold in the streets in one-third of the blocks in which dealing activity was observed. However, no conclusions can be drawn as to the relative volume of drug sales (in terms of the actual quantities of drugs sold or the actual number of sales) on the streets as opposed to house connections, social clubs, etc. Likewise no conclusions can be drawn about the nature of drug purchasers and whether the street purchaser differs in any way from the house connection client. House connections usually sold heroin or cocaine while street action may involve only marijuana, pills and/or methadone. While some street drug scenes involved 20 or 30 people on a given block at a given time, other street scenes never seemed to involve more than 4 or 5 people at any time.

Eight drug programs (including 7 methadone maintenance programs) were observed. Efforts to conduct a census of street opiate users residing in housing projects was not attempted. Street informants and subjects gave contradictory reports as to the extent of drug use among housing project residents. Some said the projects were characterized by very low levels of drug use (as compared to the tenements) while others said there were few differences between them. Some informants said housing project residents were more secretive and circumspect about their activities. Appendix A - A Methodological History -285-

2 . Attempting a Census of Opiate Users on Selected Blocks

The following procedures were proposed for drawing a systematic sample of East Harlem residents.

- (1) From the total of 266 blocks containing non-project residences, 20 blocks would be randomly selected; three blocks would be selected from among the 27 high activity blocks.
- (2) At least 2 key informants would be recruited from each selected block.
- (3) A census of drug users living on each selected block would be constructed in consultation with the key informants.
- (4) Depending upon the number of drug users that were identified by informants, either all or a sample of the drug users living on each block would be selected for inclusion in the sample.
- (5) These individuals would be approached and asked to participate in the study. A critical aspect of this phase of the research was the response rate. Prior to this effort we had worked mainly with "volunteer" subjects. This would be our first systematic effort at recruiting specific individuals into the sample from a known sampling frame.
- (6) Individuals who agree to participate in the study, if found to be opiate users, would be placed on a systematic reporting schedule (i.e., complete 33 days of reporting).

After the blocks were selected, staff chose three blocks for pilot study to determine whether the above plan was feasible. Two major efforts were undertaken to compile a list of suspected opiate users living on the key blocks. They proved unworkable for a variety of reasons. Initially, our indigenous field workers were asked to fill out census cards for all addicts or opiate users they knew on those blocks. While they tended to know a few addict groups or blocks well, they were relatively unfamiliar with others. To fill the gaps in their knowledge we moved to a "key informant" system. The plan was for field workers to approach an acquaintance on a block and ask the acquaintance to fill out a "census" card for every opiate user he knew on the block. Most individuals were reluctant to name other addicts.

counted in the census.

BDJ:4749A;723A;bj

alere -

-Trip

1010

and the second second

......

Nerse 2 -

3 8 10 1

Nr. . . .

100.0

الإغوار بيداد

Second

Sec. Star

**R.** 1998,

The Print

**\*\*\***\*\*\*\*\*\*\*\*

0.000

**87.**77 (1946)

يقصدون

T. States

The second second

E. Contractor

**S** the second

ુ છે. આ છે

The state

The second

Appendix A - A Methodological History -286-More importantly, different key informants on the same block mentioned different names (there was little overlap) and disagreed over who on the block actually used drugs and what kinds of drugs they used. Many drug users had multiple residences (e.g., with friends, family, abandoned buildings), multiple hang-outs, and multiple street names, so that it was often difficult, if not impossible, to determine just how often the same individual was being

Some key informants requested a stack of blank census cards and said that while they would not name others as drug users, they would circulate the cards and turn in those voluntarily filled out. They, of course, expected to be compensated for their effort and an amount of two dollars per completed card was decided upon. As the cards began to be returned, and the money paid out. field staff raised questions as to the validity of the information contained on the cards. Few of the claimed volunteers could be located. Staff felt that the census cards were being fictiously completed for the sole purpose of collecting the two dollars bounty. The practice was terminated.

Moreover, new subjects came to the storefront reporting that they had "heard" that we were compiling a list of addicts. The clear implication by the street grapevine was that this would be provided to police. Even though this was not our intent, the practical difficulties of conducting a census and identifying a sampling frame, combined with the growing false rumor on the streets led staff to abandon efforts to conduct a scientific sampling effort. Rather, the study would depend upon field workers to locate and recruit street opiate users from different blocks in East and Central Harlem, with efforts made to ensure a diversity of lifestyles.

In summary, this project encountered many methodological problems. Most of these problems were satisfactorily addressed, although quasi-scientific sampling was not possible. The following appendix describes more of the flavor of working with street opiate users.

## Appendix B -- Taking Care of Research Business -287-APPENDIX B

"TAKING CARE OF RESEARCH BUSINESS"\*

Experiences with Conducting Research Among Street Opiate Users

This appendix focuses upon the process by which social scientists come in contact with research subjects while doing a field study. We will focus upon the informal and frequently expedient practices which researchers may need to develop in order to implement a research design and "bring in" the data from the field. Problems that we did not anticipate arose; solutions - not always satisfactory - were developed. Maintaining an ambiance in the storefront, supervising paraprofessional field workers, maintaining safety, following professional practices, and describing respondent lifestyles are discussed in the following sections.

## A. Coping in the Ethnographic Field Station

As described in Appendix A, data collection for the Economic Behavior Project was undertaken in an ethnographic field station or storefront. During the course of the project, three different field offices were operated. The first was on 95th Street between First and Second Avenues. The second was on 105th Street between First and Second Avenues. The third and final field station is on an nearby Street (the location on 1XXth is concealed because further research is being conducted there at the current time).

\*The title is borrowed from a classic article by Preble and Casey (1969) "Taking Care of Business" which described the lifestyle and structure of the drug distribution system and focused upon the street heroin user and his world.

	Ring to the second s
	ł
stanov 15	n Norma press
	1.5
e properties	د د معمر ردیمر
an a	August Commence
i in and	$\label{eq:states} (x,t) = \sum_{i=1}^{n-1} (x^{i+1})^{i+1}$
	المغف ويبت
T. T. T. MAR	-
مېسېن ،	and an inclusion
and the second	د میشور استخ <b>میتو</b> ا
R	
	telesses and a
X. Sta	
	alayon ya
TELL TONING	
an a trainin	Alderen, Star
<b>2</b>	Samere Car
y , i na calena	allinean ann
a an	angéte <sup>r</sup> : <sub>dener</sub> .
5054 <b>8-</b> 9	ultipe and strategy
No.	
a an	milesco - J <sup>ana</sup>
1. State 1.	
	AC repair to repair
E Fridday,	a construction of the second sec
्र***T343984 4	s (1996) to a second
and the same	<del>معر</del> يد المتحورين
1 Netjejs	(#**App., ( <sup>1</sup> 79)**

Appendix B -- Taking Care of Research Business -288-

The 95th Street field station was abandoned early in the pilot years because of the quality of the storefront and the subjects it attracted. The 105th Street location was one of the more active "drug blocks" in East Harlem. An important "house connection" (i.e., a dealer selling out of an apartment) was located directly above our storefront. A number of street dealers, usually four or five at any given time, clustered on the corner of First Avenue and 105th Street. Another street dealer sat on the stoop of an abandoned apartment building diagonally across the street from our storefront where she was readily observable. Customers were constantly coming and going. Her standard procedure was to negotiate a transaction on the stoop and then take the customer through a rubble-strewn vacant lot to the back of the building where her "stash" was hidden. This young female dealer also spent a great deal of time on the stoop socializing with her female friends who often had their young children in tow.

Directly across the street from our storefront on 105th Street was an abandoned school that at one time must have been architecturally magnificent. It still retained an eerie charm, though its doors were now blocks of cement and its windows had all been broken by young marksmen hurling rocks and soda cans from the courtyard below. Fierce looking gargoyles remained poised above doorways, futilely protecting what was now simply an empty shell. The following story was constantly told to newcomers on the block.

The school had suffered the common fate of other abandoned buildings in East Harlem. It had been stripped of any copper pipes or other contents that had any resale value at all. However, there was a great deal of copper remaining on the roof. Since the roof came to a slanted inverted V-shape, the only way to reach this copper stripping was to lean out of a window and pull

Appendix B -- Taking Care of Research Business -289off whatever was in reach. The roof had been obviously denuded of copper around all of the upper windows. The story concerned an addict who thought he could get some copper that his more prudent comrades had decided was out-of-reach. This addict stretched in vain for the piece of copper, fell out of the window, and landed in the courtyard below, killing himself instantly. This story seemed to embody the frustration, futility, and tragically petty fates of so many of our subjects.

Most of our activities during the pilot years were undertaken in the 105th Street field office. The storefront was a rather dismal place. Plaster fell regularly from the ceiling. The toilet would not flush and persons using it would throw a bucket of water into the commode after use. There was seldom any heat in winter and interviewers would sit bundled up all day long. Recording data was a problem because of the cold. If the interviewer sat without gloves for any length of time, his hands would be too numb to hold a pen cr pencil. The alternative was wearing heavy gloves. However, such gloves made it difficult to manipulate a pencil and completed interview schedules often bordered on the illegible. Interviewers finally developed a system of taking their gloves off, recording interview data for about five minutes, rubbing hands vigorously and perhaps putting gloves back on for a minute or two while chatting with subjects, and then returning to the formal interview. Between interviews field staff would generally walk outside, if there was any sun, in order to warm up a bit.

These inconveniences were not the only reason that we chose to move to 1XXth Street when we began the final three years of the project. Rather, we had exhausted most of the potentialities of the 105th Street site. We had been on the block for about two years and knew it well. It no longer held any

Second of the -Sector Sector and the second second The Property of the Property o

quality of respondents discussed in Appendix A. For all of these reasons, senior staff decided to move the field station to 1XXth Street for the final three study years. This storefront, a former "bodega" (Spanish grocery store), was owned by a plumbing contractor who guaranteed heat in the winter. It had five times as much space, had two working toilets, and was well lit. In short, it was in physically much better condition, although it had much falling plaster and a storage area filled with debris left by former tenants and the owner. Because of the large number of interviews that would be conducted, staff built three interviewing cubicles which were furnished with a desk, two chairs, and a light overhead.

The move enabled us to familiarize ourselves with a new area of East Harlem and attract a new group of subjects from both East and Central Harlem. While a few of the methadone wino crowd from 105th Street discovered our new location, interviewers informed them that the study was now in a new phase and we could no longer talk with them. They soon left us alone when no more money was forthcoming.

Research staff sought to maintain a certain atmosphere in the storefront. The general mood was congenial. This reflected the personalities of field staff. Further, we wished to encourage subjects to adhere to their interviewing schedules; it was felt that subjects were more likely to return if the interview was a pleasant experience. Most subjects had negative prior experiences with bureaucracies (e.g., welfare, unemployment) and would become resentful if they perceived our attitudes or behavior as similar to those that they previously encountered from allegedly unresponsive authority figures.

Appendix B -- Taking Care of Research Business -290surprises for field staff. Even more important, however, was the deteriorating

Appendix B -- Taking Care of Research Business -291-Coffee was prepared in the storefront all day long. Field staff frequently purchased rolls or doughnuts that were available to anyone who wished them. Female subjects often brought their children with them to the storefront. At least one baby took her first steps there.

The behavior of children helped to sensitize research staff to the lives of subjects. For example, a husband and wife were both research subjects. They usually arrived for their interview together and occasionally brought their son, who was about eight years old. One day he was running about the storefront pretending that a pencil was a marijuana "joint." He would put it in his mouth and mimic the deep inhalations of a pot smoker. When somebody told him not to put the pencil in his mouth, he immediately began to pretend that it was a syringe and that he was giving himself injections into his arm.

The boy's actions helped to verify his parents' accounts of frequent heroin and marijuana use. His actions also provided researchers with an insight into the home life of the family (i.e., that the parents used drugs in the presence of the child) and into the realities of growing up in an urban slum as the offspring of addicted parents. The same little boy told us several times that "daddy hits mommy." This was confirmed one day when the mother arrived for her interview with a black eye and cursed her husband for giving it to her.

At times the children were an annoyance (e.g., being too loud, interrupting interviews). For the most part, they contributed to creating a atmosphere in which subjects felt comfortable. In this way they were conducive to the research effort.

One aspect of the storefront's congeniality was a mixed blessing. This was the presence of hangers-on. Some subjects, and even people whom we knew from the neighborhood but who were not enrolled as subjects in the study,

-------las my Contra and time me

. w. 77

A. 2965.

**\***\*\*\*

A. Contraction

**5**17 mag

......

began to conceive of the storefront as their personal hang-out. They would arrive early, stay until closing, frequently drink beer or wine throughtout the day and perhaps doze off in a chair. They would generally seek to borrow a few dollars each day and seldom refused the offer of a free lunch. The negative aspects of such hangers-on included a slow but steady drain on project financial resources for "field expenses" and the fact that they might occupy one or more of our few chairs, perhaps forcing regular subjects to stand while awaiting their interview.

Here again, however, the positive aspects of the phenomenon outweighed the negative ones. The hangers-on served many functions, often unwittingly, for the project. They provided a vital connection with the neighborhood. We were kept up-to-date on local gossip, which often concerned crime and drug issues of great interest to us. We were continually informed about homicides in the area. Their presence in our storefront, along with the exaddict field workers, legitimated our presence in their neighborhood. They would tell a new subject that we were not some sort of police undercover operation. Hangers-on frequently ran errands when it was inconvenient for us to leave the storefront (e.g., going to the supermarket to replenish the coffee supply). They sometimes assisted in recruiting new subjects or verifying the accounts given by current respondents. The security of our field staff was enhanced by having friendly local residents as a permanent fixture in the storefront. Finally, the hangers-on were utilized to pretest research instruments and occasionally as subjects in "special studies" (e.g., on the relationship between drugs and violence in East Harlem). The major difficulty with the daily interview in the field office was congestion of respondents in the storefront or on the street in front of the storefront. Most subjects showed up in the morning shortly after they had

Appendix B -- Taking Care of Research Business -292-

Appendix B -- Taking Care of Research Business -293awakened. They wanted to get in and get out rapidly to go about their daily business. Three to eight daily respondents might be waiting at the same time to be interviewed. This pressured the field workers to conduct very rapid interviews and not to write out in long hand the kind of details desired to supplement the quantitative reports. The shift to the weekly interviews relieved some of this pressure. Congestion problems were more difficult during the winter than the summer because the respondents did not want to stand outside in the cold. This made it difficult to maintain privacy and do confidential interviews in a small storefront.

Doing interviews in an ethnographic field station was a unique experience. Most social scientists conducted direct interviews with drug users and criminals in institutional settings, such as prisons or drug treatment programs. They dealt with a controlled population that was relatively sober and nonthreatening. Guards or counselors would be present or nearby.

Interviewing in the storefront was a different reality. Subjects may be inebriated, "sick" from withdrawal, or "nodding out" from narcotics. They may act surly and make demands that could not be fulfilled (e.g., asking for a large sum of money or requesting admittance into the study when for one reason or another they were not qualified). In such cases our main protection was our wits and humor, general "right guys" image, the presence of friendly subjects and the surly person's own sense of right and wrong -- as well as his/her fear or reluctance to precipitate an unpleasant incident. On one occasion we were robbed, but being robbed once in five years in that neighborhood was probably better than statistical probabilities would suggest.

The second

----

10

part and

----

1. Martin

a Baira

1.000

. ....

-

211-12-29 97

-----

line - - -

Appendix B -- Taking Care of Research Business -294-When actually conducting interviews, alcohol was the major problem. Inebriated subjects often rambled on and were unable to focus on the questions they were being asked. Further, inebriated subjects often could not remember what they had done just a few hours earlier. In a less important vein, a few inebriated subjects had annoying personal habits that irked interviewers. For example, Marcia C. consumed copious quantities of beer; during the course of a typical interview, she would emit thunderous belches in an interviewer's face. On various occasions other subjects who drank too much called interviewers names or engaged in excessive back-slapping.

Subjects who injected heroin shortly before coming to the storefront might "nod out" during an interview. They would have to be gently awakened several times before the interview could be completed. Respondents in this state had difficulty in maintaining concentration on the questions being asked. Many heroin users would awaken, come directly to storefront, do their interview and receive the ten dollar interview fee, then go out on the streets to go about their daily business of raising sufficient money to purchase drugs. This meant that on occasion we had to interview respondents who were experiencing withdrawal symptoms. Such subjects were edgy, nervous, impatient; they couldn't wait for the interview to be over. Interviewers had to be skillful to insure that such respondents did not omit important information just to speed up the interview process.

During the course of research, the professional staff had to rely heavily upon former addicts/and convicts for locating and bringing in "good" respondents from the streets. During this research, about five persons functioned in this role, as well as some subjects. Locating them was fairly easy; many subjects volunteered for and claimed great expertise in finding suitable subjects.

B. Supervising ex-addicts and ex-convicts as field workers.

Appendix B -- Taking Care of Research Business -295-

The professional staff was aware of opportunities that field workers had for "ripping off" a project. Three factors were kept in mind. First, indigenous field workers were hired because of their knowledge of and connections into the subcultures of crime and drugs. They were employed to recruit addicts and criminals into the study, to explain deviant phenomena to the senior researchers, to serve as neighborhood guides, and to clarify the argot spoken by subjects. Indigenous field workers were usually ex-addicts and/or ex-convicts. They were accustomed to "hustling squares" and seizing whatever opportunities were presented to "get over" and make a few dollars.

The second factor concerned the social and economic insecurity of such indigenous paraprofessionals. Poor, relatively uneducated, ex-addicts were being offered employment and colleagueship with more affluent and educated social scientists. They were being given the opportunity to work on an important research project. They wanted very much to demonstrate proficiency at the tasks that were assigned to them. They wanted to earn the respect of senior staff members and they wanted to keep their jobs; all of them worked hard to do so.

The third factor was that employment on the project enhanced the prestige and power of the paraprofessional amongst his peers. He was working with "the man" and "the man" had money. Prospective research subjects had to be recruited by the fieldworker in order to be eligible to participate in the study. We shall now examine how these three factors interacted to produce certain situations that were potientially detrimental to the research.

1

Appendix B -- Taking Care of Research Business -296-Social scientists have been trained to admit that they do not know some things, may be unable to do other things without skilled assistance, or that some activities could be impossible to do. Field workers were frequently unable to make such admissions. Aware that they were hired because of their extensive knowledge about and contacts with life in the streets, they were afraid that if they could not answer certain questions (e.g., how many "shooting galleries" were on a specific block) or accomplish certain tasks (e.g., recruit ten interviewees having specific characteristics) that they might lose their jobs.

- i

STORE STORE

CARRY ADDRESS

Sector Sector

resultation makes

TATION CONTRACTOR

STATES STATES

CARGE STREET

A Contraction of the second

M. Trining and

Contraction of the second second

Service 1

Control 1

1.00

of Anna

Carlo Tana

Concer.

Field workers on this project sometimes responded to our queries quickly and glibly. However, careful probing revealed that they did not know the answer to the question. They had no real intention to deceive us or to give misinformation. Rather, they were too scared or too insecure to admit that they did not know.

Early in the project, one field worker had major responsibility for the recruitment of new subjects. In an effort to construct a sampling frame, this field worker was asked to recruit subjects from specific blocks. He was asked if he knew opiate users from those blocks and assured us that he did. He claimed that he knew a few people from each block. Through such contacts, he would be able to reach a wide variety of opiate-using block residents. He did bring in a number of subjects who reported their address as being on the target blocks. However, several subjects, after they got to know and trust the senior researchers, independently confided that the fieldworker had solicited them outside of their methadone program and told them that there was a study willing to pay them ten dollars for interviews. According to these subjects, the field worker instructed them to lie about their current address.

Appendix B -- Taking Care of Research Business -297-

-

where and

and the second

------

10 - Stag

Labora E. T.

10111

1 ANN 14

----

1000 B

1.394545

1 Santa

**SE** 1997

Such a situation, if allowed to go unchecked, could have seriously distorted the data<sup>\*</sup>. Attempts to make projections about the number and kinds of drug users in East Harlem based upon block reports would obviously be based upon spurious data. For example, methadone clients might have been grossly over-represented.

Was the field worker in this incident lazy and just chose the easiest way to get the job done? Was he too insecure to admit that he did not know anyone on certain target blocks? Did he make an effort to recruit subjects from the target blocks, but was unable to do so and afraid to report failure? Senior researchers decided not to confront the field worker directly on this issue. He was valuable to us in many others ways and did gain us entry into certain groups that we otherwise might not have been able to reach. Many of the subjects he brought in were highly involved in criminal activity and heavy drug use -- but they lived on the wrong block. It was felt that confronting him on this issue might have injured his pride, or aroused his anger, and led to his quitting the project. Senior staff recognized his value and, even though he had acted deceitfully on one task, we did not want to lose his assistance in other areas. Eventually, however, this field worker took another job and left the project.

Another form of behavior aroused the suspicions of senior researchers. This involved field workers expressing the wish to do interviews in the field (i.e., not in the storefront) and insisting that it would be impossible for senior researchers to accompany them. Field workers reported knowing potential subjects that would be excellent for the study, but who would not come to the storefront. Further, they claimed, these subjects would not talk honestly in the presence of white people. Several other times, when senior researchers expressed disappointment that certain subjects had failed to complete their interview cycle, field workers reported that these subjects were

This was another reason why the effort at conducting a census of street opiate users was ultimately unsuccessful (see Appendix A).

However, they claimed that senior researchers would not be allowed in the apartment (e.g., in one case a "crazy husband" was just too violent) and that only they, working alone, could get the interview. Initially, senior research staff permitted field workers to conduct interviews outside the storefront. However, this practice was terminated for several reasons. Even though field workers were trained to conduct interviews, they manifested very different levels of competence at this task. Much of the differential appeared to be language related. Field workers were typically not accustomed to writing things on paper. Hispanic field workers often had great difficulty writing in English. Research protocols occasionally came back almost blank, or with the same items either filled in or omitted for all subjects.

When senior researchers decided to limit interviewing to those field workers who had demonstrated proficiency at the task, the result was jealousy and animosity between field workers. Here again the basic insecurity of paraprofessional workers manifested itself. A paraprofessional seeing another being granted greater responsibility was likely to feel threatened, to feel that the other field worker was becoming a more valued member of the research team, and to feel that his own days of employment might be numbered. Field workers denied the opportunity to interview began to tell tales about those who were given the added responsibility. They accused others of being police informers who were not trusted by subjects, or of using drugs again.

Another factor contributing to this animosity was that paraprofessionals doing field interviews were entrusted with greater sums of money (i.e., in order to pay subjects their interview fees). The prominence of the money issue led senior researchers to wonder whether all completed interviews had actually been done. What was to stop a field worker in the streets from filling out an interview protocol himself, pocketing the ten dollar interview fee, and simply turning in a "dummy" protocol? This question appeared

Appendix B -- Taking Care of Research Business -298ill, could not leave their apartment, but could be interviewed at home.

Appendix B -- Taking Care of Research Business -299especially relevant when the subjects involved were people whom the field worker asserted could not be seen by senior staff for one reason or another. Evaluation of the completed field interviews did little to alleviate senior researchers' anxieties on this issue. The decision was made to terminate the practice of field interviewing, unless the interview was done by a senior researcher or in his presence. All other interviews were to be done in the storefront. Unsupervised field interviews have been omitted from the data analysis.

In addition to the temptation of diverting project monies to their own pockets, field workers had the opportunity of diverting money to their friends. They exercised a certain amount of power because they chose subjects for the study. There were indications that they chose friends, both because friends were easy contacts to make and, also, because friends appreciated the opportunity to make a few extra dollars.

Such "friends" were also a genuine burden to fieldworkers trying to do an honest job. For example, when fieldworkers were instructed to only recruit subjects from specific blocks they were often hard-pressed to refuse the entreaties of old friends on the street who needed the money, wanted to participate in the study, but did not live on the designated block. The honesty of some field workers strained old friendships and compromised one of the attributes that led to their employment in the first place, i.e., popularity on the streets and a wide network of friendships.

During the major interviewing period, two exaddicts and one college student were employed on a part-time basis to conduct interviews. All had considerable difficulty in doing the interviews as fully as was desired. Two major problems were present. First, they had difficulty comprehending what the senior researchers wanted. They did not ask followup questions or probe responses which were unclear. Occasionally they had difficulty understanding how to record data about steering, touting, copping and drug sales (this took senior researchers a long time to figure out as well).

Appendix B -- Taking Care of Research Business -300-Second, their writing skills were limited. They had difficulty writing down the subject's own words into coherent narrative. They occasionally left out critical details. Although directed to write down each crime described by the respondent, many apparently interesting robberies and burglaries have no written scenario.

1

-----

والمح المحمون

اللغة المريز

1000

34.14

Third, they made many assumptions which resulted in less than full specificity. For example, a respondent might report getting two "quarters" of heroin and one of cocaine, but would not record the dollar amount which the subject really paid for it. Although this amount has a standard market value, many subjects "cop short" (buy drugs for less than the standard value). They assumed that the person bought it for the "retail" price rather than ask for the exact payment.

Despite limitations, however, such paraprofessionals were of great assistance to the research process when closely supervised and carefully trained. Senior research staff learned much from them, and they gained from their association with us.

C. Maintaining Safety and Dealing with Subjects Ethnographic field work involved many potential dangers and temptations. The main dangers involved being robbed or assaulted. No member of the research team was assaulted during the five years that the project operated in East Harlem. Staff were robbed on only one occasion, despite the fact that we were studying many persons who committed robberies and were involved in assaults.

Staff took several precautions to protect data and research equipment. After an early burglary (see below), tape recorders and similar valuable instruments were removed from the storefront every night. Completed interview schedules were forwarded to the World Trade Center office on a regular basis. Only a few articles of minor value were left in the storefront, along with the ancient chairs and tables used for interviewing and old issues of the

Appendix B -- Taking Care of Research Business -301-

Several efforts were undertaken to protect against robbery or theft of respondent payment cash. A large check was deposited in a bank account, and cash machines were used to withdraw money depending upon the projected number of interviews to be conducted during the day. Only on one occasion did a robbery take place.

One slow day in 1982, the senior interviewer (Miller) was in the storefront waiting for a field worker to bring in a research subject when three men appeared at the door. One was an ex-research subject; the others were his 'friends.' The door was unlocked and opened slightly to find out more about the person's code name. But instead of waiting for the door to be opened all the way, the subject shoved his foot and leg in the door and tried to push it open. Miller jammed his foot under the door and pushed to prevent it from being opened further. The subject demanded, "give me your money." Stalling for time, Miller asked, "why are you doing this; you get carfare occasionally". At the same time, Miller observed the subject had a hand in his pocket, and one of the friends had his hand inside his belt. The subject responded, "Because I'm a mother-fucking junkie." Miller, afraid that the friend was ready to pull out a gun, decided to comply rather than resist. He reached into his right front pocket and handed the money over. The 'subject' and his friends ran away.

After locking the door, Miller counted the money in his other pockets and determined that he had given \$40 to the robber. Afterwords, he reported the robbery to the police. Without our consent, other subjects subsequently provided this robber-subject with a not-so-gentle reminder that such behavior was not appropriate. The word got out on the street that our storefront was not to be robbed. Appendix B -- Taking Care of Research Business -302-Several items were stolen from the storefront. Among the items that were stolen at various times throughout the five years were several tape recorders, tools (e.g., a hammer, can opener), a Mister Coffee machine, and our office clock. The story of the clock theft is an interesting one in which we were able to find out exactly what had happened. Two female subjects. Mandy M. and Elinor S. came for their interview

Two female subjects, Mandy M. and Elinor S., came for their interview. Mandy was interviewed first and then sat in the front of the office while Elinor was being interviewed. At the conclusion of Elinor's interview they both left. Shortly afterwards the clock--which had been hanging on a wall in the front of the field office--was discovered to be missing.

Mandy failed to appear for her next scheduled interview. Elinor was reluctant to divulge any information about where Mandy was and why she was failing to adhere to her schedule. Finally, several weeks later, Elinor admitted that Mandy had stolen the clock. While Elinor was being interviewed, Mandy had taken the clock from the wall and wrapped it up in her coat. However, Elinor stated that Mandy felt terrible about stealing the clock and wished she had never done it. Elinor reported that Mandy was so embarrassed that she just could not return to complete her interview cycle.

Field staff handled the incident in the following way. Elinor was told to tell Mandy to return the clock. We had no hard feelings and would not hold it against her. Mandy did come by a few days later, looking very sheepish, and returned the clock. She resumed her interview schedule.

In the above episode, the theft of the clock was less important to field staff than the loss of a good respondent. The clock could be replaced. A subject with an incomplete interview cycle could not be. Further, field staff had gotten to know and like Mandy. Nobody wished her to suffer from the shame that her theft of the clock was reportedly causing.

£.

Appendix B -- Taking Care of Research Business -303-Early in the project, the storefront was burglarized. This incident was handled differently than the clock theft.

One night the storefront was broken into and a typewriter and some tape recorders were stolen. Through the street grapevine, staff members were able to ascertain the identity of the culprits. Three subjects were involved.

The alleged perpetrators had broken into our office, stolen items of real value, and had shown no contrition after the act. Further, their general presentation of self was as "slick" hustlers. Field workers were reluctant to go to the police as this would be in violation of the street code of "taking care of business" without recourse to public authorities. Also, there was no tangible evidence to support any charges. Other subjects who had heard about the burglary offered to beat up the offenders. Field staff refused to allow this. After much discussion, it was decided to terminate the three as subjects. The money that they would lose from interview payments would certainly be in excess of what they received for the stolen merchandise.

When one subject had the audacity to ask when he would be interviewed again, Preble pointedly told him, "we've learned enough about you already!" This statement clearly informed the person that his involvement in the burglary was known without making the direct accusation (which would have led to denial and further conflict), while at the same time terminating him as a subject.

A continuing problem during the study was the unwanted presence of persons who had not been invited to be research subjects or who were undeterrable "moochers." Such persons frequently exhibited anger or hostility at staff and sometimes engaged in threatening behavior -- as well as interrupted ongoing interviews. Such persons needed to be dealt with firmly and consistently. The standard procedure followed by interviewers was to tell such people that they must see a specific field worker who was authorized to accept subjects into the study. This served to mollify most persons.

1

Appendix B -- Taking Care of Research Business -304-Some, however, raised a bit of a ruckus. One person compared his experience with "getting on the program" to his experiences trying to obtain welfare benefits. He shouted angrily that whenever he tried to get some money he was always given a runaround and told he had to see someone else. In such cases friendly subjects or other neighborhood residents would tell the irked individual to "be cool" and, usually, escort him out of the field station. Another problem concerned legitimate potential subjects (i.e., those qualified to participate by the fieldworker) bringing in friends or family

qualified to participate by the fieldworker) bringing in friends or family when they appeared for their first interview. The scenario for this situation was not uncommon.

and the second s

-----

Carlos man

tre cr. .gove

nee gaeen <sup>oo</sup>nnee g

and a residence of the second s

A STATES

a manage and the second

and a start for

"**"**"

1.1.1

1000

ويعونه مست

ಷ <sup>ಸ</sup>್ಗಳನ್ನು

· ·····

and a second

A street opiate user was contacted by the field worker and the nature and purpose of the study was explained, including the fact that a ten dollar interview fee would be given. The drug user was delighted to participate and told others about it. These friends decided to get a "piece of the action" themselves. The potential subject appeared at the field station accompanied by sister, brother-in-law, crime partner, or neighbor. All wanted to be interviewed.

In such situations, interviewers must tread a very thin line. Some members of the respondent's entourage may be qualified candidates for admission into the study. Others may not be. It would be impolitic to accept some and not others. There was also a danger that a blanket rejection might irritate the legitimate subject sufficiently to cause him or her to withdraw from the study. Interviewers told the unauthorized companions that the study was "full up" right now, but that they might be contacted in the future. They were asked to leave names, phone numbers, or other means of contacting them in case a slot was to open up. The interviewer would question the field worker about those persons considered to be potential subjects (i.e., appropriate age, ethnicity, drug use). If the field workers confirmed the interviewer's impression, the person would be contacted and asked to participate. Appendix B -- Taking Care of Research Business -305-

The "moochers" were a different problem. In several cases, these were former subjects who had provided much valuable information and would be returning for additional cycles. In addition, they were active in social circles from which other respondents were recruited. In many cases they came to the storefront with important information about other subjects, an interesting story about a crime, or other similar information. In many cases, they were welcomed in the storefront and were pleasant to chat with when interviewing was slow.

They were unwelcome, however, at specific times (in the middle of the interview, when many other subjects were awaiting interviews, etc.) and in their continuing demands for "carfare," a "deuce" (\$2), "borrow five until tommorrow," "something to eat," a "loan," etc.

Several strategies were designed to deal with such types. If they interrupted an interview, they were told, "wait for half an hour until I'm through" or "see me this afternoon when things are slow." If they continued to persist in the immediacy of their demands, they were told: "You'll have to see staff member X" (who was conveniently out of the storefront at that time).

But all too often, the moochers were extraordinarily (and successfully) persistent. They located and pestered the "person you've got to see." This staff member usually had the job of walking him outside the storefront, talking with him briefly, and giving him \$1 or \$2 just to be rid of him for the day -- so that other work could be accomplished by staff without such interruptions.

Despite problems with persistent moochers, too many potential respondents, and occasional crimes, the staff were able to maintain a pleasant and congenial atmosphere for themselves and for the respondents at the storefront.

- 4

Contractions Association

1811-1

and in many or it

Grades want

-------

The research staff also faced another problem with field worker staff regarding stolen merchandise. This problem involved a direct conflict between the standards of appropriate professional behavior and the informal, but equally powerful, standards followed by the ghetto culture and economic system in which field workers lived.

A principal temptation facing field workers involved offers of contraband goods. While research staff were not offered drugs, they were frequently asked to buy stolen merchandise. Sometimes subjects would come to the storefront with stacks of shirts, boxes of film, a wristwatch, other jewelry, etc., and attempt to sell them to staff members. More often, our staff were solicited by strangers.

One finding not documented elsewhere is that research subjects who engaged in shoplifting or other forms of theft seldom sold stolen property to "professional" fences. Most often, they sold stolen property to other neighborhood residents directly. This process was graphically illustrated time after time.

Complete strangers would wander into the storefront with an armful of new designer jeans, an automobile battery, a radio, an assortment of leather belts, etc., and offer them for sale. Subjects awaiting their interview frequently examined the merchandise and made purchases, at substantial discounts, of items that they might not otherwise have been able to afford. After leaving the storefront, the sellers of such items would continue down the street, entering retail establishments (e.g., the beauty parlor, a hamburger joint), social clubs, or soliciting passers-by.

The practice of buying stolen merchandise was widespread among 'straight' members of this community as well as our subjects and field workers. Persons failed to purchase such "bargains" only because they lacked the money or didn't need the product -- not because they felt it "wrong" to do so. The clear cultural norm, and the low income received by most community residents,

Appendix B -- Taking Care of Research Business -306-D. Professional Standards and the Street Culture.

Appendix B -- Taking Care of Research Business -307permitted and even encouraged the purchase of stolen merchandise.

These common events enabled researchers to observe directly the process of selling stolen goods that was described by subjects during their interviews. Further, in those cases where the stolen merchandise was being offered by research subjects, such merchandise served to verify an account of a shoplifting or other theft in which the subject claimed participation. Field staff were able to observe at first hand the prices asked and obtained for a wide variety of items, thereby increasing their ability to assess the credibility of subjects' accounts of economic transactions. These observations also provided the basis for the estimating the "fence" factor for different crimes in Chapter XII.

Senior research staff decided early in the project not to succumb to the temptation of purchasing merchandise on the streets. Such merchandise was usually stolen property and to purchase it would involve complicity in a crime. Further, we wished to present a certain image of honesty and "uprightness" to subjects, in the hope that they would act in a like fashion and give honest answers to questions and not try to "hustle" us. It was felt that buying stolen property would compromise our image. Some research subjects respected our stance and, if a seller of stolen goods came into the storefront, the subject would invite him outside and make the purchase in the streets. However, when they came back inside, they would proudly show off their new possession to other subjects and field staff alike.

Our paraprofessional field staff were likewise accustomed to buying merchandise in this fashion. They were disappointed when the project ruled out the practice. Senior research staff were frequently amused when, after a purveyor of stolen property had left the field office, the field worker suddenly remembered an important errand or field contact that had to be attended to. He invariably went off down the street in the same direction as the seller.

Appendix B -- Taking Care of Research Business -308-The underground economy is a major component of the ghetto economy. As researchers, we were there to observe: not to judge or to enforce a morality from a different sphere of society. We could, and did, exercise our right not to participate in events that were illegal, in violation of our personal and professional standards of morality, or that could be detrimental to the research effort. However, we felt it was not within our rights, nor within our interests as researchers, to attempt to restrain subjects or indigenous field staff from behaving in a manner that was socially acceptable in their own neighborhood.

172

5 1

No. 2

----

C 1 .....

const interne

and the second

------

mar Maria

Containing and the second

and a state of the second state of the second

errai errai

and the second second

and the second

the the

- State -

and the second

Courses.

Sec.

Non

Series in

J - Li seizet

Sec.

The Economic Behavior of Street Opiate Users project has produced important and critical new information on the relationship between drug use and crime. As described earlier, the project has pioneered successfully a method that combines elements of quantitative and qualitative techniques in order to produce data with significant policy implications.

However, one small regret shared by some senior staff is that the storefront methodology led the researchers to become too rooted in formal interviews. Interviewers and senior professional staff spent most of their time at the storefront, waiting for interviewees to show up to complete their The streets were primarily "worked" by the indigenous schedules. paraprofessional field workers.

One staff member likened this research technique to the "strategic enclave" policies followed during the Vietnam war. In this policy, a "friendly hamlet" was occupied by American forces and their South Vietnamese allies. There was little attempt to venture into the countryside occupied by hostile forces.

In somewhat the same fashion, the Economic Behavior project went into East Harlem, worked in a storefront, and hired local ex-addict, ex-felon field workers, to gain the cooperation of street opiate users in our efforts.

E. Field Work -- Glimpses of the Lives of Street Opiate Users.

Appendix B -- Taking Care of Research Business -309-

On the occasions when senior researchers made direct efforts to learn about the daily lives of subjects, the results were usually both rewarding and illuminating. For example, Dr. Goldstein, the Project Director, became friendly with one subject whom he had been interviewing. The subject invited him to his apartment on 105th Street. On October 29, 1980, Dr. Goldstein visited Harold T. in his apartment. The following account is excerpted from field notes written after the visit and includes a commentary written for this report.

I walked up three dismal flights of steps. Harold greeted me graciously. His apartment contained four incredibly cluttered rooms in a row, railroad style. Upon entering, the first room was the kitchen. Clothes lines were strung across the kitchen about five feet high, to which were pinned two bags and one-third of a banana.

Harold offered me some coffee and I accepted. He then offered me some bread or cake. I declined, saying that the coffee would be fine. Harold continued to offer bread or cake. He seemed very proud of the fact that he could make such an offer. He pulled the two bags off the clothes line and showed me that one did indeed contain bread and the other cake. I had already noticed that Harold had an old dog and a kitten. I asked him if he kept the food hanging on the clothesline so his animals couldn't get at it. He stared at me. "Oh no, "he said, "not my animals. The rats." He said that he had recently gotten the kitten to keep the rats away.

Harold had an old double sink unit in his kitchen. One basin was filled with water and plants and two large goldfish. Also in the kitchen was an aquarium in which he was raising crayfish. He had about six of them and said that he fed them goldfish food. The water in the aquarium was foul, there was no filtration system, and Harold said that he would soon have to change it.

The fire escape was through the kitchen window. It had been covered with plastic sheeting and was filled with marijuana plants, greenhouse style. Harold proudly explained how he processed his marijuana. He claimed that most people didn't understand the process and, as a result, lost much of the potency of their homegrown. However, Harold admitted that he knew where to purchase more potent marijuana and that generally he sold his homegrown and used the money to buy the higher quality. He claimed that he only smoked his own pot when he couldn't afford to go to his regular dealer.

Harold was 40 years old, a former heroin addict now on methadone. His drug use consisted of regular marijuana consumption, and infrequent cocaine, illicit methadone,  $Darvon^{(R)}$ , and tranquilizer use. He reported no criminal activity during the study period other than occasional marijuana sales.

particular? groups in this area.

His continuing contacts with these youths, and later when they were adults, provided the basis for his expertise. For example, Preble offered them room and board at his house when they were without employment. When they were arrested, he would speak at mitigation hearings and maintain phone contact with them while in prison. He attended their parties and social occasions (weddings, funerals, hospitalizations). Numerous christenings made him a "godfather" to the children of these research subjects/friends from

Appendix C - A Tribute to Edward Preble -313-

university -- why did Preble turn instead to street research and heroin use in

As a graduate student with wife and five children, and a new house in the distant New York suburbs, Preble held several jobs (tennis pro, community and street center director, part-time instructor) that led him to the streets. Specifically, during the late 1950s and the era of street gangs, Preble became the director of an after-school community center at Wagner Junior High School on Manhattan's Upper East Side. One of his main jobs was to keep the peace between the "fighting" gangs from the different ethnic groups in this area.

In fall 1957, Preble recalled with humor that he returned from summer vacation to find last year's rowdy gang fighters quiet and peaceful and generally sleepy. He felt very good about the positive effect which his work had upon calming them -- until he discovered that the big "H" (heroin) was really responsible. This experience led to his lifelong fascination with heroin users in the streets. Over the years, he developed close friendships with many Yorkville and East Harlem youths, some of whom became professional criminals (safecrackers, truck highjackers, fences, con men, bookies, loan sharks, etc.), many became heroin users/addicts, and others gained legitimate employment in middle or working class occupations.

## Appendix C - A Tribute to Edward Preble -312-

## APPENDIX C A TRIBUTE TO EDWARD PREBLE, M.A. 1922-1982

. . .

-

- 1716s

· Section

-7-

· 1 \*\*\*

A. Secon

. 7.4

11.250

- 5M.-

· Land

1.1.1.1

· Yours

a internet Second

🛲 prasa

. Sec. poster

1

and the

...

į

र**रच**्

· 17 ¶

ાપ્ય

78.0

BDJ; 4750A;0723A;bj

.....

- ----

۰.

. .

**~** ...

This report is dedicated to the memory of Edward Preble, a Co-Principal Investigator of this project. He contributed greatly to this project and it would not have been possible without him. His contacts within the study community, prior research among street opiate users were essential to the initial phases of this project. Clearly this project would never have been conceived or executed without Ed (as he was called by our research staff), "Ted" (he was called by his family), and "Doc" (as he was called on the streets).

I had the distinct pleasure of being Preble's close colleague and research partner for the past seven years at the Bureau of Research of the New York State Division of Substance Abuse Services and Narcotics and Drug Research, Inc. I wish to recall several aspects of his personal and professional career which contributed to or were a part of his becoming the "grandfather" of drug research ethnography, many of which were unknown to his associates. These remarks describe several professional and personal experiences that were little known, but which had a critical influence upon developing his skills as a field ethnographer and researcher.

His training in anthropology at Columbia University and psychotherapy at New York School of Psychiatry provided fine skills in field work and understanding people from all backgrounds. - His close collaboration with Abram Kardiner (Preble wrote most of <u>They Studied Man</u> (1961) and edited Kardner's dictations which resulted in Kardiner's <u>My Analysis With Freud</u> (1977) ) provided him with a depth of knowledge that was very extensive Harold's lifestyle described above, was in many respects more stable than many of our subjects. He routinely received welfare and had his own apartment which reflected his personality and interests. He was on a methadone program and occasionally used other drugs (cocaine, pills, marijuana, etc.); he no longer used heroin. His lifestyle and his departure from the community suggest the continuing difficulties these respondents face in adjusting to society's expectations and in developing conventional patterns of living.

Appendix B -- Taking Care of Research Business -311-
Appendix B -- Taking Care of Research Business -310-He claimed that most of the money needed to satisfy his needs was raised by doing odd jobs and fixing things. The visit to his apartment was confirming the information that he had given on his daily and weekly interviews.

To remark.

1.000

-

11

5

Ę.

1.

51

511

51 J

. 1

**2**17

11

. 1

T

.....

₹Ľ į÷

27 F.

ST P

.

- S

s/

Harold led me through the remainder of his cluttered apartment. He showed me the broken chairs, tables and lamps that he had collected from sidewalk trash heaps and was now fixing up for resale. He also showed me several inoperative radios that he claimed he would soon have fixed for resale. He had an unbelievable assortment of scraps of metal, wood and plastic that he had found in garbage cans or just laying on the street. I asked him what he did with these scraps. He said that he made things and that currently he was planning to build a rocket ship. I chuckled, and said that I would certainly like to see it when it was done.

Harold led me through his bedroom. The mattress on the floor covered by rumpled dirty sheets reminded me of my student days. We walked into the last room which contained his art work. I was astonished. He was a very talented artist. He specialized in paintings of birds that appeared to my untrained eye the equal of Audubon's work. His most common subject matter, after the birds, were ) beautiful landscapes that he remembered from his youth in Puerto Rico. He proclaimed that his great desire in life was to return to Puerto Rico, find whatever family members remained, perhaps find his children and open a small business.

1

;

His wife had left him years ago during his days as an addict and returned to Puerto Rico with the children. He yearned to go back, but looked at me sadly and said," How could I?" He reported coming to the mainland United States as a young man filled with hopes of becoming successful. He was now in his forties, on methadone, on welfare, and barely scrounging a daily subsistance. He was embarrassed to return in such a state. He claimed that he was putting away a little bit of money from each welfare check and that maybe, some day, he could return to his home with dignity.

Harold kept in contact for some time, dropping by the storefront at irregular intervals to say hello and perhaps offer some item for sale. A few weeks after my visit to Harold's apartment, he arrived at the storefront toting a five foot long silver rocket ship on his shoulder. The rocket was made from scrap pieces of metal. He had installed a motor that made a loud whirring sound and provided an exhaust like an electric fan. Old christmas tree lights blinked on and off when the rocket was plugged in. Harold offered to sell me the rocket for my son who, Harold knew, was a Star Trek/Star Wars science-fiction buff. I could not refuse. Harold said the price was twenty-five dollars which I readily agreed to. I brought the rocket home. My son was delighted. We hung it on the ceiling of his room.

Then, when one of his follow-up 28 day reporting periods was nearing, and we realized that we hadn't seen him for a while, a field worker was sent to locate him. The field worker reported that Harold was no longer living in his apartment and that nobody knew where he had gone. I thought of the beautiful birds that he had painted, free wild things soaring over uncluttered landscapes. I hoped that Harold did return to Puerto Rico, with dignity, and that he found what he was looking for.

contacts, he was the only outsider routinely invited to the annual dinner of "Sing-Sing" Alumni association (for those who "did time" in what is now called the Ossinng Correctional Facility). These friends were eager participants in his research and wide ranging interviews. As a result of these personal contacts with professional criminals and Preble many heroin users over 25 years and during his research efforts, fencing of stolen developed extensive expertise about the following topics: merchandise (both professional and street level), the social culture of prison among professional criminals, the street market in heroin and cocaine dealing, the history of heroin addiction among white ethnics, blacks, and hispanics, and the role of drugs, alcohol, and homicide in the high mortality Several of his of drug abusers -- to name only a few topics of interest. papers deal with aspects of these topics. Other presentations and personal conversations revealed even more depth of understanding about these topics. As many of his colleagues recall, Ed Preble's depth of knowledge about the streets and his rich store of insightful antidotes have touched the lives of hundreds of persons in East Harlem, Yorkville, Lower East Side, and the Bronx, as well as professional colleagues who had the pleasure of working with him. Many rewarding hunches and observations by Ed were converted into formal research projects. His first research grant with NIMH (1962-64) on "Street Gangs and Drug Use," resulted in his major paper, "Taking Care of Business."(1969) This paper profoundly influenced the professional view of heroin users by advancing a thesis opposed to common assumptions of that era. He showed that heroin users pursued a lifestyle which they felt was positive and meaningful to them given their life situations.

# Appendix C - A Tribute to Edward Preble -314-

varied ethnic backgrounds (Irish, Italian, Black, and Hispanic); he maintained yearly contact with his numerous godchildren. As a result of these personal

## Appendix C - A Tribute to Edward Preble -315-

1.11

In 1975, Mike Agar and Douglas Lipton brought Ed Preble to Narcotics and Drug Research where his contributions led to an extensive anthropological and ethnographic emphasis which continues at the New York State agency. Ed pioneered a storefront methodology for studying heroin users, which permitted many persons to visit the "real" world of heroin addiction. Ed's ability to locate street heroin users as subjects and his ability to have them report details about their crimes and drug use led to the submission and successful execution of this Economic Behavior of Street Opiate Users project.

His controversial paper, "Methadone, Wine, and Welfare," (1977) grew out of research (1975-1977) on a NIDA-funded Ethnography of Drug Use among Two White Ethnic Groups (Irish and Italians). He was instrumental in establishing the "Tristate Ethnographic Project," which has subsequently become a demonstration project to show whether contingency contracting and client .governance can reduce problems faced by methadone programs. He had a critical influence upon the development of a federal research agenda on drug and crime. He helped win an five-year research agreement program from the National Institute of Justice for an Interdisciplinary Research Center for the Study of the Relations of Drug and Alcohol to Crime.

My richest memories of Ed, however, are derived from personal associations and direct observations of him at work. I was constantly amazed at his ability to converse comfortably with everyone, regardless of their social status or background, on topics of their own choosing. On a dozen different occasions, I literally observed him conversing one minute with an inebriated alcoholic or street heroin user, and the next minute on the phone with a NIDA official, college professor, or budding anthropology student. If a street opiate user knew "Blackie" on X street, Doc was sure to know someone who knew "Blackie" as well. If a NIDA or state official wanted to know about heroin on the street, Doc could give then an up to the minute

G. ...... ET. ST •**स्**च CTI 57 6 6.5 RE & 11

K29 SKills

 $\prod_{i=1}^{n} (i)$ 

a citation to support his position. response. methadone clients.

## Appendix C - A Tribute to Edward Preble -316-

report, or hand the phone to someone in the storefront who did. If some academic at a professional conference cited a source with which he was unfamiliar, Ed was well grounded in the classics of Anthropology and provided

Ed also did not like qualifications. He felt that strong statements made people react strongly, and that was vastly preferable to a "so what" Thus, in "Methadone, Wine, and Welfare" he made strong statements and got strong reactions from his funding agency, methadone treatment administrators, and academic community. But subsequent research has demonstrated that he was on target, even if not precisely correct.

Ed and I frequently disagreed about alcohol and methadone. Ed felt strongly that many heroin users became alcoholic after entering methadone treatment as they became involved in "boosting methadone" with heavy alcohol consumption. I felt that pre-existing alcohol problems continued after a street heroin user began methadone treatment. While this issue still remains unresolved, however, the research of Mary Jane Kreek at Rockefeller University is beginning to demonstrate a physiological basis for "boosting methadone." Without question alcoholism is one of the most abused drugs among street opiate users. New research currently underway at our office is documenting levels of alcohol consumption among street opiate users which approach the unbelievable, but this is as true of daily heroin users as of

Ed Preble's papers and professional presentations were filled with the insights he gleaned from others. His published papers contain a careful telling of true stories and antecdotes, appealing titles, an entertaining style, and professional value. (His resume is attached below).

## Appendix C - A Tribute to Edward Preble -317-

I know that he had at least five different books which were in his head and needed to be written. He became ill in April, 1982, however, and passed away in August. Unfortunately, no one will have the opportunity to see these books and learn from them.

He has contributed much to my life and that of others who worked with him; our association was very rewarding personally and intellectually. I trust that the memory of Ed Preble and his work will also influence the lives of other professionals in the years ahead.

> Bruce D. Johnson, New York City, September, 1983

NAME Edward Preble EDUCATION (Begin with bacc INSTITUT Lewis & Clark C Columbia Univer (Graduate Schoo and Anthropo EXCEED TWO PAGES. Career Experience: New York City. Criminals." Preble, Edward

PHS 398 (Rev. 5/82)

PRINCIPAL INVESTIGATOR/PROGRAM DIRECTOR:

#### **BIOGRAPHICAL SKETCH**

Give the following information for key professional personnel listed on page 2, beginning with the Principal Investigator/Program Director. Photocopy this page for each person.

alaureate or other initial professional e	TLE Anthropologis Drug Abuse Re ducation and include po	st In esearch stdoctoral training)	BIRTHDATE (Mo., Day, Yr.) Jan. 18, 1922
ION AND LOCATION	DEGREE (circle highest degree)	YEAR CONFERRED	FIELD OF STUDY
College, Portland, Ore. rsity, New York, N.Y. ols of Law, Philosophy ology)	A.B. M.A.	1950 1956	Economics Anthropology

RESEARCH AND/OR PROFESSIONAL EXPERIENCE: Concluding with present position, list in chronological order previous employment, exp ence, and honors. Include present membership on any Federal Government Public Advisory Committee. List, in chronological order, the titles complete references to all publications during the past three years and to representative earlier publications pertinent to this application. DO N

1975-1982 -- Principal Research Scientist, Narcotic and Drug Research, Inc. 1958-1982 -- Professor of Anthropology, The New York School of Psychiatry,

1977-1982 -- Co-Principal Investigator, National Institute on Drug Abuse, Grant No. R01-DA-01926, "Economic Behavior of Street Level Opiate Addicts." 1978-1980 -- Co-Principal Investigator, Law Enforcement Assistance Administration, Grant No. J-IAA-055-8, "Economic Behavior of Nonaddict Career

1975-1977 -- Principal Investigator, Public Health Service Grant No. R01-DA-01051 from the National Institute on Drug Abuse, "Ethnography of Drug Use Among Two White Ethnic Groups." 1971-1975 -- Associate Research Scientist, Rockland Children's Psychiatric

1968-1972 -- Coordinator, Weekend Partial Hospitalization Program, Department of Psychiatry, New York Medical College and Metropolitan Hospital. 1968-1971 -- Associate Research Scientist, Manhattan State Hospital. 1965-1968 -- Narcotics Research Associate, University of Notre Dame, Center for the Study of Man, New York City.

1965-1966 -- Clinical Coordinator, Wiltwyck School for Boys, Esopus, New York. 1963-1967 -- Street Center Director, Lower East Side Service Center, New York

1962-1964 -- Principal Investigator, Public Health Service Grant No. MH-728 from the National Institute of Mental Health, "Street Gangs and Drug Use." 1962-1964 -- Consultant on Street Gangs, Jewish Board of Guardians, Consultation Unit, Council of Social and Athletic Clubs (N.Y.C.) Youth Board. 1955-1969 -- Community Center Director, New York City Board of Education.

Representative Publications

"Problems Utilizing Ethnography in a Single State Agency."

"What An Ethnographic Station Looks Like." Both in Carl Atkins, George Beschner, and Harvey Feldman (editors), A Research Tool for Policy Makers in the Drug and Alcohol Fields. Rockville, Md.: National Institute on Drug Abuse.

An Ethnography of Nonaddicted Career Criminals: "There is No Easy Money." Final Progress Report of the Economic Behavior of Nonaddicted Career Criminals Study, National Institute on Drug Abuse (RO1 DA 02355) and LEAA (LEAA-J-IAA-005-8).

1979	"Ethnographic Research in East Harlem." in Substance Abuse and
	Ethnographic Research in New York. Community Correspondents Group,
	Proceedings, Volume II. Rockville, MD, NIDA: 17-30.
Preble,	Edward and Bruce D. Johnson
1978	Final Report: Ethnography of Drug Use Among Two White Ethnic Groups.
	(R01-DA-01051), Submitted to National Institute on Drug Abuse
Preble,	Edward, Thomas Miller
1977	"Methadone, Wine and Welfare." In Robert Weppner, (Ed.), Ethnography
_	of Drugs and Crime. Berkeley, Calif.: Sage Publications.
Casey, d	John, and Edward Preble
1974	"Narcotic Addiction and Crime: Social Costs and Forced Transfers."
	In Winick, Charles (ed.), Sociological Aspects of Drug Addiction,
	C.R.C. Press, Cleveland, Ohio.
Preble,	Edward and Marc C. Lewy
1973	"Tragic Magic": Word Usage Among New York City Heroin Addicts. The
<b>_</b>	Psychiatric Quarterly, Winter.
Preble,	Edward and Gabriel V. Laury
1972	"Glue Sniffing A Communion." In, Ernest Harms, (Ed.), Drugs and
	Youth" The Challenge of Today, Pergamon Press, Inc.
Preble,	Edward and Esra S. Petursson
1970	"Ine use of Cyclazocine in the Ireatment of Heroin Addicts." Diseses
Dw-L1.	ot the Nervous System.
rredie,	Lawara and John Lasey
1303	laking care of Businesine Heroin User's Life on the Street." Ine
Dunk1.	International Journal of the Addictions
rrebie,	Edward and Gabriel V. Laury
1901	Prastic Lement: The Ten Cent Hallucinogen." Ine International
Dwohl o	Journal of the Addictions.
rredie,	Euwaru "Sacial and Cultural Eactors Dalatad ta Namaatis Nas Amore Ducate
1300	Dicence in New York City " The International Journal of the Addictions
Drahl a	Ricans in New TORK City. The international Journal of the Augustions.
1963	"Departs of Nercotics " Bulletin of the New York State Dictrict
1050-	Repuis of Marcourts, Durietin of the New York State District Branches American Deveniatric Acen Nocomber 1963, March 1965,
1505	$\frac{1}{100}$ $\frac{1966}{100}$
Prohla	Edward and Howard. Davidman
1965	"Schizonbrenia Among Adolescent Street Gang Leaders" In Hoch Paul
1000	and Joseph Zubin (Eds.). Psychonathology of Schizonbrenia New York.
	Grune and Stratton
Kardner	Abram and Edward Preble
1961	They Studied Man. New York:World Publishing Co
1301	They beared man, new jork-more rubitoning ov.,
Sunniam	antal
Informa	tion: Miltary: 11 S. Army (Combat Engineers) 1942-46
THIOLING	Three years personal neuchoanalycis as participant in research
	project. Columbia University Psychoanalytic Clinic for Training
	and Research 1950-1953
	Research Affiliate The Association for Develoanalytic Medicine
	Scientific Associate The American Adademy of Devencenalysis
	Seminar Associate Columbia University Seminar on Drug Abuse
	Editorial Roard Contemporary Develoanalysis
	Editorial Board. The International Journal of the Addictions
	Editorial Roard The American Journal of Drug and Alcohol Abuse
	Evecutive Editorial Roard Deveniator Quantoria
	Executive Euleurian Duaru, rsychilatific Quarterly. Descident National Accordation of Ethnography and Social
	Policy national Association of Ethnography and Sociat
	ruiicy.

future research implications. In Robert Shellow (ed.), Drug Use and Crime: Report of the Panel on Drug Use and Criminal Behavior. Washington, D.C. National Technical Information Service: 215-229. 1981 Statement of Dr. John C. Ball, Temple University. U.S. House of Representatives Subcommittee on Crime June:1-17. 1982 The hyper-criminal opiate addict. Paper prepared for the Interdisciplinary Research Center, New York, N.Y: (September). 1970 The Epidemiology of Opiate Addiction in the United States. Ball, John C., and Carl D. Chambers Springfield, Ill.: Charles C. Thomas. Ball, John C., Lawrence Rosen, Ellen G. Friedman, and David Nurco 1979 The impact of heroin addiction upon criminality. In Louis Harris ne impact of nervin addiction upon criminality. In Louis narris (ed.), Problems of Drug Dependence, 1979. Research Monongraph 27, Rockville, Md.: National Institute on Drug Abuse: 163-169. Ball, John C., Lawrence Rosen, John A. Flueck, and David Nurco 1981 The criminality of heroin addicts when addicted and when off opiates. In James A. Inciardi (ed.), The Drugs-Crime Connection, Beverly Hills:Sage. 39-65. Ball, John C., Lawrence Rosen, John A. Flueck, and David N. Nurco 1982 Lifetime criminality of heroin addicts in the United States. Journal of Drug Issues 3:225-239. 1983 Continuity of crime rates and stability of crime patterns among heroin Ball, John C., John W. Scahaffer, and David N. Nurco addicts in Baltimore. January:1-35. 1983 Living Poorly in America. New York: Praeger. Beeghley, Leonard iernacki, Patrick 1979 Junkie Work, 'hustles' and social status among heroin addicts. Biernacki, Patrick Journal of Drug Issues Fall:535-551. 1983 Crime control: the search for the predators: problems with the crime problem. In Kenneth R. Feinberg (ed.). Violent Crime in America. Blumstein, Alfred Washington, D.C. National Policy Exchange. Jumstein, Altrea, and Jacquerine Jonen 1973 A theory of the stability of punishment. Journal of Criminal Law and Blumstein, Alfred, and Jacqueline Cohen Criminology. 64(2): 198-207. 1978 Deterrence and Incapacitation: Estimating the Effects of Criminal Blumstein, Alfred, Jacqueline Cohen, and Daniel Nagin, Eds. Sanctions on Crime Rates. Washington, D.C.: National Academy of Sciences.

1976 A selective review of the crime-drug literature with reference to Ball, John C. Ball, John C. Ball, John C.

# References -318-

## REFERENCES

### References -319-

Brown, Barry F., and Lester P. Silverman 1974 The retail price of heroin: estimation and applications. Journal of the American Statistical Association 69(347): 595-606.

Brunswick, Ann F.

- 1979 Black youths and drug-use behaviors. In George M. Beschner and Alfred S. Friedman, Youth Drug Abuse: Problems, Issues, and Treatment. Lexington, Massachusetts: Lexington Books: 443-490.
- Brunswick, Ann F., and John M. Boyle
- 1979 Patterns of drug involvement: developmental and secular influences on age at initiation. Youth and Society 11: 139-162.

Casey, John J. and Edward Preble.

1974 Narcotic addiction and crime: social costs and forced transfers. In Winick, Charles (ed.) Sociological Aspects of Drug Dependence. Cleveland, Ohio: 283-307.

- Chaiken, Jan, and Marcia R. Chaiken with Joyce E. Peterson 1982a Varieties of Criminal Behavior: Summary and Policy Implications. Santa Monica, Calif.: Rand Corporation.
- Chaiken, Jan and Marcia Chaiken Santa Monica, Calif.: Rand 1982b Varieties of Criminal Behavior. Corporation.
- Chaiken, Marcia R.
- 1983 Crime Rates and Substance Abuse Among Types of Offenders. Los Angeles, University of California at Los Angeles and New York, Interdisciplinary Research Center. (unpublished)
- Clayton, Richard
- 1981 Federal drugs-crime research: setting the agenda. In James A. Inciardi (ed.), The Drugs-Crime Connection, Beverly Hills:Sage. pp. 17-38.

Clayton, Richard R. and Barry S. Tuchfeld

1981 The drug-crime debate: obstacles to understanding the relationship. Journal of Drug Issues 2:153-166.

Clayton, Richard R. and Harwin L. Voss

1981 Young Men and Drugs in Manhattan: A Causal Analysis. Rockville, Md.: National Institute on Drug Abuse, Research Monograph 39.

Cohen, Jacqueline

1978 The incapacitative effect of imprisonment: a critical review of the literature. in Blumstein, Alfred, Jacqueline Cohen, and Daniel Nagin (Eds.) Deterrence and Incapacitation: Estimating the Effects of Criminal Sanctions on Crime Rates. Washington, D.C.: National Academy of Sciences. pp. 187-243.

Collins, James J., J. Valley Rachal, and Robert Hubbard

1982a The relationship between expensive drug use and income generating crime. Research Triangle Park, North Carolina: Research Triangle Institute. October.

References -320-1982b Criminality in a Drug Treatment Sample: Measurement Issues and Initial Findings. Research Triangle Park, North Carolina: Research Triangle Institute. 1982c Crime and Crime Indicators in the Treatment Outcome Prospective Study. Research Triangle Park, North Carolina: Research Triangle Institute. 1977 Non-survey observational techniques. In Louise Richards and L.B. Belvens (eds.) The Epidemiology of Drug Abuse. Current Issues. Rockville, Maryland: National Institute on Drug Abuse. 1983 Statewide Comprehensive Five-Year Plan 1984-85 Through 1988-89. Draft. 1980 Drugs and Crime: A Survey and Analysis of the Literature. National Institute of Justice. Washington, D.C.: Government Printing Office. 1976 Drug markets and addict consumption behavior. In Robert Shellow Goldman, Fred (ed.), Drug Use and Crime: Report of the Panel on Drug Use and Criminal Behavior. Washington, D.C. National Technical Information Service: 273-296 1981 Heroin and the federal strategy: a policy in search of evidence. Goldman, Fred Journal of Psychoactive Drugs 13(3):217-225. 1981 The economics of addiction. In Inciardi, James A. (ed.) The Goldman. Fred Drugs-Crime Connection. Beverly Hills: Sage.

Collins, James J., J. Valley Rachal, Robert Hubbard, Elizabeth R. Cavanaugh, S. Gail Craddock, and Patricia L. Kristiansen. Collins, James J., J. Valley Rachal, Robert Hubbard, Elizabeth R. Cavanaugh, S. Gail Craddock, and Patricia L. Kristiansen. Crawford, Gail A. Division of Substance Abuse Services Gandossy, Robert P., Jay R. Williams, Jo Cohen, and Henrick J. Harwood

Goldman, Fred and Diane Sixsmith 1982 Medical Care Costs of Drug Abuse. New York: Columbia University School of Public Health. (unpublished report).

Goldsmith, Douglas S., Dana E. Hunt, Douglas S. Lipton, and David L. Strug 1983 Methadone folklore: beliefs about side effects and their impact on treatment. Human Organization (forthcoming).

#### References -321-

Goldstein, Paul J.

1979 Ethnoeconomical approach to the relationship between crime and drug use: preliminary findings. Problems of Drug Dependence 1979, National Institute of Drug Abuse Research Monograph Number 27, pp. 156-162.

Goldstein, Paul J.

1981 Getting over: economic alternatives to predatory crime among street drug users. James A. Inciardi (editor), The Drugs/Crime Connection, California: Sage Publications, pp. 67-84.

Goldstein. Paul J.

1982a Drugs and violent behavior. Presented at meetings of the Academy of Criminal Justice Sciences, Louisville, Kentucky.

Goldstein, Paul J., Reuben Norman, and Barry Spunt

1982b Habitual criminal activity and patterns of drug use. Presented at meetings of the American Society of Criminology, Toronto.

Goldstein, Paul J., and Nina Duchaine

1979 Preliminary findings on the economic behavior of the opiate user. In Substance Abuse and Ethnographic Research in New York. Community Correspondents Group, Proceedings, Volume II. Rockville, MD., NIDA. pp. 43-67.

Goldstein, Paul J., Douglas S. Lipton, Edward Preble, Ira Sobel, Tom Miller, William Abbott, William Paige, and Franklin Soto

1983 The marketing of street heroin in New York City. Journal of Drug Issues, (accepted for publication)

Greenberg, David

1975 The incapacitative effect of imprisonment: some estimates. Law and Society Review 9(4): 541-80.

Hindelang, Michael J., Travis Hirschi, and Joseph G. Weis 1981 Measuring Delinquency. Beverly Hills: Sage.

Hughes, Patrick H., et al.

1977 Behind the Walls of Respect: Community Experiments in Heroin Addiction Control. Chicago: University of Chicago Press.

Hunt, Dana E., Douglas S. Lipton, Douglas Goldsmith

1981 Meth and medication: observations of clinic life. Report to the National Institute on Drug Abuse, New York: Division of Substance Abuse Services.

Hunt, Dana E., Douglas S. Lipton, Douglas Goldsmith, and David Strug

1982a Street pharmacology: uses of cocaine and heroin in the treatment of addiction. Paper presented at the NADAP Conference on Clinical Relevance of Research Development in Substance Abuse: New York, March.

1.145

Inciardi. James A.

Inciardi, James A.

Inciardi. James A.

Inciardi, James A.

Johnson, Bruce D.

Johnson, Bruce D.

Johnson, Bruce D. 35-53.

Johnson, Bruce D. 31-42

Johnson, Bruce D.

# References -322-Hunt, Dana E., Douglas S. Lipton, Douglas Goldsmith, and David Strug 1982b Problems in methadone treatment: The influence of reference groups. (draft). New York: Division of Substance Abuse Services, August Hunt, Dana E., Douglas S. Lipton, and Barry Spunt 1982c Patterns of criminality among methadone clients and current narcotics users not in treatment. Paper presented at the American Society of Criminology, Toronto, Canada, November. Hunt, Dana E., Douglas S. Lipton, and Barry Spunt 1983 Patterns of criminality among methadone clients and current narcotics users not in treatment. Paper presented at the American Society of Criminology, Toronto, Canada, November. 1979 Heroin use and street crime. Crime and Delinquency (July):335-346. 1980 Youth, drugs, and street crime. In Frank Scarpitti and Susan K Datesman, Drugs and the Youth Culture. Beverly Hills: Sage:175-204. 1981 The Drugs-Crime Connection. Beverly Hills: Sage. 1982 The production and detection of fraud in street studies of crime and drugs. Journal of Drug Issues 3:285-291. Inciardi, James A. and Carl D. Chambers 1972 Unreported criminal involvement of narcotic addicts. Journal of Drug Issues. 2:57-64. 1973 Marihuana Users and Drug Subcultures. New York: Wiley. 1976 The Race, Class, and Irreversibility Hypotheses: Myths and Research about Heroin. In Joan D. Rittenhouse, (ed.) The Epidemiology of Heroin and Other Narcotics. Research Monograph Series 16. Rockville, Md.:National Institute on Drug Abuse: 51-57. 1978 Once an addict seldom an addict. Contemporary Drug Problems Spring: 1979 Ethnography and Drug Policy. Informal presentation reproduced in Substance Abuse and Ethnographic Research in New York. Community Correspondents Group, Proceedings, Volume II. Rockville, Md.: NIDA: 1980 Towards A Theory of Drug Subcultures. In Dan Letteri, etal. (eds.), Theories on Drug Abuse: Selected Contemporary Perspectives. Rockville, Md: NIDA Research Monograph 30: 110-119.

#### References -323-

Johnson, Bruce D.

1981a Empirical patterns of heroin consumption among selected street heroin users. Presented at International Conference on Social Stress and the Abuse of Drugs, International Anti-Drug Abuse Foundation, New York.

Johnson, Bruce D.

1981b The criminal behavior of street heroin and cocaine users. Presented at International Institute on the Prevention and Treatment of Drug Dependence, International Council on Alcohol and Addictions, Vienna, Austria.

Johnson, Bruce D.

1981c The drug crime nexus (1979-1982): research on the drug-crime relationship, with emphasis upon heroin users/injectors as criminal repeaters. Paper prepared for the National Institute of Justice. New York: Interdisciplinary Research Center, December.

Johnson, Bruce D., Paul J. Goldstein, and Nina Duchaine

1979 What is an addict? Empirical patterns and concepts of addiction. Presented at Society for the Study of Social Problems, Boston, MA.

Johnson. Bruce D. and James Schmeidler

1981 Exploring asymmetries in the hard drug-crime relationships. Presented at Society for the Study of Social Problems. Toronto, Canada.

Johnson, Bruce D., Eric Wish, David Strug, and Marcia Chaiken

1983a Violence Among Heroin Abusers. Presented to American Sociological Association. Detroit: August.

Johnson, Bruce D., Eric Wish, and David Huizinga

1983b The Concentration of Delinquent Offending: The Contributions of Serious Drug Involvement to High Rate Delinquency. New York: Interdisciplinary Research Center. (forthcoming)

Moore, Mark H.

1977 Buy and Bust. Lexington, Massachussetts: Lexington Books.

Moore, Mark H.

1979 Limiting supplies of drugs to illicit markets. Journal of Drug Issues Spring: 291-309.

Moore, Mark, Susan Estrich, and Daniel McGillis

1981 Final Report on the Project on Public Danger, Dangerous Offenders, and the Criminal Justice System. Cambridge, Mass: Harvard University.

McGlothlin, William H., M. Douglas Anglin, and Bruce D. Wilson

1977 An Evaluation of the California Civil Addict Program. Services Research Issues Series. Rockville, Md.: National Institute on Drug Abuse.

McGlothlin, William H., M. Douglas Anglin, and Bruce D. Wilson 1978 Narcotic addiction and crime. Criminology 16:293-315.

Newman, Robert

1983 Remarks on Methadone Treatment. Outlook. Albany, N.Y.: New York State Division of Substance Abuse Services. May.

Nurco, David N., A.J. Bonito, M. Lerner, and M.B. Balter 1975 Studying addicts over time: methodology and preliminary findings. American Journal of Drug and Alcohol Abuse 2(2):183-196. 1981a Addict careers. I. a new typology. International Journal of the Addictions 16(8):1305-1325. 1981b Addict careers. II. The first ten years. International Journal of the Addictions 16(8):1327-1356. 1981c Addict careers. III. trends across time. International Journal of the Addictions 16(8):1357-1372. 1981d Narcotic Addiction: A Changing Scene? In Louis S. Harris (ed.), Problems of Drug Dependence, 1980. Research Monograph 34, Rockville, Maryland: National Institute on Drug Abuse: 206-212. 1982 The stepping-stone hypothesis--marijuana, heroin, and causality. Chemical Dependencies: Behavorial and Biomedical Issues 4(3):229-241. 1976 Young Men and Drugs - A Nationwide Survey. Research Monograph 5. Rockville, Maryland: National Institute on Drug Abuse. 1976 Drug Use and Crime: Shellow, Robert (ed.). Washington, D.C.: National Technical Information Service. 1978 Criminal Careers of Habitual Felons. Washington, D.C.: National Institute of Law Enforcement and Criminal Justice. 1980 Doing Crime: A Survey of California Prison Inmates. Santa Monica: Rand Corporation. 1982 Survey of prison and jail inmates: background and method. Prepared for National Institute of Justice. Santa Monica: Rand Corporation. 1981 Letter to Henry A. Waxman, House of Representatives. Rockville, Md.: National Institute on Drug Abuse. 1980 What an ethnographic station looks like. Problems utilizing ethnography in a single state Agency. In Carl Atkins, George Beschner, and Harvey Feldman (editors), A Research Tool for Policy Makers in the Drug and Alcohol Fields. Rockville, Md.: National Institute on Drug

Nurco, David N., Ira H. Cisin, and Mitchell B. Balter Nurco, David N., Ira H. Cisin, and Mitchell B. Balter Nurco, David N., Ira H. Cisin, and Mitchell B. Balter Nurco, David N., Ira H. Cisin, and Mitchell B Balter O'Donnell, John A., Richard R. Clayton O'Donnell, John A., Harwin L. Voss, Richard R. Clayton, Gerald T. Slatin, and Robin G.W. Room Panel on Drug Use and Criminal Behavior Petersilia, Joan, Peter Greenwood, and Martin Lavin Peterson, Mark A., Harriet B. Braiker and Suzanne M. Polich Peterson, Mark, Jan Chaiken, Patricia Ebener, and Paul Honig Pollin, William Preble, Edward

Abuse.

#### References -324-

#### References -325-

Preble, Edward and Bruce D. Johnson

- 1980 An Ethnography of Nonaddicted Career Criminals: There is No Easy Money. Final Progress Report of the Economic Behavior of Nonaddicted Career Criminals Study, National Institute on Drug Abuse (RO1 DA 02355) and LEAA (LEAA-J-IAA-005-8).
- Preble, Edward

- 1979 Ethnographic research in esat harlem. in Substance Abuse and Ethnographic Research in New York. Community Correspondents Group, Proceedings, Volume II. Rockville, MD, NIDA: 17-30.
- Preble, Edward and Bruce D. Johnson
- 1978 Final Report: Ethnography of Drug Use among Two White Ethnic Groups. National Institute on Drug Abuse.
- Preble, Edward and Thomas Miller
- 1977 Methadone, wine, and welfare. In Robert S. Weppner (ed.), Street Ethnography. Beverly Hills: Sage, pp. 229-248.
- Preble, Edward and John J. Casey
- 1969 Taking care of business: the heroin user's life on the street. The International Journal of Addictions 4:1-24.
- Research Triangle Institute
- 1976 Costs to Society of Drug Abuse, Vol. II. Maryland: National Institute on Drug Abuse.
- Robins, Lee N.
- 1973 A Follow-Up of Vietnam Drug Users. Washington, D.C.: U.S. Government Printing Office.

#### Robins, Lee N.

1974 The Vietnam Drug User Returns. Special Action Office Monograph, Series A, No. 2. Washington, D.C.: Superintendent of Documents, U.S. Government Printing Office.

- 1975 Drug use among vietnam veterans--three years latter. Medical World News pp. 44-45, 49.
- Robins, Lee N.
- 1977 Vietnam Veterans three years after Vietnam: How our study changed our view of heroin. Journal of Problems of Drug Dependence.
- Robins, Lee N.
- 1978 Sturdy childhood predictors of adult antisocial behavior: Replication from longitudinal studies. Psychological Medicine 8:611-622.
- Robins, Lee N.
- 1979 Addict careers. Chapter 31 in Robert L. Dupont, Avram Goldstein, and John O'Donnell (eds.), Handbook on Drug Abuse. Rockville, Md.: National Institute on Drug Abuse., 325-336.
- Robins, Lee N., Darlene H. Davis, and Eric D. Wish
- 1977 Detecting predictors of rare events: demographic, family, and personal deviance as predictors of stages in the progression towards narcotic addiction. In J.S. Strauss, H.M. Babigian, and M. Roff (eds.) The Origins and Course of Psychopathology. New York: Plenum, pp. 379-406.

Robins, Lee N., Michie Hesselbrock, Eric Wish, and John E. Helzer. 1977 Polydrug and alcohol use by veterans and nonveterans. In Smith, David E., Steven M. Anderson, Millicent Buxton, Nancy Gottlieb, William Harvey, and Tommy Chung (eds.) A Multicultural View of Drug Abuse: Proceedings of the National Drug Abuse Conference, 1977. Philadelphia: Hall and Schenkman, pp. 74-90.

Sells, Saul B., et. al. 1976 A national follow-up study to evaluate the effectiveness of drug abuse treatment: a report on cohort I of DARP five years later. American Journal of Drug and Alcohol Abuse 3(4):545-556.

Silverman, Lester P and N. L. Spruill 1977 Urban crime and the price of heroin. Journal of Urban Economics.

Wish, Eric, Bruce D. Johnson, Kevin Anderson, David Strug, Thomas Miller 1983 Concordance between self-reports of drug use and urinalysis test results from active unapprehended criminals. New York: Interdisciplinary Research Center.

Wolfgang, Marvin and Robert M. Figlio 1982 Seriousness Scores from the National Survey of Crime Severity. Philadelphia: University of Pennsylvania (unpublished),

\$

BDJ;8557A;bj

#### References -326-

Robins, Lee N.



# END

Ģ